Northumbria Research Link

Citation: Bradley, Gemma (2018) "Have they got rehab potential?" – An ethnographic study exploring meaning and evaluations of rehabilitation potential for an older person following an acute admission. Doctoral thesis, Northumbria University.

This version was downloaded from Northumbria Research Link: http://nrl.northumbria.ac.uk/39713/

Northumbria University has developed Northumbria Research Link (NRL) to enable users to access the University's research output. Copyright © and moral rights for items on NRL are retained by the individual author(s) and/or other copyright owners. Single copies of full items can be reproduced, displayed or performed, and given to third parties in any format or medium for personal research or study, educational, or not-for-profit purposes without prior permission or charge, provided the authors, title and full bibliographic details are given, as well as a hyperlink and/or URL to the original metadata page. The content must not be changed in any way. Full items must not be sold commercially in any format or medium without formal permission of the copyright holder. The full policy is available online: http://nrl.northumbria.ac.uk/policies.html

www.northumbria.ac.uk/nrl



"Have they got rehab potential?" – An ethnographic study exploring meaning and evaluations of rehabilitation potential for an older person following an acute admission

G Bradley

PhD

August 2018

"Have they got rehab potential?" – An ethnographic study exploring meaning and evaluations of rehabilitation potential for an older person following an acute admission

Gemma Bradley

A thesis submitted in partial fulfilment of the requirements of the University of Northumbria at Newcastle for the degree of Doctor of Philosophy Research undertaken in the Faculty Health and Life Sciences

August 2018

ABSTRACT

Rehabilitation potential is the potential of an individual to benefit from interventions which aim to optimise and restore function after a period of ill-health or new onset of disability. Health professionals are frequently required to evaluate this potential to determine who may be most likely to benefit from the provision of ongoing rehabilitation services.

This doctoral study explored decision-making related to the assessment of rehabilitation potential of older people in hospital and the recommendation of rehabilitation pathways. To explore this in real-time, and in the shared patient and professional context in which decision-making occurred, principles of ethnographic and case-study research were utilised. The case study site was one acute medical ward within a local acute hospital in the North East of England. Three phases of fieldwork were undertaken, including a 2-week orientation and mapping phase, an 8 week period centred on the practice of occupational therapists and physiotherapists in relation to five patient cases, and finally a phase which involved five in-depth interviews with individual health professionals. Data was generated through observation, interviews and the review of clinical records. The whole data-set was analysed using thematic analysis.

Key findings highlight that rehabilitation in this context was understood as a process to facilitate physical improvements and associated with an organisational aim for optimum safety rather than optimum function. And, although idealised as a phase of care, rehabilitation was often linked to a specific place, with the evaluation of rehabilitation potential subsequently linked with a hospital transfer. Furthermore, rehabilitation potential was ambiguous and poorly explained to patients and families. Health professionals recognised that their evaluation of rehabilitation potential was linked to high-stakes decisions about access to, or withholding services, and therefore the

ethical dimensions of this decision had far-reaching influence. The involvement of the older person in judgements about rehabilitation potential and pathways was minimal, and there were many critical challenges to older people receiving fair and just access to services.

The research findings conclude that there are significant tensions between the context of acute hospital care and the philosophy and ideals of rehabilitation. Furthermore, findings can assist professionals to recognise and reconcile tensions in practice and to move towards reframing rehabilitation to place the individual needs of older people at the centre of service delivery.

TABLE OF CONTENTS

Glossary...i List of Figures...iv List of Tables...v List of Appendices...vi Acknowledgements...vii Authors Declaration...viil

1. Chapter One: Introduction to the thesis...1

- 1.1. My position influences on the early development of the study...1
- 1.2. Background...4
- 1.3. Overview of policy and legislation...6
- 1.4. Research aim and objectives...10
 - 1.4.1. Research aim...10
 - 1.4.2. Research objectives...10
- 1.5. Outline of subsequent chapters...11
- 1.6. Writing conventions...12

2. Chapter Two: Literature Review...14

- 2.1. Chapter introduction...14
- 2.2. Rehabilitation for older people...15
 - 2.2.1. Defining rehabilitation...15
 - 2.2.2. Models of rehabilitation for older people...17
 - 2.2.3. Effectiveness of rehabilitation services for older people...20
- 2.3. Rehabilitation potential...24
 - 2.3.1. Definitions of rehabilitation potential...24
 - 2.3.2. The significance of rehabilitation potential...26
 - 2.3.3. Judgements about rehabilitation potential...26
- 2.4. Rehabilitation decision-making...30
 - 2.4.1. Decision-making frameworks...30
 - 2.4.2. Decision-making predicting who will benefit from rehabilitation...33
 - 2.4.3. Decision-making selection for rehabilitation...36
 - 2.4.4. Decision-making managing risk...41
 - 2.4.5. Decision-making allocating finite resources...43
 - 2.4.6. Decision-making service-user involvement...45
 - 2.4.6.1. Definitions and characteristics of shared decisionmaking...45
 - 2.4.6.2. Wider concepts participation, engagement and involvement...47
 - 2.4.6.3. Benefits and challenges of involvement for older people...48

2.5. Chapter summary...51

3. Chapter Three: Methodology and Methods...53

- 3.1. Ontological beginnings...53
- 3.2. Epistemology...54
- 3.3. Methodology...57
 - 3.3.1. Ethnography the overriding methodology...57
 - 3.3.2. Research design...62
 - 3.3.3. Identifying the 'case' and negotiating access...63
 - 3.3.4. Patient and public involvement in research...64
- 3.4. Methods...66
 - 3.4.1. Methods of data gathering...66
 - 3.4.2. Principles of observation...67
 - 3.4.2.1. Overt observation...67
 - 3.4.2.2. Participant observation insider or outsider?...67
 - 3.4.2.3. Level of participation during observation...69
 - 3.4.3. Phases of data gathering ...70
 - 3.4.3.1. Preparatory phase relationship building, negotiating access and consent...70
 - 3.4.3.2. Phase 1 orientation and social/environmental mapping...71
 - 3.4.3.3. Phase 2 ethnographic fieldwork centred on 'patient-tracking'...72
 - 3.4.3.4. Phase 3 in-depth interviews with health professionals...74
 - 3.4.4. Sampling and recruitment...75
 - 3.4.4.1. Recruitment of patients during phase 2...75
 - 3.4.4.2. Recruitment of health professionals during phase 2...78
 - 3.4.4.3. Recruitment of health professionals during phase 3...78
 - 3.4.5. Types of data...79
 - 3.4.5.1. Fieldnotes from observations...80
 - 3.4.5.2. Written data sources...80
 - 3.4.5.3. Interviews and reflexive interactions...81
 - 3.4.6. Data analysis...81
 - 3.4.7. Methods to write and present findings...85
 - 3.4.8. Methods to strengthen the quality of ethnographic research...87
 - 3.4.8.1. Credibility...87
 - 3.4.8.2. Transferability...91
 - 3.4.8.3. Relevance...92
 - 3.4.9. Ethical considerations...93
 - 3.4.9.1. Ethical approvals...94
 - 3.4.9.2. Consent...95
 - 3.4.9.3. Issues of power...97
 - 3.4.9.4. Avoiding harm...99
- 3.5. Chapter summary...99

4. Chapter Four – Research Findings: Descriptions...101

- 4.1. Introduction to research findings (chapters four to seven)...101
- 4.2. Chapter introduction...102
- 4.3. The study site...103
- 4.4. The base ward...103
 - 4.4.1. The physical environment of the base ward...105
- 4.5. Linked services and care environments...106
 - 4.5.1. The rehabilitation ward...106
 - 4.5.2. The bed-based intermediate care unit...107
 - 4.5.3. The reablement service...107
- 4.6. Occupational therapy and physiotherapy staffing for the base ward and linked rehabilitation services...108
- 4.7. Roles and working patterns within the base ward...109
 - 4.7.1. The consultant role...109
 - 4.7.2. Ward doctors...110
 - 4.7.3. Nursing...110
 - 4.7.4. Physiotherapy...111
 - 4.7.5. Occupational Therapy...112
- 4.8. A typical day...113
- 4.9. Description of key activities observed during fieldwork...116
 - 4.9.1. Multi-disciplinary team (MDT) daily handover...116
 - 4.9.2. Nursing handover...118
 - 4.9.3. Weekly multi-disciplinary team (MDT) meetings...118
 - 4.9.4. Length of stay meetings...120
- 4.10. Description of changes to study site between phase 1 and phase 3...121
- 4.11. Description of how professionals evaluated rehabilitation potential...123
 - 4.11.1. Stage: Gathering of baseline information and comparing previous and current levels...124
 - 4.11.2. Stage: Provision of curative and supportive interventions to resolve of manage issues which may impact on ability to make functional gains...125
 - 4.11.3. Stage: Provision of rehabilitative interventions and monitoring of progress...126
 - 4.11.4. Stage: The evaluation of rehabilitation potential and decision about the subsequent pathway...127
- 4.12. Description of the use of criteria for referral to rehabilitation services...130
- 4.13. Chapter summary...131
- 5. Chapter Five Research Findings: Social Construction and meaningmaking of 'rehabilitation' and 'rehabilitation potential'...133
 - 5.1. Chapter Introduction...133
 - 5.2. The meaning of rehabilitation...137
 - 5.2.1. Rehabilitation as a process to bring about improvements...137

- 5.2.2. Rehabilitation as a process to bring about physical improvements...137
- 5.2.3. Rehabilitation as physiotherapy...139
- 5.2.4. Rehabilitation as a place...143
- 5.2.5. Versions of rehabilitation the ideal versus the reality...146
- 5.2.6. Meaningful rehabilitation and time...147
- 5.2.7. The objectives of rehabilitation...148
 - 5.2.7.1. Patient-centred objectives and goal setting...148
- 5.2.8. The organisational objective of rehabilitation...149
- 5.3. The meaning of rehabilitation potential...151
 - 5.3.1. An ambiguous concept...potential for what?...151
 - 5.3.2. Potential to 'get back to baseline'...152
 - 5.3.3. Potential to get back to baseline mobility...153
 - 5.3.4. Synonymous with a judgement about transfer of care...154
 - 5.3.5. Rehabilitation potential as potential to influence discharge outcome...157
 - 5.3.6. Rehabilitation potential controversial and open to challenge...158
- 5.4. Evaluations about rehabilitation potential influences on reasoning...161
 - 5.4.1. Clinical and individual factors...161
 - 5.4.1.1. Co-morbidities...161
 - 5.4.1.2. Cognitive function...162
 - 5.4.1.3. The importance of understanding the 'baseline'...164
 - 5.4.2. Engagement with a rehabilitation process...167
 - 5.4.2.1. The influence of the response to rehabilitative interventions...167
 - 5.4.2.2. Allowing time...167
 - 5.4.2.3. Mood and motivation to engage...169
- 5.5. Chapter summary...170

6. Chapter Six – Research Findings: The ethical dimensions of evaluating rehabilitation potential...172

- 6.1. Chapter introduction...172
- 6.2. Bringing about a positive outcome...174
- 6.3. Avoiding harm...177
- 6.4. The blurring of doing good and minimising harm...178
- 6.5. Autonomy and involvement...180
 - 6.5.1. Involvement in evaluations about rehabilitation potential...rehabilitation what?...180
 - 6.5.2. Involvement in pathway decisions...181
 - 6.5.3. Patient involvement after decisions...183
 - 6.5.4. Patient involvement after family involvement...184
 - 6.5.5. The vocal and not-so-vocal...185
 - 6.5.6. Reasons for limited involvement...186
 - 6.5.6.1. Not wanted?...186

- 6.5.6.2. Rationed out?...187
- 6.5.6.3. Lack of meaningful alternatives?...188
- 6.6. Fairness and rationing...189
- 6.7. Professional tensions and discomfort...191
- 6.8. Strategies for dealing with competing ethical demands...193
 - 6.8.1. Overt strategies...193
 - 6.8.1.1. Waiting lists...193
 - 6.8.1.2. Triage...194
 - 6.8.1.3. Reflection...195
 - 6.8.2. Covert strategies...196
 - 6.8.2.1. Creating narratives about patients...196
 - 6.8.2.2. Managing demand through information-giving...197
- 6.9. Chapter summary...197

7. Chapter Seven – Research Findings: Professional roles...199

- 7.1. Chapter introduction...199
- 7.2. The traditional hierarchy...201
- 7.3. The physiotherapy role...201
 - 7.3.1. Perceptions of why the role was significant...203
 - 7.3.2. Challenges faced by physiotherapists...203
- 7.4. The occupational therapy role...204
 - 7.4.1. Focus on mobility...204
 - 7.4.2. Role in decision-making...205
 - 7.4.3. Respect for the occupational therapy role...206
- 7.5. Comparing the physiotherapy and occupational therapy role...207
- 7.6. Chapter summary...210

8. Chapter Eight – Discussion...211

- 8.1. Chapter introduction...211
- 8.2. The 'macro' influences on meanings and social constructions within the acute hospital context...212
- 8.3. The meaning of rehabilitation...215
 - 8.3.1. Revisiting definitions of rehabilitation...215
 - 8.3.2. A version of rehabilitation that values (physical) improvement...216
 - 8.3.3. A version of rehabilitation that values flow and efficiency...217
 - 8.3.4. A version of rehabilitation that prioritises optimum safety rather than optimum function...218
 - 8.3.5. A version of rehabilitation happening in a separate place...220
 - 8.3.6. A perceived gap between ideal and real-world versions of rehabilitation...221
- 8.4. The meaning of rehabilitation potential...224
 - 8.4.1. The ambiguous nature of rehabilitation potential...224
 - 8.4.2. Potential to improve...224
 - 8.4.3. A blurring of rehabilitation potential and an internal transfer...225

- 8.4.4. The subjective nature of judgements about rehabilitation potential...227
- 8.5. The ethical realities of rehabilitation decision-making...228
 - 8.5.1. Real world application of ethical principles and frameworks...229
 - 8.5.2. Autonomy...231
 - 8.5.3. Beneficence...233
 - 8.5.3.1. Is there an evidence base that rehabilitation brings about improvement?...233
 - 8.5.3.2. Were 'good' functional outcomes evident during patienttracking?...234
 - 8.5.3.3. Balancing beneficence and non-maleficence...235
 - 8.5.3.4. Balancing beneficence with autonomy who's version of a 'good outcome'...238
 - 8.5.4. Fairness, justice and rationing...241
 - 8.5.4.1. Rationing based on appropriate or deserving behaviours...244
 - 8.5.4.2. Manipulating the nature and quality of information to assist with rationing...245
 - 8.5.4.3. Is justice and fairness realistic for older people?...247
 - 8.5.5. Ethical distress for professionals...250
- 8.6. The centrality of the physiotherapy role in evaluating rehabilitation potential and making decisions about pathways...256
 - 8.6.1. Locating occupational therapy and physiotherapy within the acute care hierarchy issues of power and decision-making authority...257
 - 8.6.2. Challenges to both disciplines from systemic constraints...262
- 8.7. Chapter summary...264

9. Chapter Nine – Conclusion...266

- 9.1. Chapter introduction...266
- 9.2. Summary of key findings...266
- 9.3. Implications for practice...274
- 9.4. Strengths and limitations of the current study...280
- 9.5. Dissemination of findings...283
- 9.6. Recommendations for future research...284
- 9.7. Final reflections on the research journey...286
- 9.8. Final conclusions...289

References...290

Appendices...316

GLOSSARY

Acute Hospital – acute hospitals provide a range of services (including both physical and mental health services) which require time-sensitive and rapid intervention and are often orientated towards urgent or emergent episodes of injury or illness. In the United Kingdom, acute hospital care is a branch of secondary care (a level of care after the primary and most immediate care provider such as General Practitioner, Health Visitor or District Nurse) and often includes emergency and admission services, critical and intensive care and medical and surgical specialities. In this study, the acute hospital was a district hospital including some of these services and orientated towards physical health specialities.

Bio-medical Model – a term which is used to describe an approach to health care which focuses on the physical or biological aspects of disease and ill-health. It views the problems which arise from illness and disability as primarily related to the underlying disease pathology and therefore health interventions focus on these causes.

Gatekeeper – the term gatekeeper is used to describe those with control and decisionmaking authority about how tasks should be organised and by whom, and who often have responsibilities for access to resources. The term is used within the methodology section to discuss those with authority to enable researcher access to research resources such as sites and participants. The term is also used in the reporting of findings to represent health professionals with responsibility for access and referral to resources and services. **Health Professional –** employed members of the NHS trust, registered with relevant regulatory bodies and who have direct responsibility for patient care. They include, for example, doctors, nurses, physiotherapists and occupational therapists. For the purpose this study, this term also may refer to social workers. The term is also used interchangeably with practitioners.

Occupational Therapist – registered health professionals who provide support for people to do everyday tasks, help people to regain or maintain independence where possible, and use everyday activity to promote health. Occupational therapists within this site provided assessment and interventions for older people who were experiencing changes in their abilities to carry out everyday activities following an acute hospital admission.

Pathway – a pathway, sometimes referred to as a care pathway or a patient pathway, is the route a patient takes when entering a healthcare system. In this research, the term pathway is used to refer to the journey from entry in to the acute hospital system for one episode of care. Although the term is sometimes used to refer to standardised tools to promote equity and consistency across a care delivery process, often for conditions with a predictable clinical course (for example hip fracture), the term is used here to represent the journey rather than the set of tools or actions.

Physiotherapist - registered health professionals who help people affected by injury, illness or disability through movement and exercise, manual therapy and education. Physiotherapists within this site provided assessment and interventions for older

ii

people who were experiencing changes in their movement, strength and mobility following an acute hospital admission.

Rehabilitation – is used to describe a set of interventions designed to optimise functioning and reduce the impact of illness or disability. It differs from other sets of interventions which have other aims, such as curative or preventative interventions. A full exploration and conceptualisation of the term rehabilitation will begin in the literature review and will permeate the full thesis.

Rehabilitation Potential – an individual's capability to benefit from rehabilitative interventions. This is often a judgement or estimate made by health professionals. Again, a full exploration of this term will be a central focus of the thesis, beginning with an illustration of current knowledge and issues within the literature review.

List of Figures

- Figure 1: Summary of activities during phase 1 of fieldwork...72
- Figure 2: Summary of activities during phase 2 of fieldwork...74
- Figure 3: Inclusion criteria for recruitment of patients during phase 2 of fieldwork...75
- Figure 4: Additional inclusion criteria for recruitment of patients during phase 2 of fieldwork...75
- Figure 5: Overview of reasoning process to evaluate rehabilitation potential...123
- Figure 6: Detailed reasoning process to evaluate rehabilitation potential and reach decisions about the rehabilitation pathway...128
- Figure 7: Written criteria for reablement service...131
- Figure 8: Coding and theme development social construction and meaningmaking of 'Rehabilitation'...135
- Figure 9: Coding and theme development social construction and meaningmaking of 'Rehabilitation Potential'...136
- Figure 10: Coding and theme development ethical dimensions of evaluating rehabilitation potential...173
- Figure 11: Coding and theme development professional roles...200
- Figure 12: A taxonomy for rehabilitation in the acute hospital context...223
- Figure 13: Influences on the evaluation of rehabilitation potential...254

List of Tables

Table 1:	Literature search strategy and search terms14
Table 2:	Summary of patients recruited for patient tracking77
Table 3:	Stages of data analysis83
Table 4:	Examples to illustrate the presentation of findings102
Table 5:	Occupational therapy and physiotherapy staffing for the base ward,
	rehabilitation ward, bed-based intermediate care unit and reablement
	service109
Table 6:	Summary of occupational therapy and physiotherapy involvement during
	patient tracking140
Table 7:	Summary of omissions or inaccuracies with baseline information
	documented in clinical records164
Table 8:	Summary of information obtained from clinical records documenting
	optimum functional level of patients tracked during phase 2175
Table 9:	Examples illustrating patient involvement after decisions183
Table 10:	Examples illustrating patient involvement after family involvement184
Table 11:	Definitions of rehabilitation216

List of Appendices

- Appendix 1: Phase 3 semi-structured interview schedule...317
- Appendix 2: Sample observational checklist...321
- Appendix 3: Extract from fieldnotes...322
- Appendix 4: Clarifying the analytical position adapted from Braun and Clarke...323
- Appendix 5: Example coding code: meaning of rehabilitation...325
- Appendix 6: Faculty ethical approval...332
- Appendix 7: NHS ethical approval...333
- Appendix 8: Participant information for all core staff on base ward...338
- Appendix 9: Summary/poster for all patients/families/visitors to base ward...339
- Appendix 10: Participant information 'At-a-glance' summary for patients and families...341
- Appendix 11: Participant information extended information for patients and families...342
- Appendix 12: Patient consent form ...348
- Appendix 13: Health professional consent form...349
- Appendix 14: Process for seeking advice from a consultee...350
- Appendix 15: Consultee advice form...351
- Appendix 16: Diagram of ward layout (not to scale)...353

Acknowledgements

I acknowledge the integral and invaluable contributions made by a number of individuals during the completion of this research study and thesis.

Firstly, I would like to thank the ward community who were at the centre of this research study. Particular thanks are extended to the patients and families who gave their consent for their own care to become the focus of observations. And to the occupational therapists and physiotherapists who enthusiastically contributed to the study in its entirety.

Alongside those who were directly involved, thanks also go to the wider team and the clinical and service managers who created an open culture for observational research to take place.

Thanks to my supervisors, Dr Katherine Baker and Dr Cathy Bailey who have maintained unwavering belief in the project, and in me.

Finally, thank you to Jim, Mam, Kate, Sam, Molly and Anna.

Author's declaration

I declare that the work contained in this thesis has not been submitted for any other award and that it is all my own work. I also confirm that this work fully acknowledges opinions, ideas and contributions from the work of others.

Any ethical clearance for the research presented in this thesis has been approved. Approval has been sought and granted by the Faculty Ethics Committee / University Ethics Committee on 17th July 2015 and the NHS National Research Ethics Committee on 11th November 2015.

I declare that the Word Count of this Thesis 82,462

Name: Gemma Bradley

Signature:

Date: 30th August 2018

CHAPTER ONE: INTRODUCTION TO THE THESIS

The focus of this study is to explore the reasoning and decision-making of health professionals related to the assessment of rehabilitation potential of older people and the recommendation of subsequent rehabilitation pathways. This was initially motivated by both clinical and academic interests; by my desire to understand a reasoning process I had myself engaged in as a clinical occupational therapist; alongside an educational desire to prepare future health professionals. I was further inspired by an awareness of the imperative need for knowledge generation and critical thinking in relation to care and services for older people with complex health needs.

1.1 My position – influences on the early development of the study

The concept of reflexivity recognises that the orientations of the researcher are shaped by their own values, interests and socio-cultural histories and that this biography will permeate throughout enquiry of a qualitative nature (Hammersley & Atkinson, 2007). Making these constructs explicit is an essential part of the research process and this therefore, is the first of several sections which attempts to recognise the orientations of the researcher – at this early point, to recognise the lens through which the research was originally conceptualised.

I am a qualified occupational therapist. I practiced clinically within the NHS, mainly in roles providing services for people following stroke (in inpatient acute and rehabilitation services, alongside a community stroke team), before moving to my current role as a senior lecturer in occupational therapy where I have been for 11 years. My main responsibility in my current role is the delivery of education for pre-registration

occupational therapy students, although I do contribute to teaching students from wider health professional programmes.

I am also female, aged in my late thirties, and of White British origin. Although it was difficult to find more recent demographical information, statistics published in 2006 (Yar, Dix & Bajekal) report that at that time, women accounted for 80% of the healthcare workforce and were the largest gender in all groups except for medicine and dentistry; 68% of the workforce were aged between 35 and the state pension age; and 90% of the workforce were white. Whilst there may have been some changes in the past decade, it is hard to imagine dramatic shifts in these dominant groups. Therefore the fact that I 'fit' with the norms of a healthcare role is important to note and may have helped with acceptance and research positions in this setting.

My original interest in decision-making and rehabilitation potential grew out of my personal experience as a clinical occupational therapist. I would frequently be asked questions about a person's rehabilitation potential, often for patients following a stroke, and often at an early point in their care and pathway. I would also often finding myself asking other professionals about the rehabilitation potential of a patient, or writing the expression in clinical records, often recording statements such as 'limited rehabilitation potential'.

There were many issues which challenged me. I had often only known a patient for a short time before I was asked for this professional opinion. I frequently wondered whether my evaluation was similar to that of other clinicians. And I was acutely aware of some of the high stakes decisions which surrounded this often passing discussion or note in a clinical record, such as whether someone would be moved to a rehabilitation bed, referred for community interventions or perhaps quickly referred to a 24-hour care environment if the evaluation had been unfavourable. Transparency about my personal experience highlights that I was approaching this enquiry with professional knowledge

2

as an occupational therapist and with a personally held view that judgements about rehabilitation potential were linked to high stakes decisions and outcomes for patients.

I have never worked clinically within the NHS Trust or geographical area where the research took place. I do however have professional links with the occupational therapy team within this Trust – some of whom were previous students and many are practice educators for pre-registration students. I had a number of 'chance' conversations with occupational therapists from this Trust who expressed anecdotally that they were also interested in this topic and that they thought their Trust had an openness to research of this nature. I then utilised a professional contact with the occupational therapy service manager to begin to develop a partnership for the purpose of the research. Whilst constructing this partnership, I was grateful of existing relationships to help to open doors and to promote feelings of professional comfort (perhaps with some early awareness of some of the uncomfortable issues ethnographic research may provoke), but also keen to not face challenges of over-familiarity or blurring of boundaries that returning to a previous place of work may create.

Finally, in my educational role, I am regularly exposed to issues in occupational therapy practice through the sharing of stories from students and educators. With honesty, such stories are often focussed on issues of challenge and pressure within professional practice. Although occupational therapy spans many specialisms, client groups and sectors, because of my own clinical experience I have a familiarity of the experiences within the acute hospital setting and particularly for older people following an acute hospital admission and therefore perhaps tune in to such experiences in a different way. Through trying to facilitate realistic preparation for occupational therapy students, I have a particular interest in healthcare decision-making and the role that therapists can play in decision-making in acute care.

3

The position of insider or outsider will be commented upon in a later section although it is perhaps helpful to summarise that some of this history contributed to subsequent research positions. At the outset, my position could have been interpreted as an 'insider' within the discipline of occupational therapy and to an acute hospital system, although potentially an outsider to other disciplines and to current clinical practice in general.

1.2. Background

With an undisputed ageing population in the UK, there are well-reported pressures related to unplanned acute hospital admissions, relating to significant reduction in bed numbers alongside simultaneous increases in patients treated (The Kings Fund, 2017). Admissions for older people account for the vast majority of hospital emergency bed days, with 62% of all hospital bed days in 2014-15 occupied by people over 65, and 25% occupied by those over 85 (The National Audit Office, 2016). Bed use for this age group has also seen the greatest increases in the last ten year period (NHS Digital, 2017). Therefore, understanding pathways for this patient group is of understandable significance.

Because of reducing bed capacity and increasing demand, alongside major reconfigurations in long term hospital-based care, there has been particular emphasis on the need to understand the flow of patients through systems (The Kings Fund, 2013; 2017). The Kings Fund acknowledge that efforts to make effective use of bed capacity should focus on the relatively small numbers of older people who often stay in hospital for a long time (2017). The use of non-emergency and community services to reduce demand on the emergency system in a timely and efficient way is also receiving important attention. Older age is associated with an increased likelihood of living with multiple chronic conditions and a growing number of functional and cognitive impairments (The Kings Fund, 2012), with studies suggesting that about half of the population aged over 75 will live with three or more chronic conditions (Anderson, 2011). Older people are therefore likely to be admitted to acute hospitals for wide-ranging and often co-existing medical, surgical, psychological and social needs, not easily remedied by one, disease-specific treatment or pathway.

For emergency admissions, average length of stay in hospital for people over 65 is 11.9 days, in comparison to a population average of under 5 days (The National Audit Office, 2016; NHS Digital, 2017) and decisions about readiness and appropriateness of transfer of care to services who can meet ongoing needs are often complex. Evidence does however suggest that the way in which hospitals are organised can influence admissions, length of stay and readmissions, such as the availability of senior physicians (White, Armstrong & Thakore, 2010), dedicated assessment wards (Cooke, Higgins & Kidd, 2003), timeliness of therapist assessments (Jasinarachchi et al, 2009) and patient and carer involvement in decision-making (Henderson et al, 2011).

Evidence suggests that up to 35% of older people experience a deterioration in ability to undertake activities of daily living between baseline and hospital discharge, with 12% of these people experiencing this reduction between the time of admission and the point of discharge (Covinsky et al, 2003). Older people can often have extreme responses to even the most straightforward reasons for hospital admissions, are at risk of functional decline during and after hospital admissions, and often require a longer recovery phase of care (NHS National Institute for Health Research, 2017). A rehabilitative phase of care therefore plays a pivotal role in helping people to live and recover well, whether this be in hospital, in people's homes or in other care environments.

5

The recognition of the pivotal role that rehabilitation plays in maintaining quality of life and reducing dependency in older age, alongside reducing pressure on acute services and cost-containment across health and social care, means maximising the use of these services is of central concern (Ward et al, 2008). However, in a climate of reduced beds and increasing demand, it is the complex reasoning informing the allocation of finite rehabilitation resources which is central to this project. Burton et al (2015) suggests that it is a reality that 'healthcare professionals will have to continue to allocate resources in ways that balance political, ethical and service perspectives around their perception of a patient's **rehabilitation potential** (p1956).

1.3. Overview of policy and legislation

In 1997, the Audit Commission highlighted that investment in preventative and rehabilitative services was not adequate and that this was contributing to excessive unplanned hospital admissions and also to premature transfers to long term care. A report three years later (Department of Health (DH), 2000) suggested similar trends, and suggested that approximately one fifth of bed days were being inappropriately used simply because of a lack of alternatives or availability of supporting services.

Following this, the National Service Framework for Older People (NSFOP; Department of Health, 2001) was a key policy document in promoting rehabilitation services for older people and recognised the significance in particular of a range of services developed under the umbrella of 'Intermediate Care'. Intermediate Care was proposed as a new layer between primary and secondary health care services to help prevent unnecessary hospital admissions, promote timely and, if possible, early discharge, and reduce or delay the need for long term institutional care (Department of Health, 2001). The NSFOP also encouraged a range of providers (including health, local authority and independent providers) to develop integrated ways of working and contributed to the emergence of many models of service delivery. Recent National Institute for Health and Care Excellence (NICE) guidelines on 'Intermediate Care including Reablement' were published in 2017, which continued to place emphasis on the importance of services to deliver rehabilitation and support following hospital admissions, alongside proactively preventing unnecessary admissions to both hospital and 24-hour care environments.

Since the first publication in 2010, the annual NHS Outcomes Framework is the way by which the government monitors key health outcomes and how the NHS is performing (NHS Digital, 2018). More recently, it has also introduced similar Outcomes Frameworks for Social Care and Public Health. The 2018 NHS Outcomes framework places continued emphasis on helping older people to recover and live well following hospital admissions through measuring the proportion offered rehabilitation following acute admissions and measuring the proportion still at home following provision of reablement and rehabilitation services (NHS Digital, 2018). The similar Outcomes Framework for Adult Social Care also places importance on measuring the extent to which the need for long term care is reduced through the provision of reablement, and monitoring delays in transfer from hospital which are attributable to adult social care (Department of Health and Social Care, 2018). Whilst introduced with the aim of moving away from performance targets to establishing priorities and measuring national outcomes, the Outcomes Framework continues to require reporting of a wide range of data at a local, organisational level, such as adverse events, length of stay and mortality rates.

Consideration of how services are paid for is the focus of ongoing and significant attention, largely attributable to the passing of The Health and Social Care Act (Department of Health, 2012). However, the issues are far from new and can be traced back to the origins of Payment by Results (NHS Plan; DH, 2000) and the move to Foundation Trust status leading to scrutiny of financial viability. The Health and Social Care Act provided a comprehensive legislative framework to commence far-reaching reforms across the NHS. It outlined the evolution of a framework for restructuring how services are commissioned and paid for, and introduced Clinical Commissioning Groups (CCGs) to determine local health needs and to commission services and manage budgets accordingly. Included within this modernisation was the ongoing development of national tariffs attached to particular services and episodes of health care. Although much of this is still emerging, the Act created an environment where tariffs for episodes of rehabilitation and new providers of rehabilitation services could emerge. As older people are the largest users of NHS services, impact on services delivered for older people is inevitable.

The needs of older people with multi-morbidity are increasingly being recognised in contemporary policy and professional guidance. The condition of 'frailty' is now a frequently cited condition of modern ageing, defined as a condition which 'develops as a consequence of age-related decline in multiple body systems, [resulting] in vulnerability to sudden health status changes triggered by minor stress or events such as an infection or fall' (NHS England, 2014). Over recent years, many guidance documents have been published highlighting the specialist needs of those living with frailty and including recommendations on managing needs in primary, secondary and social care (NHS England, 2014a; NHS National Institute for Health Research, 2017). Notably, the routine identification of frailty is now part of the General Practice Contract for 2017/18 (NHS England, 2018), recognising that assessing and discussing frailty, and the associated risks for the person, are most effective before reaching advanced stages and before people are acutely unwell.

NICE guidance on clinical assessment and management of multi-morbidity (2016a) also recognise the complex needs of people who live with multi-morbidity and provides recommendations for management in primary care, alongside comprehensive, integrated, assessments during hospital admissions. This guidance also emphasises

8

the importance of establishing priorities and goals to commence integrated planning before multi-morbidity reaches advanced stages of frailty.

The Care Act (Department of Health, 2014) outlined reforms to the provision of social care for adults and older people. The Act places responsibility on local authorities to arrange service provision which assists to prevent or delay the need for care and support and to ensure social care and health services are integrated to promote health and wellbeing and to reduce or delay the need for services. It also highlights the role of local authorities in helping people regain skills, for example after a hospital admission. Considering rehabilitation as a strategy to facilitate improvement, prevent decline and reduce dependency, this places joint responsibility for rehabilitation between health and social care and emphasises the importance of integrated models of service delivery.

Service user involvement in decisions about their care is a professional requirement (Health and Care Professions Council (HCPC), 2013) and is emphasised within current national policy drivers (DH, 2010). More specifically, shared decision making is advocated by the National Institute of Health and Care Excellence, emphasising that users of adult NHS services should be actively involved, with the support of health professionals to make fully informed choices about their treatment and care (NICE, 2012).

To summarise, current policy and strategic drivers recognise the importance of services and pathways for frail older people, including rehabilitation services which play an integral part in maintaining quality of life and reducing dependence on care services. However, the policy context also illustrates the complexity of service provision with multiple funding streams, providers and models of service delivery. How practitioners are navigating the complexity which exists at this macro level, in order to make decisions about which individual patients might benefit from different rehabilitation services at a local level is a motivator for this study. The background of financial

9

scrutiny and changes to tariffs and commissioning adds additional layers to this area of enquiry.

1.4 Research Aim and Objectives

Developed at an early stage of this research for the purposes of the proposal and the necessary approvals, and refined through engagement with literature, policy and key stakeholders in the field, the research aims and objectives are outlined below:

1.4.1. Research aim

To explore health professional's reasoning and decision-making related to the assessment of rehabilitation potential of older people in hospital and recommendation of subsequent pathways.

1.4.2. Research objectives

- To explore the meaning of the concepts of 'rehabilitation' and 'rehabilitation potential' to occupational therapists, physiotherapists and patients

- To map the reasoning process of health professionals involved in evaluating rehabilitation potential and when making decisions about rehabilitation pathways for older people

- To compare the occupational therapy and physiotherapy role in evaluations of rehabilitation potential and decisions about rehabilitation pathways

- To describe influences on decisions regarding rehabilitation potential and subsequent pathways

- To explore how patients and families are involved in evaluations of rehabilitation potential and subsequent recommendations regarding treatment and care.

1.4. Outline of subsequent chapters

Chapter Two presents an overview of literature in relation to the main conceptual elements of the study – namely rehabilitation, rehabilitation potential and decision-making – all framed within the context of unplanned and rehabilitation pathways for older people. This chapter illuminates existing knowledge and current gaps.

Chapter Three introduces the methodological approach taken, including a presentation of the theoretical underpinnings, justification for the approaches and any alternative positions considered. This position is then utilised to explain, in detail, the methods used to gather and analyse data, to ensure ethical research practice and to communicate the consideration of research rigour and quality.

Chapter Four is the first of four chapters to present the research findings. This first chapter, entitled 'Descriptions', presents a thick descriptive account of the research environment and the roles, systems and practices which surround health professional decision-making. The chapter concludes with a descriptive account of the decision-making process to evaluate rehabilitation potential and reach decisions about rehabilitation pathways.

Chapter Five is the second research findings chapter and is entitled 'Social Construction and Meaning-Making of Rehabilitation and Rehabilitation Potential'. It

11

presents the meanings of concepts which are central to the research study and how these concepts were socially constructed within the setting.

Chapter Six is entitled 'The Ethical Dimensions of Rehabilitation Potential'. This chapter discusses the ethical nature of rehabilitation decision-making which emerged as significant within the analysis of findings.

Chapter Seven is entitled 'Professional Roles' and this chapter concludes the presentation of research findings. In line with the research objectives, the professional roles of occupational therapists and physiotherapists are specifically examined within the final section, with data utilised to explore similarities, differences and issues pertinent to these roles. Within chapters four-seven, participant quotes, extracts from records and observations from fieldnotes are given the important emphasis that they richly deserve.

Chapter Eight presents a critical and interpretive discussion of the findings in light of wider literature, policy and theoretical frameworks. It illuminates the ways in which meanings of rehabilitation concepts are shaped by professionals, patients and wider society; offers explanatory insights into how people are acting towards such meanings and; how this is subsequently shaping contemporary health services and professional practice.

Chapter Nine concludes this research study. It firstly revisits the findings in line with the revised aims and objectives and uses this to illustrate the unique contribution to knowledge. This section goes on to discuss implications for practice – including those that could be directly influenced by professionals, alongside wider implications for health policy, strategy and research. The chapter also recognises strengths and limitations with the current study and provides recommendations for future research.

1.5. Writing Conventions

All data has been edited throughout to remove idiosyncratic details and preserve anonymity. In relation to health professional participants, they are simply identified by their role (e.g. one physiotherapist expressed that...) or by an interview participant number (e.g. I1). Where idiosyncratic details, such as gender or grade, had the potential to identify individuals, this information has been omitted. This was important in the context of such a localised setting with a small number of participants.

The term 'patient' is used throughout to represent a person receiving care and services. Critical consideration was given to other terms such as service-user, client and customer, all of which are perhaps reflective of a more contemporary and active position for this group in today's health and care services. However, the word patient was an accepted part of language and discourse within the research environment and therefore the use of this term within this written account was a deliberate choice.

Finally, and as identified within the glossary, the term 'health professional' is used interchangeably with 'practitioner' to refer to employed members of the healthcare team including nurses, doctors, therapists and social workers. Where it has been important to attribute extracts or perspectives to particular professional roles, this has been clearly identified. In other places, the term health professional or practitioner may have been used to reflect issues affecting groups as a whole, or to maintain anonymity if this was an area of concern. I have avoided abbreviating the names of health professional roles in line with current writing conventions (such as 'OT' and 'PT'), except to indicate exact terms used by participants (for example the terms used in verbatim quotes and clinical records) or in places where this has assisted with clearer formatting (such as tables or appendices).

CHAPTER TWO – LITERATURE REVIEW

2.1. Chapter introduction

This literature review presents current knowledge and understanding relating to three overarching concepts underpinning this research: rehabilitation; rehabilitation potential; and rehabilitation decision-making; all discussed in the context of older people.

Searches were undertaken using Northumbria University Library Search, Cinahl, Medline and Google Scholar using the search terms outlined in Table 1. Hand searches were also undertaken and relevant literature was utilised to further understand definitions and policy context. In relation to 'rehabilitation potential', there was a paucity of evidence particularly pertaining to older people and therefore the search was broadened to examine the concept in relation to wider fields.

Rehabilitation	AND			Older (old*, older	
(rehab*; reablement)				people, elderly,	
				frailty, geriatric*)	
Rehabilitation	AND			Older (old*, older	
potential (rehab*				people, elderly,	
potential; potential)				frailty, geriatric*)	
Rehabilitation potential (rehab* potential; rehabilitation AND potential)					
Decision making (decision-making; decision; evaluation; judgement; reasoning)	AND	Rehabilitation (Rehab*; reablement) OR Rehabilitation potential (rehab* potential; potential)	AND	Older (old*, older people, elderly, frailty, geriatric*)	

Table 1: Literature search strategy and search terms

2.2. Rehabilitation for older people

2.2.1. Defining rehabilitation

Defining and describing rehabilitation is a complex challenge, and indeed the term rehabilitation, when considered alongside specialisms such as stroke, spinal injury, hip fracture or heart failure, results in the emergence of many different descriptions and meanings. The term can also be paired with terms such as vocational rehabilitation, cardiac rehabilitation, sports rehabilitation and offender rehabilitation, amongst others, and again, even wider variances in descriptions and models add to the complexity of the concept.

The World Health Organisation (WHO), in their recent publication entitled 'Rehabilitation in Health Systems' (2017) define rehabilitation as 'a set of interventions designed to optimise functioning and reduce disability in individuals with health conditions in interaction with their environment' (p11). This document goes on to outline that the purpose of rehabilitation is to maximise people's ability to live, work and learn to their optimum potential, and that functional difficulties associated with ageing can also be reduced through rehabilitation.

The 2017 WHO definition is broadly similar to the earlier WHO definition of rehabilitation integrated within the influential International Classification of Functioning, Disability and Health publication (ICF; WHO, 2001). This framework defined rehabilitation as 'a health strategy that aims to enable people with health conditions experiencing or likely to experience disability to achieve optimal functioning in interaction with the environment' and importantly to this project, was a definition adopted by physical rehabilitation medicine (Stucki, Cieza & Melvin, 2007). The ICF also provides insight in to the purpose and focus of rehabilitation; with a focus not on cause but on impact of impairment, and the need to address elements at the level of

impairment and dysfunction, but also at the level of accessibility of activities, environments and societies.

Other definitions focus more on a rehabilitation process with core components. Wade (2005) defines rehabilitation as an "educational, problem-solving process that focuses on activity limitations and aims to optimise patient social participation and well-being, and so reduce stress on carer/family" (p814). An alternative definition proposes rehabilitation as a "goal-orientated and time-limited process aimed at enabling an impaired person to reach an optimal mental, physical and/or social functional level thus providing him or her with the tools to change his or her own life" (United Nations, 2009). Hammell (2006) defines rehabilitation as "a process of enabling someone to live well with an impairment in the context of his or her environment, and as such, requires a complex, individually tailored approach" (p127).

With definitions of rehabilitation presenting broad-brush statements, it is perhaps the elements which go in to rehabilitation which give the concept meaning. Authors have recognised this, suggesting that recognising the 'active ingredients' of rehabilitation may make it easier to understand or research (Whyte, 2009). Examples within literature also make reference to a 'black box' of rehabilitative interventions, suggestive of difficulties characterising what it actually is (DeJong et al, 2004; Whyte & Hart, 2003). Some attempts aim to summarise the different activities such as activities to promote musculoskeletal, cardiopulmonary, or cognitive and perceptual improvements (DeJong et al, 2004). Other examples focus more on the delivery of rehabilitation, highlighting the centrality of patient goals, a multi-professional approach, the allocation of resources to hospital and community models, and collaboration with patients and family (Wade, 2015; Wade 2005; Whyte, 2009; WHO, 2017).

To add complexity to the many contexts and specialisms, the term rehabilitation is also used to describe an underpinning approach or theoretical basis, based on beliefs and

16

models of health, illness and disability, alongside theories of behaviour change (Wade, 2005). From a public and population health perspective, it is a term used to describe a health strategy with prevention, cure and support cited as other inter-related health strategies (Stucki, Cieza & Melvin, 2007). Rehabilitation research further hinders the quest for clarity with recognition that research evaluating and exploring rehabilitation themes does not always specify the activities involved and often it includes large numbers of multi-faceted interventions grouped together ambiguously (Wade, 2005).

2.2.2. Models of rehabilitation for older people

Models of specialist rehabilitation, for treatment of conditions such as stroke, hip fracture and pulmonary and cardio-vascular diseases are well accepted and established. However, it is recognised that for older people with specialist needs associated with ageing, models of care and service provision, alongside the impacts and effectiveness of rehabilitation, are less clear (Bachmann et al, 2010). Despite this, demand for rehabilitation for older people is significant, linked to demographic patterns, changes over time in orientation of acute healthcare services, and a growing evidence base that rehabilitation is an integral component of restoring health and wellbeing after periods of illness or injury.

Rehabilitation service delivery models include the use of inpatient beds in rehabilitation wards and units; rehabilitation provided in identified residential step-down units or within residential and nursing homes; community rehabilitation in the person's home; and, more recently, examples of tele-rehabilitation which could involve interventions provided through information and communication technologies. Also, the provision of rehabilitation is not just limited to wards or services identifying themselves as providers of rehabilitation, and is more realistically being delivered for older people across general wards and teams and by many and varied providers of health and social care
services. These models also have diverse funding and commissioning arrangements and can be delivered by NHS, Adult Social Care and private providers.

The NSFOP (Department of Health, 2001) received criticism for a lack of detail and reference to an evidence base relating to effective models of service delivery, and in particular, a lack of clarity about the term 'Intermediate Care' (Melis et al, 2004). In the 15 years since the implementation of the NSFOP, there have been many exemplars outlining how services have been developed and evaluated (Young et al, 2007; Cunliffe et al, 2004; Griffiths et al, 2007) although this has also resulted in challenges to practitioner decision-making about which services may best meet individual needs, and challenges to commissioners and policy makers to understand the landscape and evaluate which services offer clinical and cost effectiveness.

Furthermore, it could be argued that more recent policy and legislative developments have only continued to foster diversity in services through the encouragement of new providers (for example private companies and charities) and through the joint responsibility placed on Health and Social Care. The latest guidance on Intermediate Care from the National Institute of Health and Care Excellence (NICE, 2017), continues to recognise different types of intermediate care, although reinforces four broad categories including crisis-response, home-based, bed-based and reablement. Bed-based intermediate care and reablement services are of particular note to this study (with descriptions of the services experienced during fieldwork provided in Chapter four).

Reablement has been defined as 'assessment and interventions provided to people in their home (or care home) aiming to help them recover skills and confidence and maximise their independence. [It is] delivered by a multidisciplinary team but most commonly by social care practitioners' (NICE, 2017; pg 15). Parker (2014) recognises that there are multiple challenges in understanding the differences between

terminology such as reablement, rehabilitation and intermediate care although the time limitations, the difference between health and social care provision and the focus on adjustment rather than regaining skills represent some ways that services are differentiating at a local level. It is interesting to note that difference and overlaps are not clearly explained in recent NICE guidance (2017) perhaps representing a missed opportunity to develop clarity on this important issue.

The prevalence of cognitive impairment increases with age (Ray & Davidson, 2014) and is therefore of importance to a study which focuses on older people. In particular, recent estimates suggest that one or more of dementia, delirium or a non-diagnosed cognitive impairment as indicated by medical screening was present in almost 40% of all patients aged over 65 admitted to a general hospital, and rises to over 50% in those aged over 85 (Reynish et al, 2017). Estimates have also suggested that approximately 60% of people receiving care services in their own homes are living with cognitive impairment, and this figure is almost 70% of those living in residential care (United Kingdom Homecare Association, 2015; Prince et al, 2014).

Although the term cognitive rehabilitation is a recognised term used to describe nonpharmacological methods focussing on identifying and addressing needs and goals for those with cognitive impairment (Bahar-Fuchs, Clare & Woods, 2013), synthesising messages and implications from available evidence is complex. Definitional challenges exist with terms such as cognitive training and cognitive interventions often used interchangeably within literature discussing rehabilitation focussed on remediating the effects of, or adjusting to, cognitive impairment (Bahar-Fuchs, Clare & Woods, 2013). Once again, many examples also discuss cognitive rehabilitation in relation to specific pathways such as stroke and acquired brain injury (das Nair et al, 2016; Kumar et al, 2017). In a recent Cochrane systematic review of enhanced models of rehabilitation for those with cognitive impairment following hip fracture, only five studies were included, the studies were small and were deemed to be of low quality and none assessed the

primary outcome of the influence of the model of rehabilitation on cognitive function, instead exploring the impact of the cognitive impairment on the improvement of physical function (Smith et al, 2015).

To summarise, integrated models of rehabilitation which support older people to recover from and adjust to periods of ill-health, new onset of health conditions, or trajectories of ageing and frailty which may result in wide-ranging cognitive and physical impairments are not clearly described or supported within literature, although are supported in policy (NICE, 2017). The picture becomes particularly blurred when considering that, within research, measures of cognitive impairment are highly contestable and there are many examples where people with cognitive impairment or excluded from studies (Crocker et al, 2013). Furthermore, studies which specifically address or explore models of rehabilitation for those with cognitive impairment report outcomes often relating to physical improvements. Issues of particular relevance to this study, such as the link between cognitive function, rehabilitation potential and rehabilitation outcomes are integrated within other sections.

2.2.3. Effectiveness of rehabilitation services for older people

As outlined in the previous section, there are many different ways in which rehabilitation can be provided for older people. Even with the acknowledgement that some models of rehabilitation for older people are condition-specific and will therefore contribute to the variation (for example stroke rehabilitation, pulmonary rehabilitation and cardiac rehabilitation among others), there are still geographical differences in the availability of slower-stream rehabilitation beds, the availability and organisation of intermediate care facilities, and the capacity of community rehabilitation or outpatient provision (The Kings Fund, 2012). Although, in line with an agenda to commission services based on the needs of local populations (The Health and Social Care Act, 2012), the variation provides a challenge to practitioners, commissioners and policymakers to understand efficacy.

In line with many condition-specific delivery models, much of the research attempting to evaluate effectiveness of rehabilitation is condition-specific – for example the systematic reviews from the Cochrane Collaboration evaluating home-based rehabilitation in comparison to centre-based rehabilitation for cardiac conditions (Anderson et al, 2017), and assessing the effectiveness of models of rehabilitation following joint replacement (Khan et al, 2008). Research also reflects the fact that this in an international topic of interest although models of service delivery (mainly linked to how services are funded) are significantly different. However, there is general consensus that periods of organised, multi-professional rehabilitation are effective in promoting health and wellbeing benefits such as positive functional outcomes and reducing dependency on carers.

There are some examples of studies attempting to synthesise results from published research particularly relating to the provision of rehabilitation for older people. The Cochrane Collaboration attempted to compare rehabilitation for older people provided in care homes, hospitals and own home environments (Ward et al, 2008). The review of international published research suggested that there was insufficient evidence to compare the effects of rehabilitation in care home environments, hospital environments and own home environments and own home environments is and own home environments to a lack of clarity in descriptions and specifications of rehabilitation and the weaknesses in existing research, rather than an ability to establish no effect.

The Cochrane Collaboration also led a review aiming to determine whether nursing-led inpatient intermediate care units are effective in preparing patients for discharge in comparison to usual inpatient care (Griffiths et al, 2007). There was some evidence that patients who received care within the nursing-led unit were better prepared for

discharge although this could have been related to a near significant increase in the average length of stay for these patients. Also, although there were no statistically significant adverse events between the groups, the review did suggest that they could not rule out an increase in early mortality in relation to nursing-led intermediate care. A further Cochrane Review comparing day hospital care for older people versus alternative forms of care (Brown et al, 2015) concluded that there is low quality evidence that medical day hospitals appear effective when compared with no care, although there is no clear advantage over other medical care provision.

A large systematic review and meta-analysis of randomised controlled trials (RCTs), including 17 trials with a total of 4780 participants, evaluated the effectiveness of inpatient rehabilitation specifically designed for geriatric patients in comparison to usual care (Bachmann et al, 2010). In order to meet inclusion criteria for this systematic review, the rehabilitation provided needed to be multidisciplinary in nature (with a minimum of a geriatrician and a nurse) alongside including active physiotherapy, occupational therapy or both, and meet the principles of the ICF Rehabilitation Framework (Stucki, Cieza and Melvin, 2007). The authors noted issues related to methodological quality with many of the studies, for example only 10 of the 17 included studies reported adequate concealment of allocation and only seven studies blinded the independent outcome assessors. Despite this, the authors concluded that a period of specialised geriatric rehabilitation showed beneficial effects over usual care in that it was more likely to result in improved functional outcomes and less likely to result in a nursing home admission or in death either during the admission or at follow up.

However, when looking beyond the headlines of positive generalised benefits, there were only two areas where statistical significance was demonstrated; inpatient orthopaedic rehabilitation demonstrated significant improvements in functional outcomes in comparison to usual care; and studies with a lower mean age demonstrated statistically significant reduction in admissions to nursing homes than

those with a higher mean age. Both of these statistically significant findings are perhaps unsurprising. Furthermore, nine of the 17 studies related to specialist orthopaedic geriatric rehabilitation, with eight relating to the more heterogeneous geriatric rehabilitation which forms the basis of this study. With only four of the original studies carried out in the United Kingdom (UK) and the research publications dating from 1984-2007, the likelihood of fast-paced change to rehabilitation services and pathways within contemporary UK healthcare systems raises further questions about currency and transferability.

Despite questions about statistical significance and transferability of findings, the scale of the review by Bachmann et al does lend credibility to some of the wider observations of the study and the authors. Firstly, although all services needed to include a geriatrician and a nurse, there was wide variance in the make-up of the rehabilitation team meaning understanding in relation to the components of effective rehabilitation for older people continues to represent a challenge. Secondly, the review only identified studies pertaining to general geriatric rehabilitation or orthopaedic geriatric rehabilitation suggesting that rehabilitation targeted at particular health conditions but specialising in the needs of older people is either not being provided, or perhaps more realistically, not being researched.

Studies have also attempted to evaluate rehabilitative care for older people in community hospitals by comparing outcomes with standard care in a general hospital (Young et al, 2007). Patients who received rehabilitation in the community hospital had slightly higher independence outcomes after six months, although length of stay, mortality, discharge destination and patient satisfaction were not statistically different. A linked study which presented an economic evaluation using the same participant data, suggested there was no significant difference in cost (O'Reilly et al, 2008). The results from both studies suggest there could be small improvements in functional outcomes when rehabilitation is provided in a community hospital although many other outcomes when compared to rehabilitation provided in a general hospital were similar. It is perhaps the absence of significant differences which provide some of the key messages here with locality-based, closer-to-home care providing a meaningful alternative to acute general hospital care, with no suggested adverse effects or increased risks.

Cunliffe et al (2004) conducted a RCT including 370 patients to compare an early discharge rehabilitation team with usual care (management in hospital until fit to go home) for older people. They found that health outcomes, measured by activity limitation and psychological wellbeing were improved in the short term for those in receipt of rehabilitation from the early discharge team, with improvements also noted in carer wellbeing and no differences noted in adverse events such as readmissions or mortality rates. The authors recognised that it was potentially the organised, person-centred rehabilitation approach promoted by a team who were aware of the research trial, rather than the superiority of a home setting over a hospital setting, which could contribute to favourable results. They also recognised the resources which accompany a large RCT and that the benefits of early discharge rehabilitation may be lost if the realities of resource constraints compromise services.

2.3. Rehabilitation potential

2.3.1. Definitions of rehabilitation potential

The NSFOP (2001) recognised that 'good management' of an older person following a hospital admission would include attention paid to rehabilitation potential. Rehabilitation potential has been defined as an 'estimate of the individual's capability of cooperating with a rehabilitation programme and making measured functional gains' (Rentz, 1991). Zhu et al (2006) define rehabilitation potential retrospectively if a person has made

functional improvement or remained at home over a period of one year and discuss that an effective assessment of rehabilitation potential relates to the selection of individuals who are most likely to benefit from rehabilitation. In a study by Burton et al (2015) professionals consistently described the concept of rehabilitation potential referring to two main elements - the visible achievement of goals or outcomes over time and the observation of carry-over within and across therapy.

It has been suggested that the clinical assessment of rehabilitation potential is an everyday occurrence in hospitals (Cunningham, Horgan and O'Neill, 2000; Burton et al, 2015) yet definitions and explanations to support practitioner understanding and decision-making within literature are at times vague. Shun et al (2017) attempt to explore factors influencing perceptions of rehabilitation potential but do not clearly define how they are interpreting this foundation concept. Other studies allude to judgements about rehabilitation potential with limited explanation of what this means or entails (Kumlien et al, 1999; Kotiadis, Carpenter and Mackenzie, 2004). For example, a study by Arling, Williams and Kopp (2000) explores therapy use and discharge outcomes for elderly nursing home residents and states the exclusion of people who are deemed to have conditions which would limit rehabilitation potential but provides limited detail or examples.

Enderby et al (2017) published an expert commentary on the meaning and value of the term 'rehabilitation potential', involving professionals with expertise in stroke rehabilitation. The article raised critical insight in to the unclear nature of the concept, alongside the lack of recognised decision-making tools or algorithms. Critically, and perhaps to illustrate the ambiguity, the article does not offer a definition of the central concept.

2.3.2. The significance of rehabilitation potential

The definition above from Zhu et al (2006) relating an effective assessment of rehabilitation potential to the selection of individuals who are most likely to benefit from rehabilitation suggests a clear link between a judgement about rehabilitation potential and the allocation of rehabilitation resources. Burton et al (2015) discusses this in further detail suggesting that, at an individual level such decisions can determine when and if rehabilitation begins, the intensity of rehabilitation required and at what point rehabilitation may fail to deliver meaningful outcomes. Burton et al recognised that at an organisational level, those thought to benefit most from services can be prioritised and those with limited potential can be tracked to the most appropriate environment. They also highlighted other, more subtle, reasons why decisions about rehabilitation potential are of significance, recognising the emotional significance for those health professionals with a responsibility for making such decisions; a process which potentially involves labelling or putting people in boxes; and an awareness that decisions about potential can become self-prophesying.

2.3.3. Judgements about rehabilitation potential

Burton et al (2015) used multi-professional focus groups to discuss a hypothetical case scenario with the aim of encouraging discussion about judgements of rehabilitation potential following stroke. Participant descriptions suggest that judgements about potential tend to emerge from observing responsiveness to therapy, even through potential failure or poor outcomes, rather than from predictor variables prior to starting rehabilitation. Indeed in simple terms, this study suggests that rehabilitation needs to commence before meaningful decisions about potential can emerge.

As there are only limited examples of studies exploring health professional judgements of rehabilitation potential, yet such studies are of direct relevance to this research, it is important to attend to the limitations. Data generated through the focus groups emerged from discussion relating to a fictional scenario and the health professionals involved were from a small number of services in a geographically close area of the UK. However, these limitations add weight to researching the concept in another geographical location, using a method which explores real-time decision in context, and exploring the transferability of findings to broader clinical contexts outside of stroke services.

In a recent qualitative study in Australia, Shun et al (2017) recruited 12 occupational therapists to participate in a consensus building data collection method to identify the most important patient-related factors when considering rehabilitation potential in people following acquired brain injury. From an original long-list of 51 items, the group achieved consensus on 11 factors as essential to consider when evaluating rehabilitation potential: age, behaviour, cognitive abilities, endurance, home environment, medical status, observed improvement in acute care post-injury, physical abilities, post-injury functional status, pre-injury functional status, and patient and family expectations. However, alongside these patient-related factors, other themes emerged from focus group discussions including the organisational context (such as time and resource pressures), professional expertise, experiential knowledge, knowledge of scientific evidence and ethical considerations. Findings present as being consistent with other studies which aim to explore factors predictive of functional gains (presented in section 2.4.2), although again hypothetical cases were utilised to explore professional reasoning.

Cunningham, Horgan and O'Neill (2000) explored the clinical assessment of rehabilitation potential of older people admitted to a geriatric rehabilitation ward, by a physician, a nurse, a physiotherapist and an occupational therapist. The findings

suggested that professionals were often unsure of their own judgement about rehabilitation potential and agreement between professionals was poor. Alongside differences in approaches to assessment, the authors suggest differences in the meaning of rehabilitation potential as a contributing factor to the low levels of agreement although do not go beyond raising this as an issue for professional reflection and dialogue.

Chang et al (2011) compared judgements about rehabilitation potential amongst care home residents in Taiwan and employed caregivers involved in their care (either qualified nurses or carers). The findings found significant differences between perceived rehabilitation potential rated by residents in comparison to their caregivers. Whilst recognising the cultural and contextual differences which inevitably lead to challenges in generalising findings from this study, a study which explores a person's own perception of rehabilitation potential and the potential differences in judgements in comparison to those making decisions or responsible for taking actions based on perceived potential is worthy of note. Reasons for differences in perceptions were potentially attributed to residents being too optimistic about their potential, or caregivers being too passive in their evaluation rooted in previous frustrating experiences or ideas of cost containment.

A study by Myers et al (2009) provides insight in to a simple attempt to quantify rehabilitation potential. On admission to one of 14 rehabilitation facilities in the USA, the admitting nurse was required to use the history of the presenting condition and past medical and social information to rank rehabilitation potential using a three-point scale and this was found to have a significant relationship with functional status at discharge. However, the sensitivity of a three-point scale and the lack of information to interpret how this three-point scale was administered limit the interpretation of such findings.

With so few studies actually attempting to articulate or explore descriptions or meanings of rehabilitation potential, literature which acknowledges the absence of rehabilitation potential could provide additional insight. Arling, Williams and Kopp (2000) studied therapy use and outcomes after 90 days of 1419 elderly residents admitted to nursing homes in an identified State in the United States and stated a judgement about rehabilitation potential as an element of the study inclusion criteria. Description of the sampling method acknowledged that participants were included in the study if they were free of serious mental or behavioural problems and had no diagnosis that might severely limit rehabilitation potential. Further descriptions of those excluded from this study included people with high grade pressure sores, chronic mental health problems and an explicit terminal prognosis, among others, suggesting an interpretation in this study that the presence of physical and mental health comorbidities reduces rehabilitation potential.

Kotiadis, Carpenter and Mackenzie (2004) present an insight in to the referral and admission criteria to Intermediate Care services in a particular locality in the UK. They discuss that rehabilitation potential was stated as part of the service admission criteria, and offer general clarification of this that a person should be able to benefit from rehabilitation. The analysis reviewed the criteria against an identified number of patients using each service and interestingly they identify that this criterion was the element that the fewest number of patients met although information about how such judgements were reached is limited. This publication was not published as empirical research and therefore lacked a rigorous discussion of concepts and methods, however, the ambiguous discussion of rehabilitation potential and discussion about judgements relating to potential are reflective of wider literature. This piece also provides an insight in to the dialogue and reasoning of professionals, where the term 'rehabilitation potential' is being used and judgements are alluded to, without a detailed understanding or acknowledgement of the conceptual challenges.

All of these issues are perhaps best summarised by returning to Enderby et al (2017) who suggest that the concept of rehabilitation potential is 'imprecise, inadequately defined and influenced by the non-clinical context'(p712) and compare it to a 'guessing game' (p709).

2.4. Rehabilitation decision-making

2.4.1. Decision-making frameworks

It is globally accepted that decision-making is a constant requirement for those involved in health service delivery, from daily frontline decisions pertaining to patient care, to strategic decisions relating to how services are organised, delivered and funded. Healthcare decision-making therefore understandably receives significant focus with explorations of how decisions are made, discussions of how decisions are best made, or variances in decision-making (particularly when thinking about novices and experts) all widely researched. Debate about how theory can be used to improve practitioner decision-making is another area of important focus (Chapman and Sonnenberg, 2000).

There are many models of professional reasoning proposing to illuminate issues pertaining to healthcare decision-making. One such model is the Integrated Patient-Centred Model of Clinical Reasoning (Higgs and Jones, 1995). This model describes reasoning as a process of reflective inquiry, involving the client, and promoting an indepth, contextually relevant interpretation of the clinical problem. The model incorporates the application and integration of three core elements: knowledge, cognition and metacognition. To summarise the three core elements, practitioner decision-making requires: a strong, well-organised knowledge base; the cognitive skills to analyse, synthesise and evaluate information pertaining to the clinical problem; and an ability to reflectively monitor the reasoning process and subsequent actions taken (Higgs and Jones, 1995).

In a later representation of the same model, Higgs and Jones offered further development to recognise an additional three influences: mutual decision-making, contextual interaction and task impact (2000). The inclusion of mutual decision-making was an acknowledgement of the growing move away from dependent patients to active and informed consumers of healthcare, a concept which has only continued to grow in significance and is explored within section 2.4.6. Contextual interaction refers to the interaction between the professionals, the situation and the environment and again reflects changes in physical environments and locations of healthcare but also of wider economic and socio-cultural factors impacting on decision-making. Finally, the task impact is included as an influence on reasoning to recognise that the nature of the clinical problem can be changeable, uncertain and multidimensional (Higgs and Jones, 2000).

Predictive reasoning is a term used to describe the process of envisioning future scenarios, including the exploration of choices and the implications of those choices (Jensen et al, 1999). More specifically to rehabilitation professionals, it has been described as envisioning future scenarios based on estimated responses to therapy and linked to judgements about the potential benefit an individual may gain from rehabilitation (Edwards et al, 2004; McGlinchey and Davenport, 2015). It is a reasoning type that is recognised within literature pertaining to nursing, occupational therapy and physiotherapy (Fonteyn and Fisher, 1995; Edwards et al, 2004; Hagedorn, 1996) and can be clearly linked to attempts to evaluate rehabilitation potential. The information potentially used to predict functional outcomes following engagement in rehabilitation will be explored later in this review.

Ethical reasoning involves the reasoning process used to attempt to resolve ethical dilemmas and to balance one or more values against another in an effort to act in the best interests of service users (Chapparo and Ranka, 2008). Ethical reasoning in health and care has received much attention in literature and research (Duncan, 2010; Seedhouse, 2017) and more specifically in relation to rehabilitation decision-making and hospital discharge planning (Bushby et al, 2015; Durocher & Gibson, 2010; Levack, 2009).

Set within a landscape of many and varied ethical philosophies and principles, two particular branches of ethical theory – deontology and utilitarianism - emerge as particularly relevant to healthcare and rehabilitation (Hugman, 2005; Barnitt, Wareby & Rawlins, 1998; Levack, 2009). Deontology is based on the assumption that ethical laws are duties and should be applied equally and universally to everyone in the same type of situation as a result of rules, and in the case of health professionals, codes of conduct, being applied (Hugman, 2005). Some have suggested that at the level of individual professional and patient interactions, deontological principles should be dominant (Garbutt & Davies, 2011).

In contrast, utilitarianism is founded on rational judgements about the consequences of actions with an aim to produce the greatest benefit for the greatest number (Hugman, 2005). It is suggested that wider healthcare systems have for a long time inevitably been driven by utilitarian forces, aiming to maximise finite resources for the greatest good, with the guidance provided by the National Institute of Health and Care Excellence providing examples to serve this very purpose (NICE, 2013). To summarise the difference, deontological ethics is driven by intention to do good, whereas utilitarian ethics supports the evaluation of likely consequences as the basis for action.

Seedhouse (2005) suggests that in relation to healthcare decision-making, there is often more than a technical and scientific process of reasoning and that health

professionals will inevitably be influenced by values when reaching decisions. In fact, he suggests values often precede an examination of evidence and dominate decisionmaking behaviour in humans. Decisions which focus on who will benefit from access to rehabilitation services, and potentially considering who will benefit *more* from such services if there are competing demands, could inevitably be driven by practitioner values.

2.4.2. Decision-making – predicting who will benefit from rehabilitation

Retrospective studies in to clinical factors which predict the likelihood of functional gains are of obvious interest. Many studies have investigated factors which may predict functional outcomes for older patients including cognitive ability (Hershkovitz and Brill, 2007; Hershkovitz et al 2010; Poynter et al, 2011), increasing age (Traballesi et al, 1998), and presence of co-morbidities (Semel 2010; Press et al, 2007). There are recurrent themes within the vast literature although findings are difficult to generalise. Many studies are specific to particular patient pathways, such as hip fracture, stroke and amputation and generalising findings to an older population who may present with a number of co-existing conditions is challenging. In fact, one study which investigated predictive factors for an older population with diverse reasons for acute hospital admission identified that the main presenting condition itself did not provide an effective indication of functional gains, discharge destination or survival (Elphick et al, 2007).

Although there is general agreement about factors such as advancing age (Traballesi et al, 1998) and the severity of cognitive impairment (Hershkovitz et al, 2010) being negatively correlated to measures of functional outcome after a period of rehabilitation, functional gains are still reported for these client groups. A retrospective study of admissions to a general hospital in the UK (Elphick et al, 2007) found that successful inpatient rehabilitation – measured by duration of rehabilitation days, destination and

level of support on discharge, hospital readmissions, and death during admission or within 120 days of discharge - of patients aged 90 and over is possible and the factors influencing this success, such as higher cognitive function and better premorbid functional ability, are similar to younger populations. They did acknowledge that the presence of co-morbidities is negatively correlated to the success of rehabilitation and that this inevitably occurs more frequently in the older elderly population. However this study supports the notion that rehabilitation potential may be affected, but not eradicated, by advancing age.

Many studies report the presence of comorbidities as being negatively correlated to good functional outcome following hospital admission although specificity about this issue within the evidence is a little less clear. A few studies do develop specificity on this theme with some detailing a particular condition such as diabetes mellitus (Semel, 2010), or offering approaches to grading severity (Press et al, 2007) although they present as being small in number in comparison to the many studies which simply identify the issue in a general way.

When considering the influence of depression on rehabilitation outcomes, there are examples of studies which report a negative impact on rehabilitation outcomes (Hershkovitz et al, 2007). However, there are also examples of studies where rehabilitation outcomes between those with and without depression do not significantly differ, with authors warning against limiting rehabilitation opportunities for those who present with low mood or increased apathy (Lenze et al, 2007).

Some studies develop the themes above to not only identify factors which are predictive of functional improvements, but factors which are predictive of eventual discharge outcome. Unsworth (2001) identified that performance on stairs, bed transfers, and eating as measured by the Functional Independence Measure (FIM) could accurately predict discharge home, to nursing care or to further rehabilitation in

74% of patients admitted to acute care for lower limb orthopaedic surgery. The same study achieved similar accuracy in predicting discharge destination of stroke patients but required wider factors such as bowel management, performance in instrumental Activities of Daily Living (ADL) alongside social factors including social situation and pre-admission housing (Unsworth, 2001).

Because of the multitude of studies which present information on this theme alongside international diversity in models of service provision, studies from the UK are of particular interest. A recent study from a large acute hospital in the UK identified that age, gender, Abbreviated Mental Test (AMT) score, walking ability outdoor and indoor, anaesthetic grading, need for pre-operative medical review and delay to surgery all had a significant effect on whether an individual returned to their previous place of residence after traumatic hip surgery (Nanjayan et al, 2014). The study proceeded to identify that where all of these factors were present in a person over the age of 80, they were one thousand times more likely to not return to their pre-operative review and delay to surgery to surgery were often linked to the presence of co-morbidities and that this, alongside poor cognitive and physical function were the most significant predictors of a change in discharge destination.

Although predictive factors are clearly of relevance, the difference between rehabilitation outcomes and rehabilitation potential has been highlighted. Burton et al (2015) suggests rehabilitation outcome is dependent on many factors including the availability and receipt of clinically effective interventions and that the response to such interventions is difficult to differentiate from the responses of the research populations on which evidence for clinical effectiveness is based. Subsequently, generalisable information that predicts rehabilitation outcome is only likely to partially explain the rehabilitation potential of individuals, with wider influences on professional decisionmaking being relatively unknown.

Despite this, studies detailing predictive factors have been noted to use the terms rehabilitation potential and rehabilitation outcomes almost interchangeably. Semel et al (2010) conducted a study to explore the predictors of functional outcome after hip surgery in a large acute inpatient rehabilitation facility in the USA. The study identifies several significant patient characteristics which can give forewarning of a poorer functional outcome and then within the presentation of findings and discussion, claims about the links to lower functional outcome and lessened rehabilitation potential are used to express similar messages.

Another important issue of note is that some studies report excluding participants or patient data which may have been deemed to have a severe co-existing condition. The limited knowledge base about the types and severity of conditions alongside the potential areas of ambiguity noted above, result in challenges to services and practitioners when attempting to generalise or adopt key messages from findings.

2.4.3. Decision-Making - selection for rehabilitation

The presentation of literature in section 2.4.2 above, aiming to predict who will benefit from rehabilitation, precedes an overlapping element of decision-making to select those for rehabilitation in order to optimise the use of rehabilitation resources. Core underpinning values of the National Health Service (NHS) include the availability of services to all and that for most, these services are free at the point of need (DH, 2013). This availability of preventative and health promoting services for those who need it is also echoed in recent Social Care legislation (The Care Act: DH, 2014). However, as services are finite and demand, particularly in relation to older people, is significant and growing, judgements about access and need are fundamental to optimising the use of, and benefit from, rehabilitation services. Working with eligibility or selection criteria is a frequent requirement for health professionals, from establishing who is an appropriate candidate for surgery, recruiting to clinical trials, to decision-making about equipment provision, to name a few examples. Criteria is often developed to help services and practitioners to make decisions about who should have access to a particular resource and is recognised as a way of attempting to allocate and ration services to maximise benefit and fairness (Wade, 2003). There are examples in literature of selection criteria to assist with judgements about referral and acceptance to a variety of rehabilitation services including stroke rehabilitation units, pulmonary rehabilitation, and cardiac rehabilitation (Hakkennes, Brock and Hill, 2011; llett et al, 2010; Ambrosino and Foglio, 1996).

As part of a wider systematic review, Hakkennes, Brock and Hill (2011) found six studies which detailed selection criteria for rehabilitation after stroke, with age, prestroke functional level, and post-stroke functional level consistently cited within selection criteria. However, directions of association were not always predictable – for example, in relation to post-stroke functional level, both patients with severe and mild impairments were less likely to be accepted than those with moderate impairment, providing tentative suggestions that selectors finely judge those who can achieve maximum benefits from rehabilitation. Hakkennes, Brock and Hill do indicate that there was consensus amongst studies that absolute criteria for selection for rehabilitation would be unfeasible, with potential to include inappropriate patients and exclude some who may ultimately benefit.

Considering the fact that research presents as being more fully developed in the field of stroke rehabilitation, the identification by Hakkennes, Brock and Hill of only six studies over more than a 40 year period is illustrative of under-developed empirical research in this area. The review also highlights the quality of included studies was generally poor leading to a call for more rigorous research.

Wider literature reports challenges with selection and eligibility criteria in healthcare including the fact that it is not always used (or documented) when it is recommended to be; that generic (and sometimes untrained) staff are required to apply criteria for access to specialist (and sometimes not fully understood) services; and the important fact that unless sensitivity, specificity and predictive value is proven, criteria potentially excludes people who may benefit from a service and includes those who are potentially inappropriate (Lynch et al, 2016; Wade, 2003).

Perhaps an emerging, but unquantifiable challenge, relates to who has responsibility for the application of criteria as responsibility can lie with the referring service or the accepting service (Lynch et al, 2016). A policy context of payment by results and national tariffs could lead to significant differences in the motivations of these referring and accepting services, with services potentially being motivated to reduce length of stay or to prioritise people who may reflect positive outcome measurement for a service. There is a suggestion that decisions about transfer of care from one service to another may sometimes simply be underpinned by a desire to 'shift responsibility' (Dodier & Camus, 1998).

Although professional bodies and policy makers attempt to provide guidance on eligibility and selection criteria, the fact that there is theoretically an infinite demand for a free and open service inevitably leads to those involved in the day-to-day delivery of care and services making decisions to manage this challenge. Lipsky (1980) suggests that it is the decisions and routines of individual workers and teams, influenced by various factors and pressures, which effectively become the public policies which assist to manage workload and demand.

The decision-making process and factors influencing decision-making has also been explored in relation to the selection for rehabilitation after stroke. Putman et al (2007) explored clinical and non-clinical factors involved in decision-making concerning

admission to six European stroke rehabilitation units. Data was gathered through analysis of assessments and outcome measures relating to patients admitted to the identified units, alongside questionnaires and interviews with medical consultants. Findings revealed that factors influencing decisions could be grouped in to factors related to the patient, factors related to the network between facilities, and factors related to the referring hospital, and inconclusively, but perhaps predictably, the most influential factors differed across sites and countries.

Again in the field of stroke, llett et al (2010) explored the process of selecting patients for stroke rehabilitation across seven acute sites in a specific region in Australia. They found variations in practices between different acute hospitals, and suggested that acute sites may be influenced by organisational factors such as numbers of beds, availability of outpatient resources and financial incentives related to shortening length of stay. A small study by Kennedy et al (2012) revealed that rehabilitation physicians were influenced by a range of individual factors such as prognosis, age and social support, alongside organisational factors such as availability of specialist services and staffing issues when reaching decisions about selection for rehabilitation following acute stroke. The same study also attempted to investigate levels of agreement between rehabilitation physicians regarding the assessment of rehabilitation potential based on a sample of case vignettes. They found high levels of agreement for some scenarios alongside divergence for others and suggested variability is likely in this area of practice.

Hakkennes et al (2013) aimed to identify factors that assessors considered important in decision-making regarding suitability for inpatient rehabilitation after severe acute stroke in Victoria, Australia. A questionnaire was developed from a comprehensive review of the literature, and included 17 factors in total, spanning prognostic indicators, alongside social and organisational factors. The questionnaire utilised a 10-point visual analogue scale to rate the importance of each factor (from 0 – not important at all to 10

- very important) and indicate if the factor was positive (increased the likelihood of selection for rehabilitation) or negative (decreased the likelihood of selection for rehabilitation). Results highlight that, for those accepted to rehabilitation, items linked to pre-morbid status (such as pre-morbid cognition, mobility and communication), were the most influential factors but conversely, for those not accepted, factors linked to post-stroke status (current mobility, cognition and social support) were most important. Interestingly, social factors, such as level of social support and carer advocacy for rehabilitation, did not emerge as among the most important factors across the study, although were highlighted as influential in some scenarios. Additionally, organisational factors were not highlighted as significant in either those accepted or not accepted for rehabilitation. The use of an unvalidated visual analogue scale affects the reliability and validity of results. Also, as the study was limited to one state in Australia and only physicians were involved as participants in the research, caution should be used when generalising to wider countries and professional groups. However, as there are limited studies providing insight in to the decision-making process, the results add insight to the topic as an area for further enquiry.

Zhu et al (2006) explore the use of a computer algorithm to support clinical decisionmaking relating to rehabilitation potential of older people and whether this could be more effective than the use of a current standardised protocol. The 'K-nearest neighbour (KNN)' algorithm was used to simulate expert clinical reasoning through a sophisticated database to enable comparison against a large number of past clinical cases and compared with the use of a clinical assessment protocol used within the identified sites. The same items relating to physical functioning, comprehension, health status indicators and functional potential were used by both assessments. The findings suggest that the use of the KNN algorithm had lower false positive and false negative rates, therefore making more reliable predictions about whether an individual would make functional gains or maintain the ability to live at home.

In relation to professional roles in the evaluation of rehabilitation potential, many studies outline service delivery models where there is an inpatient assessor (often a physician) who assesses rehabilitation potential and therefore determines eligibility for rehabilitation services (Conroy, DeJong & Horn, 2009; Kennedy et al 2012; Hakkennes, 2012). In other service delivery models, the evaluation of rehabilitation potential may be determined by health insurances with limited involvement from healthcare teams (Putman et al, 2007). Within UK professional guidelines (such as the NICE guideline for Intermediate Care and Reablement), general statements advocate assessment for the need for rehabilitation services and indicate this could be carried out by a range of professionals including therapists, nurses or social workers. Studies which discuss the role or involvement of rehabilitation professionals such as physiotherapy and occupational therapy are fewer in number (Cullen, 2007; Shun et al, 2017) although do highlight involvement worthy of exploration.

2.4.4. Decision-Making – managing risk

It could be understandably suggested that practitioners make decisions about the type of rehabilitation based on assessment and management of associated risks. For example, if a practitioner is considering recommending transfer to an inpatient rehabilitation ward, as opposed to a transfer home with referral to a community rehabilitation team, it is potentially the assessment of the extent to which risks can be managed at home which will heavily influence the decision.

Literature searching revealed a paucity of specific examples of studies exploring the influence of risk and risk factors on decision-making about rehabilitation potential and accessing rehabilitation. However, if decisions about rehabilitation are often linked to a transfer of care, themes from literature about transfer of care and discharge from acute hospitals will have obvious overlap. The management of risk is said to be at the centre

of decisions about transfer of care and discharge in an acute hospital context (Atwal, Wiggett & McIntyre, 2011; Huby et al, 2004) and is the focus of research studies and national policy (Goncalves-Bradley et al, 2013; The National Audit Office, 2016).

Whilst it is beyond the scope of this section to present a detailed analysis of this policy and literature, key themes about the influence of risk on transfer of care decisions can be summarised. It has been suggested that the Hippocratic tradition continues to dominate decisions about care transfer within and from acute hospitals, with a riskavoidant approach of doing no harm being apparent (Moats & Doble, 2006). It is also purported that professionals develop a risk-avoidant approach to their practice in this area because of fear over accountability and blame if negative consequences do arise (Atwal, McIntyre & Wiggett, 2011).

There is a great deal of complexity when considering how conceptualisations of risk, risk assessment and risk management influence decisions about transfer of care and discharge. The focus on early discharge means that patients, families and health professionals have limited time to carry out comprehensive assessments or to fully understand implications, which could potentially contribute to decisions favouring risk avoidance (Moats & Doble, 2006). In the case of older people, and particularly where a person may lack capacity for decision-making, family members may be granted proxydecision making responsibilities and evidence suggests they may be likely to default to a position of protecting elderly relatives (Moats & Doble, 2006; Denson, Winefield and Beilby, 2012). Literature also suggests that health professionals may default to paternalistic and protective positions, perhaps because resources result in a lack of meaningful alternatives, although coercion is sometimes used to elicit acceptance of the best option (Moats & Doble, 2006; Huby et al, 2004). Huby et al (2004) also highlight that when risks present as difficult to manage within a transfer of care decision, the decision may be delayed or deferred through referral to another service, rather than being explicitly addressed. Add to this the acceptance that risk is a

subjective construct (Atwal, McIntyre & Wiggett, 2011; Densen, Winefield and Beilby, 2012) and, similarly to the concept of rehabilitation potential, professional perspectives may vary (Clemens & Hayes, 1997).

2.4.5. Decision-making – allocating finite rehabilitation resources

Although policy makers attempt to provide guidance on how to manage supply of, and demand for, healthcare, the fact that there is theoretically an infinite demand for a free and open service poses critical challenges for all involved. Inevitably, those involved in the day-to-day delivery of care and services are required to make decisions which ration and manage access and utilisation.

Priority setting and rationing is a key component of strategic decision making within healthcare and public health, with high profile examples particularly pertaining to funding of medications and surgical procedures open to evidence-based scrutiny and ethical debate (NICE, 2018). However, of relevance to this study, is the priority setting and rationing which occurs on the frontline of clinical practice; including whether someone should be transferred to an inpatient rehabilitation bed or receive intensive community rehabilitation with outcomes of such decisions potentially meaning another person may miss out. This link between practitioner decision making and rationing at a frontline clinical level is acknowledged in literature (Lipsky, 1980; Vryonides et al, 2015).

Lipsky (1980) proposes that often public servants such as health workers enter their chosen roles with ambitious goals orientated towards beneficence and social justice. They often aim to make decisions which respond to the needs of the individual but in the reality of their practice, they must develop techniques which respond to their service users on mass in order to optimise resource allocation and practice fairly.

Lipsky goes on to suggest that it is the decisions and routines of individual workers and teams, influenced by various factors and pressures, which effectively become the public policies which assist to manage workload and demand. He describes this in his seminal work on 'Street-Level Bureaucracy' (Lipsky, 1980), and discusses that bureaucrats practicing in public service roles develop techniques to manage compromises and conflicts over demand, resources, access and worthiness. At best, such techniques help them to provide the best and fairest service under prevailing circumstances and pressures although at worst, the techniques can be open to favouritism and stereotyping.

Allen, Griffiths and Lyne (2004) found evidence of 'Street-Level Bureaucracy' in an ethnographic study of resource allocation for adults requiring continuing care and rehabilitation following stroke. They found that, where there were increasing pressures on resources such as funding for care packages or bed availability, staff would renegotiate issues of need and responsibility for care to find an option that was potentially 'easier' to deliver and organise, and this did not always coincide with the preferences of patients or families. Examples of easier options included giving the option of family members providing care or equipment, or attempting to reassess and redefine the nutritional needs of a patient because a community care package could not be found to meet the need. Allen, Griffiths and Lyne go on to suggest that challenges were particularly heightened when there was clear differentiation in funding and responsibility between health and social care and attempts to navigate these challenges were not always led by clinical need (2004).

Rationing can happen explicitly in routine clinical practice where identified criteria is used to screen in and out those who may benefit from a service. However, along with the suggestion from literature that criteria to support selection for rehabilitation is not always used in clinical practice (Hakkennes, Brock & Hill, 2011), theorists also recognise that there are challenges for public servants about openness in relation to

priorities and rationing. Being open about withholding benefits to some and not others is understandably seen as socially divisive and for clinicians, being seen as withholding rather than care-giving is uncomfortable and counter-intuitive (Scheunemann, 2011; Daniels and Sabin, 1998). For these reasons, the decision-making which supports priority setting and service rationing can sometimes become guarded or implicit.

2.4.6. Decision-making - service user involvement

2.4.6.1. Definitions and characteristics of shared-decision making

Health professionals have a responsibility to involve patients in decisions concerning care and treatment. This is outlined in professional codes of conduct, is an ethically accepted approach to practice and is an integral part of contemporary policy and legislation (HCPC, 2013; DH, 2010). There are several reported benefits of patients being more actively involved in decisions about their health care including reduced anxiety, improved knowledge, reduced conflict with professionals, improved treatment adherence and improved satisfaction (O'Connor et al, 2003; Loh et al, 2007).

In their seminal work exploring healthcare decisions, Charles, Gafni and Whelan (1997) highlight and compare prominent models of decision-making – particularly focussing on the paternalistic model, the informed decision-making model and the professional-as-agent model and comparing them to a model of shared decision-making. The paternalistic model places the patient in a passive role with the physician or professional placed as expert. Examples of this can range from the obvious scenarios of a professional recommending tests or treatments or informing the patient what will happen and when it will be initiated. However, more subtle examples could include a professional providing selective information or options and encouraging patients to take a recommended course of action.

Charles, Gafni and Whelan proceed to discuss the informed decision-making model and the professional-as-agent model, suggesting these two approaches are sometimes misjudged as being similar, or interchangeable, to a model of shared decision-making. The informed decision-making model adjusts the asymmetry between professional and patient, but which results in imbalance in the opposite direction. Here, the patient has ultimate responsibility for the decision and the role of the professional is one of information transfer to enable patients to make decisions which reflect the best scientific knowledge and are aligned to their preferences. Whilst also having an important place within health care, particularly in relation to self-management of long term conditions, research has highlighted that there is a difference between the desire for information and understanding, and the desire for ultimate responsibility for decisions (Ryan, 1992 cited in Charles, Gafni and Whelan, 1997).

The professional-as-agent model reverts back to the professional assuming responsibility for decisions, but with a responsibility to elicit client's preferences and perceptions in relation to issues such as quality of life, perception of risk and the meaning of health and wellbeing. Again, the contrast with a model of shared decision making is obvious, with responsibility being placed with the professional and reliant on information sharing and giving which may have varying degrees of success.

Charles, Gafni and Whelan (1997) subsequently develop principles of shared decision making which are frequently referred to in more recent research examples. They identify four key principles including: the need for at least two participants; the need for both parties to take steps to participate; the need for information sharing as a prerequisite; and the need for a treatment decision which both parties agree to. Critics suggest that this seminal discussion, followed by more recent definitions of shared decision-making often lack a specific focus on collaborative goal setting although this should be a core component of the concept (Vermunt et al, 2017).

The focus on involvement and shared decision-making in professional standards and health policy has led to attempts to establish the extent to which this is being achieved in practice. Evidence suggests that, despite patients wanting a more participatory role, the majority of patients are still not experiencing a shared approach to decision-making (Dierckx et al 2013). However the challenge of conceptualising what shared decisionmaking means to different groups and the subjective measures by which it is evaluated are often cited as challenges for practitioners and researchers and raise inevitable questions about trustworthiness and generalizability of evidence.

The reported distance between policy and practice has prompted exploration of strategies for improving the adoption of shared decision-making principles by health professionals although a recent Cochrane Collaboration systematic review found a paucity of studies and resulted in difficulty reaching meaningful conclusions (Legare et al, 2014).

2.4.6.2. Wider concepts – participation, engagement and involvement

More recently, the terms participation, engagement and involvement (among others) have been commonly used to reflect a broader agenda for service users, families and carers to become active participants, not just in decisions, but in care and services in their entirety. Once again definitional challenges emerge, with the terms being used loosely to describe concepts relating to patient actions or behaviours, retention within or adherence to services, self-management of conditions or referring to the interaction between patient and provider, among other uses (Bright et al, 2015).

A recent systematic review on the topic of engagement in healthcare and rehabilitation proposed a definition of engagement as 'a co-constructed process and state....[which] incorporates a process of gradually connecting with each other and a therapeutic program, which enables the individual to become an active, committed and invested collaborator in health care' (Bright et al, 2015; p 650). Interestingly, out of 28 studies included in the review, the majority of studies were carried out in relation to mental health contexts (with others referring to discipline specific understanding in social work, or speech and language therapy) leaving the understanding in relation to physical rehabilitation a little less clearly developed.

2.4.6.3. Benefits and challenges of involvement for older people

In relation to rehabilitation, participation and involvement often happen within a goalsetting process (Rose, Rosewilliam and Soundy, 2017). Studies have suggested that involvement in a goal-setting process can increase patient satisfaction, improve motivation, and creates a greater sense of ownership of the rehabilitative process (Rose, Rosewilliam and Soundy, 2017). For these reasons, goal-setting is widely discussed as a fundamental part of a collaborative rehabilitation process.

Vermunt et al (2017) carried out a systematic review on the topic of collaborative goal setting with elderly patients with chronic disease or multi-morbidity. Whilst providing some evidence for the health and wellbeing benefits of goal-setting, such as improved functional outcomes, improved motivation and improved adherence to interventions, overall conclusions suggested compelling evidence was lacking. They mainly attributed this to practice variations and methodological challenges, suggesting that goal-setting is often poorly defined and described, and is often part of multi-factorial interventions making it difficult to attribute outcomes to particular elements. They concluded that the review was relevant, yet premature, with conceptual clarity an important pre-requisite.

Despite supporting evidence for goal-setting to contribute to a multi-factorial rehabilitation process, older people are less likely to be involved in goal-setting and decision-making and are more likely to actively or passively relinquish involvement in decisions about care (Schulman-Green et al, 2006; Foss, 2011). Challenges to

involvement and shared decision-making specific to an older population are regularly cited, including frailty, sensory impairments and cognitive impairment (Lally & Tullo, 2012). And although some studies highlight a proportion of older people who may not want to be involved in decisions about their health and care (Ekdahl, Andersson and Friedrichson, 2010), authors and researchers warn that it is simplistic and inaccurate to assume a generalised position that older people prefer to be told what to do or to have decisions made for them (Lally & Tullo, 2012; Bastiaens, 2007). A large scale study, involving 406 older people across 11 European countries, highlighted that older people do want to be involved in decision-making about primary health care (Bastiaens, 2007), something echoed by other studies in a range of generalist or specialist health care contexts.

If the view that older people may not want to be involved in decisions about their care is agreed as simplistic, the more realistic and accurate reflection is that the interpretation of what involvement and shared decision-making means is complex. Bastiaens (2007) found that involvement was interpreted by a caring relationship and receiving appropriate information rather than necessarily having a role in decisions themselves. Similarly, and specific to an acute care context, Ekdahl, Andersson and Friedrichson (2010) explored the experiences and preferences of frail elderly patients for participation in medical decision-making in hospital. Qualitative interviews with 15 participants aged over 75 revealed that participation in decision-making was more realistically about effective communication, information giving and the opportunity to be listened to, rather than responsibility for a particular choice.

Although some of the challenges to involvement and shared-decision making involving older people have been noted above, it is perhaps also pertinent to this area of enquiry to acknowledge the challenges with achieving shared decision-making in an acute hospital environment. Such challenges include the dominance of a biomedical model and a Hippocratic tradition which places value on beneficence, the fear of liability and

the potential difficulty in offering options due to time and resource limitations (Moats, 2006). There are also valid reasons for diminished abilities to engage in decisionmaking during acute admissions (Scheel-Sailor et al, 2017).

Data gathered through qualitative interviews with occupational therapists involved in discharge decisions for older people highlighted the prevalence of professionallydominated decisions in an acute context, and discuss the influence of limited time to work towards true client-centredness (Moats, 2006). The authoritative influence of physicians who can make or override a final decision was also noted in this study.

Lally and Tullo (2012) suggest that, to respond to variation in preference for, and meaning of shared decision-making, the foundation of working towards this for older people should be an attempt to establish the preferred extent and nature of involvement. Bridges, Flatley and Meyer (2010) highlight the difference between power and choice being willingly 'given up' and it being 'taken away' by professionals in acute care, and discuss the sharing of information to establish whether an individual wants to be involved, and the way in which they would like this to happen, are issues of significance. The principle of information sharing as a pre-requisite, identified in the seminal work of Charles, Gafni and Whelan (1997) underpins this important message.

To develop the importance of information sharing, Huby et al (2004) provide an insightful way to approach the issue of involvement of older people in decision-making, suggesting that rather than looking at autonomy (that is, the person making the decision for themselves), attention should be placed on trust. They go on to suggest that trust is an active decision in itself and for older people making decisions about their treatment and care, this involves a decision to share responsibility for their own welfare with somebody else in situations of uncertainty and risk. For an active decision to trust a professional, Huby et al (2004) summarise that an individual fundamentally needs

sufficient information to understand the system of care, which includes open communication about risk, resources and responsibilities.

2.5. Chapter Summary

The literature review has attempted to define and conceptualise key themes central to this research, including rehabilitation, rehabilitation potential, and decision-making. The review has highlighted that despite being a part of a professional lexicon, the term rehabilitation potential is notably ambiguous and judgements lack reference to criteria or structured approaches. However, there are consistently reported patient-elements which may predict those likely to benefit from rehabilitation – such as age, presence of co-morbidities, and severity of cognitive impairment – and these elements influence either formal or informal criteria.

There is a general consensus that rehabilitation for older people is effective in improving functional outcomes. However, on closer examination of research, findings provide broad and general support for the many different models of bed-based, community-based and multi-disciplinary rehabilitation, with wide variances in service provision a reality for patients, practitioners and commissioners.

Across all areas of literature, methodological challenges have been consistently noted. Many areas of research relate to specific clinical pathologies or pathways and many studies relating to rehabilitation decision-making rely on retrospective reflection or hypothetical cases.

Synthesis of these issues has therefore focussed this research to explore areas where there are gaps in knowledge or a need for clarity. Rehabilitation potential, and processes used to evaluate rehabilitation potential, are poorly understood and research which explores this in relation to older people is particularly deficient. There is also a compelling case for research methodologies which enable exploration of decisionmaking in real-time and in the context in which it occurs. Finally, a focus on roles within decision-making, particularly the roles of professionals outside of medicine, such as occupational therapy and physiotherapy, and the roles of patients and families will add important understanding to such enquiry.

CHAPTER THREE – METHODOLOGY AND METHODS

This chapter outlines the philosophical perspectives and approaches adopted to inform the methodological position of this project. Research design and methods are then discussed and the chapter concludes with an exploration of strategies used to enhance quality.

3.1. Ontological beginnings

Ontology is the study of being, concerned with what is real and the nature of existence (Crotty, 1998). It is an important part of any research that the basic set of assumptions of what can be taken as real are outlined in order for the development and utilisation of methods to be fully understood (Gonzalez, 2000). From the research aim above, it could be suggested that there are certain, perhaps simple, realist assumptions, for example that the hospital system is a reality. However, Crotty suggests that whilst 'the existence of a world without a mind is conceivable...meaning without a mind is not' (1998; p11) and it is here where a more constructionist ontology emerges.

Therefore, from this constructionist position, the research is based on the assumption that reality is social in nature and is created through engagement in social relationships and interactions (Alvesson and Skoldberg, 2009). Secondly, there is an assumption that meanings are developed within social contexts and that, irrespective of whether these meanings are held individually or collectively, people will act towards external realities (in this case, act by way of making decisions) based on such meanings (Blumer, 1969). Furthermore, the research also acknowledges boundaries as socially constructed realities; boundaries between professions such as occupational therapy and physiotherapy; boundaries between professionals and patients; and indeed boundaries between the researcher and the research. And finally, there is assumption
that the research relies on interdependence between the researcher and the context; in the words of Gonzalez 'a seed will not grow unless it is planted' (2000; p634).

3.2. Epistemology

Epistemology refers to the nature of knowledge and is concerned with providing the philosophical basis for deciding what knowledge is possible and how it can be accessed. Social Constructionism offers a starting point to present the philosophical basis underpinning this research. The main premise of social constructionism is that what people know about the world depends on how they approach any given phenomenon, and how they approach it will depend on the social relationships and constructs which they are part of (Gergen, 2009). Furthermore, if ways of knowing and understanding are influenced by social relationships, such ways of knowing, categorising and understanding concepts and phenomenon are also historically and culturally specific (Burr, 2003). People construct ways of understanding through their social interactions and through shared language. And because knowledge and social action go together, the ways in which individuals, groups, organisations and communities act and behave are again socially constructed and framed in a set of physical, temporal, social and political circumstances (Burr, 2003).

A social constructionist paradigm is perhaps in opposition to a more realist paradigm and realist or positivist approaches to research. Social constructionism does not deny that things are real, but instead suggests that when people are defining reality, they are doing this from their own particular perspective; a perspective which has been shaped by the social relationships which have constructed their world (Gergen, 2009). It is such initial insights that create the important alignment with qualitative research.

Qualitative research is appropriate where the issue is emerging, complex or ambiguous (Bowling, 2009). This type of research can not answer research questions about frequencies, effects, causes or strength of relationships but instead can develop understanding of social processes, experiences and behaviours (Barbour, 2014). It is such insights that also underpin a founding epistimelogical position of this research, that knowledge will be created through the understanding of this phenomenon within the naturalistic social context in which it occurs.

Thinking in more depth about social constructionist theory, for individuals and groups it could be suggested that social constructions of reality gain their significance from their social utility and how they can then be used or applied (Gergen, 2009). That is to say, if constructing a shared meaning leads to a perceived positive or satisfactory outcome, then people begin to perpetuate and preserve the shared meaning and the opposite can be seen when social utility is lost or becomes less obvious.

Social constructionist theory does not view knowledge as something held by an individual, but instead is seen as something that is generated within a social context. Utilising principles of social constructionism to inform social research leads to a critical stance towards meanings, language, assumptions and norms, and to examine the social processes which are constructing ways of knowing, thinking and acting. Such foundations begin to offer important theoretical principles for this research; that people will make decisions and take actions based on shared assumptions and norms of what rehabilitation offers and what an evaluation of rehabilitation potential can mean.

However, this viewpoint would also purport that there is little benefit in looking for oneoff, once-and-for-all descriptions or explanations as social life and relationships are continually changing and exposed to new and different influences (Burr, 2003). This therefore supports another important epistemological assumption – that the research

should involve looking for descriptions and meanings at different periods in time with different contextual influences.

Under a broad grouping of traditions which place value on social reality, Symbolic Interactionism provides further theoretical understandings of the ontological, and therefore epistemological, stance. When attempting to understand Symbolic Interactionism, Blumer (1969) presents regularly cited core principles. Firstly, that people act towards things, including each other, on the basis of the meanings they have for them. Secondly, that these meanings are derived through social interaction and interpretation with others. And thirdly, that these meanings are managed through an interpretive process that people use to make sense of the objects that constitute the world they live in.

It is this suggestion that people act towards things – including other people, objects, and language – that supports an additional important epistemological assumption. Although similar to the broader church of social constructionist research in the agreement that meaning making is a social process, it is the insight that humans can be best understood through what they do. Dewey insightfully captures the essence of this by suggesting that 'the test of ideas, of thinking generally, is found in the consequences of the acts to which the idea leads' (1929; p136). Drawing from this, this research will use the vehicle of decisions and actions (decisions about rehabilitation potential and rehabilitation pathways) to closely examine social processes and meaning-making.

3.3. Methodology

3.3.1. Ethnography – the overriding methodology

From a perspective that ways of knowing and acting can be understood as socially constructed, ethnography was identified as a methodology which enables the study of cultural groups in their natural setting to understand the realities of actions within social contexts (Cresswell, 2009) and therefore presented an important starting point. This commitment to a methodological position which placed importance on understanding of meanings and activities within a given 'field', through participation in real time and in the authentic context, embdedded the principles of ethnography as guiding principles for this research.

Ethnography involves the researcher participating in the daily lives of participants, or the daily pattern of a setting, over an extended period of time and collecting a broad spectrum of available data in order to illuminate taken for granted and everyday behaviours, rituals and practices (Brewer, 2000; Hammersley & Atkinson, 1995). Ethnography commits to 'telling it like it is from the inside' (Brewer, 2000; p17). It has also been suggested that it is the way in which ethnography makes links between the micro and the macro, between everyday occurences and wider cultural, social and political influences that is one of the most clear distinguishing features of this methodoloy (Savage, 2006). Importantly, there are contemporary examples of ethnographic research exploring decision-making and care processes in hospital contexts (Poole et al, 2014; Waring et al, 2014).

Hammersley, one of Britains recent and foremost authors on the topic of ethnographic research suggests that it includes the following features:

- People's behaviour is studied in everyday contexts rather than unnatural or experimental conditions created by the researcher.
- Data are collected by various techniques but primarly by means of observation

- Data collection is flexible and unstructured to avoid pre-fixed arrangements that impose categories on what people say or do
- The focus is normally on a single setting or group and is small-scale
- The analysis of the data involves attribution of the meanings of the human actions described and explained

(Hammersley, 1990)

Traditions and branches of ethnography are described in different ways by different authors, such as naturalist, realist, critical and feminist ethnography, among others (Skeggs, 2001; Savage 2006; Schwandt, 2007; Hammersley, 1998). The branch of critical ethnography approaches enquiry from a social-ethical perspective, aiming to scrutinise taken-for-granted social, economic, cultural and political assumptions of societies or organisations (Schwandt, 2007). The insight that critical ethnography often focuses on specific practices within specific institutions (Schwandt, 2007), suggests some alignment with this study.

Organisational ethnography is also discussed within literature pertaining to social research (Gaggiotti, Kostera & Krzywozeka, 2016; Neyland, 2007; Brewer, 2000) although as sociologists understand the term organisation to mean structures which manage and organise life and behaviour, it could be argued that all ethnography is organisational in nature (Brewer, 2000). That being said, organisational ethnography often refers to the study of more formal organisations such as workplaces and institutions and has been identified as an approach to social research which can assist to understand occupational careers and identities, alongside the maintenance of power and control and examples of decision-making within organisations (Brewer, 2000). This last example of organisational ethnography to explore reasoning and decision-making within organisations is particularly pertinent. There are examples of research aligned to the umbrella term of organisational ethnography which explore discretion, rule-making

and interpretation of organisational influences when making decisions, and specifically examples researching health professionals operating within institutions and organisations (Becker et al, 1961; Chambliss, 1996).

More specifically, medical ethnography is often seen as a branch of organisational ethnography studying the cultures and everyday pracitces within healthcare systems and workplaces. However, it is also perhaps seen as broader than simply a branch of organisational ethnography, not only taking place within healthcare instituions, but exploring issues of health, illness, and disability across a range of settings, cultures and groups (Bloor, 2001). Influential ethnographic studies within this tradition include the study of 'Asylums' to examine the social construction of mental illness from within an institution (Goffman, 1975), and the culture and socialisation of medical students in the seminal research by Becker et al (1961) entitled 'Boys in White'.

Medical ethnography has been used to explore the meaning of every day or taken for granted phenonena in relation to health and healthcare systems, alongside examining healthcare interactions and professional decision-making. It has increasingly been recognised as a way to analyse inter-professional and intra-professional interactions which create consensual and collective agreements of patients or their difficulties and establish particular courses of action (Bloor, 2001). There are examples of ethnographic research which illuminate how meanings of health concepts such as rehabilitation can be shaped differently for different audiences, and under different social, temporal and organisational influences (Gubrium & Buckholdt, 1982). Furthermore, medical ethnography is recognised as often having a powerful impact on professionals due the rich descriptions of healthcare systems and routines and the accessible links to everyday practice (Bloor, 2001). Features such as this have illuminated the synergy with this research.

Despite recognising positive alignment with the key features of ethnography, there were other perhaps more 'problematic' issues requiring critical consideration. For

example, ethnograpy as a methodological position places emphasis on immersion and extensive data collection; an area where I had pragmatic concerns due to time Iwas able to commit to data collection and role-immersion. Opportunities to be flexible, serendipitous and to examine the social world from all angles would indeed need to be boundaried, not just due to time but by the desire to set parameters and stay focussed on specific research objectives.

There are also other methodological positions which place value on lived experience, leading to methods which support contemporaneous data gathering in context. Critical consideration was given to branches of phenomenology to explore the meanings of a phenomenon (namely rehabilitation potential), and the lived experience and emotional response to decisions and actions relating to this phenomenon (van Manen, 1990). Aligned with phenomenology, the use of audio or written diaries, used as adjuncts or alternatives to indepth interviewing, provide methods to capture real-time thinking and reflections (Alaszewski, 2006). However such a position, and the use of such methods, would not have enabled the attention to the wider contextual and cultural influences on reasoning, with emphasis instead placed on learning from individual interpretations and experiences.

Similarities have been drawn between ethnography and grounded theory (Brewer, 2000; Charmaz and Mitchell, 2001) and again this was given consideration. Although often discussed as more of an approach to data analysis, it is also discussed as a methodologyy for developing theory, grounded in data, which is systematically gathered and analysed (Brewer, 2000). Key features of grounded theory include the use of constant inductive analysis to build theory from the ground, theoretical sampling to test theories which emerge from the data on an ongoing basis, and reaching theoretical saturation which describes a process of searching deliberately for understandings until no new data or examples emerge. It is often discussed as

informing the development of qualitative analysis software packages such as NVIVO, aiming to enhance objectivity in qualitative analysis (Gobo & Molle, 2017).

Those who recognise the synergy between grounded theory and ethnography suggest principles of both can work to overcome problems if adopting one approach in a more purist sense. To summarise some of the potential benefits, a combined application can lead to more focussed time within the field and assist the researcher to move beyond what is often viewed as unsophisticated description towards theoretical development (Charmaz & Mitchell, 2001). However, others argue that rules and procedures can lead to a constrained analytical lens and more subtle processes are required for teasing out meaning and interpretation (Brewer, 2000). Whilst recognising that some of the structured guidelines and procedures may be of assistance in the case of this research, it was a desire for a more sensitive and naturalist approach to data generation and analysis that encouraged a move beyond this. Also, from a pragmatic view, I was also aware that some of the key features of grounded theory (such as theoretical sampling and data saturation) may not have been easy to achieve in the context of a study requiring rigid ethical approvals alongiside the necessity to comlete data collection within a set time period as a part of a part-time PhD programme. Because grounded theory is built on such principles, and the scientific rigour that this aims to achieve, there was a danger of being left with an unsatisfactory version of this methodology which neither aligned with the strengths, nor answered the criticisms.

Continued exploration beyond the expansive literature regarding ethnographic traditions led to important alignment with case study research. Although some commentators suggest case studies are primarily a method, others suggest the principles communicate more than just a method, instead suggesting that it can be considered as an over-arching methodological position or paradigm which communicates assumptions about the nature of the social world and how that world can be studied and understood (Hammersley & Gomm, 2000). The case study

paradigm has an overriding commitment to understanding a case or phenomenon in depth and in authentic contexts and appreciates the case as a bounded entity. Generalising to other cases or examples of a phenomenon is not a primary aim, although rich description and illumination of the features of the case will enable readers and scholars to judge generalisability for themselves (Yin, 2009; Yin 2012). Case studies have been reported as having a central tendancy to illuminate decisions or sets of decisions, why they were taken, how they were reached or implemented and with what result (Schramm, 1971; cited in Yin, 2014) and again, commonalities with this research which aimed to explore decision-making in a healthcare context are apparent.

3.3.2. Research design

Both ethnographic and case study research support the use of multiple data collection methods and a flexible and creative approach to data gathering, in order to look beyond the known and the anticipated and to be able to respond in a serendipitous and adaptable way (Rock 2001; Yin 2009). Rock goes on to suggest that ethnography which attempts to prescribe firm research designs or instruments may result in blinding the researcher to the social world, preventing effective responses to what could be discovered. Although stages and methods of data collection were provisionally planned and indeed, such plans were a requirement of ethical approval and negotiating access, it was also important to approach the collection and generation of data with flexibility and openness.

Although recognising some similarities in the use of multiple methods and the flexible, responsive design, it is also this element of research design where there are differences between conventional ethnographic research and a case study design. Furthermore, it is perhaps here where a pragmatic hybird of principles from both positions was adopted.

3.3.3. Identifying the 'case' and negotiating access

Stake (2000) suggests a case is whatever bounded system is of interest to the research. It can be a common or everyday phenomenon, or the case can be an individual, oranisation, process or an event (Yin, 2009; Yin 2012). In this study, the case was initially one ward within one identified institution. However, from the theoretical and empirical picture (illustrated within the literature review), the research aim and objectives expressed an interest in specific health professionals (in this case occupational therapists and physiotherapists) and learning from specific patient examples. In addition, although one ward represented a bounded entity, the boundary between the case or phenomenon and it's context can be blurred, and therefore the research design needed to take in to account wider physical, temporal, spatial and social dimensions (for example services, professionals or other wards which linked to the main ward). In relation to literature guiding case study design, this has been explained as 'nested units of analysis' or 'embdedded subcases' within a single case study (Yin, 2012) and therefore the focus on occupational therapists, physiotherapists and specific patient cases could be described as examples of such nested units of analysis within the case of the identified ward and system. It is perhaps through this need to develop pragmatic boundaries around the case and areas of interest where the principles of case study design influenced the research, although the study remained committed to an overarching ethnographic methodology.

Preliminary support for the project was sought through an existing contact (as part of my role as a university lecturer) and through developing a contact with the NHS Trust Research and Development department. Both before and after gaining institutional and NHS ethical approval, an extensive period of meetings (with operational managers, therapy and nursing managers and senior therapy clinicians) took place to negotiate where the research could be situated. These meetings were undertaken with two main aims; to introduce the research sensitively and professionally to key stakeholders and

gatekeepers and to narrow down the focus on an identified ward and subsequently a smaller group of professionals. Although presence was initially legitimised through NHS ethical approval and a small number of identified contacts within the organisation, identifying other (sometimes powerful) gatekeepers who could sponsor and legitimise the research was important and is well recognised as a lengthy process in ethnographic research (Pope, 2005). Email contact was also initiated with two consultant geriatricians and although attempts to meet were not feasible due to workloads, it was essential that they did not hear about the research project or developments second hand.

The final decision to identify the ward was strongly guided by service managers. Reasons for this guidance mainly related to stability of staffing and considerations about timing of data collection (mainly linked to movement or rotation of staff). At this point, it was critical that I maintained clarity about the aims and objectives of the research in order to notice and consider how these preferences expressed by managers or services had the potential to shape or drive the research with different organisational agendas or perspectives.

3.3.4. Patient and public involvement in research

Involving the public, including patients, in the planning, design and delivery of research, and not solely as participants, is an explicit and strategic aim of the Health Research Authority (HRA, 2013). I was aware from the outset that I had not encouraged public or patient involvement in the research idea, the generation of the research aim or question or the early design of the project - all recognised as important stages for involvement (National Institute for Health Research, 2012).

Involvement did begin at the stage of ethical approval as there was a service-user lay representative sitting on the NHS Research Ethics Committee panel. To further

develop patient and public involvement, I progressed with an intention to establish a research advisory group to aid with planning and implementation, with representation from physiotherapy, occupational therapy and service users and/or carers. I asked for volunteers from the physiotherapy and occupational therapy clinical teams within the Trust but who were not directly involved in the research. A member from each discipline volunteered and were invited to this meeting.

Repeated attempts were also made to partner with a service user and carer representative through contact with the Trust Research and Development Department, the Patient Advice and Liaison Service (PALS) and Service Managers. Despite these attempts, I did not receive any specific guidance on how to contact a service user representative and this group were subsequently not represented on the research advisory group. This was not only disappointing for the project but reflective of accepted barriers to meaningful public and patient involvement, not just in health research, but in health education, policy and indeed practice (Snape et al, 2014).

Instead of delaying the project, I progressed with an initial advisory group meeting where the physiotherapy and occupational therapy representative reviewed research design elements, such as how patient participants would be identified, how I would find out about important professional-patient interactions (in order to be in the right place at the right time), and the process of giving research information and gaining consent. The professional representatives also read examples of participant information and gave feedback. Due to one of the group members subsequently changing roles and becoming more directly involved as a research participant, this group did not meet again.

The issue of service user and carer representation was critically discussed within research supervision, with supervisors encouraging reflection on ways to overcome this. Current approaches to public and patient involvement have been criticised for

being tokenistic (Snape et al, 2014) and often driven by a consumerist model rooted in market research which aims to gain feedback in order to improve the product (Ocloo & Matthews, 2016). This reflection assisted me to move forwards and I subsequently moved away from trying to create a one-off mechanism to seek feedback, instead using more naturalistic opportunities within the orientation period of fieldwork to talk to, and involve, patients and carers. I showed copies of information to gain feedback about accessibility of format and information and chatted to patients about whether there were issues they would like to understand further about healthcare decision-making.

From both the advisory group meeting, and the informal collaboration with patients during the orientation stage of fieldwork, there were no major changes to the research design or changes in the direction of the study. However, I did feel that this helped me to prepare for subsequent periods of fieldwork, particularly the period where I would begin approaching patients for involvement as participants.

3.4. Methods

3.4.1. Methods of data gathering

Tradtionally, ethnograhers immerse themselves in daily lives and routines for an extended period of time, gathering structured and unstructured data from a range of sources (Hammersley and Atkinson, 2007). At the outset of the study, it was envisaged that the study would involve four main approaches to generate data; field observations (either focussed on activities such as meetings, or focussed on spaces such as the nursing station or a ward bay), review of secondary data (clinical records), individual observations (of health professional practice and health professional-service user interactions) and individual interviews (with health professionals and service users). However, it was here that the commitment to an overriding ethnographic methodology

was influential by emphasising direct observation as the primary method of data collection.

3.4.2. Principles of observation

3.4.2.1. Overt observation

A simple distinction between overt and covert observation is that overt observation is conducted openly and with the researcher's identity being clear to participants, whereas covert observation has not gained full consent and is not conducted with full knowledge (O'Reilly, 2005). However, O'Reilly recognises that the realistic distinction is perhaps less clear with participants potentially forgetting, not understanding the true purpose, or with motivations and interests emerging throughout the process of the research. For this study, I adopted an overt position – introducing myself as a researcher, wearing a name badge identifying myself as such, and providing information both before the research and during the course of research activities to remind participants about the purpose and focus of the research.

3.4.2.2. Participant observation - insider or outsider?

The term 'insider ethnography' was mainly used in the past to distinguish ethnography undertaken in familiar and 'home' settings, in contrast to those who travelled some distance to research cultures, societies and groups who were significantly different to their own (O'Reilly, 2005) However, there has been more recent recognition that this is not only simplistic, but the very introduction of such categories is problematic (Hammersley & Atkinson, 2007).

I reflected on elements which could assist me to move between both positions - as an insider within the occupational therapy profession, and an insider (albeit previously) within an acute hospital system. However, I was aware that I could be perceived as an

outsider to other professions, or to current clinical pracitce. I was also aware that some of the choices made during fieldwork would influence this. One significant example was choice of dress – not wearing a uniform and instead wearing a professional dress which adhered to hospital guidelines and with a visible hospital identity badge. This attracted one comment from a physiotherapist during the early stages of fieldwork that I 'looked like one of the doctors'. Coffey (1999) recognises that the presentation of self and forms of dress can communicate several different subconsicous or conscious positions, potentially declaring affinity or creating distance. I had consciously considered the distance I was creating from clinical occupational therapists and physiotherapists by not wearing a uniform, although the potential proximity to others (potentially medical staff) that this created was unanticipated.

Being aware of proximity to certain groups was something I attended to throughout fieldwork. Ethnographers are encouraged to adopt a 'marginal position' (Hammersley & Atkinson, 2007) which strikes a balance between access to participants and opportunities, but minimises the danger of over-rapport. I kept this in mind during fieldwork, for example, although I had an area to leave personal belongings within the occupational therapy department and was required to sign in and out of this department as a health and safety requirement, I deliberately did not visit the department over lunch and break times representing an attempt to draw some boundaries for this group in relation to what consituted an obvservation or a research activity.

Much of the critique of insider and outsider positions has focussed on the potential influence this then has on the susequent research process (Coffey, 1999; Hammersley & Atkinson, 2007; O'Reilly, 2005). Through engagement with this critique, I became more aware of issues affecting this position, aiming to maximise the advantages of both which would provide opportunities for acceptance and shared understanding, whislt maintainng objectivity. O'Reilly insightfully summarises this, encouraging

ethnographers to embrace the tensions of moving between insider and outsider and using this to strive to make the familiar strange, and the strange familiar (2005; p98).

3.4.2.3. Level of participation during observation

The level of participation was something I needed to consider and plan before entering the field, something that was required for the process of ethical approval. Initially, parallels were drawn with the concept of 'passive observation' (Spradley, 1980; p59) which mainly involved being a spectator within social situations, finding unobtrustive and detached positions, both physically and emotionally. This was mainly with the wellintentioned, although perhaps inexperienced, motivation of keeping any influence on patient care to a minimum, and many examples of observations in the early days of fieldwork (particularly in the orientation phase) were charachterised in this way.

The process of ethical approval encouraged me to plan for situations where I may be required to take a more active role. I was encouraged to prepare for situations such as observing an unsafe moving and handling situation, or becoming aware of a safeguarding issue. I was therefore aware that instances such as this may increase participation to 'moderate' or 'active' participation (Spradley, 1980) and that insider elements of my role such as my own Code of Ethics and Professional Conduct (Royal College of Occupational Therapists (RCOT), 2015) and my awareness of safeguarding legislation would mean I may need to intervene within such observations. Part of this planning included the identification of a 'line manager' within the organisation to whom I would report such issues and seek support.

In reality, some of these more extreme examples of participation did not emerge. However, what did emerge was that participation moved beyond a passive role in many less obvious ways. For example, I found myself moving obstacles when observing a physiotherapist mobilising with a patient, covering patients with blankets which had

fallen on the floor, or pushing the notes trolleys back to the ward after a multidisciplinary team (MDT) meeting. Coffey (1999) recognises fieldwork observations as an inevitably embodied activity, and that performing parts of the roles of those being observed is an important part of the whole sensory experience.

This move from a more passive to a more active role encouraged me to tune in to my influence within the field and encouraged me to consider issues of reciprocity; that is the extent of my involvement in give and take (Baumbusch, 2010). I considered that I was asking participants to share time, experience and reasoning with me, and reflected on the extent to which I could give things back that could assist their own practice. Whilst doing this in small ways (such as carrying things for them, or returning trolleys to the ward after a meeting), I needed to carefully recognise when this could pose risk. For example, on one occasion it was highlighted that there would be no occupational therapy representation at a meeting and a team member asked whether I could take notes for them. I needed to consider the risks of this activity (for example if I gave inaccurate information), the interruption to the status quo (who would take notes for the absent team member on any other occasion?), and importantly when it would detract from the focus of my own observations.

3.4.3. Phases of data gathering

3.4.3.1. Preparatory phase – relationship building, negotiating access and consent

The extensive period of negotiating access outlined above was also a foundation for subsequent data collection stages. It is recognised that the relationships built with people during the early stages of ethnographic research can have important consequences for way the research subsequently develops (Hammersley & Atkinson, 1995). Ethnographers need to trust those they are working alongside and vice versa (Brewer, 2000) with people in the field often being more concerned with what kind of person the researcher is, rather than the research itself (Hammersley & Atkinson, 1995).

In the weeks preceeding the beginning of fieldwork, I visited the base ward on several occasions, mainly under the premise of meeting key people and gaining written consent for observations. I also completed mandatory training which was a requirement of the organisation (such as information governance, good clinical practice in research and infection control), further legitimising my position and helping move from the position of 'outsider'. Whilst having practical reasons helped to legitimise my presence, it also helped me to consider other elements which had yet to receive consicous attention, such as dress code. This time was an important time in making the transition from 'outsider' and 'visitor' to ethnographic researcher.

Following this preparatory phase to negotiate access and to build relationships, there were then three main phases of data gathering which are discussed in detail below.

3.4.3.2. Phase 1 – orientation and social/environmental mapping

This first phase of fieldwork was carried out over a 2-week period in May 2016 with the purpose of developing an understanding of roles, processes and an orientation to the physical, social and temporal environment. As per the research protocol, I then came away from the field for a period of 1 week to differentiate between the different phases. A summary of the activities undertaken in this 2-week period can be seen in Figure 1.



Figure 1: Summary of activities during phase 1 of fieldwork

*Included attendance at 3 'Length of Stay' meetings, time spent with members of the occupational therapy and physiotherapy teams outside of the immediate ward therapists, and time spent with other services including a representative from a community reablement service, a therapist from the local intermediate care unit, a representative from a hospital based team who facilitated hospital transfers and discharges, and time spent with a bed manager.

3.4.3.3. Phase 2 – ethnographic fieldwork centered on 'patient tracking'

The next phase of fieldwork subsequently took place over an 8-week period between May and July 2016. A patient-tracking approach (adapted from Waring et al, 2014) was utilised whereby five patients were recruited and consented to be involved in the study and fieldwork activities such as observations of health professionals, attendance at meetings and review of clinical documentation focussed on the identified patients.

The two main aims of this patient-tracking approach were to provide a structure to move from more generalised observations during phase 1 to focussed observations during phase 2, and to also to give a central position within the research to the patient journey and experience. Awareness of both of these areas was developed through engagement with relevant literature. The problem of what and how to observe is regularly articulated in relation to ethnographic and case study research (Gobo & Molle, 2017; O'Reilly, 2005) with whole texts dedicated to attempts to give structure to the process (Spradley, 1980). Alongside this, the case for service user involvement as a central tenet for the research was strong.

Activities and interactions which contributed to data collection during this period of fieldwork are summarised in Figure 2. This included 10 short interviews (or reflexive interactions) with health professionals about one or more of the identified patients.



Figure 2: Summary of activities during phase 2 of fieldwork

3.4.3.4. Phase 3 - in-depth interviews with health professionals

Following a period of initial familiarisation and analysis of emerging themes, the final stage of data collection involved in-depth semi-structured interviews with five health professionals, carried out between April-May 2017. Interviews ranged in length from 40 minutes to 80 minutes, and all were conducted in private rooms away from the clinical area. The outline structure was developed after an initial period of data analysis following phase 2 and on further consideration of the study aim and objectives. An outline of the questions can be viewed in Appendix 1.

3.4.4. Sampling and recruitment

3.4.4.1. Recruitment of patients during phase 2

Using principles of purposive sampling, and adapting an approach described by Waring

et al (2014), inclusion criteria for patient tracking was used (and outlined in Figure 3) to

guide the recruitment of patients at this stage:

Figure 3: Inclusion criteria for recruitment of patients during phase 2 of fieldwork

Inclusion criteria to identify patients:

- Over 65
- Treated in the identified ward of the acute hospital for an unplanned admission
- Current functional level below pre-admission functional status as determined by the healthcare team
- Medically fit to be approached to participate as determined by the healthcare team
- Able to hear and speak to respond to interview questions

In addition, and utilising principles of theoretical sampling in order to facilitate

exploration of emerging issues from the review of literature, additional criteria were also

developed (Figure 4) with the aim of sampling people with different experiences of

healthcare decision-making.

Figure 4: Additional inclusion criteria for recruitment of patients during phase 2 of

fieldwork

Additional criteria:
 At least 2 participants aged 85 or over;
 At least 2 participants between 65-84
 At least one participant with cognitive difficulties (as determined by asking the health care team at initial screening)

Attendance at MDT morning handover meetings and weekly MDT meetings was the main means to identify patients, with subsequent follow-up discussions with the healthcare team utilised to determine appropriateness to approach to participate in the research. As almost all patients admitted to the base ward could have been potentially appropriate for the research, final decisions were based on the recommendations of the healthcare team and to allow exploration of cases with different features.

Initially, it was proposed that only one patient would be involved at any one time to enable research activities to focus on their specific experience and journey, potentially visiting different wards and services across the pathway. However, due to pragmatic limitations such as time, availability and the desire to become involved in potentially rich patient cases, in reality there was some overlap of tracking more than one patient at one time. But once again, decision-making about which participants to approach was made on review of how many other patients were involved at that time. For example, although many patients met the study inclusion criteria, they were not approached to participate as I did not feel I could dedicate sufficient time to their patient tracking at that particular time.

Throughout this 8 week period, 7 patients were approached to participate and 5 patients (or family members) provided consent. A summary of these cases is provided in Table 2. Reasons for not wanting to participate in both cases related to patients not wanting to be involved in (or in the words of one participant, 'can't be bothered by') what they perceived as bureaucratic processes – presenting as being deterred by lengthy information or consent forms.

		T			1		1		1
Patient	M/ F	Age	Social Situation	Reason for Admission	Pathway before base ward	Length of Stay on base ward	Pathway after base ward	Total length of stay	Other
1	F	94	Lives with daughter	Aspiration Pneumonia	1 day at emergency care hospital	24 days	27 days on rehabilitation ward. Died on this ward	52 days	
2	F	89	Lives with son	Fall at home and increased confusion	1 day at emergency care hospital	7 days	Transferred to rehabilitation ward. Still on rehabilitation ward at end of fieldwork (after 29 days). At this point, decision made for 24 hour care.	>37 days	Patient deemed to lack capacity to consent
3	М	76	Lives alone	Scrotal abscess Admitted to hospital with regional specialism	40 days at out of area hospital	25 days	N/A – died on base ward	25 days (+ 40 days out of area)	
4	F	79	Lives with son	Pneumonia	6 days at emergency care hospital	9 days on base ward	Discharged home with family support. Recommendation of care package although declined	15 days	Readmitted 3 days after discharge -died during subsequent admission
5	Μ	80	Lives with wife	? Seizure, ? worsening of Parkinson's Disease, ? Stroke	1 day at emergency care hospital	22 days on base ward	Transferred to rehabilitation ward. 14 days on rehabilitation ward and discharged home with follow-up reablement package	37 days	

Table 2: Summary of patients recruited for patient tracking

3.4.4.2. Recruitment of health professionals during phase 2

During the preparatory phase and phase 1 of fieldwork, I attempted to anticipate which health professionals would be most directly affected by patient tracking fieldwork activities and had gained written consent from many of these staff for their practice to be observed and to be present during their interactions with patients. In the main, this affected physiotherapists and occupational therapists, but also involved nursing, social work and medical professionals.

It was explicitly stated in the research proposal and ethical approval that full written consent for everyone who may be present during a research observation was unrealistic, mainly due to the use of temporary staff within hospital wards and the transient nature of some roles (for example, a nurse visiting from a mental health liaison team). Issues of consent pertinent to this are outlined in section 3.6.8.2.

3.4.4.3. Recruitment of health professionals during phase 3

Utilising principles of purposive sampling, seven health professionals (4 occupational therapists and 3 physiotherapists) had emerged as significant contributors during phase 2 of fieldwork and had already provided full written consent at that stage. On return to the field at this stage, it was established that one occupational therapist had since left this role within the Trust. Six participants were approached to be involved in this stage, with five - three occupational therapists and two physiotherapists – progressing to giving consent to be involved. One physiotherapist did not respond to requests for involvement.

The occupational therapists and physiotherapists interviewed were all either in Band 7 ('Highly specialist') or Band 6 ('Specialist') roles and, during phase 2 of fieldwork, had a primary remit to deliver services to either the base ward or the rehabilitation ward

(although roles had subsequently changed for some staff). One participant was male and four were female. As some of these idiosyncratic details (such as grade and gender) could potentially identify individuals within the organisation, detailed information or characteristics about participants is deliberately not presented. This however does subsequently limit the ability to discuss the influence of gender, experience or length of service in relation to the research findings.

3.4.5. Types of Data

3.4.5.1. Fieldnotes from observations

Fieldnotes are acknowledged as the main means of recording observations in ethnography (Hammersley and Atkinson, 2007). There are many ways to approach this important activity and there are not necessarily pre-determined approaches or techniques (Brewer, 2000; Spradley, 1980; Hammersley & Atkinson, 2007). And whilst essential to the quality of the research, taking contemporaenous fieldnotes has been recognised as an activity which influences the social setting, can be obtrusive and can affect a researcher's ability to engage in true participant observation (O'Reilly, 2005; Brewer 2000). Before entering the field I generated a list of elements to attend to during observations and when completing fieldnotes (Appendix 2).

Becker et al (1961) suggests that it is a good starting point to note everything – the mundane as well as the interesting – and in particular, fieldnotes developed during phase 1 were largely descriptive and concrete, noting many elements during observations such as time of day, people present (and not present), noises and interruptions, physical layout, amongst many other things. Some early field-notes also included diagrams – such as an illustration of the physical layout of the ward environment. Through immersion in the environment, developing during phase 1 and continuing in phase 2, other elements such as interactive patterns, language, and signs

of beliefs, values or motivations were added to the layers of information (Spradley, 1980). An extract from fieldnotes is included in Appendix 3.

Some of the above information was captured in short-hand in real time, although at some point during every day short-hand notes were then translated and developed in to long-hand. Care was taken to attribute information to different participants where possible, using quotation marks where I was confident of exact recall, or simply attributing ideas to people where substance was recalled although exact detail was not (Lofland et al, 2006). At this long-hand stage, I also included initial analytic ideas, and personal reflections although again, took care to identify these elements as such. These two different types of fieldnotes – that is the short-hand type produced in the participatory context, and the long-hand often completed in non-participatory, reflective (and often solitary) contexts – are a recognised part of documenting ethnographic research (Madden, 2017).

3.4.5.2. Written data sources

Daily handover sheets and the review of clinical records also contributed important sources of data. Clinical records could not be removed from the clinical environment and therefore similarly to the process of generating fieldnotes from observations, fieldnotes were also made when reviewing clinical records, noting timings, chronology, and written language and expression used by different contributors.

The daily handover sheets were anonymised and stored as an additional source of data, mainly providing contextual information about bed status on each day, numbers of planned discharges, an overview of dependency of patients, and written language used to handover information during meetings. Supplementary information about discussions within the handover meeting were added in short-hand to the handover sheet and then again, integrated in to long-hand fieldnotes. During phase 1, referring to

the daily handover sheet was used to assist with socialisation to the setting (for example understanding terminology, abbreviations, contributors), although as the research progressed in to phase 2, I then used these sheets to provide focussed information relating to the patients involved in tracking, or the handover of information pertinent to emerging issues of interest.

3.4.5.3. Interviews and reflexive interactions

During phase 2, short interviews were carried out with health professionals to discuss aspects of their involvement and decision-making in relation to identified patients being tracked. Although these interactions could be perhaps termed short interviews, they were often used to describe routine ways of seeing and doing, and to make visible the usually taken for granted and have been described in other ethnographic research as reflexive meetings or interactions (Liberati, 2015). During phase 2, 10 short, unstructured reflexive interactions were carried out with five different occupational therapy and physiotherapy professionals, eight of which were audio recorded and subsequently transcribed. Reasons for not audio-recording two of these short interviews was due to them taking place in an ad-hoc, spontaneous way which was not conducive to recording. However, long-hand notes were written immediately after these interviews. During phase 3, the five in-depth interviews were all audio-recorded and transcribed verbatim.

3.4.6. Data Analysis

There are many different types of qualitative data analysis aligned to differing theoretical and epistemological positions and providing a range of frameworks, approaches and tools. Examples include Conversation Analysis, Narrative Analysis, Discourse Analysis and Interpretative Phenomenological Analysis. In addition, insight was developed (and discussed on page 60) to the systematic approach to data analysis supported within the tradition of grounded theory.

After considering a wide range of approaches, thematic analysis (adapted from Braun and Clarke, 2006) was chosen as the analytical method in this study. In comparison to other forms of qualitative data analysis aligned to particular traditions, thematic analysis is often seen as a more general approach, compatible and sharing principles with, a wide range of theoretical positions. It has been defined as 'a method for identifying, analysing, and reporting patterns (themes) within data. It minimally organises and describes [the] data set in detail...and interprets various aspects of the research topic' (Braun & Clarke, 2006). It was this over-arching desire to identify, report and interpret patterns which informed the use of thematic analysis within this project.

As part of the data collection method was organised around patient journeys (the method of 'patient-tracking'), a biographical form of analysis – such as narrative analysis – to report the 'stories' of these cases, was considered. However, narrative analysis is recognised as placing scrutiny on the structure and content of individual cases and stories (Silverman, 2014) and this was not the focus of the research aim or objectives. Adapted principles of narrative analysis were integrated in to the overall approach to thematic analysis (in that examples from the entire data set were pulled out and related to individual patient cases), however a purer form of narrative analysis had the potential to become reductionist and detract from the core and wider focus of decision-making. A more general approach to thematic analysis instead helped to give attention to themes and patterns across the whole data so as to reflect and represent the context as a whole.

The wide use, compatibility and resultant flexibility of thematic analysis is seen perhaps as one of the significant strengths, alongside one of the most important challenges to

address. One area of criticism of thematic analysis is that it can be poorly demarcated in published work, with a lack of agreement about what it is and how to do it, and limited levels of detail in relation to decision making guiding analysis. To counter this challenge, Braun and Clarke (2006) provide a comprehensive and regularly cited guide for performing thematic analysis which has provided an outline framework for this project.

Firstly, Braun and Clarke suggest researchers reflect and respond to a series of questions to clarify their analytical position and reflective responses to these questions, originally prepared to discuss this position with supervisors, are included in Appendix 4.Subsequently, Braun and Clarke offer a framework to guide the steps of analysis, summarised in Table 3.

Stage	Examples
 Familiarisation with the data 	Extensive time spent whilst in the filed re-reading short hand notes, translating to long-hand field notes and adding analytical notes and reflections
	Critical reflection in research supervision between phases 1 and 2, and phases 2 and 3 to assist with sense-making
	Cataloguing and annotating field notes throughout – page numbering, cataloguing episodes in relation to patient tracking, and annotating with questions and memos
	Transcription of audio-recorded data
2. Generating initial codes	Initial codes included: - Meanings - Roles - Decisions - (Patient) Cases/Stories
	New catalogue of units of data in relation to initial codes Reviewing whole data set in light of research aims and objectives – for example (sometimes referred to as structural coding (Guest, MacQueen, & Namey, 2012) : - Examples of the use of the term 'rehabilitation potential' - Examples which illustrate the decision-making process

Table 3: Stages of data analysis (continued overleaf)

	 Examples relating to physiotherapy Examples relating to occupational therapy Reviewing whole data set with reference to other interesting features – for example reoccurring language. Further catalogue of units of data in relation to structural codes
3. Searching for themes	 Emergent themes included: The meaning of rehabilitation The meaning of rehabilitation potential Decision making about rehabilitation potential about the rehabilitation pathway with patients and families
4. Reviewing themes	Use of data clinics within supervision Recognition of the importance of understanding the concept of 'Rehabilitation' before understanding 'Rehabilitation Potential' (unanticipated and an example of 'thinking flexibly' discussed by Lofland et al, 2006). Recognition of ethical dimensions of rehabilitation decision making and the impact this was having on professionals Re-review of whole data set in relation to above
5. Defining and naming themes	 Descriptions Social Construction and meaning -making The ethical dimensions of rehabilitation decision making Professional roles in decision making

To summarise the table above, the analytical process began immediately through translating short-hand notes in to long-hand notes which also included researcher reflections and analytical memos. Any audio-recorded interviews (both the short interviews during phase 2 and the in-depth individual interviews during phase 3) were also personally transcribed as soon as possible after the interview to enable immersion and early analysis of this data. This was integrated in to a full data set with other examples of secondary data (such as documentation from daily handover meetings) and catalogued firstly in chronological order, and then secondly, in relation to the five patient cases.

Adopting a deductive or theoretical approach to coding, (Braun and Clarke, 2006), I then coded the full data set using codes firstly generated from the research aim and objectives. A further process of structural coding then took place. The codes developed at this stage are outlined in Stage 2 of Table 3. Examples of early coding and theme development can be found in Appendix 5.

Stages 3-5 outline the development and refinement of four overarching themes. This process evolved through ongoing immersion with the full coded data set, and was supported by elements of analyst triangulation with research supervisors outlined in the section below. The four themes are discussed in detail in chapters four-seven.

Computer assisted tools to aid data analysis were considered (for example the software package NVivo) and particularly had potential to assist to organise, categorise and code a large data set. However, as all fieldnotes were handwritten, the task of transposing this to electronic records would have been time-consuming. Instead, I utilised my own manual approach to organising and categorising the data set, which enabled early and ongoing immersion in the data.

3.4.7 Methods to write and present findings

The process of writing and presenting qualitative, and more specifically ethnographic, data is the topic of much debate (Brewer, 2000). However, from reading methodological guidance and example ethnographic studies, key features were identified as methods to aid writing and presentation and will be utilised within subsequent chapters.

Firstly, thick description of context and generic topics are common in ethnography, including physical settings, key individuals, key activities, schedules and patterns of order (Brewer, 2000). These principles have been utilised within the analysis of the

theme 'Descriptions' and the subsequent writing of this chapter (chapter four). Additionally, because the base ward was the primary 'case' or unit of analysis, a thick description of this enables a detailed understanding of context and assists the reader to substantiate analysis and interpretation in subsequent chapters.

Embedded cases of individual patients and professionals are described in part, although instead of devoting lengthy descriptions or sections to these cases, chapters have instead been dedicated to cross-case themes. Information from individual cases have been dispersed across chapters and supported by abbreviated vignettes (Yin, 2014). The use of short vignettes is supported as a technique to pick out specific players or events and written up for special description in order to present a microanalysis of an identified feature within the data (Brewer, 2000). Vignettes (placed within text boxes within the presentation of findings) were chosen because they exemplified particular phenomena at the centre of the identified part of the analysis.

Another principle embedded in the presentation of findings include the use of verbatim extracts, paraphrasing or reported speech (Woods, 1999). From discussions with patients and health professionals during all phases of fieldwork, I have either been able to present exact verbatim extracts (from transcribed audio recordings) or extracts from fieldnotes (although acknowledging such extracts are open to recall-bias and my own interpretation in fieldnotes). Exact verbatim extracts also have the additional benefit of presenting findings using the natural language of the culture, another important feature of ethnographic writing (Cresswell and Poth, 2018). The different ways of presenting such extracts are outlined on page 102. Such extracts were selected in order to illustrate breadth (where themes or topics occurred across different players, different activities or at different points in time), whilst also selecting more lengthy extracts for purposes of quality and detail (Woods, 1999).

3.4.8. Methods to strengthen the quality of ethnographic research

After long resistance to adopting terms such as validity and reliability in qualitative research, there is general agreement that such terms about whether methods achieve their desired aim, and whether findings are what they claim to be, do not cross over easily to researching the social world. Some question the need to assess the quality of qualitative research in the first place, and therefore deem the principles or terminology a distracting irrelevance. Instead, they suggest acceptance that there are no standards by which claims to knowledge can be judged and knowledge generated through qualitative research should simply be accepted as one account within multiple possible realities (Hammersley, 1998).

For those who feel quality standards are a necessary part of qualitative research, replacement terminology has led to multiple adaptations and new proposals, with credibility, truthfulness, relevance, plausibility, representativeness and legitimacy all being mentioned (Hammersley, 1998; Silverman, 2000; Brewer 2000; Gobo & Molle, 2017). This seems to add unnecessary complexity to an issue which should be relatively simple; to what extent can the researcher satisfy participants, readers and scholars that the representation of the social world reflects an accurate version of reality. I have chosen three concepts of credibility, transferability and relevance which offer explanatory frameworks for many of the strategies used to ensure quality within this study.

3.4.8.1. Credibility

Concepts of credibility have arisen from the counterpart in the positivist paradigm of validity, the extent to which the account accurately reflects the phenomenon it represents (Hammersley, 1998). It is also closely linked to the concept of truthfulness, although this term is yet again controversial within interpretivist research for being based on an assumption that a 'truth' exists (Gobo & Molle, 2017).

O'Reilly (2005) suggests that ethnographers are traditionally good at dealing with issues of credibility – due to the ongoing nature of engaging with subjects and developing areas of enquiry as the research unfolds. During this process, participants will often, sometimes subconsciously, point out misunderstandings or misrepresentations, and guide the researcher back to their own lived experience. Within the context of this research, strategies to enhance credibility included the importance of voice, the use of triangulation (both of method and of researchers), member checking, and a clear audit trail to understanding decision-making.

Voice

Constructivists embrace subjectivity and aim to deeply understand specific cases within particular contexts. To do this, it is important to give proper emphasis to the voice of participants in order to represent the phenomenon in the words of those who directly experience it and to take seriously the responsibility to communicate authentically the perspectives of those encountered during the research process (Patton, 2015). During fieldwork, I utilised a voice recorder wherever appropriate and if not, tried to develop sufficiently detailed fieldnotes which enabled words to be captured, or perspectives to be clearly attributed to particular people within the field. In the subsequent reporting of findings I have presented verbatim extracts wherever possible and even when recall may have hindered the use of exact words, have attempted to attribute statements and perspectives to participants.

Triangulation

Triangulation, can involve using multiple data collection methods, collecting data at multiple points, or utilising multiple researchers in data collection, all with the intention

of gaining a multi-dimensional view of the phenomenon. This is one of the proposed strengths of ethnographic research in that it openly encourages triangulation of methods, researchers and theoretical frameworks (Brewer, 2000). Looking at the social context using different methods and lenses, is seen not as a way of influencing validity, but as an alternative to validation (Denzin & Lincoln, 1998a).

Methods of observation and interviewing, alongside the use of reference documentation such as medical records and handover sheets were seen as a way of gaining the most rounded picture of the social context and reality and are all recognised as common methods in ethnographic and case study research. Indeed, it is the spontaneous way in which the researcher can move between research methods, providing creative methods of triangulation in order to look in to new interests or provide new perspectives on existing interests (Denzin & Lincoln, 1998b) that is seen as a particular strength of this research. If interest was raised on a particular issue during a patient interaction for example, I could then examine the clinical record to look at how this had been documented, or ask a health professional a particular question about this in a short interview.

An important aim of the use of in-depth interviews during phase 3 was to look again at the issues under exploration in order to provide different dimensions and understandings at a different point in time, and therefore utilised to triangulate between data sources. Although perhaps a little simplistic, comparisons can be drawn between Schön's model of 'reflection in action', and 'reflection on action' (1991) where interactions initially took place in real time and in context (during phase 2), and then retrospectively and in an environment away from the site of fieldwork (during phase 3). Perhaps in an ideal world, I would have liked to have been able to explore ideas from data analysis through further and more extensive periods of fieldwork. Undertaking a further period of patient tracking, or carrying out further patient tracking following the period of organisational change, were both considered and discussed within
supervision, although were primarily restricted by pragmatic workload demands of carrying out a part time PhD alongside a full time role.

Finally, analyst triangulation was also employed - a method whereby multiple analysists are used to review findings (Patton, 2015). During fieldwork, I discussed early fieldnotes within supervision meetings in order to assist with initial analytical processes. At later stages, supervisors also independently reviewed anonymised interview transcripts and developed their own coding suggestions and analytical notes to share and pose questions. Although recognising that such meetings did not happen regularly and supervisors could not achieve the level of immersion required if carrying out full data analysis, the role of 'critical friends' has been noted as an important role which can be performed by people who fully understand the context of the work but can ask provocative questions and examine work through an alternative lens (Patton, 2015). Notes from such processes were included in data analysis.

Member-checking

Member-checking is proposed as a strategy to work towards credibility and truthfulness by revisiting descriptions and meanings with participants and reflecting on how they understand and respond to the researcher's construction of these concepts (Gobo & Molle, 2017). Alongside a way of extending the data, the phase 3 interviews were also utilised as a form of participant validation or member-checking to strengthen credibility. Themes from the initial analysis of the data were discussed with participants to encourage reflection on congruence, resonance or to open up issues for challenge. Critical consideration was given to the outline of the semi-structured interview schedule (Appendix 3) and discussed in depth within research supervision to place emphasis on co-interpretation rather than looking for agreement or validation, which can sometimes be seen as a danger of member-checking.

<u>Audit trail</u>

Throughout this thesis, and particularly within this Methodology and Methods section, I have presented information about decisions made and activities undertaken with supporting justification. This links to the concept of dependability, and underpins credibility, in that it is the researcher's responsibility to convince the reader that the process was logical, traceable and documented (Patton, 2105). Examples have also been provided within appendices for the audience to also view examples of fieldnotes, interview schedules and stages of data analysis.

3.4.8.2. Transferability

Qualitative researchers have long faced the challenge that they cannot claim to generalise without the inbuilt features of quantitative research, mainly relating to sizes of samples (Gobo & Molle, 2017). Some create distance between qualitative research and the idea of generalisation (Denzin, 1983), while others recognise there is merit in discussing qualitative generalisation, often using the term transferability (Gobo & Molle, 2017).

Firstly, I hold the belief that instances and situations observed during one local context are of interest in their own right and therefore claims about transferability are perhaps less important than the account holding value for this context. However, Gobo & Molle (2017) suggests that knowledge manifests in particular ways in local situations and it could therefore be hypothesised that, with sufficient knowledge of that local situation, an audience can make a judgement whether this knowledge may then apply to situations with shared characteristics. I have therefore attempted to describe in detail some of the features of this local context for people to judge how this knowledge could be transferred to wider settings. Also, because many of the issues are not unique to the local context (such as issues pertaining to hospital use, ageing populations and

financial pressures to name but a few) where appropriate, tentative suggestions are made about issues which may affect wider groups, disciplines or organisations.

3.4.8.3. Relevance

Hammersley, a leading author in qualitative ethnographic research, suggests that one of the main standards by which ethnographic research should be judged is that of relevance (1998). He recognises this as perhaps a lesser talked about criterion although discussed that whether the research is of actual or potential relevance should be a primary concern of researchers. This links to concepts of usefulness and originality cited by other authors (Charmaz, 2006). Within this, Hammersley suggests there are two main aspects for a reader to judge; the importance of the topic and the contribution of conclusions to existing knowledge (1998).

It was the purpose of the background and literature review section to outline the importance of the multiple layers of this topic, and the rigour of a research process – in this case a PhD research process involving ethical approval, supervision and annual progression points – to ensure I revisited this issue with regularity and confidence. It is also the purpose of the literature review to illustrate a picture of what is already known, and the purpose of the discussion and concluding sections to explore relationships to existing studies in an accessible way. However, researchers and readers alike must take care not to associate the level of contribution with the extent to which findings confirm or contrast with existing knowledge. Particularly relevant for social constructionist researchers, the commitment to deep understandings of specific and localised contexts should not be forgotten (Patton, 2015), and should not confuse this with a wider search for relevance (Hammersley, 1998).

Whilst the overall case for relevance is yet to be read, and will develop through the subsequent discussion of findings, interpretations and conclusions, another insightful

although perhaps less obvious way to consider relevance links to the concept of praxis. Praxis refers to the transformative power of the research and the researcher, and the relevance this had for the practice area (Baumbusch, 2010). Baumbusch recognises that the presence of the researcher can contribute in small and incremental ways to either individual, service or team development and that having the clinical background to appreciate this is an important component.

There were a number of instances where I did feel that participation within the research was having a contemporaneous impact on practice. On one occasion, after a research discussion about their involvement with a particular patient, one physiotherapist commented on how interesting it was and that 'he might do a reflection on it'. And another participant, at the end of the phase 3 in-depth interview concluded:

"...Because you don't really think that you're doing all of these things as you're doing them. But when you talk about them and you reflect them, it is quite indepth isn't it...it is a lot to think about (laughs)" (I4, Line 352).

Although I cannot make causal claims that my research involvement led to such reflections, or that this reflection may not have occurred irrespective of participation in the research, I could not help but feel that engagement in the research process was emphasising the importance of reflective processes. As reflective processes are linked to learning from experiences and looking for alternatives (Howatson-Jones, 2013; Johns 2013) and have been linked to quality within health care practice, this was a salient issue linked to relevance.

3.4.9. Ethical Considerations

As in any research, ethnographers have a responsibility to protect research participants from harm, placing participant rights at the forefront of research endeavours (Gobo & Molle, 2017). When conducting research in healthcare settings, initial activities are

often centred around gaining necessary ethical approvals meaning that considerable time and attention is given at this early stage to anticipating ethical issues and outlining how they would be addressed if they arise (Goodwin et al, 2003). Detail in this next section outlines the development of such processes in relation to anticipated ethical dilemmas and issues, alongside discussion of those dilemmas that arose spontaneously.

3.4.9.1. Ethical Approvals

Ethical approval was initially granted by my own institution (Appendix 6), and subsequently, following submission of extensive supporting documentation (particularly attending to issues of involving vulnerable people as research participants and presenting a protocol which met requirements for clarity but also allowed for flexibility of design and methods), granted by the NHS Research Ethics Committee (Appendix 7). Permissions were then granted from the local Trust Research and Development Department to be on trust premises as a recognised visitor and researcher.

Many procedural aspects of sound ethical research practice were required to be addressed for the purpose of ethical approvals. Rather than focus on many of these procedural aspects (such as the storage of data, the maintenance of confidentiality and anonymity), key elements of establishing consent and issues of position and power will be explored in this section.

3.4.9.2. Consent

Health professional consent during phases 2 and 3

The stages outlined in relation to negotiating permissions and access were integral to the consent process. It was vitally important that potential gatekeepers provided their own verbal consent for me to enter their service or to talk to their teams.

Prior to phase 1, senior clinicians (namely the ward manager and senior therapists) provided help to identify any member of staff who was likely to be directly affected by research observations. For occupational therapists and physiotherapists (and including assistants), information sessions were held in small profession-specific groups, written information was provided and the groups were given the opportunity to ask questions. Consent forms were left with individuals and were returned to the researcher individually.

For nursing staff and nursing assistants, as it was difficult for people to have time away from the ward in groups, the ward manager provided the staffing rotas during the preparatory phase and I attempted to meet with all staff, either individually or in small groups as the demands of the ward would allow. Again, written information was provided and all participants were given the opportunity to ask questions. Participant information and consent forms can be found in Appendices 8-13.

There was also a recognition that there would be many other people who may become indirect participants within the research – such as visitors to the ward, and transient members of the healthcare team, such as chaplains, pathologists, or on-call doctors. For this reason, large print notices were displayed in public areas and left on patient tables, to highlight that observational research was being undertaken on the ward. Furthermore, during every interaction or observation, I was diligent to introduce myself as a researcher, explain the purpose of the observation and ask for verbal consent from those involved to be present.

During phase 3, health professionals were approached to participate in an in-depth interview, were resent participant information, and, in part due to the length of time between phase 2 and phase 3, were asked to sign a new consent form.

Patient consent during phase 2

For patients who were approached to participate in the patient tracking phase, I approached the patient independent of any introduction from a member of the healthcare team. This was a deliberate first step to avoid issues of perceived alliances with professionals. At this first stage, the researcher provided verbal and written information, including an 'at a glance' written summary and read information out-loud if required. The patient was given opportunities to ask questions and left for a minimum of 24 hours to consider their involvement in this study. Although initially hesitant that this would compromise my ability to be involved in real-time acute health care decisionmaking, this 24-hour time period was a requirement of the University and NHS Research Ethics Committee approval. After 24-hours, I returned to the patient to judge whether the person had understood and retained the information and if they were able to give consent. For those able to give consent and willing to participate, the written consent form was completed at this stage. However, although written consent was obtained prior to any involvement, a 'process-consent' (Dewing, 2002) approach was also adopted with all patient participants, whereby at each point of contact, information about the purpose of the observation or interaction was revisited and permission sought to continue on that given day.

Patients who were not able to give consent

When potential cognitive impairement (either short term related to an issue such as delerium, or longer term impairment) had been identified by the healthcare team, I still approached the participant with information about the study as a first stage. This was in

line with principles of the Mental Capacity Act (Department of Health, 2005) which stipulates that capacity is specific to each decision, and should be assumed to be present until assessed otherwise. Initial discussion and sharing of information enabled me to determine whether the person was able to understand and retain the information either at the end of the first interaction, or on return after 24-hours.

Out of five patients, four provided consent following this process. For one patient, she was unable to understand or recall information at the end of the first interaction, therefore the process for seeking advice from a consultee was followed (Appendix 14). In this case, the patient was still given information about the study and assent was sought, and her son acted as a consultee and advised that he thought his mother would have no objection to being involved. The outcome of this consultee advice was recorded using the consultee form (Appendix 15) and this patient was included in the study. It is important to note that, for the four patients who were able to consent, capacity to consent and ability to communicate wishes was not always obvious mainly due to frailty and hearing problems. For one patient where this was particularly notable, her daugher was present during all interactions and gave her own support of involvement in the research.

3.4.9.3. Issues of Power

At the research planning stage, I had reflected that my role as a qualified health professional and a legitimised researcher within the setting potentially introduced power dynamics within my relationships, particularly in my relationships with patients. This was particulary pertinent as I would be interacting with patients at a time of significant vulnerability during an acute hospital admission. Issues to minimise the issue of a potentially powerful research postion were given signifcant consideration during planning and ethical approval (with many of the strategies such as introductions which were independent of health professional interactions and adopting a processconsent approach outlined above).

A power issue I had not anticipated in as much depth was the issue of power created through my role as a university lecturer and the past relationships I had with some of the occupational therapists when they had previously been students. During an observation, approximately mid-way through phase 2 of fieldwork, I observed one such occupational therapist undertaking an assessment with a patient. As we walked away she shared her own reflection that 'it's like being a student again'. Although we briefly laughed and continued talking, this encouraged me to question whether she, and others in this position, may be experiencing pressure because of the perception that their practice was being observed by a university tutor. Whilst not being in a position to eradicate the emotional responses arising from this shared history based on student-tutor relationships, it did encourage ongoing sensitivity to this issue. It reminded me of the importance of continuing to revisit the purpose of my role and of negotiating consent so as not to capitalise on, or exploit these relationships (Goodwin et al, 2003).

Another issue of power that I have reflected upon is one that is documented in critique and commentary of ethnographic research as 'privilege' (Coffey, 1999). In ethnographic research, the observer is often cast as a privileged 'other' who has the luxury of time and resources to explore the social world, and claim representations as their own. During the course of fieldwork, I recognised the privileged position I was in to be able to afford time and cognitive resources to such in-depth observation and analysis. This was particularly felt when contrasted with the busyness of the ward and of the time and resource pressures which were being experienced by all (explained in more detail in the reporting of findings). Whilst elements of this were inescapable, I became mindful that the representation of findings should not be detached from participants, but should aim to resonate with participants in the field.

3.4.9.4. Avoiding Harm

Many of the examples of which patients to recruit, processes to gain consent, and approaches to difficult conversations or approaching ambiguous topics were driven by an overriding concern to avoid or minimise harm. Although activities within fieldwork were planned to some extent, some activities and elements were serendipitous and therefore unpredictable. In other ethnographic accounts within healthcare, it is discussed that in unpredictable situations, professional codes of conduct should be used to guide research practice (Goodwin et al, 2003) and avoidance of harm is widely cited as a founding principle of healthcare ethics (Beauchamp and Childress, 1989) and of my own code of professional conduct (RCOT, 2015).

The fact that it is often reported as 'first, do no harm' (McLean, 2016; General Medical Council, 2015) provides support that this ethical principle overrides others and explanatory offerings for my own decision-making, particularly in relation to selection of patients to approach for involvement, and managing the potential for illuminating challenges with decision-making. However, the extent to which this principle reflects contemporary healthcare which encourages positive risk-taking and attempts to move away from professionally-led decisions and interventions has received critique (Walton & Kerridge, 2014).

3.5. Chapter summary

This chapter began with a discussion of social constructionism and how this underpins the research. A commitment to naturalistic and qualitative research emerged, promoting the exploration of meanings, interactions, social constructions and behaviours in the context in which they occur. Principles of ethnographic and case study research were presented. A critical, yet pragmatic overview of methods chosen and applied was discussed Because of emphasis given to observation within ethnographic research, detailed and specific consideration was given to this as a method to generate research data, focussing on research decisions such as the type of observation utilised, the level of participation adopted, and the essential activity of recording notes during episodes of observation. The chapter concludes with sections discussing qualitative validity and strategies to ensure ethically sound research.

In its entirety, the chapter (and supported by detailed appendices) presents an audittrail of decision-making, an insight in to alternatives considered at each stage, and balanced evaluations when reaching decisions in real-world research.

CHAPTER FOUR – RESEARCH FINDINGS: DESCRIPTIONS

4.1. Introduction to research findings (chapters four to seven)

This is the first of four chapters presenting research findings. Writing an ethnographic account can take many forms including: realist 'tell-it-like-it-is' approaches; the creation of 'ethno-dramas' of possible worlds; or documenting internal dialogues and commentaries from the researcher (Brewer, 2000). Principles of what has been termed a 'post post-modernist' approach to ethnographic writing have been adopted which aims to adopt an authoritative voice, but acknowledging the naturally selective nature of the instances chosen and phenomenon represented (Brewer, 2000). Brewer suggests strategies such as presenting persuasive and evocative narratives and use of exemplary vignettes can enable a rational presentation and assessment of findings whilst not disguising them as infallible or absolute.

Where there is certainty about exact words and who they were spoken by (by referring to audio recordings), this is indicated clearly within the presentation of findings by use of double quotation marks and attributing the verbatim extract to an interview participant. Where the researcher was relying on her own recall and translations within fieldnotes, single quotation marks are used within the text although where possible, this is again attributed to an identified person within the field. If fieldnotes did not identify quotes or extracts as attributable to particular individuals, but instead reflected general discussions or perspectives from multiple players, this is again denoted. Where narratives are used from patient tracking, the extract has been placed within a text box. Finally, where a topic represents my own impression formed within the social context, this will be attributed as such. Table 4 illustrates how these different elements are represented within the findings chapters.

Table 4 - Examples to illustrate the presentation of findings

Example	Explanation	
"the word rehabthey're going to a rehab ward and there's physio's there" (I4)	A direct quote, supported by a transcription of an audio-recorded interview or vignette, attributed to interview participant 4.	
And this issue was discussed by the base ward physiotherapist, highlighting that a patient could potentially make 'good progress with intensive rehab but don't feel we have that here'	Attributed to the base ward physiotherapist, although the researcher cannot be fully confident of exact wording due to issues of recall and her own translation in field-notes	
therapists on the rehabilitation ward discussed that they can experience a lack of clarity, and sometimes anger, from patients or relatives linked to unrealistic expectations of rehabilitation	Concept not attributable to one identified person within the field, but discussions with groups or multiple people reflected in fieldnotes.	
Patient 5 was transferred to the base ward 2 days after his emergency admission for a suspected stroke or seizure but what was later determined to be a worsening of symptoms associated with Parkinson's Disease	Example taken from tracking of patient five, with references within the text box to objective information taken from clinical records alongside discussions and observations with patients, family members and professionals.	
It was my impression that this normally resulted in daily support for more junior medical staff, and new patients and urgent issues being reviewed in a timely way	The text represents a personal impression formed by the researcher, linked to features within the social context.	

4.2. Chapter introduction

Fieldnotes from the all phases of data collection are used in this chapter to present a thick description of environments, roles and activities, all with the aim of building a detailed understanding of this context. 'Thick description' (Geertz 1973, cited in Lincoln & Guba, 2000) – an appropriately explained base of information about the salient

features of the context - assists the inquirer and reader of research findings to make judgements about transferability and fittingness to their own areas of interest. Within ethnographic research, explanations and theory-building cannot be attempted before phenomena are defined and described (Hammersley, 1998).

This first findings chapter will therefore begin with description of the study-site, before describing the main professional roles, and an outline of a typical day. Changes to the study site between phases of data gathering will also be described in this section, illuminating salient issues affecting practice and decision-making at different points in time. This section will then conclude with a descriptive presentation of the stages involved in evaluating rehabilitation potential.

4.3. The study site

The research was situated in a large NHS Trust, providing hospital and community services (in addition to some locality based social care services), serving a large population and geographical area. At the time, hospital provision occurred across an emergency care hospital, alongside 9 other inpatient sites including general hospitals, community hospitals and smaller units. The particular focus of the study involved fieldwork within one of the general hospital sites.

4.4. The base ward

One ward was identified as the main focus for the study, hereafter referred to as the base ward. Patients were admitted to this ward (mainly from the emergency care hospital) for treatment and management of general medical issues. The hospital had other established pathways for the treatment and management of condition-specific admissions such as stroke, fractured neck of femur and cardiology conditions,

therefore the most common reasons for admission to this ward included treatment for co-existing conditions and co-morbidities including falls, infections and delirium. The ward was referred to most commonly as a 'Care of the Elderly' ward, although was also referred to as an 'Acute Medical' ward and there was a recognition that it had a remit which combined both of these types of service. Although the ward accepted people over the age of 18, due to the demographics of those requiring inpatient hospital services, the majority of individuals were over 65, with a high proportion of patients over 85.

The majority of admissions to the ward originated from the emergency care hospital site. There were occasional admissions from other wards within the same site, and occasional direct admissions from out of area hospitals. The ward did not accept direct admissions from the community or from other services such as the intermediate care unit (described below).

During fieldwork, many staff shared perspectives about the current nature of the ward, particularly since the opening of the emergency care hospital. Anecdotally, staff shared reflections that the dependency, medical instability and complexity of patients had recently increased. Bed managers also acknowledged that they tended to now use this ward for people who were more medically unwell, but who needed to move on from the time-limited emergency care site.

Staff also shared the perspective about how this had influenced the purpose and function of the ward. On a couple of occasions, staff started conversations or reflections with 'when we were a rehab ward', with the subsequent discussion suggesting that rehabilitation was no longer a primary focus due to the dependency and complexity of patients. Staff suggested that when patients moved in to a phase more primarily focussed on rehabilitation, they would often be transferred elsewhere.

4.4.1. The physical environment of the base ward

The ward had physical space for 28 beds (with four six-bedded bays and four single cubicle rooms), although during fieldwork, was operating with a capacity of 26 beds. The beds utilised depended on the amount of male/female patients, the need for single rooms and other individual factors such as the need for visibility from the nursing station due to falls risks. A diagram of the layout of the ward (not to scale) is provided in Appendix 16 to illustrate the spatial layout of the ward. Some important features of the physical layout are described below.

There was no dedicated area on the base ward for therapy or rehabilitation activities. Patient bays, corridors, beds and bathrooms were utilised for assessment or practise of mobility, transfers, and personal care. Patients were accompanied off the ward to other environments for activities such as step or stair practice, or to carry out activities in an assessment kitchen or bathroom.

The nursing station was a central, and public area, and was staffed by the ward clerk between 8.30am-4.30pm. Many functions were performed at the nursing station, including receiving and making telephone calls, admitting ward visitors using the intercom and using desk space to complete clinical records. It was in close proximity to a treatment and medication room and to many of the patient cubicles and bays. These features contributed to it being a place of congregation with many interactions and opportunities for information exchange.

In front of the nursing station was a large whiteboard listing the name and allocated bed of every patient on the ward and also indicating individual information about issues such as mobility status, falls risk, and an Early Warning Score system (EWS) based on vital signs. This was another central and public point within the ward environment and a place which encouraged convergence and information sharing.

On the periphery of each bay was a notes trolley containing patient medical notes for those within the bay. Notes trolleys could be moved around the ward for the purpose of ward rounds or Multi-Disciplinary Team (MDT) meetings. However, most often they remained static in the position adjacent to each bay and again, provided another junction for small meetings and information sharing. Staff could often be observed standing and writing records, using the trolley as a substitute for desk and office space. Small groups would often take opportunities to converge and it would be common for relatives to attempt to discuss issues with staff at these points.

4.5. Linked services and care environments

4.5.1. The rehabilitation ward

This was a 24-bedded ward on the same general hospital site although again, bed numbers fluctuated with flexibility to increase and open beds in response to pressures. Most admissions to this ward were internal transfers from other more acute wards on this site, and most transfers were based on the premise that this ward provided a slower stream and more intensive approach to rehabilitation for older people with complex needs.

The physical environment differed to the base ward in that it had a larger number of single cubicles and some 2-bedded and 4-bedded areas. It also had a dedicated therapy area on the ward with physical rehabilitation equipment such as parallel bars, steps and pedals.

4.5.2. The bed-based intermediate care unit

This unit was in the same locality area as the general hospital, although not on the same site. The unit was jointly funded by NHS and Social Care funding, with staff provided through different funding and contractual arrangements, and premises managed by the Local Authority. The overall unit consisted of 2 areas – a downstairs area with 12 beds allocated for orthopaedic admissions and an upstairs area with 13 beds for general care of the elderly admissions.

All admissions to these beds came from one of the NHS Trust hospitals and although historically, the unit had accepted direct admissions from the community, this was no longer permitted. The unit had the explicit aim of being a 6-week assessment and rehabilitation service although staff acknowledged that this specific timescale was not always possible to achieve or enforce.

The unit had dedicated nursing cover between 7.30am and 8pm although overnight there was no qualified nursing staff onsite. Consultant geriatricians from the NHS Trust also provided medical cover for this unit although did not visit every day.

4.5.3. The reablement service

The reablement service was a short term support service, funded by Adult Social Care. The service had a stated purpose of supporting rehabilitation and convalescence following a period of illness, hospital admission or following the onset of a new disabling condition. Similar to the residential intermediate care unit, the service provided short-term support for a maximum of six weeks.

The service consisted of 'locality leads' in co-ordinator roles - who had responsibility to carry out initial assessments and establish and review plans and packages of support appropriate to needs – and supported by scheduling officers and a large team of

support workers. There was no occupational therapy or physiotherapy personnel within this service.

Referrals could be made by any health and social care professional although in reality, the hospital social workers had the main responsibility for referring a patient who was being discharged from a hospital ward, and this was discussed mainly for pragmatic reasons in that they had access to the local authority systems. If screened to be appropriate, a short term reablement package could then be put in place quickly (normally within 24-48hours).

Professionals discussed the potential benefits of the reablement service in terms of accessing packages of home support within short time scales, alongside promoting ongoing work towards functional goals, such as returning to outdoor mobility, or becoming more confident with a new walking aid. Some frustration was expressed with the limited opportunity to communicate specific active rehabilitation goals, and also with the absence of rehabilitation professionals integrated within the team. An example of this discussed by a physiotherapist was that if specific physiotherapy input was required to review range of movement following a fracture, a separate referral to community or day-service physiotherapy would also be required.

4.6. Occupational therapy and physiotherapy staffing for the base ward and the linked rehabilitation services

The occupational therapy and physiotherapy staffing for the base ward, and in comparison to other rehabilitation services is presented in Table 5.

Table 5: Occupational Therapy and physiotherapy staffing for the base ward,

rehabilitation ward, bed-based intermediate care unit and reablement service.

	Base Ward	Rehabilitation Ward	Intermediate Care Unit	Reablement Service (Community)
Occupational Therapy	1.0 Band 6 Access to pool of assistants Weekend cover from hospital-wide service	 1.0 Band 6 0.5 Band 6 Access to pool of assistants Weekend cover from hospital-wide service 	1.0Band 61.0AssistantBoth staffcovered bothorthopaedic andelderlyrehabilitationunitsNo weekendcover	No dedicated service
Physiotherapy	1.0 Band 7 (also with service lead responsibility) Access to pool of assistants Weekend/ evening/night cover from hospital-wide service	0.6. Band 6 1.0 Band 5 1.0 Assistant Weekend/ evening/night cover from hospital-wide service	 1.0Band 6 (orthopaedic unit) 1.0 Band 6 (elderly rehab unit) 1.0 Assistant covering both units No weekend on on-call cover 	No dedicated service

4.7. Roles and working patterns within the base ward

4.7.1. The Consultant Role

Two dedicated 'care of the elderly' consultants had overall medical responsibility for the patients on the base ward with the aim of at least one consultant visiting the ward on a daily basis. Consultants did not allocate specific patients (or bed spaces) between them, but instead whichever consultant visited the ward on the identified day, would

see any new patients, review issues of medical urgency, and discuss any queries or plans with the ward medical, nursing and therapy team. It was my impression that this normally resulted in daily support for more junior medical staff, and new patients and urgent issues being reviewed in a timely way. Both consultants had many other additional clinical, teaching and leadership roles across the trust.

4.7.2. Ward doctors

The base ward had four junior doctors allocated to cover all medical issues on a daily basis. In reality, with these staff also on on-call rotas for night and weekend cover, or away from their duties to attend teaching or sit exams, there were many occasions when there were less than four, sometimes when there was only one on duty. Ward doctors normally worked between 9am-5pm, with on-call staff covering at other times.

4.7.3. Nursing

Nursing care was provided by a team consisting of 14 qualified staff (Bands 5-7) and 11 healthcare assistants (Bands 2-3). 24 hour cover was organised around three main shifts – early, late and night shift – with staff also occasionally working long days, or 'office hours'. During fieldwork, the ward had between 2-4 qualified nursing staff on duty during early shift and most often 2 qualified staff on duty for late shift and night duty, ideally supported by 4 healthcare assistants. In reality, staffing was often reduced by planned and unplanned leave, sickness, and the requirement to attend training or meetings.

Nursing staff worked in teams, covering an identified 'side' of the ward. Each side incorporated two of the six-bedded bays and two single-bedded cubicles. In general, I observed attempts to keep staff on these identified sides on different shifts to promote continuity of care, however, there were many occasions where this was not possible.

4.7.4. Physiotherapy

The physiotherapy team covered the base ward between 9am-5pm. There was no formal referral system and instead the physiotherapist monitored all patients within the base ward and made judgements about prioritisation. The physiotherapist discussed that a referral system had been considered, although used the example that the team did not want to be in a position where they found out key information, such as a patient having stairs, at a later stage. Therefore, the team operated a 'blanket referral system', meaning they screened all patients, and then made judgements about appropriateness for, and level of, involvement.

The ward physiotherapist discussed the model of working and general rules that informed workload management and prioritisation. These general rules included the aim to see all new patients on their first working day within the ward, and aiming for every patient to be seen every weekday by a member of the team (which included assistants). However, the team recognised that this was often not realistic or appropriate, and on a daily basis, priorities were reviewed and negotiated, normally based on information from the morning handover and from continuous information sharing during the day.

Outside of the normal working hours described above, on-call respiratory physiotherapy cover was available during evenings, nights and weekends, and 9am-5pm cover for a standard physiotherapy service was also available at weekends through a rota of existing staff. Reasons for being seen by physiotherapy at weekends included assessment for issues which could facilitate discharge (for example mobility and stair assessments), to continue to promote carryover of progress (for example to practice standing or transfers), or to assist in non-urgent management of respiratory issues (such as supporting ward teams to ensure that patients at risk of respiratory complications were sitting out of bed during periods within the weekend days).

Physiotherapists did have departmental office space in another area of the hospital and also utilised a physiotherapy treatment area a short walk from the base ward. The allocated physiotherapist also spent time on many other wards providing supervision and cover for other staff. Despite this, they could often be found on the base ward and discussed largely basing the working day within this location.

In their own words, one physiotherapist described this role as:

"working with them [patients], and goal setting with the patients to just try and get maximum function and independence really. And...with...from a physio point of view then that's, you know, transfers, mobility, things like that" (I1, line 32).

4.7.5. Occupational therapy

The allocated occupational therapist for the base ward provided services between 9am-5pm, Monday-Friday. The occupational therapist operated a referral system from ward staff whereby ward staff used a referral book to highlight when a patient required occupational therapy and to provide basic information about the patient's situation. A box on the daily handover sheet would indicate if the patient had been referred to occupational therapy. The occupational therapist was also observed to operate outside of this system and to initiate the referral themselves if they felt this was warranted either from the morning MDT handover, or from information received through informal ward interactions.

No evening or overnight occupational therapy provision was provided. The wider occupational therapy team operated a rota-based weekend service from existing staff, with weekend referrals initiated from either ward based occupational therapists on a Friday for urgent involvement over the weekend, or from ward-based staff referring newly admitted patients. Priority for weekend involvement focussed on those patients who had the potential to be discharged during the weekend, or early the following week if the assessment and planning took place over the weekend.

The occupational therapist was also normally present on the ward on a daily basis, alongside having a departmental base in another area of the hospital site. They were observed to work more regularly from the departmental base than the physiotherapist, and this was acknowledged by the occupational therapists themselves, with the difference in the way of working attributed to the increased need for visits to patient's homes, the use of facilities within this departmental base (such as the assessment kitchen and bathroom areas) and the need for a quieter environment to liaise with family, carers and a range of services through telephone contact.

Again, in the words of one of the occupational therapists who participated in this study, insight in to the role is provided below:

"a lot of our role does centre around discharge facilitation...completing initial assessments to see how they were managing in terms of mobility and such like. How they were managing with ADLs, possibly doing further assessments...kitchen assessments, washing and dressing assessments etc. And em...my...and then considering whether they need support at home, any equipment, adaptations, that sort of thing". (I5, line 6)

4.8. A typical day

In the context of this ward, activities were observed as happening at different times of day and night. However, for the purpose of this section, activities that were observed occurring regularly throughout Monday-Friday daily routines are summarised.

Early morning activities included a shift handover between the nursing staff on night

duty and the nursing staff beginning an early shift. This would take place at

approximately 7.45am, before night staff finished their shift at 8am. Following this,

breakfast was provided to all patients, some patients were assisted with personal care

and toileting, and observations and medication administration commenced (although this could also commence later depending on staffing, the need to meet hygiene and toileting needs and the urgency of any other issues).

Doctors, occupational therapy, and physiotherapy staff normally began arriving between 8.30am and 9am and if there in advance of the 9am handover, could often be seen at the nursing station, often reviewing records. The MDT handover took place at 9am on Monday-Friday and staff were largely diligent about observing the timetable for this activity. This would normally be completed by 9.30am.

Between 9.30am-12pm, I observed a contradiction of activity. From one perspective, staff presented as being very busy with a multitude of activities happening simultaneously and with significant speed. An example of this could be nursing staff performing quickly and simultaneously the administration of medications, responding to patient-call alarms, holding conversations with patients and professionals, assisting with observations or other procedures and responding to telephone queries from relatives about how a patient had been overnight. Ward doctors and therapy staff also discussed that the time before lunch was the most effective time to make progress with direct patient contact: discussing that this time would not be interrupted by visitors, and the completion of assessments and investigations at this time could often facilitate afternoon discharges. Consultants often attended the ward during this time to lead medical ward rounds, although this attendance was less predictable with the ward round often taking place without the consultant, and attendance later in the day used to support and review issues with the ward doctor and wider team. Rehabilitative interventions could also be observed during this time, such as patients engaging in walking practice with a member of the physiotherapy team, or being assisted with transfers out of bed by nursing staff and therapists.

From another perspective, the contradiction was seen when focussing observations on any one particular patient, where the activity involving each individual could be limited and fragmented. Within this time an individual could be typically assisted with personal care, be visited by the ward doctor during the ward round, visited by the physiotherapist, and provided with a mid-morning hot drink. However, these interactions characteristically tended to be short (perhaps no longer than 15 minutes and potentially as short as a few seconds) and interspersed with much longer periods with no interaction. As many patients were frail and still in the acute stages of a hospital admission, many could be observed sleeping or sitting quietly during this time. Some were able to engage in self-directed activities such as reading a newspaper. Later in this period (between 11am-12pm), nursing staff were scheduled for timed breaks and once essential routine care activity was completed (for example personal care, medications, observations and procedures), breaks would commence, again contributing to a quieter time for individual patients.

The period between 12pm-1pm was identified as a 'protected meal time', a practice which is used to enable patients to eat meals without interruptions from hospital staff and to free up staff to assist patients with feeding and eating. It was my perception that a commitment to protected mealtimes was to be observed in the main and therapy and medical staff often used this period to have their own break.

The period between 1pm-2pm was observed to be an extension of the earlier period, where assessments and routine care activities resumed. A nursing shift handover would also take place around this time between staff working an early shift, and the late shift commencing at 1.45pm. At 2pm, the first of two 1-hour visiting periods began, although staff often commented that visitors would often attempt to arrive earlier and a queue could often be observed at the door from approximately 1.30pm. During the visiting period, many staff could still be seen to be present on the ward although alongside the routine care activities, communication with relatives also became

significant, and professionals who emphasised their role in understanding home and social situations – namely occupational therapy and social work – were perhaps more often seen at this time. Staff also used this time for indirect patient activities such as record keeping and telephone calls, discussing their awareness of the importance of visiting time and the desire to keep professional interruptions to a minimum.

Other ward and hospital activities were also scheduled during afternoons, such as the weekly MDT meeting, the three times weekly Length of Stay meetings, and staff supervision and development activities. After 3pm, and after the end of the first visiting period, some assessments and care activities resumed although this period after 3pm did not have the same sense of busyness or activity level, with professionals discussing that the highest priority activities often took place earlier in the working day to provide the maximum time to resolve issues or facilitate a transfer. This sense of reduced activity continued until late afternoon and evening time with the protected evening meal time happening between 5pm-6pm, and the second visiting period between 6.30pm-7.30pm. This then led in to the nursing handover and the commencement of a quieter evening routine.

4.9. Description of key activities observed during fieldwork

4.9.1. Multi-disciplinary team (MDT) daily handover

The MDT handover took place on the base ward at approximately 9am every week-day morning. It lasted approximately 30 minutes and was led by a qualified nurse. In addition to the nurse(s) who took the lead role for this meeting, it was attended by core members of the MDT including ward-based medical, occupational therapy and physiotherapy staff. Alternative occupational therapy and physiotherapy staff sometimes attended if the identified therapist was not on duty and there were

sometimes occasions where there was no representation from identified disciplines. One of two identified social work staff were also often in attendance, although notably less frequently than other staff. On less frequent occasions, one of the allocated consultants or health professional students also attended.

The handover meeting was held in a corner of the ward; the environment was discussed by the ward manager as being chosen to promote privacy and confidential discussion, although still within view of the nursing station and the white-board. Although chosen for reasons of privacy, this was not a closed area and was within very close proximity to some patient areas.

The meeting was structured around a daily printed handover sheet (see Appendix 17 for an example) and each professional normally collected a copy of this from the nursing station. Nursing staff on night shift were responsible for generating this new handover sheet on a daily basis, populating with the most up to date information about every patient on the ward. The handover sheet layout utilised a structure of 'Situation, Background, Assessment and Recommendation' (SBAR) – a recognised and structured technique to promote information exchange between individuals or within teams (NHS Institute for Innovation and Improvement, 2010). On some occasions, the SBAR framework was used as a verbal structure to discuss patients during the meeting, however, more often, each patient was discussed using a more narrative approach. No formal records were taken although individual professionals would make their own notes either on the printed handover sheet or in the case of the ward doctors and the physiotherapist, by using a 'jobs' book.

4.9.2. Nursing handover

These meetings represented an information exchange between nursing shifts. Again, the printed handover sheet was used to exchange key information, and these meetings tended to focus on medical and nursing tasks. Examples included the need to get a Midstream Specimen of Urine test (MSU), or to ask medical team to review Intravenous Antibiotics (IVABs). Although heavily focussed on medical issues, this handover also included the sharing of information about wider issues such as contact with family and the need for wider referrals. Information from this handover often informed information then updated on the printed sheet or information shared at the 9am handover meeting.

4.9.3. Weekly multi-disciplinary team (MDT) meetings

Weekly MDT meetings were more formal in nature than the daily handover meeting. Led by one of the consultants, the MDT meetings were attended again by all core members of the ward-based team (namely nursing, occupational therapy, physiotherapy and social work), and lasted between 1.5-2.5 hours. On occasion, additional professionals attended this meeting, for example a link nurse from the psychiatry of old age service, or a 'nurse assessor' (a nurse with responsibility for continuing healthcare assessments).

The consultant, supported by the ward doctor, recorded entries in the medical notes. Nursing and therapy staff tended to make informal notes on their own printed handover sheet or within log books. The regular attendance by the consultant, the use of a closed room removed from the patient area (a shared patient and staff 'day room'), and more formalised entries in medical notes all contributed to the impression my perception of this as a more formal meeting, in comparison to daily or shift handovers. Each patient case would be presented and discussed in more depth than the daily handover and plans formulated accordingly. Although the structure of each case varied, commonly the consultant would summarise important background information pertaining to the admission, with the detail varying depending on an implicit judgement of how well the team knew the patient and how much information they needed to review. This summary would then lead to a presentation of current status, including issues such as skin integrity, continence, nutrition and pain, which mainly involved a dialogue between the consultant and the nurse. As the physiotherapist was involved in almost all cases, they would also contribute to this review and add an update on current mobility. Occupational therapy and social work professionals would contribute to cases where they were involved.

Plans made within the base ward MDT presented as being focussed on assessment and stabilisation of acute issues -for example trial without a catheter (TWOC) or carry out a Mini Mental State Examination (MMSE; a standardised screening tool to assess for the presence of cognitive impairment (Folstein, Folstein & McHugh, 1975)) to monitor acute delirium - and plans which could facilitate transfer or discharge - for example, referral to a nurse assessor; or to list for the intermediate care unit or rehabilitation ward. However, it would also be common for plans to be much less explicit, with the use of statements such as 'needs more time', 'more physio', or 'home later in the week'.

Although some of these examples did link to patient goals, goal-setting was experienced as a more implicit part of the process, with specificity about desired outcomes and timescales often not clearly stated. Timescales for plans or goals were often left unconfirmed and the use of question marks within the medical entries from MDT meetings was a common occurrence (for example '? home at the end of the week' or '? rehab bed'). Patients and family members were not present at weekly MDT meetings.

4.9.4. Length of stay meetings

These meetings were hospital-wide reporting meetings which took place three times weekly. They were held outside of the immediate clinical areas in various rooms in the base hospital building. The attendance and format at the three meetings I attended varied greatly although core membership presented as being a directorate manager who chaired the meeting, alongside a representative from physiotherapy, occupational therapy and a service involved in facilitating discharge. The varied format experienced presented as being linked to changes in personnel, particularly to the allocated chair of the meeting.

In advance of the meeting, a list was circulated to these core members which included details of every patient within the hospital with a ward length of stay (LOS) over 14 days. Members then had identified responsibility for gathering information from specific wards and reporting back to the meeting. The nature of the updates regularly included examples such as 'ongoing rehabilitation', 'awaiting nurse assessor' or 'awaiting a package of care'.

Professionals discussed a feeling that issues could potentially be misrepresented at these meetings with the potential for blame to be apportioned. They discussed the desire to avoid examples which could be attributed to delays with their own profession (for example 'awaiting stair assessment', or 'awaiting home visit'). Language used to describe patient choices within the meeting was also of interest with statements such as 'refused a bed on the rehab ward', or 'failed the home visit' being noted during these meetings.

4.10. Description of changes to the study site between phase 1 and phase 3

Although phase 3 involved interviews with health professionals and therefore did not involve any direct work-based observations or interactions, it is important to note that the researcher became aware of some significant service changes that had happened during the period between phase 1/2 and phase 3. These changes are important to the context of the research and will therefore be summarised in this section although, as I did not carry out fieldwork observations during this phase, my understanding of these changes is solely based on the perspective of professionals.

During phase 2, staff received information that the intermediate care unit would be closing (this information was made available to staff in June with a proposed closure date of September of the same year). When the researcher returned in April of the following year for the further period of data collection, the intermediate care unit had closed, and rehabilitation beds had been provided in a different location. The new site had previously been a nursing care environment and the rationale for the re-provision included the building being more accessible and fit for purpose. There was no longer a split between orthopaedic rehabilitation and general elderly rehabilitation, with instead a reduced number of beds provided for all general rehabilitation. Some therapy and care staff had moved across from the intermediate care unit although staffing was reduced due to the reduced number of beds.

Another significant change between phase 2 and phase 3 was a change in function of the previously identified rehabilitation ward. During phase 3 interviews, this was discussed by professionals as no longer being an identified rehabilitation ward, but instead labelled as a 'discharge ward'. It was discussed by professionals that if all active interventions had been completed and a patient was, for example, only waiting for a care package, they could then be transferred to this ward. Professionals discussed that overall bed numbers within the hospital remained broadly the same although medical treatment and active rehabilitation was now all provided within the base ward, and on other wards with similar purposes (for example another elderly medical ward, the orthopaedic ward, respiratory ward among others) until the patient was deemed to be at a level safe for discharge. Professionals discussed that a philosophy of co-located medical and rehabilitation treatment underpinned this, with the rationale of reducing unnecessary internal transfers which had the potential to delay interventions or complicate communication. Physiotherapists discussed that the idea was that they would no longer need to provide as much active therapy for patients on the new discharge ward (and therefore able to concentrate staff resources in fewer geographical locations) although discussed becoming quickly aware that patient needs were potentially not being met in terms of maintaining current levels of function and mobility (for example a person's ability to mobilise could deteriorate whilst they were waiting for the care package). They discussed that there were examples where this had subsequently affected wider discharge planning (such as equipment and care package requirements) and resulted in further delays.

One further change discussed was the introduction of a 'Discharge to Assess' multidisciplinary team, operating in to the acute wards over the hospital site. This was only briefly mentioned by professionals and therefore understanding of this initiative is limited. However, professionals discussed that at an early stage of an acute admission, patients were now being referred to this team who would carry out a generalised assessment of safety to be discharged home. Following this, profession-specific assessments including full physiotherapy and occupational therapy assessments, would then be carried out in the community. Enhancements to the community rehabilitation provision were also under discussion.

There were likely to be many other organisational and service-level changes in the period between phase 2 and phase 3 of fieldwork. However, the above changes

presented as being important to note and are referred to in later interpretations and discussion.

4.11. Description of how professionals evaluated rehabilitation potential

The section below provides a description of decision-making in relation to rehabilitation potential, based on data gathered from general fieldwork, specific observations during patient tracking and discussions with a range of health professionals. Decisions or judgements about a patient's rehabilitation potential could happen at any time and therefore the process was not necessarily a linear one, with discreet or separate stages. However, there were commonalities observed and discussed in the section below. Figure 5 represents a simplified version of this process, with Figure 6 providing explanatory detail.

Figure 5: Overview of reasoning process to evaluate rehabilitation potential

Gathering baseline information (reasoning: to compare current and previous levels and to set implicit goals)

Provision of curative and supportive interventions (reasoning: to start rehabilitation at optimum point; to rule out other reasons for changes or limited progress)

Provision of rehabilitative interventions (reasoning: to promote and monitor response and change towards desired goals)

The evaluation of rehabilitation potential and decision about the subsequent pathway

4.11.1. Stage: Gathering baseline information and comparing current and previous levels

The understanding of baseline information was discussed by health professionals as one of the first steps in evaluating rehabilitation potential. 'Baseline' was a term used across all professionals in the setting and interpreted to relate to information about a patient's previous level of health, disability, function and social support. Through the review of clinical records, this information gathering was understood to commence during the initial assessment – mainly the initial nursing assessment at the emergency care hospital – and then continued through a series of slightly different, although overlapping, discipline-specific assessments, sometimes at the emergency care hospital or sometimes initiated on arrival at the base ward. This baseline information was often revisited formally (for example, through questions asked during physiotherapy or occupational therapy initial assessments), or more informally (for example, through discussion with professionals who may have known the patient from a previous admission).

Information about particular aspects relating to pre-admission function were included in both physiotherapy and occupational therapy assessment tools (for example, preadmission mobility and transfer status were documented in particular places within the discipline-specific assessment documents). Nursing and medical staff also took opportunities to gather information although, with the exception of the initial nursing assessment completed on admission, any further information was documented in an un-structured way within general entries in the medical notes. The responsibility to gather this information was not always clear in the patient cases observed.

An element which presented as influencing this stage of the process was if the person was already known to the team. For example, during a discussion with the base ward physiotherapist about patients being listed for weekend physiotherapy, the

physiotherapist highlighted that 'she is likely to be here for a while...know her from a previous admission and she can normally walk approximately 15 metres with a zimmer frame'.

4.11.2. Stage: Provision of curative and supportive interventions to resolve or manage issues which may impact on ability to make functional gains

The team discussed the impact that acute medical issues had on reaching judgements about rehabilitation potential and that attempts were therefore made to resolve and treat particular issues which could potentially be resolved. The team inferred that this in turn informed and improved the judgement about rehabilitation potential. For example, the consultant summarised a patient in an MDT meeting on the base ward by indicating 'he is now eating and drinking much better and on a more normal diet and fluids...this can make a big difference to rehabilitation'.

The physiotherapist on the base ward reflected on a past case during one discussion, suggesting that 'if I'd seen him in writing as to when he first got here...I would have thought he's going to take a week or two...but it all depends on...getting rid of the delirium'. And the same physiotherapist, when discussing Patient 2, mentioned '...some of the pressure issues she's got on her legs, so that's going to obviously affect any mobility progression. But that's now under treatment which is a medical issue'.

Although curative, supportive and rehabilitative interventions were observed being delivered and facilitated simultaneously, it was implied that the impact of rehabilitative interventions, and therefore meaningful judgements about rehabilitation potential, could not be evaluated until the team had worked towards potential resolution of acute problems. This was not only because acute medical issues which changed, improved or deteriorated, could rapidly change the functional presentation of a patient; but also related to the fact that acute medical issues often meant that patients were not well
enough to engage in the type of rehabilitation activities (for example mobility, transfer or stair practice) which may help to facilitate the desired functional improvements.

An example was discussed within an MDT meeting on the base ward where the occupational therapist asked the team if they needed to prepare the family that the patient was unlikely to make functional improvements and therefore a return home was unlikely. At this point the consultant stated 'she has been unwell...needs more time. [We'll] review in a week and see what kind of progress we're making.

4.11.3. Stage: Provision of rehabilitative interventions and monitoring of progress

Following comprehensive information gathering and the provision of curative and supportive treatments, the importance of engagement in a rehabilitation process in order to make meaningful judgements about the likelihood of making functional gains was observed. In relation to Patient 2, the physiotherapist on the base ward discussed 'it's early days yet and we'll see what happens...so until I see that progress, that initial this is where we are...I can then make that decision (about the patient's rehabilitation potential)'.

In the case of Patient 2, during the first MDT meeting following her transfer from the base ward to the rehabilitation ward, the consultant verbally discussed that she was 'not sure what her rehab potential is...give her 2 weeks and review'. She also added that the team would be guided by 'OT and physio' during this time. This was reinforced by a written record of the meeting which stated 'more time for rehab – review 2/52'.

4.11.4. Stage: The evaluation of rehabilitation potential and decision about the subsequent pathway

Although not a linear process, once the team had worked to stabilise acute medical issues, and evaluated the likelihood of further functional gains, this then subsequently informed a more confident evaluation of likely progress and decisions about where a person would go next or which (if any) services they would be referred to. This evaluation of likely progress and the decision about the pathway are presented as one stage because they were difficult to separate – something that will be discussed in later sections.

Decisions about the pathway from the base ward observed during the period of fieldwork included: patients remaining on the base ward for a period of rehabilitation; patients transferred to the rehabilitation ward; patients transferred to the intermediate care unit; patients transferred to a specialist unit for cognitive assessment and rehabilitation; patients discharged home with referral to another service who could provide rehabilitation (most frequently the reablement service although occasionally other services such as referral to a day hospital or community physiotherapy service); or patients discharged home or to another care environment (for example 24 hour nursing care) with no further rehabilitation. At this stage, the consideration of access to finite rehabilitation resources – such as availability of beds or waiting lists for specific services – was observed as being influential in the reasoning of practitioners. More detail about all of these stages are illustrated visually in Figure 6.

Figure 6: Detailed reasoning process to evaluate rehabilitation potential and reach decisions about the rehabilitation pathway (continued

overleaf)

Stage 1: Gathering Baseline information and comparing current and previous levels						
 Professional assessments of current medical, physical and functional status Primary medical condition Secondary issues (e.g. pain, delirium, pressure) Mobility Functional level and care needs (e.g. self-care, toileting and continence) Cognitive ability Mood 	COMPARED TO	 Information gathering about previous level Comorbidities Previous mobility Previous functional level and care needs (informal and formal) Previous cognitive ability Home environment (physical and social) (NB if previously known to services, use this information) tting (which differences can be improved?)				
Evaluate the differences between current and previous and implicit goal setting (which differences can be improved?) Early judgement about rehabilitation potential – informing decisions such as need for professional involvement (e.g. referral to OT), need for weekend therapy etc.						
Stage 2: Provision of curative and supportive treatment and care to resolve or manage issues which may impact on ability to make functional gains	Alongside or followed	Stage 3: Provision of rehabilitative interventions and monitoring of progress				
For example to treat infections, mal-nutrition, constipation, manage pain Evaluate curative/supportive interventions and review as required	And Or	 For example mobility and/or stair practice, introduction of new aids or equipment, education regarding self-management of hygiene, toileting. Evaluate rehabilitative interventions- are they bringing about desired change (is the person making gains)? If not, evaluate reasons – medical reasons, motivation, 				
		carryover, environment, resources?				

Stage 4: The evaluation of rehabilitation potential and decision about the subsequent pathway

More detailed judgement about rehabilitation potential – continues to inform decisions such as need for professional involvement (e.g. referral to OT), need for weekend therapy, identification of appropriate bed/ward, and referrals to ongoing services (e.g. reablement team, community physiotherapy etc). These elements then inform the pathway decision which is a combination of consideration of rehabilitation needs/potential and the availability of rehabilitation resources/beds.

Pathway decisions observed during fieldwork:

A Rehabilitation	В	C	D	E	F
on base ward:	Rehabilitation on	Rehabilitation within	Rehabilitation at	Transfer (home or to	No Decision about
	other inpatient	intermediate care	home (with	24 hour care) with no	rehabilitation
Reasons for	hospital ward:	unit:	reablement and/or	rehabilitation services	
decision observed/			supported by other		Reasons for
discussed:	Reasons for	Reasons for decision	community services)	Reasons for decision	decision observed/
	decision observed/	observed/ discussed:		observed/ discussed:	discussed:
Has rehab goals	discussed:		Reasons for decision		
		Has rehab goals	observed/ discussed:	No rehab goals	Medical needs are
Professional	Has rehab goals				prioritised – not
judgement		Transferring with	Has rehab goals	Judged that the	appropriate to reach
suggests desired	Transferring with	assistance of 1		person/family can	judgement about
changes will be	assistance of 2		Current needs and	manage their own	ongoing care or
quick	people (or more)	Less nursing/medical	risks judged to be	rehab progression	transfers
•		needs than 'B'	manageable with		
Occasionally if	Seen as 'complex'		maximum home care	Any achievement of	Palliative Care on
patient vocal about	(e.g. complex		package	goals perceived as	base ward
not wanting a move	medical, social,			unlikely to influence	
•	behavioural)			outcome	
	,				

4.12. Description of the use of criteria for referral to rehabilitation services

There was no written criteria for referral to the rehabilitation ward observed to be in use. Decisions about transfers to this ward were observed to be made drawing on tacit knowledge and reasoning processes from different members of the MDT at different times. It was discussed that the decision to list for the rehabilitation ward was ultimately the decision of the consultant, although they themselves acknowledged that they drew on information and judgements from other members of the team, mainly the physiotherapist.

For the intermediate care unit, no formal, written criteria for this service was observed in use during fieldwork, although professionals working within the base ward, rehabilitation ward and within the intermediate care unit itself did frequently verbally refer to an understanding of eligibility criteria – mainly the need for patients to live within the locality area, to be medically stable, and to be able to mobilise or transfer with one person. Professionals demonstrated an understanding of this criteria through their own stories and ways of knowing, rather than referring to written or explicit statements.

Written criteria for eligibility for the reablement service was available (see Figure 7). Again, there were no occasions where I observed professionals directly referring to this, and indeed one social worker (despite the social work role essentially acting as a gatekeeper for this particular service) discussed that there was no criteria for this service. Instead of making direct reference to the criteria, professionals referred to examples and stories where they had found out the criteria by default.

130

Figure 7: Written criteria for reablement service

Customers must:

- Be 18 or over
- Be residents of **** (locality area) NB those of 'no fixed abode' will be included if deemed eligible through the assessment process
- Be eligible for support in line with current **** (locality are) eligibility criteria for Fair Access to Care Services
- Be medically stable
- Be safe to be at home between visits
- Be in agreement to a referral being made
- Have the potential for improvement, benefiting from a reablement programme
- Have the motivation to participate in the programme and agree goals

Although services had been reconfigured by the time of phase 3, discussion of how criteria can influence decisions arose during interviews with health professionals. One professional reflected on an example of implicit criteria; for example because the new rehabilitation facility did not have standing hoists, it had become implicitly understood that this represented a form of exclusion criteria in that people who needed this equipment could not be managed in this environment.

The presence of rehabilitation potential itself did form part of explicit and implicit criteria for rehabilitation pathways. For example, the requirement to have the potential for improvement was stated within the written criteria for the reablement service. And, in relation to the new rehabilitation facility which opened following the closure of the intermediate care unit, the base ward physiotherapist reflected that "one of the things I always say to people when they send them to (new rehab facility) is 'they've got to have rehab potential' (laughs)".

4.13. Chapter Summary

This chapter describes the location and purpose of the 'base ward', situated within a large NHS system covering a large geographical area. The descriptions of wider

services linked to this ward have also been presented in order to understand the most commonly utilised services for ongoing rehabilitation pathways. The features of the base ward and the linked services, including the roles of different professionals and the array of professional activities, act as a physical, social, cultural and temporal backdrop for decision-making.

The working patterns of the different professionals are of interest and will be referred to in subsequent chapters. Through the tracking of patient cases, and through time spent in numerous interactions and activities with professionals, the process of evaluating rehabilitation potential has been described. Although this has been presented as a series of stages, in reality this was not necessarily happening in a linear way.

Professionals described that the provision of rehabilitative interventions was an important stage in evaluating rehabilitation potential, in order to assess carryover and progress. This is interesting to hold in mind in light of themes developed within future sections which illuminate the many challenges to providing rehabilitative interventions experienced by professionals on the base ward.

CHAPTER FIVE – RESEARCH FINDINGS: SOCIAL CONSTRUCTION AND MEANING-MAKING OF 'REHABILITATION' AND 'REHABILITATION POTENTIAL'

5.1. Chapter introduction

This chapter will explore how concepts and processes integral to the research objectives were being social constructed. Again drawing on extensive fieldnotes from observations and interactions, alongside interview transcripts and extracts from clinical records, I will present instances and examples where key terms were used, discussed and represented.

This section will develop definitional elements – outlining how the concepts of rehabilitation and rehabilitation potential were constructed within the setting. Whilst Hammersley (1998) suggests that it is unlikely that the main claims of ethnographic accounts will be definitional, definitions will form an important part of the substructure to build explanations and shape meanings.

It was an initial objective of the research to explore the meaning of the term 'rehabilitation potential' within this context although it quickly became obvious that exploration of the term 'rehabilitation' almost preceded understanding relating to rehabilitation potential (and was reflected in revised research objectives). Therefore the chapter will begin by discussing the social construction of rehabilitation, before discussing the concept, and influences on the evaluation, of rehabilitation potential.

An overview of how initial codes and examples from data led to the development of themes and sub-themes is provided in Figures 8 and 9 (NB social construction and meaning-making of rehabilitation and rehabilitation potential began as two themes and were later combined).

Early Coding:

Began as one merged theme of 'Meanings' and all data was reviewed using the code 'Meanings'. All data then reviewed again using code of 'Rehabilitation Meaning'.

Rehabilitation meaning:

- A process to bring about improvements
 "To get stronger"
- A process to improve mobility...but changed to a process to achieve safety
- Associated with physiotherapy
- A place
 - \circ Happening in specific beds
 - \circ But should be a phase of care
 - A place to go when others don't know what to do
 - \circ A way of moving within a system
- Something interchangeable with other terms (e.g. Re-ablement)
- Something not living up to an ideal
 - $\ensuremath{\circ}$ Should be multidisciplinary
 - ${\rm \circ}$ Should be focussed on patient goals
 - \circ Should focus on more than mobility
 - \circ Should have more time
 - \circ Should be intensive
 - \circ Should have an enabling environment
- A process to improve quality of life
- An overwhelming idea for older people
- Misunderstood by others
 - Physio's rehab people
 - 'all-singing', 'all-dancing'
- Contrasting objectives

Examples from the data:

Rehabilitation:

- A process to bring about improvements:
- I4: "building or regaining skills"
- P5 (Fieldnotes p199) "to get stronger"
- Improving Mobility:
- I1: it always does seem to go down to right, transfers, mobility
- Physiotherapy:

Patient tracking examples – physiotherapy involved earlier and more frequently.

Focussed on Safety (linked to organisational objectives):

I3: "And now it's become safe only"

A place:

Fieldnotes p141; I1 'waiting for rehab'

Fieldnotes p19; p42; 'listed for rehab'

Not ideal/not happening:

I1; I2: versions of 'proper rehab'

Needing time:

I5: "the way I look at rehabilitation is having time to work with something...somebody..."

Final themes and Sub-themes:

Theme: Social Construction and Meaning Making of Rehabiliation (combined with social-construction and meaning of rehabilitation

potential on next page)

Sub-themes:

Meaning of Rehabilitation

- Rehabilitation as a process to bring about improvements
- Rehabilitation as a process to bring about physical improvements
- Rehabilitation as physiotherapy
- Rehabilitation as a place
- Versions of rehabilitation ideal versus reality
- Meaningful rehabilitation and time
- The objectives of rehabilitation

 Patient-centred objectives
 and goal setting
 Organisational objectives

Early Coding:

Began as one merged theme of 'Meanings' and all data was reviewed using the code 'Meanings'. All data then reviewed again using code of 'Rehabilitation potential meaning':

- Not sure! (difficulty defining/explaining)
 A guess?
 - Subjective
 - Influenced by many factors (became separate theme)
- Different professional perspectives:
 - Potential to get back to baseline/back to 'normal'
 - \circ Potential to get back to baseline mobility
 - \circ Potential to achieve goals
 - \circ Potential to be safer
 - Potential to influence discharge outcome
- Limited when known to have had rehabilitation previously
- Identified as having rehabilitation potential when others don't know what to do
 - \circ The judgement can be controversial
- Linked to engagement with a rehabilitation process
- Therapists want people to have rehabilitation potential
- Potential for what?
- Inextricably linked to a judgement about the pathway

Examples from the data:

Rehabilitation Potential :

Difficult to define - 12: "one of those holy-grail phrases we use in the NHS"

Potential to get back to baseline - Fieldnotes p13; p91; p99 'the ability to improve back to baseline'

Potential to improve mobility - Fieldnotes p33; p46; p99; 'whether someone's current mobility is different to their baseline mobility'

Potential to achieve goals - I1: "It should always be goal-specific"

Potential to be safer: I1: "the rehab potential would be to get to a safe point"

Potential to influence discharge outcome: Fieldnotes p32 "Already in sheltered plus so what else would ward ** do?"

Limited potential when previously known to have had rehabilitation:

When others don't know what to do: Fieldnotes p64 "get sent people with "Rehabilitation Potential" when other wards don't know what to do with them"

Therapists want people to have potential - I1: "I think your natural instinct within the physio profession to always maximise somebody's mobility"

Open to challenge – 14: "people come and you're kind of like well, what is it that we're actually doing here".

A best guess - I2: you're sort of like, picking a point in the distance and saying this is where we want to be

Potential for what? – Fieldnotes p228 – links with mental capacity. what

Final themes and Sub-themes:

Theme: Social construction and meaning making of rehabilitation potential (combined with social construction and meaning-making of rehabilitation)

Sub-themes:

Meaning of Rehabilitation potential

- An ambiguous concept...potential for what?
- Potential to get back to baseline
- Potential to get back to baseline mobility
- Synonymous with a transfer of care
- Potential to influence discharge outcome
- Controversial and open to challenge
- Influences on reasoning

5.2. The meaning of rehabilitation

5.2.1. Rehabilitation as a process to bring about improvements

In simple terms, the word rehabilitation could be seen as being attributed to a process

which facilitated improvements from the current state. When asked directly within

interviews to express the meaning of rehabilitation in their own words, a number of

responses reflected a process which aims to bring about improvements:

"For me rehabilitation is somebody getting back something that they've lost. So if they've had an ability before...em...then...my aim, if I was rehabbing them, would be to try and get them back to that point to where they were before" (I2, Line 74)

"Working towards getting them back to that level of function that they were previously at. Em...obviously sometimes it may not be...we may not be able to get them back to that baseline so that functional level would drop down. So therefore, you're rehabilitating back to their maximum potential" (I3, Line 8)

5.2.2. Rehabilitation as a process to bring about physical improvements

Despite these overview statements indicating rehabilitation as a process which aims to bring about different types of functional improvements, observational and interview data suggested a more narrow focus on rehabilitation; conceptualised as a process which aimed to improve physical abilities, and more specifically, mobility. During a ward round on the base ward, one of the consultants was observed concluding a patient interaction with a statement that the team would 'get the physio's to get you moving'. Also, from review of records during patient tracking, 'ongoing physio' was frequently cited as a plan or recommendation from ward rounds and MDT meetings, and 'ongoing mobility work' or 'decreased mobility from baseline' were cited as reasons for listing for a bed on the rehabilitation ward. In relation to one patient, the occupational therapist asked the team 'does she need rehab for mobility?', and concerning another patient, a discussion in an MDT meeting raised whether someone needed the reablement team 'for mobility practice'. And although rehabilitation goals were often very general or implicit on the base ward, it was common for the MDT meeting to include planning around mobility issues, such as 'aiming to progress to mobility with a stick', or 'needs stair assessment'. This was also reflected by Patient 5 when asked why he was going to a rehabilitation ward, he responded 'to get stronger'.

It was also recognised by one professional that this was perhaps part of how the meaning of rehabilitation was more broadly interpreted in society:

[people think of rehabilitation as a] 'getting soldiers to walk again kind of place'. (I4, Line 61).

Professionals were aware of this dominant focus on mobility when discussing this issue within interviews:

"so, we need to have goals...but within the team, and that's the whole team, it always does seem to go down to right, transfers, mobility, that seems to be where everyone hones in on" (I1, Line 82).

They also offered reflections and explanations for this focus, particularly linked to what

rehabilitation means in the context of acute hospital care:

"Because I think mobility is quite key in terms of all of the other activities of daily living...for the majority of people, [mobility] is their number one priority of being able to manage something themselves...I think it really is the cornerstone of everything else going on" (I2, Line 95).

"All the other areas of rehab have dwindled away...there isn't the time to look in to other areas of ADLs that would need a bit more time, that we could have assistance to support patients in those roles. And then, mobility is really the last thing that we're left with perhaps" (I5, Line 39)

Although this focus on mobility and physical abilities presented as being a dominant interpretation of the focus of rehabilitation, this was not to the total exclusion of other areas. An example was noted during tracking of Patient 3, where the documented notes from an MDT meeting on the base ward identified the management of 'own stoma bags and catheter' as part of the plan. Another example included a patient with Motor Neurone Disease whose upper limb function was recognised as affecting washing and dressing and toileting abilities.

Another such example of wider functional goals being part of a rehabilitative process was illustrated when a patient on the base ward was reviewed by a liaison psychiatry service for a review of mood. The liaison psychiatry service recommended that staff on the ward promote '1:1 activities and engagement with staff'. However, after this, some of the staff discussed that this would be 'nice to do, but not realistic'. This echoed the sentiment shared within 15 that when time is pressured, other areas of rehabilitation have 'dwindled away'.

5.2.3. Rehabilitation as physiotherapy

Referring again to quotes illustrated above, it was common for handovers and MDT discussions to identify 'ongoing physio' as a general statement about the status of the patient and their admission, or as a reason for transfer to the rehabilitation ward. During patient tracking, it was identified from records that most patients received physiotherapy from the outset of their admission to work towards improvements in their mobility. Occupational therapy provision generally commenced later and was less frequent (summarised in Table 6).

Table 6: Summary of occupational therapy and physiotherapy involvement during patient tracking (continued overleaf)

Patient	PT Initial assessment	OT Initial assessment	PT contact on base ward	OT contact on base ward	PT contact on Rehab ward	OT contact on Rehab ward
1	At emergency hospital – on day of admission	On base ward – on day 6	Transferred on day 2, assessment on day 3 10 contacts documented between day 3 and day 24 (with a focus on mobility and respiratory issues)	No further contact on base ward	After transfer on day 24 (Friday), seen on day 27 (Monday) Multiple contacts with a focus on mobility	After transfer on day 24, seen on day 44
2	At emergency hospital – on day of admission	Not assessed	Transferred on day 2 and seen for PT assessment on same day. 2 further contacts documented between day 2 and day 9	N/A	After transfer on day 9 (Friday), seen on day 13 (Tuesday) with a further 5 contacts documented until day 29	N/A
3	On base ward (as transferred from out of area hospital). Assessed on day 4 (Monday after transfer on previous Friday)	On base ward (as transferred from out of area hospital). Assessed on day 5 (Tuesday after transfer on previous Friday)	Following initial contact on day 4, 3 further contacts focussing on mobility documented until day 21	No further contact	N/A (Remained on base ward)	N/A (Remain ed on base ward)
4	Unclear from records	Unclear from records	Transferred to base ward on day 6 and seen for PT assessment on day 7	Assessed by OT on base ward on day 13	N/A (Discharged home from base ward)	N/A (Dischar ged home from

						base ward)
5	At emergency hospital on day 2	At emergency hospital on day 2	Transferred to base ward on day 2. Assessed by PT on day 3 Further 13 entries documented between day 3 and day 25	Assessed by OT on day 17. Further 1 entry documente d between day 17 and day 25	Transferred to base ward on day 25 (Friday). Seen by PT on day 28 (Monday). Further 7 entries documented between day 25 and day 36	Seen by OT on day 32. Further 3 entries docume nted between day 32 and 36.

Although the data set is small and does not capture contextual information or influences (such as therapy staffing levels, waiting lists, dependency levels of patients), this information does highlight general trends that physiotherapy tended to be involved earlier and more often than occupational therapy within this context.

Looking more closely at the information provided in Table 6, from the five patients who became the focus of patient tracking, three patients moved from the base ward to the rehabilitation ward (Patients 1,2, and 5) and were therefore perhaps in a more obvious rehabilitative phase of care. For the three patients who were transferred to the rehabilitation ward during their stay, all three continued to receive physiotherapy quickly after the transfer and there was documentation which supported work towards mobility goals during this period. In contrast, only two of the three patients received occupational therapy on the rehabilitation ward and this commenced between 7 and 20 days following transfer. It is important to note however that both of these patients had been seen by occupational therapy at an earlier stage.

In the case of Patient 2, this person did not see an occupational therapist at any point during their admission. Interestingly there were discussions in MDT meetings,

supported by entries in the medical records which suggested that occupational therapy would play a part in the rehabilitation of this patient - for example, medical notes documented a plan of 'More time with OT/PT', and the consultant discussed in an MDT meeting that the team were 'not sure of her rehab potential...to be guided by OT and PT'.

Physiotherapists presented as being aware that other members of the MDT closely associated rehabilitation with physiotherapy interventions. During interviews, one physiotherapy participant suggested:

"I think if you looked at the wider team and you said 'what's rehab?' to somebody other than the physio's, the OTs etc., those therapists involved in rehab...I think they would give a very simplistic idea of what rehab is...oh yeah you'll walk with the physio and then you'll be better" (I3, Line 107)

And an occupational therapy participant reflected that "the physios did more rehab" (I2,

Line 27) and linked this to the level of physiotherapy staffing.

Participants also reflected that patients, families and perhaps wider society held similar views:

"The word rehab...they're going to a rehab ward and there's physio's there" (I4, Line 331)

"People see physio, they think 'physio – rehab', that's...I think that's everybody in the population in general. It usually starts with the guy running on the football pitch with the sponge, oh that's what physio does...physio rehab people... the vast majority of people's perception of what rehab is is mobility work, so who mobilises patient's?" (I3, Line 620)

This way of interpreting the meaning of rehabilitation was not just present in the consciousness of professionals and patients, but predictably filtered in to other elements of the hospital system and decision-making. During interviews in phase 3, professional's described that some wards in the hospital were beginning to implement a new initiative around identifying each patient on specific wards as having 'green days' or 'red days' with a green day utilised if the person was receiving active interventions

from any member of the healthcare team, and a red day utilised if the person had completed active interventions and was now waiting for, or ready for discharge services. An interesting insight shared during an individual interview suggested that if a patient was receiving active physiotherapy, they were identified as having a 'green day', although if they were receiving active occupational therapy they were labelled as a 'red day' and aligned this to the interpretation that occupational therapy was seen by the team as a discharge planning service rather than a provider of active rehabilitation.

Reflecting this, an extract from an interview with an occupational therapy participant highlighted that when they did try to focus on aspects of rehabilitation wider than mobility, this was not readily appreciated by others:

"I've recently had someone that...was physio safe and I was kind of pushing for them to have further time on the ward because they couldn't manage bed transfers and initially it was kind of like, well, can we not just get a carer" (I4, Line 87)

A final interesting insight here was that if something was labelled as a provider of rehabilitation but did not include physiotherapy, professionals struggled to conceptualise this as rehabilitation:

"We've got the community rehab team as well...it's completely new...basically...when the [intermediate care unit] closed down the teams were made up of support workers who worked...so they've all got a really good background of rehab. The problem that we've got with that team at the moment is that it doesn't have any kind of physio that's...em...connected to the team. So at the moment we're not using them that much from the hospital." (I1, Line 397)

5.2.4. Rehabilitation as a place

Many different discussions alluded to rehabilitation as a particular place, including

when someone was 'waiting for rehab', or 'listed [or coming off the list] for rehab'.

Furthermore, when staff on the base ward conceptualised rehabilitation as a place,

they also clearly conceptualised that the place they had in mind was not their own

ward. The expressions of 'waiting for rehab' or 'listed for rehab' suggested that it was the next place of care, and hence the move away from the base ward, which would result in the provision of rehabilitation. And this issue was discussed by the base ward physiotherapist, highlighting that a patient could potentially make 'good progress with intensive rehab but don't feel we have that here'. Most often, the statements about waiting for a bed referred to a move to the rehabilitation ward based on site, or a move to the intermediate care unit.

The medical team on the base ward alluded to transferring people for rehabilitation who were more medically stable with one of the ward doctors suggesting that when he was 'only doing smaller things...[it] means they can manage her on [the rehabilitation ward]'. This links to rehabilitation being conceptualised as the phase of care which follows medical treatment and stabilisation, and in this case characterised by a movement in place.

Developing this idea, the change of location was linked to freeing up a specialist bed, with rehabilitation being perceived as being provided in a more generalist location. Examples included an extract from an interview with one of the occupational therapists:

"Now that I'm on ward (respiratory), we talk about patients needing further rehab and they are referred to a care of the elderly ward... because our ward is specifically medical for respiratory patients, they need the beds for those patients and then patients would go to a care of the elderly ward which we also refer to as rehab wards. And...it's no different...there isn't any increased support there" (I5, Line 68).

The final part of this extract suggested little difference in the resources and support provided by different wards, despite a movement of place often being conceptualised as a move to more intensive rehabilitation. A discussion with the physiotherapist on the base ward echoed this, discussing that a move to the intermediate care unit is often suggested as a move for intensive rehabilitation, but in reality it may not be that different. Linked to this, a member of staff from the intermediate care unit suggested there were elements where this unit actually had less resource for rehabilitation, highlighting in particular that there was no weekend therapy input.

Despite honest acknowledgements that wards were often not dissimilar in terms of staffing and resources, there was still a strong narrative implied on the base ward that the rehabilitation ward and the intermediate care unit would have increased resources to provide more intensive rehabilitation (in terms of time, staff and facilities), and hence why the concept of rehabilitation was intrinsically linked to these locations. When discussing Patient 5, the occupational therapist on the base ward reflected that 'they've [the rehabilitation ward] got more time'. When reflecting on the reasons for internal transfers within the hospital, another occupational therapist during an interview suggested that staff hold views such as 'we'll send them to [the rehabilitation ward] because they'll have more time to sort them out'.

Linked to this, I observed that when a patient was listed, and waiting, for a bed on a rehabilitation ward, he or she potentially became a lower priority for rehabilitative interventions on the base ward. An example of this was noted with Patient 5 who, from the first documented entry which indicated listing for the rehabilitation ward, waited 4 days to be transferred to the rehabilitation ward. However, in this time (despite earlier entries in his clinical record focussing on improving consistency with mobility and transfers), the base ward physiotherapist discussed 'handing him over to the assistant', implying that this was linked to the fact that he would not be discharged imminently and therefore became lower priority. This was echoed by the base ward occupational therapist who became involved with Patient 5 at a point when the ward team were discussing discharging him directly home, although she subsequently 'prioritised him out' when he was listed for the rehabilitation ward.

145

5.2.5. Versions of rehabilitation - the ideal versus the reality

Many staff discussed features of their ideal version of rehabilitation. During the early stages of the study, the base ward physiotherapist discussed the idea that rehabilitation should be a phase of care and ideally involve a 24-hour approach with all staff working towards rehab goals. At various points within fieldwork the same physiotherapist indicated an incongruity between an ideal state of rehabilitation and what could be provided on the base ward with statements such as 'we're doing the bare minimum' and 'we don't have that [intensive rehabilitation] here'.

Interview extracts implied incongruity between what therapists thought rehabilitation should be and what it actually was in the context of their practice:

"I think we all want this ideal text book thing...you know, goal-led, get the carers, get the family members, get the patient, do it all together. But it's not always as easy as that. Unfortunately." (I1, Line 258)

"The way I look at rehabilitation is having time to work with something...somebody...on some meaningful activity....It's working with them daily, having daily input with that person, practicing tasks, so that they can...possibly become independent or improve in their independence in that area that we're looking at. But that relies on having the time to do that." (I5, Line 22)

Words like 'should' and 'meant to' were frequently used when reflecting on ideas and

ideals of rehabilitation, being suggestive of a difference between what professionals

would like rehabilitation to be, and what it was in reality. The use of the word 'proper'

added another interesting insight, which again was suggestive of discrepancies

between ideal and reality but with language that suggested that alternative versions of

rehabilitation may be in some way improper:

"So then I ended up getting shouted at because people hadn't got the moving and handling plans done and hadn't had their site visits done because I'd been doing proper rehab" (I2, Line 39).

11: "It's a multi-disciplinary...and goal-led...that's what I tend to think...if we're looking at proper rehabilitation" (I1, Line 54).

The idea of rehabilitation as a phase of care was highlighted within an interview extract, although again suggested that this was an aspiration rather than an established way of working:

"[Rehabilitation] could be a place...or a philosophy of...what do I do in this rehab? I've started to mark out patients who are getting towards that rehab part of their stay and em...and then getting the healthcare's etc. involved. So everything that patient does is then focussed towards function and rehabilitation...that's what I want to create on the ward and it takes that whole emphasis about...oh physio, they're rehab, they're mobility. It's not...it's everybody's job to get involved with that" (I3, Line 78).

There was an awareness that this discrepancy between ideals of rehabilitation, and

what rehabilitation actually constituted, particularly on the rehabilitation ward and

intermediate care unit, could perhaps lead to challenges in managing patient and family

expectations. During fieldwork observations, professionals on the base ward

acknowledged that the rehabilitation ward can get 'sold' to people as a place for

intensive rehabilitation, and therapists on the rehabilitation ward discussed that they

can experience a lack of clarity, and sometimes anger, from patients or relatives linked

to unrealistic expectations of rehabilitation. This was echoed within interviews:

"I sometimes feel it's kind of sold to people like it's rehab and people...you know people who aren't in the profession or kind of work with elderly people have big expectations of what a rehab hospital or ward would be like. And in reality, it's not kind of all singing, all dancing you know" (I4, Line 58).

"Because if you say to a patient that you're going to rehab, the family, they really cling on to that and they're expecting some intense rehabilitation" (I5, Line 89).

5.2.6. Meaningful rehabilitation and time

Time is mentioned in many areas of the research findings; for example the time needed when evaluating rehabilitation potential, the pressures on time for certain rehabilitation activities and the narrative of 'more time' as a recommendation when situations were unclear or emerging. It is important to note here that professionals did associate the allocation of time as being linked to the quality of a rehabilitation process. One professional, when asked to describe what they meant by the concept of rehabilitation highlighted the importance of time as their first reflection:

"the way I look at rehabilitation is having time to work with something...somebody..." (I5, Line 22).

5.2.7. The objectives of rehabilitation

5.2.7.1. Patient-centred objectives and goal-setting

An explicit goal-setting process was not observed on the base ward. During, daily handovers, discussions were centred around professionally-led actions rather than patient-centred goals. Goals were a little more obvious during weekly MDT meetings on the base ward – with examples such as 'aiming to progress from mobilising with a 4-wheeled walker to mobile with a stick'. However, from researcher observations, this applied in only some cases within MDT meeting discussions, and again the discussion tended to focus on professionally-led actions.

Goals were evident within some examples of clinical records, although difficult to find at times. Physiotherapy and occupational therapy entries sometimes included information about goals (for example 'aim to increase exercise tolerance and aim for independence with mobility [physiotherapy entry for Patient 1]; 'aim to progress to independence' [physiotherapy entry for Patient 1); 'transfer practice' [occupational therapy entry for Patient 5]). This was sometimes contained in profession-specific assessment documentation, and sometimes mentioned in general continuation records. In some examples of clinical records, there was no explicit documentation of goals.

I also did not observe explicit discussion with patients about their goals, with instead goals discussed implicitly with patients ('we'll get the physio's to get you moving'; 'we'll try you with a frame without wheels'). Discussion with family members about goals were also not explicitly observed. I was aware that some patients had planning meetings although the timing and reasons were not always fully clear and the attendance of different members of the MDT was recognised by the team as unpredictable. I attended a planning meeting for Patient 5 which was attended by family members, the social worker and the occupational therapist (but without the patient who had been invited but decided not to go) and again discussions focussed on professionally-led actions such as the referral to reablement, the requirements of the care package and the organisation of a key-safe to enable carer access to the home.

Finally, in relation to goals, a physiotherapist shared the reflection that when they got to the point of thinking about goals, patients tended to leave the base ward to go elsewhere.

5.2.7.2. The organisational objective of rehabilitation

To conclude this section which examines the conceptualisation of rehabilitation, over the course of the data collection period (more specifically during the course of the 12 month period between commencing observations and completing in-depth interviews), professionals expressed a shift in the way the organisation was communicating the purpose and desired objectives of rehabilitation.

During phase 1 and 2 of fieldwork, I understood one of the primary objectives of rehabilitation as promoting functional improvement, most often an improvement linked to mobility or physical abilities. The expression 'back to baseline', most often in relation to baseline mobility, formed part of discussions and observations several times every day, in relation to most patients. The purpose of rehabilitative interventions therefore presented as aiming for improvements towards the person's previous baseline.

It became very quickly apparent during phase 3 (interviews with occupational therapists and physiotherapists), that the narrative had changed in relation to this. "So...we would be looking at...obviously...em...their level of mobility prior to coming in is a factor, so that we had an idea of what their b...we're not supposed to use the word baseline now" (I1, Line 23).

"We're not allowed to say we're getting people back to baseline" (I2, Line 79).

"I was going to mention baseline function...this is now a bit of a nasty word within the trust...baseline" (I3, Line 304).

Through trying to understand this issue further, professionals described the reason for

this as the Trust encouraging a movement away from working towards achieving a

return to baseline function, towards encouraging a focus on achieving safety for

discharge, which may or may not be achieved through the achievement of functional

gains. Again, this was expressed strongly in a number of interview extracts:

"So it used to be, like we said earlier, baseline or better. Then it became baseline only. And now it's become safe only rather than...so...an example would be someone independently walks with a stick when they came in, they are now independent with a zimmer frame, and at that point they are safe to go home" (I3, Line 135).

"What they're saying is that we...it's all about...em...us not looking at getting people back to their baseline. It's about people being discharged at a level that is safe" (I1, Line 40)

Professionals did express a general understanding in relation to this, with an

awareness that this was being driven by organisational objectives to reduce length of

stay and facilitate patient flow, both of which enabled the Trust to ensure the most

acutely unwell patients would be able to access the services they needed. However,

although able to recognise the rationale, professionals expressed concerns and

counter-arguments in relation to this:

"Fair enough, we can all see that. But the problem that we have is that if they're not back at their baseline...em...we don't really have a huge about of community services to ensure that people then can progress in the community" (I1, Line 42).

"The problem that I have with that is that I sort of feel that we're failing patients because....patients aren't getting rehabbed back to baseline...in terms of mobility and things. That then has a huge impact on their other functional activities of daily living. And if there's nobody picking up that baton in the community, and pushing forward to try and get them back to where they were before, then they're

stuck on that point and we've missed that window of opportunity and, 2 months time they get readmitted again and are we doing the same thing again. They're becoming more infirm and more dependent on other people" (I2, Line 80)

5.3. The meaning of rehabilitation potential

5.3.1. An ambiguous concept...potential for what?

When directly asked to explain the concept of rehabilitation potential in their own words during interviews, a number of the interview participants began by recognising the ambiguity and complexity of the term, almost to explain their own lack of clarity or potential subjectivity:

"Em...well...it's a really broad term isn't it...It's quite a difficult thing...you can't put it down in one sentence" (I1, Line 148).

"I think it's one of those holy-grail phrases that we use in the NHS" (I2, Line 190)

"It's kind of one of those words that kind of get banded about and people use it, and there's no definitive reason as to why someone has, or someone hasn't" (I3, Line 117).

Alongside the above extracts which illustrate the difficulty in articulating what

rehabilitation potential was, it was also apparent during fieldwork and interviews the

term rehabilitation potential was often used globally and without specificity.

Expressions such as 'what's his rehab potential?', or '? rehab potential?' were regularly

part of team discussions or statements noted in clinical records.

Extracts from interviews suggested an awareness that it would be helpful to be specific

in relation to which areas a person had capacity to improve within, although reflected

that this is perhaps not always the case:

"When we're looking at rehab potential it should always be goal-specific. But I don't know whether it always is" (I1, Line 245).

"We probably don't do it [explain rehabilitation potential] enough, explicitly enough in the notes" (I2, Line 346).

Linked to this, in the absence of specificity, it was highlighted that perhaps professionals reverted to assumptions about rehabilitation potential, or the default way of understanding rehabilitation potential in this context:

"It could be something that the OT's working with but it would just be termed 'rehab'. And I think...as I've mentioned on a handover if we see that, we just assume it means they're not back to their baseline mobility (laughs)" (I5, Line 158).

A social worker drew a comparison between evaluations of rehabilitation potential and evaluations of mental capacity with both terms being used as 'umbrella...catch all' terms. The social worker continued to discuss that when professionals indicate that someone does not have mental capacity...they need to be saying 'capacity for what?', and similarly instead of just saying someone does or does not have rehabilitation potential, professionals should instead be asking the question 'potential for what?'.

5.3.2. Potential to 'get back to baseline'

One of the physiotherapists linked their understanding of rehabilitation potential to a hypothetical patient, with that patient having 'the ability to improve back to a point of baseline or beyond, [with] baseline being where they were prior to coming in to the hospital'. Wording was echoed by other physiotherapists when describing this concept, with one suggesting 'it is about a judgement about getting as close to baseline as possible' and the other considering whether '[the patient has] the potential to get back to that baseline'.

Occupational therapists used similar descriptions with one occupational therapist indicating that the judgement about rehabilitation potential linked to whether a patient 'still has a little bit more further input to reach their new baseline', with another discussing that they consider 'what they're achieving now and where they were before...and what's the likelihood of them getting any better'.

The meanings constructed by occupational therapy and physiotherapy were of particular interest in this study, although meanings discussed by other professionals within the environment were also significant and contributed to the wider social construction of concepts. A doctor on the base ward simply described rehabilitation potential as 'the potential to get back to baseline', and the consultant (in relation to patient 2) discussed that 'she is quite a way off baseline...mainly indicated by the variance in her mobility'.

The later phase 3 interviews occurred at a time when the organisation was attempting to encourage a change in the narrative in relation to thinking about baseline. Despite this, when asked to describe rehabilitation potential in their own words, the interview extract below illustrates that the approach to conceptualising rehabilitation potential as the potential to get back to baseline continued to be significant:

"I suppose it's whether we think that they have the ability to improve on the state that they are in at that point of time. So, it might not be getting back to their baseline (laughs)...which we're not supposed to be saying anymore...but actually to have an improvement in their physical and mental status so that they can em...we can see an improvement" (I1, Line 167)

5.3.4. Potential to get back to baseline mobility

Perhaps unsurprisingly, linked to significance of physical improvement, which emerged through conceptualisations of rehabilitation, this also emerged in relation to the ways in which professionals considered rehabilitation potential. When discussing influences on rehabilitation potential one of the physiotherapists discussed considering what 'might affect potential for either muscle strength or power being built [and] whether their co-morbidity may be around exercise tolerance'.

Notably, although again unsurprising in the context of the dominant discourse amongst the professionals, was that occupational therapists also attributed similar meanings and examples to their own explanations of rehabilitation potential. After discussing their ideas of the concept of rehabilitation potential, the first example given by one occupational therapist was to suggest that 'if a patient's mobility is rather poor...we still have the potential to improve...to a more stable mobility level'. The same occupational therapist did go on to recognise other functional aspects, although this came after their descriptions about working towards improvements in mobility status and they then concluded the discussion by saying that 'if there is any further rehab potential, then that can be looked in to in the community and the physiotherapists can follow up' suggesting rehabilitation potential links to the potential to make physical gains and is therefore addressed by physiotherapy.

5.3.4. Synonymous with a judgement about transfer of care

A judgment about the rehabilitation potential of a patient on the base ward was often intrinsically linked to a judgement about appropriateness for a transfer to a less acute rehabilitation bed. However, it almost became impossible to distinguish these elements from each other and the judgement about whether someone had capacity to make gains through rehabilitation became reframed with the adjunct of the judgement about the appropriateness for the transfer or movement within a service.

A member of the hospital to home team – a team with a remit to support discharge planning and care transitions - discussed that their team could assess rehabilitation potential and used this particular term. However, on further discussion, this assessment involved gathering information about medical stability and level of current dependency on medical and nursing interventions, in turn using this information to inform a judgement about the appropriateness of an internal transfer to a less acute (rehabilitation) bed.

Also, during an interview, one participant response suggested blurred lines between the judgement about rehabilitation potential and the judgement about the appropriateness of a transfer:

"There was a patient who got sent up to one ward and then brought back again because their rehab potential...I would say that the therapists were probably right there but they got over-ruled by the medical team" (I3, Line 581).

One of the critical challenges suggested during fieldwork observations was that the evaluation of rehabilitation potential could potentially be used as a strategy to promote movement within the system. This was observed during an informal observation on the base ward when nursing staff were discussing how to free up a bed for a medically unwell patient who needed to be transferred to this ward from elsewhere in the hospital. The discussion highlighted one patient to transfer to the rehabilitation ward although they discussed that they thought the primary aim was a 'social sort out'. A member of staff subsequently discussed that '[the rehabilitation ward] can do that, we used to do that when we were a rehab ward'.

Movement from one ward to another was linked closely to the length of stay figures for each ward; particularly pertinent as length of stay for each patient who had been on a ward for longer than 14 days was discussed and reported to organisational management.

The below extracts were expressed during interviews although were expressed in a third person style – indicating that professionals were aware that this may occur although did not express this in relation to their own practice:

"Sometimes I think it was said, well, if it was going to be quite a complex discharge if there was a lot of problems, it seemed they'll be like they need more time so we'll send them to rehab when there wasn't really a rehab need. They just wanted...rid...that sounds awful...or they kind of wanted them moved elsewhere or the problem" (I4, Line 221).

"[it would be said by others] "we don't really know what to do with this patient...there's a lot of complexities...we'll send them to [the rehabilitation ward] because they'll have more time to sort them out" (I2, Line 21).

Whilst recognising that a positive evaluation of rehabilitation potential could influence a transfer to a rehabilitation environment and therefore whether a person was afforded more time within the hospital environment, the reverse of this was also observed in that a negative or low evaluation of rehabilitation potential could equate to decisions not to pursue options for a patient which involved further inpatient or bed-based

management.

Interestingly, once in a rehabilitation environment, the evaluation of rehabilitation

potential was still linked to movement within the system although in a critically different

way. That is to say, in a rehabilitation environment, a positive evaluation of

rehabilitation potential was equated to a desire for the person not to move, and linked

to a lack of confidence in community services to continue to develop the identified

potential. Reflecting on experiences from the rehabilitation ward:

"Well generally if somebody has rehab potential, I would usually be trying to fight to keep them where they are at, to keep them continuing on that pathway. Because, as I said, the services aren't available once they leave hospital" (I2, Line 482).

Finally in relation to this, rehabilitation potential in some instances almost became a

criterion for the internal transfer, although the extract below also suggests the critical

challenge that this presents in terms of ambiguity and subjectivity:

"One of the things I always say to people when they send them to (new rehab facility) is 'they've got to have rehab potential' (laughs). So it's probably em...one of those things, those outcomes or one of those features that people have to have that people find it hard to find it hard to say, well, have they or haven't they got rehab potential" (I3, Line 416).

5.3.5. Rehabilitation potential as potential to influence discharge outcome

During a base ward MDT meeting, a physiotherapist asked 'even if [the patient] improves, do we think it will change where he ends up going?', and this provided an insight that rehabilitation potential was not only being considered in relation to likelihood of making physical or functional improvements, but also the likelihood of influencing discharge outcome. Similarly, and again in an MDT meeting, an occupational therapist queried whether the team were listing a patient for the rehabilitation ward, although it was raised that the patient was already in an extra-care sheltered scheme and 'what else would [the rehabilitation ward] do?'.

This perspective that rehabilitation potential equated to a potential to influence discharge outcome was perhaps most notably illuminated by an agreed approach to referring to occupational therapy where (with a few exceptions, mainly related to moving and handling risk assessments), the occupational therapy team did not accept referrals for people who already resided in residential or nursing care. Through discussion, they directly linked this to judgements that any intervention would not result in an influence on discharge destination.

During interview responses, professionals suggested an awareness of the presence of a link between rehabilitation potential and an evaluation of whether any functional gains had the potential to subsequently influence discharge outcome.

"But I think because of...pressure to get people moved out of hospital, it's kind of...ok, well why should we continue to try and get this person to walk if they can go to nursing care and they can look after them. So I think rehab potential does differ in that respect" (I3, Line 202)

However responses also suggested professional tensions and discomfort with this:

"Especially if someone is always in residential or nursing care. If they come in and they were previously mobile and then...now they're transferring with 2, it seems like they're kind of quick to say well there's people at home who will be able to manage their mobility with 2 people. But to me, if it was me, I would think well, why are they not entitled to have some therapy input to try and improve their mobility further. Just because they are already in a residential care setting...I don't really agree with that...personally" (I4, Line 103).

5.3.6. Rehabilitation potential - controversial and open to challenge

A subtheme of this section, summarising many of the issues outlined above, was that the concept of rehabilitation potential was witnessed sometimes to be controversial and open to challenge. Observations during fieldwork highlighted that professionals perceived that some wards or services might 'bounce patients back' if there was a difference in the evaluation of rehabilitation potential, or that judgements would be questioned. Indeed this was witnessed during an observation on the rehabilitation ward when, on a couple of occasions very soon after a patient transfer from the base ward, the team were observed to ask 'is [the patient] rehab-able?', and reflected that 'the physios [on the base ward] think he has rehab potential'.

Those evaluating rehabilitation potential and recommending a patient for further rehabilitation services demonstrated awareness that receiving services would be likely to question this evaluation and also that managers or other professionals may question this evaluation if this increased the length of stay in the hospital system. The below interview extract also suggests that when on the receiving end, they also judge such evaluations themselves.

"And one of the other physio's this morning just said to me 'why did you say she was good for the [new rehab facility]...she's just laid in bed all of the time and she won't get out'...and then you're thinking in your own mind, 'ahhh...should I have said she was for [new rehab facility] and am I going to send her and...she's actually got no rehab potential" (I1, Line 429).

The subtle differences in how different players, at different times, conceptualised rehab potential, in the main simply presented as an interesting observation although one that remained relatively unseen. The vignette summarised below involves a patient who was not part of the patient tracking (therefore their notes were not accessed and no idiosyncratic details were used), although I became aware and involved in many discussions due to my routine attendance at handovers, meetings and general ward activities.

After multiple transfers within the hospital system, compounded by an in-hospital fall which led to a fractured neck of femur, this patient was receiving care and treatment on the base ward. After returning from a period of leave, the physiotherapist shared insights that the ward staff may not see improvements because the patient was still using a full body hoist, although the physiotherapist felt he was less windswept in his posture and his standing with the standing hoist was also improved. It became obvious that a judgement about this patient's rehabilitation potential was important, and the physiotherapist shared that this patient could make improvements although did not know whether these improvements, he may require long term nursing care.

During subsequent morning handovers and MDT meetings, the team – although with the main protagonists being the nursing staff and the physiotherapist – formulated a plan to continue with rehabilitation on the base ward (mainly deciding this because of the history of multiple transfers between wards and aiming to minimise further transfers if this could be avoided) in conjunction with referring to a 'Nurse Assessment' team who would assess for the need for 24-hour nursing care against agreed criteria. Both nursing and physiotherapy staff agreed that this would enable a further period of care and rehabilitation (whilst waiting for the referral to be allocated and the assessment to be undertaken), although still make progress with discharge planning, aligned to the goals of the organisation to optimise the length of hospital stay.

Over subsequent days, the physiotherapist expressed further reflections on this case, reflecting the belief that further improvements could occur and these improvements, such as improved or less dependent transfers, could significantly affect quality of life. The physiotherapist frequently used the expression 'the clock is ticking', communicating some personal pressure that quick improvements were desirable in order to justify reasoning. The physiotherapist also shared personal frustrations that this patient could potentially benefit from more intensive therapy although this was at odds with how the organisation wanted physiotherapists to prioritise patients, with the suggestion that priority should be given to those who could be discharged quickly. An awareness of the tension between physiotherapy and nursing staff was also communicated indicating that the nursing team 'see a file [medical notes] this thick and sees that he has been the same [full body hoist] for weeks. But that doesn't tell the story...he hasn't been here for the same reasons and he has had multiple things going on'.

Six days after the referral, the 'Nurse Assessment' took place which was highlighted as quicker than the ward team would normally expect (the team discussed two weeks being 'an average' expected time between referral to assessment). The outcome of this assessment was that the patient did not meet the criteria for NHS funded nursing care. The assessment also resulted in an opportunity for the patient's wife to highlight that she would like him to return home and the detail of the discussion involved her highlighting that she did not want him to sleep downstairs and would like him to manage the transfers on and off the stair lift in situ. A reflection of the researcher at this point was that she was also conceptualising his rehabilitation potential as the potential to make gains in function and mobility towards his previous baseline.

The patient's case was discussed during the next weekly MDT meeting and nursing staff and the physiotherapist were observed presenting information about the case. The physiotherapist emphasised the history of the admission, the multiple issues treated during the admission, the fact that they were only now achieving stability with issues such as nutrition and continence, and that in physiotherapy, improvements had been noted in posture, weight-bearing and range of movement. This information was summarised by stating that it was the physiotherapy opinion that this patient had not 'had a fair crack of the whip' and acknowledged later that it was a 'big thing' when the Consultant agreed.

The case was summarised in the MDT meeting by the ward doctor who indicated that 'physio feel is improving therefore list for ward ** [the rehabilitation ward]'. This was reiterated during the morning MDT handover meeting the next day where the ward Doctor again emphasised that 'physio thinks he has potential'. The team presented as placing value on the physiotherapy opinion, although in this case, the pressure on the physio to evidence and justify this reasoning was also significant, perhaps most notably through the frequent reference to the 'clock ticking'.

Following the transfer of the patient to the rehabilitation ward, the Ward Sister asked 'is he rehabable?' and one of the physiotherapists on the ward discussed that her first impressions were that he may have 'limited rehab potential' and she was left 'wondering what he has come here for'. The physiotherapist on the base ward also acknowledged that despite the patient's wife wanting him to be able to transfer on and off the stair lift, 'this may not be realistic'.

Alongside the differences noted between how different professionals subjectively conceptualise rehabilitation potential – differences in this case between physiotherapy and nursing - this example also illustrates many of the themes developed in this section – most notably the construction of rehabilitation as a process which improves mobility and strength, the meaning of rehabilitation potential as the potential to improve mobility back towards a baseline level, and the controversial nature of evaluating rehabilitation potential which can be open to challenge.

5.4. Evaluations about rehabilitation potential – influences on reasoning

5.4.1. Clinical and individual factors

5.4.1.1. Co-morbidities

Information relating to a patient's previous health and current health presented as being an important influence on the reasoning of therapists. When discussing influences on their evaluation of rehabilitation potential, one of the physiotherapists discussed considering 'whether co-morbidity may be around exercise tolerance such as some lung diseases...[and thinking about] frailty.' This physiotherapist went on to add 'it's looking at the baseline, what has got them to that baseline or if that is just their norm for a long time and whether they're likely to get back to that taking in to mind factors such as co-morbidities and cognition'.

The consideration of co-morbidities as factors which may predict whether someone may get back to, or beyond, their pre-admission functional level, was also obvious within the wider team. One ward doctor during a weekly MDT meeting acknowledged that one patient 'has a long history of problems...not too significant individually but when added up might suggest limited progress'. And a therapist at the intermediate care unit suggested that 'people with multiple conditions tend to have lower potential'.

During interviews, professionals highlighted current and previous medical and health conditions as a significant influencing factor in their professional reasoning, suggesting this information assisted them to understand whether someone was likely to improve; whether changes were likely to be chronic and enduring in nature; or indeed whether a person was likely to deteriorate. And in essence, they suggested this was integral to judgements about rehabilitation potential:
"Always considering what their norm is before they came in to hospital, and what they've been through in the hospital, you know if it's a surgery or other illness, what medication they're taking, what is expected from the doctors as well, does somebody have a prognosis, are they a palliative patient...because like I say, if I know that patient is palliative or whatever, I know I can expect less of a rehab potential you know" (I5, Line 164).

One participant went on to offer insight that information about health, medical

conditions and co-morbidities is particularly influential at an early stage of professional

reasoning:

"Because if you put a timeline on it, at initial offset you might say that comorbidities and cognition and medical problems may be the things that are focussing on your mind at that particular time. But, as they move away from those, it suddenly becomes what's their functional level like, have they got a package of care..." (I3, Line 369)

5.4.1.2. Cognitive function

There were many examples observed where professionals discussed how cognitive function of an individual influenced their reasoning about rehabilitation potential. Professionals discussed associating decreased cognitive function with a person having 'limited carryover', and recognising that the ability to retain information relating to their rehabilitation and rehabilitation goals could influence outcomes. However, professionals also discussed that having dementia or cognitive impairment would not rule people out from having rehabilitation potential, with one of the social workers particularly linking this to people making good progress within familiar home environments.

Although professionals indicated that they would aim to flexibly interpret the influence of cognitive function on rehabilitation potential, it was also observed that professionals could create their own, often concrete, rules and ways of thinking about this. One patient was referred to the nurse assessor particularly quickly after the acute admission, with one of the reasons discussed during the morning handover being that 'she has known dementia, an MMSE of 9 out of 30 and she wasn't coping prior to admission'. This presented as being a quick judgement that this patient would not make further progress through rehabilitation, with a question following this relating to whether the team would just be 'taking up a rehab bed' if they referred her for ongoing rehabilitation.

A similar example involved the team considering the rehabilitation potential of a patient being discharged home in relation to the appropriateness of a reablement package, although instead decided a long term care package would be more appropriate due to her MMSE of 12 out of 30. Another example was noted during a morning handover when nursing staff highlighted that 'with an MMSE of 10 we can't really put reablement in as she won't learn anything'.

During interviews, some professional's highlighted cognition explicitly as one of the most important influences, linking this to the ability to learn and retain information (often termed as carryover), and the ability to understand a rehabilitation process:

"I think definitely with the cognitive aspect of it, I would say that's a key thing. Em...so will they be able to understand what we're...you know, you can get the patient who can't even follow an instruction so you think well how are we ever going to rehab this person" (I3, Line 334).

"But I think the cognition side of things...if somebody...if somebody can say this is where I was a month ago, and this is what I'm like now, and I don't want to stay like this, I want to get back to where I was a month ago" (I2, Line 454).

However, one participant did express that poor cognition should not be seen as a factor which rules out rehabilitation potential, but acknowledged this may happen, and linked this to level of experience:

"And I think...I've worked with a lot of [junior staff]. And I think because I've worked in rehab for a long time...I think they're much quicker to write people off, than I am. Because they'll be...ah, the cognition, they've got no idea...and I think that they can be 'oh, we'll just monitor them, I don't think we're going to get anywhere with them'. Em...but people can surprise you actually...it could be that, they've got some delirium that is resolving, it could be that they've had a UTI, and

em...within...give them a few days, and they're a completely different person" (I1, Line 285).

5.4.1.3. The importance of understanding the 'baseline'

The above factors – understanding co-morbidities and cognitive function - helped professionals to establish a picture of what they frequently referred to as the patient's 'baseline', a term used continuously within the field and one which has already received attention within this chapter. And within interviews, all participants reflected the influence of information about the baseline or a previous norm as an integral part of their reasoning about whether someone had the capacity to make gains from their current level.

Despite the importance of baseline information to professional reasoning, there were many examples of gaps or inaccuracies in baseline information. Examples uncovered, mainly through review of documentation during patient tracking, are summarised in the Table 7.

Table 7: Summary of omissions or inaccuracies with baseline information documented in clinical records (continued overleaf)

Case	Examples of information taken from clinical records	Issues noted
Patient 1	Day 0 – nursing assessment with baseline function completed at emergency care hospital	Baseline information not fully gathered at emergency care
	Day 1 – Transferred to base ward	hospital
	Day 2 – medical entry (by consultant)'? usual level of function' documented.	Delay in understanding pre-admission functional level
	Day 2 – assessed by physiotherapist on base ward. Physiotherapy entry: 'History from family still needed'.	(baseline)

Patient 2	Day 0 - nursing assessment completed at emergency care hospital. Documentation of cognition 'no problems'; continence 'no problems'. Medical entry: 'Need collateral history from son regarding current cognition – is she at her baseline ? ? may need CT head'.	Inconsistent and contradictory information (particularly relating to mobility, cognition
	Day 1 – Documentation of discussion between medical team and son 'Patient's memory has been poor with a decline over the past few months. Has had reablement package previously which has been helpful'.	continence and previous care package) which continued for approximately 2 weeks.
	Day 1 – Physiotherapy entry on base ward 'Normally mobile with stick'	
	Day 7 – Documentation of discussion between nursing staff on base ward and son 'Lives in a bungalow, regularly incontinent, walks with a tripod. TDS (three times daily) carers who help with ADLs'	Delay in gaining an accurate understanding of pre- admission functional level (baseline)
	Day 14 – Documentation of MDT discussion on rehabilitation ward – '? whether has had reablement or long term care package'	
Patient 3	Day 0 - Transferred from an out of area hospital following specialist surgery. Letter of referral outlined pre-admission information and social history: 'Functional level before admission - independent mobility no aids'.	Inconsistent information about pre- admission mobility level.
	Day 4 – Physiotherapy entry on base ward – 'Previous mobility independent no aid – fall 6-7 weeks ago. Struggled with mobility – occasional use of stick'.	Delay in understanding whether the patient was back to the same level

Professionals presented as being aware of the regular deficits in the accuracy of this baseline information. During fieldwork, one of the base ward doctors highlighted that 'the baseline is not always clear and can be an emerging picture...particularly difficult with no family'. During this same conversation, the doctor discussed that baseline information should be available from (the emergency hospital), but also discussed that it is not always available and it is not always accurate.

The base ward team regularly referred to gaining 'collateral history' (a term used for a full history provided by an informant) and used different strategies such as verbal discussions with family members or use of a questionnaire completed by a relative or carer. Responsibility to gain this collateral history was observed as being shared amongst the team although multiple challenges were observed with this: different professionals gaining varying degrees of information; instances where professionals wrongly assumed others were taking responsibility; limitations with how this history was documented in notes, and examples of professionals on new wards (following a transfer from the base ward) repeating the gathering of background information from a relative or carer.

Another area observed related to how the difference between current and baseline levels could be open to subjective interpretation. This was illustrated during the tracking of Patient 5:

The initial physiotherapy assessment for Patient 5 was completed at the emergency hospital the day after admission and the patient's mobility was documented at this time as 'Assistance of 1, typical PD gait' and recorded the judgement of 'likely baseline'. Despite this early impression, the patient remained in hospital (with a short stay in the Emergency Hospital and longer stays on the base ward and rehabilitation ward) for over 5 weeks, with 'more physio' and 'continue progress mobility' regularly cited in the clinical records and with a statement 13 days after admission 'for home when physio feels is at baseline'.

Other examples noted throughout fieldwork suggested differences from current and baseline levels. For example, a patient recorded as previously 'independent furniture walking' was assessed during admission as 'mobile with a handhold assistance of one'. Similarly, a patient who was noted as being 'mobile independently with a wheeled zimmer frame' and assessed during admission 'mobile with a wheeled zimmer frame of 1'. When the researcher asked professionals what they perceived the differences to be, they acknowledged that this could be open to interpretation.

5.4.2. Engagement with a rehabilitation process

5.4.2.1. The influence of the response to rehabilitative interventions

Professionals were continuously reviewing their judgement and evaluation of rehabilitation potential based on how the person was engaging with the rehabilitation process. The physiotherapist on the base ward highlighted this during a discussion about Patient 2 where he discussed that it was 'early days' and went on to say that 'until I see that progress, that initial "this is where we are"...I can then make that decision [about rehabilitation potential]'. An occupational therapist on the base ward also emphasised this saying that she 'would have to see them' in order to make a judgement about rehabilitation potential, suggesting that such judgements cannot be reached on background information alone but instead are influenced by a patient's response to, and involvement in, therapeutic interventions.

A critical issue for practitioners on the base ward lay in their reflections that they often did not feel they could deliver rehabilitative interventions, despite recognising that delivering this influenced their judgement about a patient's rehabilitation potential and therefore in turn, significantly influenced what happened next for patients. Illustrative examples of this have been provided in other sections although include professionals discussing the base ward as 'more about discharge planning than rehab', and also captured by a nostalgic discussion with one member of staff on the base ward which began 'when we were a rehab ward...'.

5.4.2.2. Allowing time

On both the base ward and the rehabilitation ward, professionals often expressed needing time to make decisions about rehabilitation potential and pathways, alongside time for rehabilitation itself. Frequently within morning MDT handovers and weekly MDT meetings, a plan for 'more time' was communicated verbally and in writing. Specific examples included professionals indicating that they needed 'more time' to assess appropriateness for transfers to other wards; plans from weekly MDT meetings documented as 'more time' (or sometimes more specifically 'more time with PT' or 'more time with OT/PT'); and a patient with fluctuating delirium discussed in a weekly MDT meeting with the summarising comment stated as 'watch and wait'.

As with the base ward, the team on the rehabilitation ward also expressed similar statements about planning 'more time', and this seemed to be discussed anew on transfer to this ward. However, it was noted that on the rehabilitation ward, the expressions were often more specific and time-framed than the examples observed on the base ward. For example, at the first weekly MDT meeting following Patient 2's transfer, the consultant verbally discussed that she was 'not sure what her rehab potential is...give her 2 weeks and review'.

This desire for more time was often understandably linked to the need to assess and judge how patients responded to rehabilitative interventions. Time was also seen to equate with the provision of a rehabilitation process of an appropriate quality. For example, when asked to describe rehabilitation, one interview participant expressed a direct link between the concept of rehabilitation and a process allowing and affording time:

"The way I look at rehabilitation is having time to work with something... even if you look at a ward handover it just says 'rehab'...which it just sort of...we all understand it that they need more time with the physio or the OT, that's how it's understood" (I5, Line 22).

If professionals felt that affording time was linked to the quality of rehabilitation, this perhaps contributed to feelings of compromised or improper rehabilitation when time was inevitably pressured. During a discussion with the base ward physiotherapist, a desire to provide intensive rehabilitation was expressed but coupled with a suggestion that they did not have the time for repeating activities such as standing practice or passive movements several times per day. Feelings that the amount of time allowed for rehabilitative interventions was unsatisfactory was also reflected during interviews:

"It's working with them daily, having daily input with that person, practicing tasks, so that they can...possibly become independent or improve in their independence in that area that we're looking at. But that relies on having the time to do that" (I5, Line 24).

5.4.2.3. Mood and motivation to engage

Professionals regularly discussed the importance of mood and motivation as factors which influence their judgement about rehabilitation potential. One physiotherapist discussed her thinking that 'a big part of it is...motivation and [the patient] understanding what we're trying to achieve...if we can do that things work a lot better'. And when reflecting on Patient 2, the same physiotherapist mentioned that her mood was very low, mentioning this before other things that may be influencing what she was thinking about the progression and potential of this patient.

This also seemed to be the first thought of other professionals on other occasions – for example during an interaction with an occupational therapist at the intermediate care unit, when asked to bring to mind people who had good rehabilitation potential, the first thing she mentioned was 'someone who is motivated'. Another physiotherapist, through her own words, placed emphasis on the importance of mood suggesting that 'her mood has got a massive amount to play in it [her rehabilitation potential]'.

Interview extracts also emphasised this as a prominent feature of professional reasoning:

"I still think this [motivation] is a really big one" (I1, Line 329).

"I think the main thing is people's engagement and how, you know, focussed and willing they really are to want to engage in rehab" (I4, Line 170).

A related issue noted during fieldwork was the extent to which professionals felt that low mood and motivation was something which could be improved in order to positively influence rehabilitation potential and subsequent rehabilitation outcomes. In the earlier section relating to the meaning of rehabilitation, it was noted that a rehabilitative approach to mood and anxiety problems, such as increasing 1:1 time with staff or social activities within the ward, was seen as something that would be 'nice to do...but not realistic'.

Another example, in relation to Patient 2 during her time on the rehabilitation ward, involved the physiotherapist reflecting that 'it might be better if can try and get her moved in to the bay where she's actually got some company'. Although the physiotherapist then followed this by adding 'unfortunately a lot of people in the bay need to be in the bay....but it's something to just keep an eye on'. The move in to a bay did not happen for this patient whilst they remained on the rehabilitation ward and was not documented as part of a proposed plan in her clinical records.

5.5. Chapter summary

This chapter has presented a multi-dimensional understanding of rehabilitation which emerged from the thematic analysis. Although the data relied heavily on time spent and perspectives shared with occupational therapists and physiotherapists, the dominant discourse amongst all professionals was that rehabilitation was a professionally-led process, predominantly led by physiotherapy, to promote improvements in physical abilities. This was linked in part to physiotherapy being involved earlier and more frequently in the delivery of rehabilitation interventions. Professionals described aspirations or 'gold-standard' versions related to rehabilitation although often realities did not correlate to these ideals. The examples and wider implications will be critically examined in the discussion chapter. Rehabilitation potential was recognised as an ambiguous concept which in practice often lacked specificity. Interestingly, the overlapping nature of evaluations of rehabilitation potential and decisions about transfer of care contributed to the evaluation of rehabilitation potential bearing responsibilities of gatekeeping, which subsequently linked to the decisions being controversial and open to challenge. Through speaking in the third person, interview extracts provided insight that professional ownership of these challenges may be difficult. This leads in to the next findings chapter which presents the ethical dimensions and tensions of professional reasoning in relating to rehabilitation decision-making.

CHAPTER SIX – RESEARCH FINDINGS: THE ETHICAL DIMENSIONS OF EVALUATING REHABILITATION POTENTIAL

6.1. Chapter introduction

The influence of ethical dimensions on practitioner reasoning emerged as significant during data analysis and is presented as one of four over-arching themes. Subthemes will explore how practitioners considered the consequences of an evaluation of rehabilitation potential – such as the desire to bring about a positive outcome for patients and the desire to minimise harm – and how values placed on such consequences were enacted in practice. Furthermore, it was an initial objective of this research to understand patient involvement in the evaluation of rehabilitation potential and subsequent pathway decisions, and sub-themes pertaining to this important element will be developed. Finally, how professionals were managing ethical dimensions will be discussed. Once again, the process of using early codes to develop themes and subthemes is illustrated in Figure 10.

Early Coding:

Whilst using the initial code of 'decision-making', different ethical dimensions were recognised within the data. A new code of 'ethical dimensions' was then utilised to review the whole data set.

Ethical dimensions:

- Aiming for a good outcome
 - Giving people a chance
 - Not 'writing people off'
 - (Absence of) goal setting
- Avoiding harm/managing risk
- Blurring of doing good and avoiding harm
- Involvement
 - Is desired
 - \circ Not routine
- Patient Involvement
 - Secondary to family involvement
 - \circ If disagree question capacity
- Family involvement
 - o Surely they should know?
 - \circ Default position?
- Barriers to involvement
 - \circ Time
 - \circ Lack of alternatives
 - \circ Not wanted?
 - \odot Those who speak up are privileged
- Rationing
 - \circ Involvement becomes rationed
 - \circ Record keeping rationed
 - o Rehabilitation as a rationed resource
- Fairness to older people
- Professional experiences of ethical distress

Examples from the data:

- **Ethical Dimensions**
- **Aiming for a good outcome/giving a chance:** Fieldnotes p70 'giving people a chance'
- **Not writing people off** Fieldnotes p163 "I don't think you can write people off".
- **Avoiding harm –** Fieldnotes p168 "It would be a shame for her to go home with a full-body hoist"
- **Blurring of doing good and avoiding harm** Fieldnotes "It's a bit of a fine line when somebody's getting very very upset though and saying that they desperately want to get home"
- **Involvement desirable** I1 "I think we all want this ideal text-book thing"
- **Not routine** 12 "sometimes we give the illusion of involvement in that decision but I think some of the time that decision has been made"
- **Patient involvement secondary to family:** Fieldnotes p188 "Providing family are happy and she is happy".
- **Surely they should know** Fieldnotes 190 "family have seen her everyday, surely they should have seen this?"
- Default position I5 "We definitely speak to the family more"
- **Barriers** 15 "I don't really think patients and families have a lot of say in that, no. I guess...there's limited resources"
- **Ethical distress** I4"It really makes me uneasy when I have to like compromise my therapy for bed pressures"

Final themes and Subthemes:

Theme: Ethical dimensions of evaluating rehabilitation potential

Sub-themes:

- Brining about a positive outcome
- Avoiding harm
- The blurring of doing good and avoiding harm
- Autonomy and involvement
- Fairness and rationing
- Professional tensions and discomfort
- Strategies for dealing with competing demands and ethical tensions

6.2. Bringing about a positive outcome

The desire that the provision of rehabilitative interventions would lead to eventual improvements for individuals, and therefore positive outcomes, was clear through all aspects of fieldwork. Professionals regularly expressed their desire to bring about functional improvements in terms of giving people 'a chance'; that is to say, they wanted to optimise the likelihood of improvements through the provision of rehabilitative interventions and the creation of rehabilitative conditions. Professionals also linked a positive evaluation of rehabilitation potential as a gateway to the provision of a rehabilitation resource, and therefore as a key part of increasing the likelihood of improving outcomes.

When talking about the judgement about rehabilitation potential for Patient 1, and reflecting on the reasons for listing her for the rehabilitation ward, the physiotherapist on the base ward discussed wanting to give her a chance to make further improvements. Similarly, when discussing beginning to reduce physiotherapy input for Patient 2 towards the end of her admission on the rehabilitation ward, the physiotherapist discussed that she would listen out at the morning handovers for any sign of improvements because she always wanted to 'give people a chance'. This was echoed by wider team members when one social worker discussed how she thought 'everyone deserved a chance'.

Although goals of rehabilitative interventions were not always explicitly stated, the primary purpose of a rehabilitative phase of care, understood through many observations and interactions during phase 1 and 2 of fieldwork, was to work towards functional and mobility improvements to achieve as close to baseline function as possible. This represented an implicit understanding that this represented a positive outcome for the patient.

174

At this point, information about functional status available from the five patient cases tracked during phase 2 of fieldwork is worthy of note. Table 8 below summarises information in relation to the functional level at the point of initial therapy assessments, the optimum or best functional level noted during the whole episode of care, and the outcome at the end of tracking.

 Table 8: Summary of information obtained from clinical records documenting optimum

 functional level of patients tracked during phase 2 (continued overleaf)

	Notes taken	Notes taken	Optimum level	Outcome at
	from	from	during	the end of
	documentation	documentation	admission	tracking
	of initial PT	of initial OT	documented in	
	assessment	assessment	medical records	
Patient 1	Sit to stand independently Assistance of 2 for standing balance Mobile 4m with Assistance – requiring oxygen	Due to existing carer support from daughter and existing equipment, no OT needs identified at time of initial assessment	Optimum sit to stand noted as independent on admission. For mobility, minimal assistance of 1 and walking aid noted throughout admission	Died on rehabilitation ward (length of admission 52 days)
			Ongoing reduced exercise tolerance	
Patient 2	Sit to stand with assistance of 2 and 4 wheeled walker Mobile 5m with assistance of 1 and 4 wheeled walker	N/A	Sit to stand with maximum assistance of 1 Mobile 6m with maximum assistance of 1 and wheeled frame	Inpatient on rehabilitation ward. Awaiting 24 hour nursing care No longer receiving active therapy Using full body hoist for transfers
Patient 3	Sit to stand with supervision Mobile with delta frame 40m	Home equipment in situ. Query whether at baseline	Sit to stand independently Mobile independently with a frame Independent bed transfers	Died on base ward (length of admission 26 days)

Patient 4	Mobile 6m with wheeled frame and assistance of 1 but shortness of breath on exertion noted	Bed transfers with assistance of 1. Query about bathing transfers	As noted during initial PT and OT assessments	Discharged home (length of admission 12 days) NB – readmitted and died during subsequent admission
Patient 5	Assistance of 1 lying to sitting Sit to stand with assistance of 1 Mobility with assistance of 1	Issues noted with transfers and carer support. Assistance of 1 for bed transfers and manoeuvring in bed	Lying to sitting to standing independent with raised bed and bed lever Sit to stand from chair with assistance of 1 Mobile independently with delta frame	Discharged home from rehabilitation ward at optimum level.

It is important to emphasise the small number of patient cases, the limitations of using narrative information from clinical records (as opposed to valid outcome measures at points of care delivery and transfer), and the many influences on patient outcomes linked with frailty and co-morbidities. And this is presented as a partial account only, due to the fact that I could only ever understand the parts of the episode of care that I observed, or the representation that was captured in clinical records. However, this summary highlights that three of the five patients had died by the end of tracking and one was documented to be at a lower functional level and awaiting 24-hour nursing care, and suggests that reasoning which was highly influenced by a desire to give people a chance of positive outcomes through a period of rehabilitation warrants critical reflection.

6.3. Avoiding harm

Practitioners were aware that an evaluation of rehabilitation potential, potentially influencing whether a person would then receive a further rehabilitation service – either bed-based or otherwise – also had the potential to cause harm. Similarly to the way a positive evaluation of rehabilitation potential was attributed particular vocabulary around being given 'a chance', a negative evaluation of rehabilitation potential was often discussed in terms of people getting 'written off'. In simple terms, practitioners placed value on the ability of rehabilitation to bring about improvements and therefore were aware that withholding or denying a person access to this could result in a missed opportunity for improvements and potentially a negative outcome. This was expressed by one physiotherapist in relation to Patient 2 whilst reflecting that she had reached a plateau although they did not want to conclude their interventions at this point because 'it would be a shame for her to go home (with a) full body hoist'. This mobility status of using a hoist – which was different to how this patient mobilised or transferred prior to admission – was perceived as a negative outcome and one that could be potentially remediated through further rehabilitation.

The consideration of the rehabilitation pathway was also influenced by awareness of harm and risk. The below vignette illustrates how the decision about the rehabilitation pathway also become a decision about how to manage risk:

Patient 5 was transferred to the base ward 2 days after his emergency admission for a suspected stroke or seizure but what was later determined to be a worsening of symptoms associated with Parkinson's Disease. He spent 21 days on the base ward with a main focus on stabilising the symptoms of Parkinson's Disease through medication management and optimising his mobility and transfers. During this time, there was significant discussion in MDT meetings and morning MDT handovers about whether he would be discharged home directly from the ward, with one of the main concerns being the ability of his wife to meet his care needs as this had been raised as a concern prior to admission.

A planning meeting was held 9 days after his admission (7 days after his admission to the base ward) where a potential care package on discharge was discussed although no firm timeline or pathway decisions were made. Notes about this planning meeting consisted of a brief entry in the medical records.

He was subsequently listed for the rehabilitation ward 18 days after his arrival on the base ward and when this decision was made, the occupational therapist reflected 'from his condition and just from what the nurses have said, I didn't think he probably had that much [rehabilitation potential]. His Parkinson's is meant to be better managed now and from what I saw, his mobility and function were still quite poor. I don't think there's possibly going to be that much change and my main concern around him going home would be the wife'. This was echoed in discussion with the physiotherapist on the base ward who indicated that 'if [Patient 5] lived on his own, we may have been able to get him home with a big care package'.

6.4. The blurring of doing good, and minimising harm

Professionals did recognise that this ongoing desire to work towards improvement could potentially become negative. One physiotherapist raised the professional challenge of at 'what point do you draw the line?'. On further discussion with this professional, tensions were discussed between the desire for improvement potentially becoming harmful for the individual (for example by contributing to a longer hospital stay or unrealistic expectations), or potentially harmful for those individuals not receiving services because the time of the physiotherapist was being allocated to this person. Such discussions provided insight in to the pressure felt by individual professionals to maximise positive benefits and minimise harm when making decisions about continuing and concluding rehabilitative interventions.

Observations and discussions suggested that giving people 'a chance' was often associated with a longer hospital stay because of a professional belief that this represented the best chance to bring about a better outcome. This was captured by the physiotherapist on the rehabilitation ward in relation to Patient 2: 'It's a bit of a fine line when somebody's getting very very upset though and saying that they desperately want to get home...sometimes you think, is it better to get someone home as quickly as possible even if it does mean them being hoisted and not reaching their potential or is it better to keep them...to work on that and to try to get them better. I think...if we could keep her for a bit longer, I think we probably will get there'.

A positive evaluation of rehabilitation potential in order to access a rehabilitation bed to afford time and manage risk was seen to have the potentially harming effect of raising expectations for patients and families. When Patient 5 was asked by the researcher what he understood about why he was transferred to the rehabilitation ward he indicated 'to get stronger'. Following this, on the first working day following his transfer to the rehabilitation ward, the new physiotherapist involved in his treatment discussed planning to assess him that day to 'look at bed transfers and to try with a delta frame' which presented as indicating a new assessment of potential and potentially new rehabilitation goals. As outlined in the vignette on page 177, this presented as being out of line with the reason for transfer to the rehabilitation ward communicated by the therapists involved on the base ward which was primarily to ensure his care needs and social situation could be managed safely. I did not observe this reasoning discussed verbally between professionals on the base ward and the rehabilitation ward and it was not explicitly documented within clinical records. In comparison, the patient's understanding was that he was being transferred in order to work towards getting stronger, and the assessment commenced by the new physiotherapist also presented as being aligned to this.

A similar issue was noted with another patient when, during a discussion with the base ward physiotherapist before his transfer to the rehabilitation ward, it was highlighted that 'his family want him to use the stair lift...this might not be realistic but it's something for [the rehabilitation ward] to look at.

179

6.5. Autonomy and involvement

To begin this sub-theme, it is important to note that patient and family involvement was seen as a desired and 'ideal' feature of rehabilitation, although it was also understood that this was perhaps not always happening in this desired way. This was summarised during an interview:

"I think we all want this ideal text book thing...you know, goal-led, get the carers, get the family members, get the patient, do it all together. But it's not always as easy as that. Unfortunately" (I1, Line 258).

6.5.1. Involvement in evaluations about rehabilitation

potential...rehabilitation what?

The concept of rehabilitation potential did not appear to be one that was explained or translated explicitly to patients and families, or indeed a concept that patients and families were encouraged to consider. Therefore I did not observe patients or families making an explicit contribution to this judgement. Instead, it was observed that when rehabilitation potential was deemed to have been reached, this was then subsequently explained to patients or families.

This was observed in the example of Patient 1 who had a lengthy hospital admission (approximately 7 weeks) with periods of acute medical issues including aspiration pneumonia and cardiac events. During this time, there were discussions amongst the team questioning and reviewing rehabilitation potential. However, it was approximately 6 weeks in to the admission that an explicit entry was made in the patient's medical notes to record a conversation between the occupational therapist and the patient's daughter that further progress through rehabilitation was unlikely.

Although the content and timing of this discussion was logical, and did fit with the process presented in Figure 5 (which outlined that time is needed to treat and

overcome medical issues and to monitor the extent of progress through engagement with a rehabilitation process), it was not made explicit within documentation that the earlier stages of this process (that is to say the time needed to evaluate potential and that the team would be looking for specific progress towards identified goals) were communicated to Patient 1 or her daughter.

Linked to this, when rehabilitation potential was deemed to have been reached, this was not always fully explained to patients or families. For example, the physiotherapist from the rehabilitation ward discussed that Patient 2 had no further rehabilitation potential and when asked whether this had been discussed with family members the response was 'No [not] at this stage...but they have seen her every day...surely they should have seen this'. This suggested that the family in this example were left to interpret for themselves that further functional gains were unlikely.

A potential outcome of patients and families implicitly understanding such judgements was expressed during an interview, where the health professional expressed a difference between the professional evaluation of rehabilitation needs and potential and that of a family member:

"Now his mum was in her nineties and was really quite frail and you think...well how could you possibly expect that your mum could cope with 2 hours of physiotherapy a day. Because that's just completely unrealistic" (I1, Line 499).

6.5.2. Involvement in pathway decisions

During observations, I did not witness a clear approach to involving patients and families in decisions about their pathway, transfers to different locations, or what services they would receive. There were places where this was happening, and many professionals could potentially be involved in different ways at different times, although the potential for gaps and variances was also noticeable.

A movement or transfer within the pathway often involved an onward referral (for example to list for another ward or to refer to another service such as social work or the reablement team) and therefore one of the first considerations relating to involvement in such decisions was the issue of consent to such referrals. Consent to list for other wards or to refer to another service such as social work was not witnessed clearly during observations, or explicitly found to be documented in any examples of records viewed during patient tracking. During an informal discussion with a social worker, she reflected that often the ward do not tell people they have been referred to social work and reflected that one consequence of this is that it is then difficult for her to introduce herself. Other staff also acknowledged that consent for such services tends to be assumed, linked to an assumption that people want whatever it takes in order to get home.

Outside of consent for onward referrals, discussions with patients and relatives about transfers to other wards were also not always apparent. In the case of Patient 1, during her 23 day stay on the base ward, there were many times that decisions were reviewed in relation to her pathway and potential transfer of care. Transfer to the rehabilitation ward, transfer to the intermediate care unit, and discharge directly home from the base ward were all verbally discussed and documented in records, normally preceded by the '?' symbol in the written records. However, there were no clear entries in the medical records to confirm that these alternatives were discussed with the patient or her family members.

In the case of Patient 2, a transfer from the base ward to the rehabilitation ward was first discussed 4 days after admission and documented in the medical notes from the MDT meeting as a possibility (noted : '? Ward ** [the rehabilitation ward]'). Following a ward round with the consultant 2 days later, this decision was confirmed and the patient was listed for transfer. Discussion with the nursing team indicated that the ward clerk would then contact the bed managers to list for the transfer and if a bed became

182

available, the ward clerk would then contact family members to inform them that she was being moved. However, the fact that this was delegated to the ward clerk and occurred at the point of transfer, implied that this was to inform them of the move rather than to discuss the reasons for the move, or their wishes in relation to this.

6.5.3. Patient involvement after decisions

One of the issues noted most frequently was that during discussions about potential transfers to other wards or services, the decision or action tended to be made first with the views of patients and families noted afterwards (see examples in Table 9)

Context	Extract from Fieldnotes
A patient discussed in a base ward MDT meeting	Consultant: 'Here or [the intermediate care unit]if they have beds at [the intermediate care unit] and she is happy to go this seems like a plan'
A patient discussed in a base ward MDT meeting	Consultant: '[Name of intermediate care unit] if patient agrees'
A patient discussed in a base ward MDT meeting	Consultant: 'Can list for [rehabilitation ward] if she's happy'
A patient discussed in a rehabilitation ward MDT meeting whether would be a candidate for the specialist unit for cognitive assessment and rehabilitation. Team discussion that the Modern Matron had asked if she can just go straight there.	Nursing: 'Hang on – no-one has discussed this with the lady or family'

Table 9: Examples illustrating patient involvement after decisions

In all of the examples above, the decision was noted or stated first, and the issue of

consent from the patient was a secondary point. And the final example illustrates that,

at least on this occasion, the team were aware that this was happening.

6.5.4. Patient involvement after family involvement

There were also similar examples where the views of the family were acknowledged

first, with the views of the patient either mentioned afterwards or not at all (outlined in

Table 10).

Table 10: Examples illustrating patient involvement after family involvement

Context	Extract from Fieldnotes
A patient discussed during an MDT morning handover	Nursing: 'Patient listed for [intermediate care unit] and family are in agreement'
A patient discussed in a rehabilitation ward MDT meeting whether would be a candidate for the specialist unit for cognitive assessment and rehabilitation (as above)	Consultant: 'Provided the family are happy and she is happy [the unit] seems reasonable'.

The essence of these issues was perhaps insightfully illustrated by a discussion in a base ward MDT meeting about whether to repatriate a patient who was out of area to her local general hospital. After lengthy discussion the consultant asked the team 'What does she want?', to which a member of the nursing team responded 'don't know...we haven't asked her'.

The issue of involving family members – either before, or instead of, patients

themselves, was also reflected during interviews:

"I think we're all guilty sometimes of just...homing in to the family member and...kind of forgetting that that person's there and is taking it all in. I would like to think that the wishes of the patient are taken as top priority but I think, especially in elderly care, quite a lot of the time, it's what the families think and kind of not what the patient either wants or feels" (I1, Line 576).

6.5.5. The vocal and not-so-vocal

There were some occasions where patients or family members were observed to vocalise their wishes about their pathway and place of care and this directly influenced outcomes and decision-making. For example, a patient who was deemed to have rehabilitation potential by the team, and that a transfer to the rehabilitation ward would be appropriate, subsequently made her wishes known to the consultant that she would prefer to stay on the base ward. Following this, the team decided that it would be appropriate for her to stay and she remained on the ward for a further period of rehabilitation and discharge planning. Similarly, there was an example where the team on the base ward proposed listing a patient for a continuing health care bed in a local nursing care environment although at a morning MDT handover meeting, a member of the nursing team discussed that when this was mentioned to family members they said 'no way...they think it's filthy'. Again, this subsequently altered the course of action and led to the team considering alternative pathways.

Within interviews, when asked about involvement of patients and families in decisionmaking, the stories that were most forthcoming were about situations where patients – or more often families – became vocal. There were two reoccurring perspectives within these stories: that those who were vocal often became so because of issues of a more negative nature; and that those who were vocal were often more involved or more listened to than those who were not. The extract below illustrates these issues:

"Often, I almost feel like the patients who might stay on a ward longer than perhaps the MDT would suggest, it's because their family have stomped their feet and actually they might kick up a bit of a fuss if the patient was to be discharged home in whatever functional status they're at. So, it's probably those families and patients who get listened to more than others who perhaps don't kick up a fuss and don't em...create problems" (I5, Line 36).

A particular extract develops this further, with language suggesting that this type of involvement can become confrontational and challenging:

"But families are sort of quite demanding about, you know, my mum needs to be in 24 hour care, there's no way she can cope at home. Em...like an example I could give of that from a few weeks ago...and [at one point] I had basically been dragged in to an inquisition by the daughter and the son with the Physio. They'd turned up on the ward and demanded that we had a planning meeting and we were the only 2 people there so we sort of got sent in like lambs to the slaughter (laughs)" (I2, Line 736).

Despite this, there were other examples within observations where patients and families attempted to vocalise a preference in relation to their place of care and this did not influence the outcome. A patient on the base ward, who had been transferred from another local hospital and also had internal transfers between wards during his admission, was discussed as being appropriate for another transfer to the rehabilitation ward. It was discussed at the morning MDT handover on the base ward that he did not want to go to that ward although a member of the team indicated 'it doesn't really matter to him...it's just a bed' and he was subsequently listed for the transfer.

6.5.6. Reasons for limited involvement

6.5.6.1. Not wanted?

Although there were examples where patients and families demonstrated vocal involvement in decisions, some patients and families were observed to be more passive, or to occasionally actively avoid involvement. During fieldwork observations, the team on the base ward shared examples where relatives would not return calls, or not attend meetings on the ward and the team linked this to not wanting to be involved in decisions or to potentially creating barriers to decision-making.

Patient 5 declined to attend a planning meeting being held on the rehabilitation ward to plan his discharge, instead suggesting that his son and wife would attend and represent him. When asked about this, this patient indicated that he would find it difficult to attend a meeting where all of the discussion was about him. In the case of Patient 2, there were multiple times when family members presented as experiencing difficulties taking responsibility for particular decisions – asking for delays to a decision about resuscitation status and not wanting to take the responsibility for consent to psychiatric assessment or commencement of anti-depressant medication.

Professionals shared reflections about why patients and families may not want to be involved in decision-making. Participants indicated that patients and families may revert to understanding of traditional hierarchies and feel that professionals who hold specialist knowledge should also have responsibility for decision-making:

"You get those people who simply say 'fix me', I don't care what you do, you just fix me" (I3, Line 708).

"And some families seem to have that old fashioned view that the doctor knows best and if the doctor tells you then that's what you do, or the professional...and they think you've got the right answers because you're the professional and they don't need to weigh in" (I5, Line 403).

Another extract from an interview suggested that involvement in healthcare decisionmaking can feel unfamiliar and overwhelming and again, this may explain why some patients and families present as not wanting to be involved:

"Some families I think...the whole hospital environment is quite alien to them and they don't really know how it all works" (I5, Line 402).

6.5.6.2. Rationed out?

From fieldwork observations, I became aware that meaningful involvement of patients

and family members involved time-intensive activities - such as discussions, joint

therapy sessions and telephone calls - and in a highly time-pressured environment, an

obvious consequence was that such activities may subsequently be rationed.

This impression was supported when professionals were encouraged to explore this

issue during interviews, with one participant in particular highlighting the issue:

"Because if you've got 28 people on a ward, you haven't got time every day to be going to see families and to update...it just isn't possible you know....Sometimes, it's just not as easy as...as you think you know. And if you...if you do a joint session with a family, that could take you an hour really. Because you'll be explaining lots of things, and so...you couldn't do that every day, do you know what I mean?" (I1, Line 523)

Also, one participant suggested that it may not only be activities involving patients and family members which are rationed due to time, but also, the documentation of these

activities:

"So I guess you do tend to sometimes fall in to, I'm in a rush, there's no point in me writing 3 pages when I can write 2 lines" (I2, Line 354).

6.5.6.3. Lack of meaningful alternatives?

Another explanation for why meaningful involvement in decisions pertaining to

rehabilitation pathways was challenging in this context was that there were a perceived

lack of alternatives which may have resulted in involvement of patients and families

feeling futile. Professional perspectives shared during interviews suggested that

historically, discussions with families may have involved the presentation of different

alternatives, although those alternatives may now not be available and therefore the

discussion itself may not take place:

"I would have always considered if there was going to be beds available but now we know that it's so constrained, that there is such limited availability that sometimes you can't even consider that as an option because you know it's not an option" (I5, Line 227).

6.6. Fairness and rationing

Evaluations of rehabilitation potential and the subsequent recommendation about an appropriate pathway inevitably involved decisions relating to resource allocation and fairness. At the level of the organisation, it could be perceived that the closure and reprovision of the bed-based intermediate care unit and the re-provision of the identified rehabilitation ward in to a 'discharge ward' were linked to the allocation of finite resources and the setting of organisational priorities.

Individual practitioners were continuously making decisions in order to allocate finite resources. Both physiotherapists and occupational therapists discussed how they prioritised their daily caseload, how they delegated work to assistants and how they prioritised patients for therapeutic input at weekends, all informed by implicit evaluations of who had potential to gain most and underpinned by attempts to distribute and maximise finite resources.

Wider examples of resource allocation and rationing have been illustrated in earlier sections. For example, onward referrals for rehabilitation or reablement for those with significant cognitive impairment could sometimes be ruled out based on decisions that these individuals may not benefit from this limited resource. Therapeutic interventions to address mood and social interaction were deemed to be desirable aspects of rehabilitation, but areas that practitioners felt unable to prioritise. Similarly, the approach for managing referrals to occupational therapy meant that any patient already living in a nursing or residential care environment was judged not to be a priority and therefore either not referred, or referred back to the original referring agent. Also, activities which involved relatives and carers – either in direct therapy sessions, or in discussions to aid decision-making – were discussed on some occasions as not being a priority. Importantly (and particularly related to this study which utilised care records

as a means of gathering information) limited detail in care records was also linked to rationing by some professionals, with time being prioritised towards other activities.

Some examples of rationing were also so implicit that it was only through brief interactions that they became momentarily detectable. During one interaction, I was discussing prioritisation for weekend physiotherapy with the base ward physiotherapist. The physiotherapist acknowledged there were patients who would ideally be listed for weekend physiotherapy if they had more staff, however because this decision not to list had unnoticeably already been made, it subsequently was not documented nor discussed with the weekend team. And perhaps most importantly, the patient did not know that they had been considered for, or could have received, this intervention.

The specific issue of allocating time for the rehabilitation of elderly patients was raised as a critical challenge during one interview, suggesting rehabilitative interventions for this population took more time than rehabilitation for a younger population and this led to questions about the fairness to others of allocating time in this way:

"You might have rehab with a younger person that takes you half an hour, forty minutes...and the same rehab session with an older person might take you the entire morning...I think...if you have twenty...I think it was 29 patients I had...you know, you have to try and balance what's happening with everybody" (I2, Line 45).

The influence of the organisation can be seen in one example previously presented (page 159) where the base ward physiotherapist reflected on the prioritisation of daily workload and how seeing those patients who could be moved on in the shortest time was more of a priority than seeing new patients on the day of admission.

6.7. Professional tensions and discomfort

Throughout the phases of fieldwork, the problematic nature of balancing the inevitable tensions of the needs of the individual, the needs of the organisation and professional values and integrity were frequently discussed. On many occasions, professionals described visions of what they would like to happen in comparison with what was happening in reality (examples noted in section 6.2.5).

Frustrations due to limitations with time and resources and the subsequent perceived failings with the system were frequently expressed – sometimes directed at the system as a whole and at other times, directed at particular services or professions. A ward doctor captured overall frustrations with care of older people in general when he shared that the "most likely outcome…home with increased care and some OT equipment, back within a fortnight and go to 24 hour care. That's my cynical care of the elderly assessment". And one physiotherapist discussed being acutely aware that 'it all comes back to money and resources' and expressed the tension with doing what is best for individual patients.

On other occasions, such frustrations were directed towards identified services or individual professions. For example, one professional expressed an opinion that length of stay meetings had come about by delays or deficiencies in the social work service stating 'this is why you have length of stay meetings...social, social, social, social'. And when discussing the involvement of a team to provide mental health liaison services, during a base ward MDT meeting it was commented that 'all they have done is stick an assessment from May in the notes'.

The theme of professional frustration – and sometimes associated guilt - was then communicated strongly during individual interviews when perhaps the act of discussing this in a private space and away from clinical duties, promoted reflection on these issues in a more personal way.

191

"I think we are all guilty of that sometimes (focussing on talking to families because it is quicker or easier to do that collaborating with patients)...I always feel awful when I've done that...but sometimes it just happens" (I1, Line 576).

"And the problem that I have with that (rehabilitation aiming for safety rather than improvement back to baseline) is that I sort of feel that we're failing patients because....patients aren't getting rehabbed back to baseline...in terms of mobility and things. That then has a huge impact on their other functional activities of daily living" (I2, Line 80).

One participant expressed the direct link between the allocation of resources, the

options that could be available to people and the professional frustrations experienced:

"I think resources again...frustratingly...is something that we really have to think about in terms of pathways. I would have always considered if there was going to be beds available but now we know that it's so constrained...sometimes you can't even consider that as an option because you know it's not an option" (I5, Line 224).

Responses suggested that such frustrations could also be linked to conflict with others:

"Because, if everyone's on the side of this person doesn't need to be here any longer and you're the lone voice saying 'hang on a minute I still think they have potential...we should give them another few days'. Then sometimes you're the person who's spitting in the wind basically" (I2, Line 567).

At other times, interview responses suggested feelings which went beyond frustrations

with self, others or systems, with instead responses suggesting feelings of professional

compromise and distress:

"It really makes me uneasy when I have to like compromise my...compromise therapy for bed pressures...I don't know I just don't like talking about it and using all the terminology...I feel like patients are just numbers and it's not very patient centred" (I4, Line 138).

6.8. Strategies for dealing with competing demands and ethical tensions

6.8.1. Overt strategies

6.8.1.1. Waiting lists

The most explicit use of waiting lists could be observed in relation to patients waiting for another service within the system – such as a bed on the rehabilitation ward, the intermediate care unit or less frequently, placed on a waiting list for a service such as community physiotherapy. For bed-based rehabilitation, once the team had decided that a patient had ongoing rehabilitation potential and rehabilitation needs, a telephone referral was normally made by the ward clerk on the base ward to the bed management team in order to place the patient on a list for the identified ward or unit. Bed managers held lists for both the rehabilitation ward and the intermediate care unit (among many other wards within the hospital system) and when a bed became available, discussion with bed managers suggested that they would normally prioritise those for transfer using the date order in which they were listed.

One area of interest about waiting lists was that, on occasions, professionals were observed to reach a judgement that a person would benefit from a service but decided against placing on a waiting list because of a tacit understanding of how long the person could potentially wait and the consequences for others also listed who may be more of a priority. The base ward physiotherapist discussed deciding not to refer a particular patient for community physiotherapy mainly because the waiting list was too long and the patient would likely have improved themselves during the waiting time.

193

6.8.1.2. Triage

One of the most obvious examples of where triage was employed was in the

prioritisation of patients for weekend therapy illustrated in the example relating to

weekend physiotherapy cover below:

The researcher was present with the base-ward physiotherapist during the prioritisation of patients to be listed for weekend physiotherapy. The physiotherapist talked through all of the patients who were deemed to be appropriate to be listed for weekend physiotherapy and discussed an individually developed traffic-light system.

Patients who the physiotherapist categorised as 'red' were those where weekend physiotherapy was deemed as essential. Two patients were identified in this category and included a patient whose oxygen saturation had been dropping when mobilising although this had improved progressively over the previous days and the therapist did not want her to go backwards. The second patient was new to the ward that day and 'would probably stay in bed all weekend' if she was not seen and a moving and handling plan initiated. The physiotherapist acknowledged that patients known to be needing active respiratory physiotherapy would also be categorised as red but discussed that they did not have any patients categorised in this way on this particular day.

Patient's categorised as 'amber' included a number of patients who would benefit from further mobility progression and another new patient who had not been seen but was 'likely to be there for a while'. When asked to discuss this judgement further, it was based on previous knowledge of this patient from a prior admission.

One patient was identified as 'amber' for stair practice, but that this would become 'red' if a decision was made that the patient could be discharged over the weekend.

All other patients were identified as 'green' and were discussed as patients who were not benefitting from active physiotherapy (for example they were medically unwell) or where established plans were in place (for example nursing staff would be following established mobility or moving and handling plans). The physiotherapist discussed that wards all had the ability to refer to the weekend team if patients changed, or if respiratory physiotherapy was required.

The physiotherapist did discuss that if he/she was the physiotherapist providing cover at the weekend, and time allowed, some patients on this ward may be seen in addition to those listed – with additional discussion that they were already known to this particular therapist.

The discussion concluded by reflecting that each of the wards which were categorised under the 'care of the elderly' service would go through a similar process and that each therapist or team of therapists may approach this task slightly differently. The list from this ward would then be reviewed and negotiated with the wider care of the elderly team, with patients changed (or more specifically, the example was given that they could be removed or the status changed) depending on how realistic the full list would be for the provision of weekend input.

A similar strategy was utilised by occupational therapy, with priority given to those patients where discharge could be facilitated over the weekend period, or where a delay in assessment or intervention may mean a longer hospital stay.

Triage was also utilised as an ongoing strategy to prioritise workload during the normal working week although – perhaps because it did not have to be communicated to a weekend team, was less explicit, and less easily observed. However, because occupational therapy and physiotherapy did not operate waiting lists for assessment or therapeutic intervention, they inevitably were required to prioritise who to see on a daily basis. On more than one occasion, the base ward physiotherapist shared an anecdote that '80% of patients are likely to get better on their own' and that time and priority should be given to the remaining 20%.

Many professionals – including occupational therapists and physiotherapists, but also the ward doctors and members of the social work team – utilised the formal and more informal opportunities for information exchange in order to assist them to prioritise their own workload and manage competing demands.

6.8.1.3. Reflection

Although this was voiced relatively infrequently, professionals did share the importance of reflective practice as a strategy to deal with some of the complexities associated with decision-making in their roles. In relation to one patient, the physiotherapist on the base ward shared that his own judgements about the best course of action, and negotiating this within the team, had been personally challenging and stated that it could be the focus 'of a reflection'. The opportunity to reflect was also discussed as an element of engaging in the individual research interviews:

"No, its interesting isn't it, thinking about things...because you don't really think that you're doing all of these things as you're doing them. But when you talk

about them and you reflect them, it is quite in-depth isn't it...it is a lot to think about" (I4, Line 348).

6.8.2. Covert strategies

6.8.2.1. Creating narratives about patients

When tracking identified patients, and during wider time spent in handover and MDT meetings, narratives were used to describe patients, relatives or situations. Favourable language was used to describe some patients ('she's lovely', 'she's my favourite'), with less favourable language used to describe others. For those described less favourably, often a narrative was created which related to non-compliance with health professionals – with examples of patients refusing to get out bed, refusing bladder scans, refusing walking aids or refusing care packages all noted either through information either exchanged verbally or in clinical records. Such examples were also discussed in Length of Stay meetings (see examples on page 120).

One potential example involved a discussion in a morning handover where a patient was described as 'delightful' during the morning handover meeting and it was also highlighted that she was going home that day with reablement support. During a subsequent interaction with the allocated social worker, it was discussed that this patient was not deemed to have reablement needs but that the long term care package could not start until the following week. In this case, the rules and criterion for accessing reablement services presented as being interpreted flexibly for this patient.

In contrast, for Patient 4 it was documented and discussed in a morning handover meeting that this patient had previously 'refused' a care package and when asked by nursing staff to evaluate whether the patient would benefit from short term reablement services, both the allocated occupational therapist and social worker judged a referral to reablement as not appropriate for this patient due to the presence of long term care needs (for which the care package had been previously declined). Although justified in line with the purpose of reablement, this example contrasted with the earlier example in this section (where reablement was used to bridge a gap until the start of a long term care package) and illustrated how professionals were interpreting rules flexibly and subjectively in different situations.

6.8.2.2. Managing demand through information-giving

It was common that information about a service or intervention would not be communicated to patients unless it had already been decided that the person would benefit from this service. For example, during observations and, supported by responses during interviews with health professionals, information about a transfer to the rehabilitation ward or the intermediate care unit was often only either communicated after the person was identified as being appropriate and listed for a bed, or in some cases, only at the point that the bed became available and the move was imminent. In this way, patients deemed not suitable for an intermediate care or rehabilitation bed would often not be aware of the existence of these service. Moreover, for those who were waiting but a bed took longer than anticipated, the decision could sometimes be reviewed or changed without the knowledge of the patient and family.

6.9. Chapter summary

Ethical dimensions of decision-making emerged as clear and significant and manifested themselves in many ways. The understanding that an evaluation of rehabilitation potential related to maximising positive outcomes and minimising harm was evident but it was also evident that these principles were not mutually exclusive.
The analysis of fieldwork data pertaining to patient and family involvement highlighted many salient ethical issues. The concept of rehabilitation potential was not clearly explained to patients or families, presenting basic yet fundamental challenges to meaningful involvement. There were also critical challenges with meaningful and informed consent noted during observations and interviews such as consulting family members first, gaining agreement (rather than consent) after a decision was made, or attending to those who shouted loudest.

Practitioners communicated a range of emotional responses to the ethical dimensions of evaluating rehabilitation potential and making decisions about rehabilitation pathways – most notably, frustration in general, frustration directed at others and guilt. Observations also suggested different overt and covert strategies used by professionals to manage the ethical dimensions, professional responsibilities and emotional responses associated with rehabilitation decision-making.

CHAPTER SEVEN – RESEARCH FINDINGS: PROFESSIONAL ROLES

7.1. Chapter introduction

Although there were many different disciplines involved in evaluations of rehabilitation potential and recommendation of rehabilitation pathways, it was an objective of this study to particularly understand the role of occupational therapists and physiotherapists. Up until this point, whilst giving emphasis to occupational therapy and physiotherapy, many themes are developed across the two main professional roles and set against the backdrop of a multi-disciplinary social space.

Because of the stated commitment to understanding the occupational therapy and physiotherapy roles in depth, the final element of analysis emphasises issues affecting the two professions at the centre of this enquiry. The difference in professional roles is developed as an important theme, with instances used to develop issues of similarity and divergence.

Figure 11 provides the final example of how codes were developed and final themes and subthemes confirmed.

Early Coding:

All data was reviewed using the code 'professional roles'

Professional roles and involvement:

- The Role of the Physiotherapist
 - \circ As perceived by others
 - o Early involvement
 - Model of working
 - The allocated Physiotherapist
- Role of Occupational Therapist
 - \circ Later involvement
 - \circ Model of working
- Challenges for Physiotherapists
 Pressure
- Challenges for Occupational Therapists
 - Lack of understanding
 - Challenge to assert role
 - o The focus on mobility
- Traditional hierarchies
 - $\circ \, \text{Medicine}$
 - Physio
- Alliances
 - \circ Physio and medicine
 - \circ Physio and OT
 - Providers of rehab

- Other roles

- $\ensuremath{\circ}$ Troubleshooters and gatekeepers
- \circ Wider MDT

Examples from the data:

Professional roles and involvement

Perception of the physiotherapy role: Fieldnotes p100 "Physio's have main responsibility for decisions about rehab potential".

Early involvement: I3 "So we'll do chest physio, we would still sit them out".

Model of working: I1: "We're seeing people every day"

The **Physiotherapist:** Fieldnotes p142 "But the physio's who are here today are only covering".

Pressure felt by physiotherapists: Fieldnotes p196 'Clock is ticking'.

Role of Occupational Therapist/Model of working: 12 "I often don't get involved as much until we've almost reached the end of the rehab potential".

Lack of understanding about the Occupational Therapy role: I4 "just kind of refer to OT as...when people need equipment, or when they don't really know what people need"

Challenges for Occupational Therapists to assert their role: 15 "I don't think people look to the OT as much".

The focus on mobility: 15 "It seems like it's simmered down to this one thing, and the physiotherapists are the therapists that come to mobilise the patient'.

Hierarchies – I3 "They got overruled by the medical team"

Alliances – Fieldnotes p221 "One of the big things was getting the Consultant's agreement"

Troubleshooters – I2 "Challenged by one of the discharge team"

Final themes and Subthemes:

Theme: Professional roles

Sub-themes:

- The traditional hierarchy
- The Physiotherapy role in decision-making
- The Occupational Therapy role in decision-making
- Comparing the
 Physiotherapy and
 Occupational Therapy role

7.2. The traditional hierarchy

Before discussing and contrasting occupational therapy and physiotherapy roles, it is important to note that traditional hierarchies within healthcare decision-making were observed. The team on the base ward discussed that the consultant tended to have the final say about transfer to other wards, and bed managers discussed that it was a consultant responsibility to list patients for beds on other wards (despite in reality this happening through nursing staff or ward clerks). When discussing the case noted on page 159, the base ward physiotherapist reflected that 'one of the big things was getting the consultant's agreement' which suggested that this physiotherapist recognised their own opinion would be valued, but only if in accordance with senior medical staff.

Another example to illustrate this was noted during an interaction on the base ward between Patient 4, two family members, the occupational therapist and the social worker. Following the interaction, I discussed with the family members whether they understood what was likely to happen next, with the response 'we will check with the doctor before we leave tonight'. Although the occupational therapist had discussed plans for discharge and a care package at home, the family perceived the doctor to be the decision-maker in relation to these issues.

7.3. The physiotherapy role

The role of the physiotherapist presented as being significant to the evaluation of rehabilitation potential. In relation to the evaluation of rehabilitation potential, different team members placed importance on the assessment and opinion of the physiotherapist. During a discussion with one of the doctors on the base ward when explaining the focus of the research on understanding decisions about rehabilitation

potential he shared that 'physio's have the main influence and responsibility...if they think they can improve [a patient's mobility]'. Another example emerged during an MDT meeting on the base ward, where the base ward physiotherapist was not present and nursing staff raised in relation to a particular patient that they needed to 'speak to physio's to find out if he has potential'.

Physiotherapists presented as being aware of the value that the wider MDT placed on their role and their professional judgement, illustrated in interview discussions:

"I do think we do [have a key role in decision making] which I think is really good...as I keep saying, it is an MDT approach but I think, the focus is often on how are they transferring, how are they mobilising, so then that immediately then comes down to us" (I1, Line 459).

The significance that the team placed on the physiotherapist's evaluation of rehabilitation potential was then also seen to carry through to the significance the team placed on their recommendation about an appropriate pathway. During an MDT meeting on the base ward (the same meeting above where the physiotherapist was not present) the team discussed whether a particular patient would go home from the ward or go to the rehabilitation ward. The consultant concluded 'see what physio make of him during this week'. Another example during a base ward MDT meeting involved a direct discussion between the consultant and physiotherapist with the consultant summarising at the end of a discussion 'we need to liaise with each other...let us know if you think [the intermediate care unit]. In a further example, the occupational therapist asked 'Shall we list her now [for the rehabilitation unit]?', with the consultant's response 'No...wait until physio see her', suggesting that the opinion of the physiotherapist was valued above others.

7.3.1. Perceptions of why the role was significant

Physiotherapists in this setting recognised that the public awareness of physiotherapy, and the construction of physiotherapy as the main profession who focus on mobility contributed to the role being seen as integral to rehabilitation and rehabilitation decision-making. However, the physiotherapists discussed that there were other influences on why their role was integral. Within this, they discussed the likelihood of physiotherapy knowing most, if not all, patients (related to blanket referrals) and the likelihood of involvement at an early stage due to their role in acute and medical interventions. This was also noted during fieldwork observations; that physiotherapists contributed to MDT daily handover and weekly meetings with a range of information about different phases of care and with more frequency than occupational therapy.

"And...it's not until people come in to hospital and see the wider aspects of physio you know, you think about our other roles, putting NIVs on people, doing chest physio all that sort of stuff....So probably that's why they come to us to ask...and just going back to that respect of the professions, when people realise that we have got skills in other areas...that we have a good previous medical history, that we understand what they are socially from...a rehab perspective...And because we've got that underpinning knowledge I think that does add to why people think, oh well actually they kind of know what they're talking about" (I3, Line 623).

"I suppose because we're just a key part of the MDT really aren't we. And we're seeing people...I think if you were to compare us to occupational therapy, we're seeing people every day. And so we get to know people really well. So we're seeing them from the day they come in to the ward and we're seeing them...sometimes more than once a day, several times a day" (I1, Line 466).

7.3.2. Challenges faced by physiotherapists

Although interventions to promote improvements with mobility are clearly within the

scope of practice of physiotherapy, the physiotherapists themselves expressed

reflections that this in itself was a narrow representation of their role:

"...the wider team just say physio, yeah, they mobilise people" (I3, Line 43).

And one physiotherapist suggested that they were aware of, and indeed frustrated by this reduced focus, and were trying to attempt to influence this narrative:

"I think mobility is probably the bulk of our work but I'm trying to push away from that...trying to think more functional...I try to deter people from just saying we mobilise people. We should be actually saying we functionally rehabilitate people" (I3, Line 27).

Physiotherapists also shared insights that this integral role in decision-making can lead

to them experiencing pressure. During observations, physiotherapists discussed that

other team members would look to them to solve problems that they could potentially

solve themselves and on a couple of occasions, used expressions such as 'the clock's

ticking' to describe how they felt when the wider team were awaiting positive outcomes

from a rehabilitative process.

The further extract below highlights the pressure experienced by physiotherapists

linked to expectations of family members:

"And [a family member] was absolutely livid because he said 'I was told that my mum was coming to this [rehab] unit to get 2 hours of physiotherapy a day...and what's happening...she's not getting anywhere near this" (I1, Line 496).

7.4. The occupational therapy role

7.4.1. Focus on mobility

Occupational therapists did recognise the dominant focus on mobility within the rehabilitative process and how this influenced their own role. In part, they rationalised this through the recognition that mobility was a foundation to improvements and safety in other functional areas (such as dressing and toileting). However, occupational therapists did express finding this particularly challenging:

"Particularly as we're OTs it should be around all areas of life but it just doesn't seem to be that way in hospital, it's whether people can get up and walk or not" (I5, Line 44).

7.4.2. Role in decision-making

In contrast to physiotherapy, fieldwork observations suggested that occupational therapists were less involved in evaluations of rehabilitation potential and less involved in the decisions about pathways. Similar examples to those above where the physiotherapy opinion was sought to support decisions, were notably absent in relation to occupational therapy. In MDT meetings on the base ward, the occupational therapist was observed to provide feedback on issues such as equipment needs or environmental considerations, although presented as having limited involvement in evaluations of likely improvements, or appropriate pathways. Indeed, it was observed that on some occasions, important decisions were made before occupational therapy became involved with one nurse expressing during an MDT meeting that '[a patient is] not going to rehab. Needs OT involvement and a big package of care'.

Occupational therapists were aware of such issues, particularly aware of the impact of their model of working and the limited involvement at an early stage on their involvement in rehabilitation decision-making:

"I think it's probably because [the physiotherapists] do have the time to see every patient every day and I don't. So, they're going to have more understanding of what they...you know, they're going to have more an opinion than myself that's never met someone" (I4, Line 264).

Indeed, perhaps because the occupational therapist had more limited involvement with some patients, there were some understandable examples where they relied on an alliance with the physiotherapist for information or judgements. It was not uncommon for the occupational therapist on the base ward to ask the physiotherapist whether identified patients needed occupational therapy involvement. Also, during a morning handover meeting when there was no physiotherapist present, the occupational therapist provided verbal feedback that 'physio feels may be plateauing', and the action agreed was that the occupational therapist would check back with the physiotherapist for further information.

7.4.3. Respect for the occupational therapy role

Interview extracts from occupational therapists indicated that there was a perceived difference between how they understood the importance of their role in relation to rehabilitation and subsequent decision-making and how the role was perceived by others:

"With the occupational therapy role it's not as kind of, not respect...I wouldn't use the word respect...well I guess respected as a physiotherapy input in the MDT. I think sometimes people look to the physios more than they would to the OT in the first instance and what we're working on" (I4, Line 46).

"we are very much a key player but I don't think people recognise that. Because I've been told that people...you know...they don't have rehab potential and they just need this piece of equipment. And I think, some of that's possibly down to people not fully understanding what our role is as well" (I2, Line 643).

Many of these issues presented as being frustrating for occupational therapists, and this frustration and potential need for ongoing assertion of their value and role was communicated through the language used in discussion. One occupational therapist reflected that people didn't look to them as much for their opinion and that it is her job to "pipe up" with important information. Another occupational therapist discussed being "adamant" that a person would achieve the functional gains she was working towards. Indeed, language almost became suggestive that the ongoing need to assert their value and role sometimes led to positions of confrontation:

"Well generally if somebody has rehab potential, I would usually be trying to fight to keep them where they are at" (I2, Line 482).

7.5. Comparing the physiotherapy and occupational therapy role in decision-making

An initial area of comparison has already been highlighted, in the sense that physiotherapists tended to become involved with patients at an earlier stage of an admission, and they tended to see patients more frequently than their occupational therapy counterparts. Physiotherapists did not simply get involved earlier and more frequently because they had the capacity to do this or due to a different pattern of working, but that their involvement linked to the treatment of acute medical issues during acute stages. This involvement then translated to knowledge about the majority of patients on the ward at any given time and potentially to professional respect within the wider team.

An occupational therapy participant discussed awareness that medical knowledge and involvement in medical treatment could also account for the significance placed on physiotherapy opinions and decisions, but instead of relating this to professional respect, suggested more candidly that this translated to a perception of power and was rooted in different philosophical foundations:

"I think physios have more power but I think it goes back to the medical model, social model. Because I think we ascribe more to the social model rather than the medical model, although I would say we have a foot in both camps" (I2, Line 697).

An extract from another interview reflected that, instead of simply assuming that the dominant focus on mobility contributed to the different value placed on occupational therapy and physiotherapy professional opinions, it could indeed be the way occupational therapists work that may have contributed to this dominant focus on mobility and therefore they themselves have inadvertently contributed to hierarchies within the team:

"It seems like it's just simmered down to this one thing [mobility], and the physiotherapists are the therapists that come in every day to mobilise with the patient, they are the ones who are looked at to see if they have any potential. Some OTs might see patients when they hit the ward, I don't do that, I'll see them a little bit after and then I'll speak to the physio about rehab potential and get their insight in to it. So maybe we've created it a bit...I don't know..." (I5, Line 316).

Interestingly, as decision-making reached more progressed stages – particularly in relation to decision-making about pathways for discharge home or alternative care environments – the influence of physiotherapy was observed to change and become less significant. It was also observed that increased responsibility was assumed by the occupational therapist in some examples. In relation to Patient 5, the occupational therapist was noted to attend a planning meeting although the physiotherapist did not. During a subsequent discussion, the physiotherapist reflected 'I didn't go to the planning meeting yesterday because it wasn't particularly physio...we haven't got the staff to go in to every single planning meeting'. But from this, when asked whether physio staff had had input in to the decision about referral to the reablement team, the physiotherapist indicated 'I haven't actually...no...because I haven't really had a chance. But often [the rehabilitation ward OT] will come and say...do you think it would be a good idea if reablement went out and did outdoor mobility and things like that'.

In the case of Patient 2, during a discussion on the rehabilitation ward (approximately 4 weeks after the original acute admission, and approximately 3 weeks after the physiotherapist on the rehabilitation ward first assessed this patient), the physiotherapist reflected that the patient was not making any progress and it was felt the patient did not have further rehabilitation potential. When asked if the physiotherapist had talked to the family about this, the response was 'No...at this stage I tend to leave that to nursing or social work although I would certainly speak to the family if they asked to'.

The occupational therapist on the base ward demonstrated insight in to the fact that the physiotherapist's assessment was significant to the evaluation of rehabilitation

potential, and perhaps more significant than their own evaluation. This was illustrated during a discussion about Patient 5: 'I just didn't think there would be that much potential for him to get better. But obviously, I think [name of physiotherapist] had seen him more than me and felt like could get a little bit better'.

Language utilised to reflect on their own roles was notably different between occupational therapists and physiotherapists. It has been highlighted above that language used by occupational therapists was indicative of challenges to voicing and asserting their role and value, and on occasion this indicated confrontation. In contrast, observations of, and interviews with, physiotherapists suggested a confidence in their professional role. This was captured during one of the interviews:

"9 times out of 10, in my thinking about sending someone ongoing to somebody else to rehabilitate them then...I would be quite confident in what I was sending them for, and sending them to" (I3, Line 502).

Interestingly, physiotherapists also communicated a need to assert the occupational therapy role and shared attempts to do this on their behalf. One physiotherapist discussed trying to get away from 'rehab being just about physio...it's also about OT'.

Despite the differences, it was evident within fieldwork observations and through interviews that there was an allegiance between occupational therapy and physiotherapy. If one was unable to attend a handover or MDT meeting, there were instances where the other would feedback on their behalf or take responsibility to share information afterwards. Similarly, if a different therapist was providing cover (most often due to leave), they would seek out the ward-based occupational therapist or physiotherapist to support their assessments. There were also observed examples where occupational therapists and physiotherapists conducted their own informal handovers, sometimes after the more formal morning handover had finished, which presented as serving the purpose of consolidating information and prioritising daily tasks between the professionals. This observation that occupational therapy and physiotherapy professionals recognised similarities and shared ways of working was expressed during interviews:

"Between things like occupational therapists and physiotherapists. I think we work along very similar lines" (I1, Line 210).

7.6. Chapter summary

This section illustrates the significance of the physiotherapy role in evaluating rehabilitation potential, in making decisions about rehabilitation pathways, and in the general contribution to, and value within, the acute healthcare team in this context. This was observed and discussed as being influenced by the dominant rehabilitation objective of improving patient mobility (traditionally a central domain of physiotherapists), alongside physiotherapy involvement in acute medical management, and the approach to involvement with every patient.

In contrast, occupational therapists had a less evident role in evaluations of rehabilitation potential and pathway decisions, which they, and others, attributed to not being involved with all patients and generally being involved at later stages. Occupational therapists also recognised that their own practice was being shaped by the dominant focus on mobility in this context. Both of these elements – the value placed on their role and the way the role was being shaped by context – presented as increasing feelings of frustration and challenge for those involved.

CHAPTER EIGHT – DISCUSSION

8.1. Chapter introduction

The research findings presented in chapters four-seven provide an early layer of interpretation, in that I have chosen instances of field observations, vignettes and verbatim extracts to illustrate themes and subthemes. Evident from the presentation of findings are over-arching hypotheses that now become the focus of further analytical discussion.

Firstly, the meaning of rehabilitation in acute care is complex and highly influenced by context, yet it takes on a simplified and reductionist form which often means it gives precedence to organisational rather than patient-centred objectives and is often significantly different to ideals. Secondly, the meaning of rehabilitation potential is ambiguous, not explained to patients, again takes on a reductionist form, but alongside this the concept is evaluated subjectively and becomes synonymous with judgements about movements within the healthcare system. Thirdly, in part because of this overlap between an evaluation of rehabilitation potential carries multi-faceted ethical dimensions about access to, or withholding of, services which contribute to ethical distress for those making decisions and to real-world tensions when working towards sound ethical practice for older people. Finally, physiotherapists are key protagonists when evaluation potential and reaching decisions about rehabilitation pathways which promotes value and esteem for this discipline, although at the same time can be a source of challenge for occupational therapists.

This chapter will firstly present the wider social and cultural context which acts as a backdrop to meaning-making and decision-making for those within the field. According to Bourdieu, to explain any social event or pattern, a writer must closely examine the

social constitution of the agents and the makeup of the social universe, including all of the conditions in which they operate (1989; cited in Wacquant, 1998). The chapter will then focus on the four central tenets outlined above to develop further and more critical layers of reflexive interpretation, drawing on multiple sources of knowledge to weave a more holistic web of cultural structures, knowledge and meanings which, superimposed on to one another, form a deeply layered cultural and social script (Seale, 2018).

8.2. The 'macro' influences on meanings and social constructions within the acute hospital context

The methodology chapter highlighted that a distinguishing feature of ethnography is the way in which it makes links between daily occurrences and wider social, cultural and political influences (Savage, 2006). An aim of this study was to understand decision-making in relation to rehabilitation potential and rehabilitation pathways and therefore integral to this sense-making is critical analysis of the wider social and political background against which these daily decisions are made.

Although many models of rehabilitation continue to rapidly develop, and often with an emphasis on specialisms, alongside community, long-term, and closer to home models - a multi-faceted, multi-professional rehabilitation phase of care in the context of acute hospital admissions continues to be recognised in current policy and guidance (NHS England, 2016a).

However, whilst rehabilitation based in acute hospitals continues to be recognised, it is important to re-examine the surrounding context in relation to acute admissions for older people. Average length of stay for older people following an emergency admission has decreased in recent years from 12.9 days (in 2010-11) to 11.9 days (in 2014-15), although this has happened at the same time as a growth in the number of

older people experiencing an emergency admission to hospital (increasing by 18% between 2010/11-2014/15) and overall bed days from such admissions increasing by 9% in the same period (Department of Health, 2016). Simply put, there are more of this population in hospital, but their individual length of stay is shorter. Simultaneously, and significantly, there has been a well-communicated projected 'funding gap' and necessary efficiency savings (NHS England, 2014b). Services which require time and interventions of high frequency, whilst experiencing high demand, will inevitably be under pressure.

Acute hospitals are increasingly organised around common clinical conditions and anatomical systems and this was clearly evident within this site. Stroke units, coronary care units, and dedicated orthopaedic wards are all common specialisms within acute hospitals, with larger sites often hosting other regional specialisms for areas such as oncology, major trauma and paediatrics. The alignment with a biomedical model of health is recognised as an overriding framework – that is to say, systems and pathways organised around a clinical condition (such as a fractured neck of femur) or an anatomical system (such as cardiology). This overriding paradigm of a biomedical model of health at a macro level is important to note and is another element of the backdrop against which professional thinking and decision-making was set.

It is important to acknowledge that the organisation of healthcare systems and specialisms aligned with clinical conditions is based on sound evidence. Where teams and services orientate environments and expertise to common presenting needs, and the more experienced they become in managing these needs, evidence supports better outcomes for patients (NICE, 2016b; Stroke Unit Trialists' Collaboration, 2017; Smith et al, 2015). However, it is also here where a critical paradox emerges for those with a non-uniform presentation of disability not easily categorised by one condition – notably the older and frail population who were the main users of the base ward at the centre of this study. Despite being the largest users of health and care services, a picture begins

to emerge of a system which is orientated away from their needs and struggles to identify the most appropriate services and locations (Bail and Grealish, 2016). Examples from fieldwork where patients were moved within the system because of a perception that services did not know what to do with them, or because they required a general or vague 'sort out' are indicative of this challenge.

It is perhaps influenced by this trend of specialisms attracting resources and profile that rehabilitation itself has also attempted to become a specialism. Specialist societies of rehabilitation medicine such as the British Society of Rehabilitation Medicine (BSRM, 2017), scholarly journals and specialist commissioning guidance (NHS England, 2016a) all allude to this specialism. The organisation of the system during the first period of fieldwork, with a separate ward for rehabilitation was reflective of this. However, Wade (2016) highlights a critical challenge that has emerged from this – that rehabilitation as a separate specialism carries an implicit assumption that patients receive rehabilitation or medical care, with the manifestation of this being that patients physically move locations between medical services and rehabilitation services. This was particularly evident within the study context where rehabilitation was associated with place, and that a move to this place was associated with the commencement of a rehabilitative phase of care. Wade goes on to suggest that this intrinsically carries risks - such as rehabilitation or medical care being absent of lesser priority when the patient is in the boundaries of the alternative service; it can lead to disagreements over responsibility; and lead to reduced efficiency due to duplication or gaps in interventions or communication (2016).

It was not only in the organisation of the hospital system where a biomedical model of health was evident. Notably, the focus on rehabilitation to bring about improvements in physical abilities, and more specifically mobility, emerged as a significant way in which rehabilitation was constructed. Social theorists have long recognised: the value of bestowing social power according to body and physique; that the body is a marker of

social class and; bodily function signifies worth (Hammell, 2006; Bourdieu, 1984). Alongside this, a historical perspective of physical rehabilitation supports an emergence from medicine and surgery at the time of World War 1 (Linker, 2016), and the quote from a phase 3 interview suggesting rehabilitation is perceived as a 'getting soldiers to walk again type of place' emphasises that this historical perspective still rings true.

8.3. The meaning of rehabilitation

8.3.1. Revisiting definitions of rehabilitation

As outlined in the background to this project, definitions of rehabilitation are wide and varied, particularly when the word is attached to particular contexts or specialisms. Extracts presented in Table 11 highlight the understandings from study participants of what rehabilitation is and what it is not, and enable comparison to both lay and professional definitions. Notably, what rehabilitation means to professionals in this setting bears close resemblance to lay definitions, although extracts from discussions about what they would like rehabilitation to be suggest aspirations towards a more multi-dimensional version, more reflective of professional definitions and ideologies. Findings also indicate other layers to the meaning of rehabilitation which are not necessarily verbalised in participant definitions, but instead are communicated in the versions of rehabilitation which are enacted in practice.

Table 11: Definitions of rehabilitation

Examples from study participants – what rehabilitation is	Examples from study participants – what their version of rehabilitation is not	Example lay definitions	Example professional definitions
"getting back something that they've lost" "getting them back to that level of function that they were previously at"	"I think we all want this text book thingyou know, goal-led, get the carers, get the family members, get the patient, do it all together. But it's not always as easy as that" "the way I look at rehabilitation is having time to work with somethingsomebodyon some meaningful activity. Say it'swashing and dressing for example, if someone can't manage that. It's working with them daily, having daily input with that person, practicing tasks, so that they canpossibly become independent or improve in their independence in that area that we're looking at. But that relies on having the time to do that"	The action of restoring something that has been damaged to its former condition (Oxford English Dictionary, 2017) The process of returning something to a good condition (Cambridge Dictionary, 2017)	A health strategy that aims to enable people with health conditions experiencing or likely to experience disability to achieve optimal functioning in interaction with the environment. (WHO, 2001). A complex process which enables individuals after impairment by illness or injury to regain as far as possible control over their own lives (Kings Fund, 2001).

8.3.2. A version of rehabilitation that values (physical) improvement

The extracts in Table 11 suggest that, at a foundation level, professional's associate rehabilitation with facilitating improvement. Furthermore, findings highlight that in this context, rather than rehabilitation being a 'black-box' with multiple ingredients to bring

about physical, social, emotional and environmental improvements, more realistically rehabilitation was often characterised by a one-dimensional ingredient of improving mobility. The influence of social theory discussed above which places value on physical strength, and of a biomedical model associated with treating a bodily impairment to restore it to a former condition, are significant. Professional perspectives (represented strongly within data) and patient perspectives (represented less strongly but with important insight) both conceptualised rehabilitation as a process to improve bodily strength and function.

Hammell (2006) recognises that the aim to enhance physical function has become a preoccupation for rehabilitation professionals although suggests this is inadequate for people with deteriorating or chronic conditions where remediation of physical function may be an insufficient or unrealistic goal. Furthermore, research suggests that older adults themselves feel that rehabilitation in acute settings is not meeting their holistic needs (Atwal et al, 2007). In this setting, professionals recognised that they had limited time to address wider functional, psychological, and social needs and were being driven to focus on needs that would quickly influence safety for discharge. Attempts to meet wider occupational needs ('can we not just get a carer [for bed transfers]' – page 143) or social and psychological needs ('nice to do...but not realistic' – page 139) were out of alignment with the way of working.

8.3.3. A version of rehabilitation that values flow and efficiency

A key implication of much of the background context about hospital admissions and length of stay is that, for practitioners delivering care and treatment in the acute context, this arguably contributed to a value on fast-paced trajectories maintaining efficient flow within the system. This view is suggested within the research findings and reported in other published studies (Bail & Grealish, 2016). The example on page 159 where the physiotherapist reflected a change in reasoning to prioritise those they could move on more quickly, rather than those who required more intensive therapy illuminated this issue. Indeed, to develop this further, authors have suggested that is it not only the progressive changes to demographics and usage, but also the application of neo-liberal principles to health care systems, where the focus has shifted from quality of care towards attaining performance standards in relation to risks and efficiencies, which is directly influencing the thinking and reasoning of health professionals (Durocher et al, 2016). The emergence of a tension between rehabilitation as a process which requires time for collaboration and learning to optimise improvements, and rehabilitation as part of an acute care system is obvious.

Alongside frequent discussions about time pressures, professionals were also aware of the significance of the measurement of length of stay, with 'length of stay meetings' used to publicly report this outcome, and examples where a large set of notes (page 159) was seen as a negative indicator of a patient's progress or the ward being reflected in a negative light.

8.3.4. A version of rehabilitation that prioritises optimum safety rather than optimum function

The threshold aim for rehabilitation for the institution was becoming safety for discharge, rather than optimum improvement towards baseline function (which had been part of the narrative during phases 1 and 2 of fieldwork). This was communicated strongly during phase 3 interviews in relation to how the organisational hierarchy wanted to discourage use of the word 'baseline' (which had been observed to be an embedded part of a shared language), and therefore discourage professionals from using a patient's previous baseline as the aim of any rehabilitative interventions. This not only meant that optimum safety superseded optimum function as the primary aim of

rehabilitative interventions, but also meant goals of aiming to address the broader fundamental principles of rehabilitation of increasing physical, social, psychological and emotional wellbeing implied in their own discussions (Table 11) and professional frameworks (WHO, 2001; WHO, 2017) were becoming further removed from real-world practice.

This challenge of not being able to focus on wider facets of rehabilitation was noticeable during observations (for example professionals felt that they could not give priority to interventions which focussed on improving mood or social interaction), and the narrative shared during phase 3 interviews suggested this had only grown in significance. Interestingly, this is perhaps another example of the influence of the biomedical model of illness, where body and mind are considered separately (Wade and Halligan, 2017).

Professionals recognised that optimum safety in wider activities of daily living is often linked to improvements in mobility (page 138) and therefore the value placed on physical improvement and working towards safety could be interpreted as part of the same overall values of rehabilitation. However safety can often be achieved earlier in a rehabilitation process than optimum function (through strategies such as environmental modification and carer support) and therefore this focus on safety was assisting the organisation to shorten length of stay and free up beds and resources to meet increasing demands. Statistics presented at the beginning of this chapter illustrating shorter hospital admissions despite increasing demand (Department of Health, 2016), and examples which suggest inpatient rehabilitation is being reduced to a process of 'destination triage' (Durocher, Gibson and Rappolt, 2016), suggest that this is a reality not just in this local context, but at a national and international level.

An additional area of challenge was that to facilitate discharge at an earlier stage and a lower functional level, professionals needed confidence in community services to

address ongoing rehabilitation needs, and this was something that was of real concern (pages 143; 150).

8.3.5. A version of rehabilitation happening in a separate place

The influence of rehabilitation being conceptualised as a separate specialism could also be understood to be shaping the construct of rehabilitation in this acute hospital setting. The strong narrative of rehabilitation as a place and as happening outside of the base ward was illuminated through direct fieldwork experiences and participant quotes. This had far-reaching implications for both professionals and patients: professionals discussed that patients could be 'bounced back' to the base ward (often from the intermediate care unit) if medical needs were deemed to be difficult to manage; professional assessments recommenced with different professionals on arrival to the rehabilitation ward often resulting in duplication; and patients could become less of a priority for physiotherapy and occupational therapy whilst on the base ward because they were waiting for a transfer to the rehabilitation ward.

Another implication for professionals was that, if rehabilitation was seen as a separate specialism existing in a separate location, then it was therefore not being provided on the base ward. The occupational therapists and physiotherapists delivering services on the base ward - professionals, who traditionally are accepted as core providers of rehabilitation and associate their professional identity with this (Colquhoun, et al 2016; Atwal et al, 2007) - indicated discomfort and tension with this. This was yet another source of the frustrations and ethical challenge facing these professionals.

It is therefore an interesting point to note that the rehabilitation ward was renamed and reconfigured during the period between phase 2 and 3, with professionals discussing in interviews that part of the rationale for this was the integration and co-location of a

rehabilitative phase of care with medical care and management. This notion is supported by Wade (2016) who presented risks and critique of rehabilitation claiming specialisation in its own right. And although in relation to the specialism of stroke, the co-location of acute and rehabilitation services was found to be equally as effective in improving functional outcomes as traditionally separately located services and more efficient in terms of hospital bed utilisation by reducing some of the challenges and duplications associated with waiting times and internal transfers (Chan et al, 2014).

Although the local site within this study was reportedly moving towards this model for older people's services, a separate off-site rehabilitation unit (which although not likefor-like, was a replacement of the previous intermediate care unit) was still in existence which meant risks and problems associated with rehabilitation in a separate location could also still present potential issues. It was beyond the parameters of this current study to explore these changes in more detail.

8.3.6. A perceived gap between ideal and real-world versions of rehabilitation

With pressures on the front end of the system, alongside less beds for the purpose of rehabilitation, and coupled with an organisation influencing staff to prioritise transfers of care and discharge, it is not surprising that professionals described a gap between ideals of rehabilitation and their perceptions of the realities of service provision. Importantly, the value on fast-paced trajectories and prioritising safety led to compromises in terms of time available, consideration of holistic needs, and meaningful patient involvement – all seen as desirable and best practice features of rehabilitation, but expressed as being impossible to deliver in reality.

A multi-professional approach is highlighted in literature and policy (Wade, 2015; Wade 2005; Whyte, 2009; WHO, 2017) as a best practice feature of rehabilitation and one where the realities of service provision may have also been perceived to be falling short. Although a multi-disciplinary approach was evident (in the sense that doctors, nurses, therapists and social workers, among others, all contributed to rehabilitation and decision-making at different times), a uni-disciplinary paradigm was implied in that the physiotherapy role was implicitly and sometimes explicitly acknowledged as the main protagonist in a phase of care associated with rehabilitation. Equally, the lack of multi-professional representation within the community reablement team resulted in a struggle for professionals to fully conceptualise this team as a provider of rehabilitation.

Because of the many layers of disparity between ideals and realities, links can be made between concepts which acknowledge disconnection between workers and the outputs of their labour: for example the Marxist theory of labour alienation (Marx, 1844; cited in Durocher et al, 2016) which purports that individuals may experience alienation if their employment does not afford the control, decision-making or creativity they desire from the role. Polatajko et al (2007) echo such sentiments in more recent work by discussing that occupational alienation may occur if external forces result in occupational roles that no longer fit the individual's potential or aspirations. An increasing awareness of systemic external pressures in relation to resources, coupled with discussions which implied a difference between a desired model of rehabilitation and the reality of what could be provided in practice were clearly observed within this research and highlight a real and ongoing challenge facing a rehabilitation workforce in the context of acute care.

To summarise this section, Figure 12 presents the main ways in which the concept of rehabilitation was observed and interpreted from the research findings. Although this representation does not perhaps do justice to some of the complexities relating to, and emerging from this issue, it does provide an at-a-glance insight in to some of the main

challenges to patients, families and professionals of such conceptualisations and provides insight in to a contemporary and real rehabilitation taxonomy present within discourse and practice.

REHABILITATION AS			
Example	Main Influences	Main Risks	
IMPROVEMENT	Desire for positive change	May not be realistic for frail older people	
PHYSICAL IMPROVEMENT	Physiotherapist as main protagonist Value on bodily strength and function	May not be realistic for frail older people May not attend to wider aspects of emotional.	
		social, spiritual functioning	
PHYSIOTHERAPY	Physiotherapist as main protagonist Later and less frequent involvement of Occupational Therapy (and potentially other rehabilitation professionals)	High expectations of (and pressure on) the physiotherapist from patients, families and other professionals Low involvement/value of other professionals	
PLACE	Association that different wards and units would have more time/staff/facilities	High expectations of the identified place from patients, families and professionals Provision of rehabilitation equated with a desirable move within a system	
TIME	Gold standard of rehabilitation associated with time intensive activities (such as goal setting, family involvement and opportunities for learning)	Incongruent with time pressures of acute care	
NOT HAPPENING/IDEAL	Many discrepancies between ideals and realities of rehabilitation	Not meeting expectations (leading to dissatisfaction, distress, alienation) of patients, families and professionals	

Figure 12: A taxonomy for rehabilitation in the acute hospital context

8.4. The meaning of rehabilitation potential

8.4.1. The ambiguous nature of rehabilitation potential

The initial review of literature, mainly related to the concept of rehabilitation potential following stroke, highlighted rehabilitation potential as an ambiguous concept and this lack of clarity has been extended within this current study to judgements relating to rehabilitation potential of older people following an acute hospital admission. Even if the findings simply hold a mirror to this issue for professionals who regularly use the term in their professional practice and encourages a more detailed articulation of meaning, this is an important development. The simple yet critical question of 'potential for what?' highlighted by one professional in the field crystallised some of the many challenges with this ambiguous term.

8.4.2 Potential to improve

The findings presented in section 6.3.4 highlight that, linked to the concept of rehabilitation being conceptualised as a process to facilitate improvement, it followed that rehabilitation potential was conceptualised as the potential to make improvements through engagement with a period of rehabilitative interventions. Once again, a narrow and reductionist worldview is evident which does not fully acknowledge rehabilitation as something which can optimise and maintain function through adaptation and learning. This narrative could contribute to practices where patients who are unlikely to improve physically, but could make quality of life gains through compensation, adaptation and learning, are deemed to have limited rehabilitation potential with rehabilitation

8.4.3. A blurring of rehabilitation potential and an internal transfer

Lane et al (2017) discuss that the concept of potential benefit is an important criterion when considering resource allocation and pathway decisions, something which was uncovered in this study. Such judgements reflect high-stakes outcomes for individual patients, leading to the allocation of a bed, or the access to (or withholding of) an intervention. Indeed, the National Audit Office's review of the National Stroke Strategy (2010) reported that 30% of stroke units excluded patients for a bed on the basis of 'no rehabilitation potential', a practice described as unacceptable.

This clearly emerged from the research findings; that a judgement about rehabilitation potential almost became synonymous with a judgement about the appropriateness for an internal transfer of care. That is to say, a positive evaluation of rehabilitation potential often accompanied a judgement about transfer to, or maintenance within, a bed in a rehabilitation environment. And conversely, a negative evaluation of rehabilitation potential often equated to judgements involved in ending the episode of inpatient or bed-based care or to question the appropriateness of onward referral to a rehabilitation potential with giving people a chance, and conversely the label of limited rehabilitation potential recognised as often linked to denying services and potentially doing harm (Enderby et al, 2017). It was perhaps here that the ambiguous, interpretive nature of the concept of evaluating rehabilitation potential was open to potential manoeuvring by professionals with subsequent high stakes decisions for patients and ethical tensions for professionals.

Whilst recognising that internal transfers are an important part of bed-based care in a modern hospital system (Bail and Grealish, 2016), it is the entwined nature of an internal transfer with a positive evaluation of rehabilitation potential which poses critical challenges. This was most acutely emphasised by staff on the rehabilitation ward and

intermediate care unit discussing that they often received people identified as having 'rehabilitation potential' although their own perception was that they received patients that other services 'don't know what to do with'.

This implied shifting of responsibility within hospital systems has been recognised in other studies (Dodier and Camus, 1997). One reason it was perhaps particularly pertinent in the environment at the centre of this this study was that individual wards were required to report their own length of stay. It was implied by health professionals that if a positive evaluation of rehabilitation potential increased the likelihood of a transfer to a rehabilitation bed, this positive evaluation also resulted in a positive outcome for the ward in terms of managing length of stay and promoting flow. Changes to reporting length of stay, such as evaluating a so-called 'superspell' – an overall length of stay across hospitals and other bed-based providers (Chalk & Pitt, 2015) - or more fundamental review of the necessity to report in this way at all, could reduce the desire to move patients who are viewed as problematic.

But it is the implications for the patient and family which are of equal, if not greater, significance. When communicated to patients and families (and this in itself cannot be assumed to have been routine practice as it was often not observed or not documented), transfers to the rehabilitation ward or intermediate care unit were discussed in relation to likelihood to make functional improvements ('to get stronger' – Patient 5). Professionals reflected that transfers to rehabilitation environments were 'sold' to patients and families in terms of time intensive interventions and better facilities (page 147), and yet resources suggested this would be an unlikely reality. The more realistic motivations of moving patients to a less acute bed to afford time to resolve wider issues of risk and complexity were not observed as being articulated to patients in a transparent way. Critical voices have suggested that this represents an allegiance with employers rather than patients and being resource- and target-driven rather than needs led (Hammell, 2007).

Current thinking suggests that if acute and rehabilitation services are co-located and a rehabilitation phase begins on day 1 alongside any curative and supportive phases of care, then the judgement about potential becomes less significant (Chan et al, 2014) and in this case, less tethered to some of the wider influences on reasoning which are linked to a transfer of care. This principle could be recognised in some of the strategic decisions which took place in this study site, with professionals recognising that the change of function of the rehabilitation ward was in part to negate the need for unnecessary hospital transfers and that instead rehabilitation would be co-located with acute care in existing acute wards.

8.4.4. The subjective nature of judgements about rehabilitation potential

Findings that an evaluation of rehabilitation potential is influenced by an interplay of patient-related and system-related factors which are context-specific and open to subjective interpretation is not surprising and this study focussing on a heterogeneous population of older people supports themes which are reflective of other acute specialisms. Important patient-related influences on reasoning in this context included previous level of (or baseline) function, presence of co-morbidities, and cognitive function. In terms of organisational factors, the availability of rehabilitation beds and services also influenced practitioner reasoning about rehabilitation potential and pathways. The fact that baseline information was not always clearly communicated or documented (illustrated in Table 7), and that different professionals interpreted baseline differently (example on page 166) suggests that an evaluation of rehabilitation potential reliant on this information is fundamentally compromised, and is one of many areas where subjectivity begins to emerge.

In their study to explore dimensions of evaluating rehabilitation potential amongst professionals working in stroke rehabilitation, Burton et al (2015) suggest that

assessment tools are deemed insensitive to rehabilitation potential, which perhaps helps to explain why assessment tools or structured criteria were not used to assist in the evaluation of potential in this current study. However, this practice is potentially perpetuating subjectivity, contributing to tensions and leaving such important judgements open to the influence of a wide range of clinical and non-clinical 'noise' (Enderby et al, 2017).

With professionals attempting to assess and evaluate something which is inherently subjective and imprecise, an awareness that this can facilitate or withhold access to important services which could hold the key to ongoing improvements, and an understanding of cost and resource implications scrutinised by public reporting, it is unsurprising that professionals experienced pressure. It could be hypothesised that the judgement itself could be influenced by the extent of pressure professionals were feeling at different times and from different directions, introducing another area of subjectivity. Pressure from vocal families, pressures during high demand for beds, or pressures due to low staffing were all examples noted during fieldwork. Burton et al (2015) discuss that judgements about rehabilitation potential carry an emotional labour and require emotional resilience which leads to the next section of this discussion about the ethical realities of evaluating rehabilitation potential and the strategies professionals used to meet competing demands of their role.

8.5. The ethical realities of rehabilitation decision-making

Patient related factors such as age, comorbidities, cognitive ability and pre-and postadmission functional status were all theorised within the literature review as influencing practitioner reasoning. However, broader ethical dimensions were found to be influencing and permeating practitioner reasoning. This emerged as an important and less anticipated theme and will therefore receive emphasis within this next section of interpretive and critical discussion.

8.5.1. Real-world application of ethical principles and frameworks

Discussion of deontological and utilitarian ethical frameworks were developed in the literature review (section 2.4.1), with the former being driven by the intention to do good for all, and the latter aiming to produce the greatest good for the greatest number. The desire that everyone deserved a chance at rehabilitation was a frequently shared view, suggesting that the desire to do good for all was an overriding value. However, ways of working often were driven by utilitarian principles – for example not listing people for weekend therapy because of an awareness of others who were more in need and occupational therapy not accepting referrals for patients already residing in residential and nursing care because of higher priority patients and perceived limitations in what could be achieved.

Because resources are increasingly prominent within public discourse in relation to health services, it follows that health professionals are therefore also increasingly aware of this and feel obligated towards a more utilitarian way of working (Garbutt & Davies, 2011). This push and pull between deontological and utilitarian principles is interesting although not remarkable, and indeed has been commented on by other authors (Levack, 2009). Insights shared by professionals within this study that it is the resources 'that win', and 'it all comes down to money and resources' suggests utilitarian principles are dominating this debate on the ground. This is echoed in other studies where non-clinical factors such as resources and time, as opposed to clinical factors based on need, are more dominant in professional decision-making (Kimmel et al, 2017). It is likely that this once more places pressure on professionals, potentially asking them to put aside their commitments to individual patients and their duty of care,

an issue strongly suggested by professionals (page 192) and supported by wider research (Blackmer, 2000).

When considering rehabilitation as a wider concept, it could be proposed that utilitarian principles influence why rehabilitation may attract a relatively low profile and fewer resources. That is to say, evaluation at a macro-level may suggest that the benefits are not justified in terms of the outcomes achieved or the numbers they are achieved for. This is perhaps particularly significant for rehabilitation of an older, frail population where the evidence is not strong and any perceived pay-off is unlikely to reap long term benefits when set against the potential assumptions about remaining expected life, or quality of life (Wade, 2015; Levack, 2009). Also, because these benefits may not obviously accrue to health services (Wade, 2015), due in part to a lack of integration in health and social care, there may be a lack of motivation from decision-makers to invest in rehabilitation for older people. Again, although some of this wider macro context was beyond the reach of this project, it is unavoidable to ignore the reduction of rehabilitation beds during the period of fieldwork and how such organisational influences may weigh on the reasoning of practitioners.

Chapter six presented analysis of ethical dimensions influencing professional reasoning and cited examples linked to doing good, avoiding harm, involving patients and families and promoting fairness. It is here that a frequently cited ethical framework emerges as significant – that of the 'four principles' outlined in the seminal work of Beauchamp and Childress (1989) and defended and developed extensively by Gillon (1994; 1995; 2003; 2015). This model suggested that the four principles of respecting autonomy, beneficence, non-maleficence and justice provide a "simple, accessible and culturally neutral approach to thinking about ethical issues in health care" (Gillon, 1994; p184). In simple terms, this model encourages health professionals to consider and balance respect for autonomy and choice of the individuals they serve whilst aiming to maximise net benefit when balanced against minimising risk, and acting on the basis of

human rights and fairness, particularly when there may be competing claims for resources.

Perhaps almost on par with the wide citation and application of these principles in practice, research and policy over several decades, has been a strong and often vehement critique. Commentators suggest that the principles merely represent a checklist rather than reflective or analytical guidance for ethical dilemmas or situations of conflict between principles (Clouser and Gert, 1990). Seedhouse further suggests that each broad heading is open for such wide interpretation that the principles themselves subsequently lack meaning (2017). Theorists also propose that the model continues to reflect a paternalistic approach to healthcare with professionals as experts and decision-makers and that the principles do not reflect critical challenges such as making decisions for sustainable healthcare (Hugman, 2005).

It is the extent to which these four principles continue to pervade contemporary practice, debate and, research – including work to critique and contest – that informs the application to the next section of this discussion. These principles are also applied in recognition that they underpin professional codes of conduct (Edwards et al, 2011). However, to represent the critique of this framework, I will ask critical questions throughout with the aim of developing layers of reflective enquiry, challenging simplistic representations and taken for granted perceptions.

8.5.2. Autonomy

Despite a shift in rhetoric and emphasis towards patient-centred practices in healthcare, professionally-led decision-making processes continue to be reported (Leach et al, 2010; Sugavanam et al, 2013). Observations and interviews suggested that the rehabilitation process was not explicitly centred around patient goals, and where implicit goals were set, this tended to be a professionally-led activity. There was an absence of patient and family input in to weekly multi-disciplinary meetings and a lack of focus on, or responsibility for, advocating for the patient at these meetings in their absence. Alongside this, ad-hoc use of family planning meetings, unclear methods of involving patients and families, and unclear processes for gaining consent for onward referrals were all suggestive of professionally controlled decision-making processes.

There are well-recognised challenges to meaningful involvement and engagement for older people and for people within acute hospital systems and authors caution against a 'one-size' approach to promoting engagement and involvement (Bright et al, 2015). In particular, many of the features of more patient-orientated goal setting processes – such as building trust and rapport, answering questions, and discussing concerns, priorities and wishes (D'Cruz et al, 2016) – are time intensive activities and practitioners in this setting discussed that these are activities which may be rationed when facing time and resource pressures. However, rather than simply accepting the deficiencies in, and challenges to, current practice there are some practical strategies which could assist, including the use of specific goal-setting forms or tools, which have been found to enhance patient and family engagement (Rosewilliam, Roskell & Pandyan, 2011). And indeed, the 4-stage model presented in Figure 5 could provide a framework to assist with patient and family discussions, assisting to translate what is sometimes an implicit reasoning process, in to explicit dialogue which can be communicated both verbally and in writing.

8.5.3. Beneficence

8.5.3.1. Is there an evidence base that rehabilitation brings about improvement?

Professionals and patients expressed a fundamental belief that rehabilitation was a process that facilitated improvement. Furthermore, the value that a positive evaluation of rehabilitation potential and access to a rehabilitative phase of care would create the conditions to optimise the likelihood of positive outcomes, were values that were evident throughout fieldwork. Therefore, the intention of professionals to bring about a positive outcome through rehabilitation was clear.

Research findings pertaining to the many models of rehabilitation for older people were synthesised in the literature review and, despite many challenges to transferability, an overriding theme emerged that an organised rehabilitative phase of care following a hospital admission for an older person does result in improved outcomes. Therefore, the professional value obvious during fieldwork that a phase of rehabilitation could potentially lead to positive outcomes was broadly supported by evidence. Interestingly however, there were no instances where practitioners themselves made direct reference to evidence during the period of fieldwork.

Returning to issues first illuminated in the literature review, practitioners face many challenges when attempting to look beyond headline general findings, to interpret in more detail the optimum conditions for rehabilitation for older people. Variances in national and international service provision (including the variance in comparisons to usual care and variances in funding), the poor descriptions of what constitutes rehabilitation services, the lack of statistical significance for both positive and negative outcomes, and the inherent challenges relating to the heterogeneous population are all widely reported in relation to the quality of the evidence base. This shines a light on the significant issues faced by researchers, strategists, policy-makers but most importantly
practitioners, when aiming to grow, and learn from, a high quality evidence base to support decisions about rehabilitation for older people following an acute admission.

However, research supporting positive outcomes from rehabilitation related to particular specialisms is strong, for example in Stroke, Major Trauma, Spinal Cord Injury and Chronic Obstructive Pulmonary Disease to name only a few. In reality, policy makers present as strengthening a limited evidence base for rehabilitation outcomes for generalist pathways for older people, by generalising from a strong evidence base for specialisms (which by default are likely to *include* older people) and by implication, asking practitioners to do the same. This raises important insights for critical reflection and implications for further research. All of these insights may offer an explanation as to why practitioners did not explicitly make links to evidence during fieldwork observations, with limited confidence or limited understanding potentially realistic consequences of evidence that is challenging to navigate.

8.5.3.2. Beneficence - were 'good' functional outcomes evident during 'patienttracking'?

Guidelines and evidence provide broad support for practitioner intentions to facilitate positive outcomes through periods of bed-based rehabilitation. However, the narratives of patient journeys tracked during fieldwork (summarised in Table 8) suggest that periods of rehabilitation in these cases (either as part of an episode of care on the base ward or through a joint episode on the base ward and the rehabilitation ward) failed to bring about functional improvements for most patients. It is a reality of health service provision that older people are at high risk of death and functional decline during or shortly after an unplanned hospital admission (NHS Digital, 2016; Milton-Wildey & O'Brien, 2010). The many reasons for admission, coupled with the likelihood of comorbid health conditions and a life-stage often characterised by frailty and susceptibility to further decline, means that older people can have worse function at the end of a hospital admission than they did prior to admission (Covinsky et al, 2003).

'Failure to Rescue', referring to the death of a hospital patient after a treatable complication (Silber et al, 1992) and 'Failure to Maintain', referring to the functional decline of a hospital patient following failure to prevent avoidable complications (Bail and Grealish, 2016) are two concepts used to describe and explore failings in acute hospital systems; the former with an international evidence base which has contributed to drivers to improve the recognition of the physiologically deteriorating patient (Massey, Chaboyer and Anderson, 2017). 'Failure to Improve' may also be a controversial reality of rehabilitation within acute hospital systems, referring to a lack of improvement towards expected functional goals. This poses yet another philosophical challenge to those professionals who see the facilitation of improvement as an integral part of their role.

There are examples of studies attempting to identify which patients are most at risk of functional decline and mortality during a period of inpatient rehabilitation. Evidence suggests that the use of a comprehensive and valid frailty index with values given to co-morbidities, number of medications, dependence in activities of daily living, cognition and deficits such as previous falls (among other items) can help to predict quality of patient outcome and mortality (Singh et al, 2012). Interestingly, the explicit use of a frailty assessment was not observed within this setting.

8.5.3.3. Balancing beneficence with non-maleficence

As discussed in section 6.2, the evaluation of rehabilitation potential was frequently discussed alongside a desire to give people 'a chance' and this word provides important insight in to the probabilities and uncertainties that professionals were contemplating. Indeed, the word itself conjures images of a professional gamble. It was

also a reflection of the researcher that the word chance was again indicative of ambiguity: that professionals used the word in the context of their desire to give people the opportunity of a positive outcome but that they often did not know what this positive outcome would mean in reality. Linked to this, the comparison between evaluating rehabilitation potential and a guessing game has been the focus of recent professional discourse (Enderby et al, 2017).

Fieldwork highlighted that a positive evaluation of rehabilitation potential and giving people 'a chance' often equated to a longer stay in bed-based services. Professional discourse suggests that, in this respect, practitioners err on the side of 'beneficent interventionism' (Singh et al, 2012; p245). However, practitioner reflections and researcher observations (particularly from patient tracking during phase 2) suggested that this may carry risks such as secondary medical complications (for example infections), functional decline, psychological deterioration, and even death. Indeed, the risk of becoming a lower priority for rehabilitation interventions, due to being located in a medical bed but whilst waiting for a rehabilitation bed was also noted during patient tracking and is a risk which has been expressed within published commentary on this subject (Wade, 2016).

Wade (2009) acknowledges wider risks associated with rehabilitation and rehabilitation decision-making. These include risks to the individual of investigating what people can and cannot do and therefore bringing attention to weaknesses and failures; loss of motivation or raising expectations if goals are not relevant or not attainable; rehabilitation inducing passivity; and communicating rehabilitation decisions (such as the decision to stop active therapy) in an ineffective way. Elements of all of these issues were observed or discussed during fieldwork: a patient being transferred to the rehabilitation ward with family having unrealistic expectations (page 147); patients associating rehabilitation with wanting more professionally-led therapy rather than

interpreting it as a phase where they can help themselves (page 142); and limited communication about the decision to discontinue physiotherapy (page 181).

Another risk, although one which is hard to quantify and articulate, is the general risk to quality of life of spending an increased proportion of remaining life in a bed-based phase of care. The extent to which honest dialogue took place to enable patients in this context to decide whether this opportunity was worth 'a chance' is explored in the section below and related to the concept of autonomy.

If a positive evaluation of rehabilitation potential most often equated to additional time within a bed-based rehabilitation service, one additional outcome worthy consideration is patient experience of such services. Indeed, if patient experience is positive, this can contribute to the achievement of good outcomes, although if negative, this could represent an additional area of risk or potential harm. Unfortunately, due to the limited direct insights gained from patients and relatives in this study (and that this was not a primary area of focus for this research), inferences from this study cannot be made. However, patient experiences of bed-based rehabilitation reported in wider research suggest that rehabilitation may not adequately meet needs or promote independence, there is a lack of involvement in goal setting and that information giving about the purpose of the service is inconsistent (Ariss, 2014; Atwal et al, 2007). It is not within the scope of this current study to suggest whether these themes reflected the patient experiences may represent an additional risk which merits consideration when aiming to achieve an important balance between maximising positive outcomes and minimising harm.

If improvements and positive outcomes cannot be assumed following a period of bedbased care for an older person, questions about adverse outcomes must also be examined, although the evidence discussing adverse events and harms is minimal (Wade, 2009). Wade (2009) suggests that one reason for this is that practitioners are

not routinely measuring or documenting harms associated with rehabilitation and rehabilitation decision-making, and similarly researchers are not routinely focussing on this within rehabilitation research.

Within the base ward, although the ward was seen to be measuring and reporting adverse events and harms in line with national drivers (NHS England, 2016b), this was generic in nature, measuring one-off events such as falls or infections. Ways of measuring and documenting general functional decline were not explicit, nor was it easy to understand how some of the softer examples of harm (such as raised expectations) were being examined or understood. However, on critical consideration of her own research aims, it must be acknowledged that adverse outcomes or harms were not an explicit focus (within research objectives and therefore not the focus of observations of interviews). It could be suggested that here the social construction of rehabilitation as a process linked to improvement was influencing her own research worldview.

8.5.3.4. Balancing beneficence with autonomy – who's version of a 'good outcome'?

An element of the values held by professionals established within previous sections was that rehabilitation was a process which facilitated improvements in function (often focussed on physical function and mobility) and that an evaluation of rehabilitation potential was based on a judgement about the likelihood that engagement in a rehabilitative process would bring about desired changes in mobility and function. However, due to lack of an explicit goal-setting process, particularly on the base ward, this represented an imbalance between beneficence and the autonomy of the individual, and perhaps was more of a well-intentioned paternalistic approach to determining a good outcome.

A multi-centre cohort study exploring patient goal-setting in post-acute geriatric rehabilitation found that goals relating to mobility were overwhelmingly the most frequently reported by patients (Kus, 2011). Therefore in this context, it could be suggested that practitioner reasoning was based on well-intentioned generalisations from evidence. It is perhaps an understandable (although not unavoidable) reality that, within the constraints of acute hospital care where an explicit goal-setting process would be time intensive, particularly with frail older adults often with additional challenges of cognitive or sensory impairments, practitioners were relying on pattern recognition and population-based understandings to formulate goals for rehabilitation. Practitioner extracts presented in interviews did recognise that this was not truly patient-centred although expressed sentiments that they still believed the focus on improving mobility was in the best interests of patients. Notably however, practitioners also highlighted that this was becoming a default reasoning process.

Linked to the above, professionals placed a value on promoting mobility gains through rehabilitation and equated this version of rehabilitation as a process requiring time and intensity. Although not always, this did often subsequently equate to time within bedbased rehabilitation and often related to perceived deficiencies or waiting times for community based rehabilitative services. This then leads to another critical question as to whether, if meaningful alternatives and consequences were explained, time spent in bed-based rehabilitation would be how an older person would want to spend their time when, a trajectory over time will, inevitably for some, be one of functional decline.

This is unquestionably a challenging question for patients, families and professionals alike although one where honest dialogue between all parties could assist. This honest dialogue was observed in the example of Patient 1, where a long discussion between the occupational therapist and patient's daughter acknowledged limited recent progress and the risks associated with prolonging a hospital stay. However, this took place on day 44 of the admission, after time had been spent treating acute medical issues whilst

simultaneously trying to facilitate functional improvements which ultimately were not consistently achieved. Honest dialogue at an earlier stage, and involving the patient themselves where appropriate, could begin to develop understanding on this challenging issue. That is to say, in simple terms, whether time spent in hospital – often (but not always) as a person nears the end of their life – is the way people would choose to spend their time if offered an honest appraisal of the risks and potential gains.

Parallels can be drawn with a model of 'Advance Care Planning'; a model to facilitate advanced discussions about wishes and priorities for care, established in palliative care and now growing within dementia care (NHS, 2007; NHS England, 2017). The advance care planning principle is that a voluntary discussion (with documentation of the outcomes) can happen between any person and a health professional about priorities and wishes for the end of a person's life. However, in reality the model tends to be used when it is known someone is approaching the end of life, or it is anticipated they will lose capacity to make decisions.

Recent surveys highlight only 5% of people over 65 have been offered an opportunity to discuss an advance care plan, although a third of respondents indicated they would be interested in public sessions or discussing this with their GP (NICE, 2016c). This suggests that there is public interest in developing an open dialogue on this issue although both survey results and the local examples would suggest that this is happening at a very late stage of life, if at all. It is perhaps reflective of larger philosophical debates about the medicalisation of ageing and death, and the orientation of a health service underpinned by a belief that most causes of death or decline can be resisted, postponed or remediated (Al-Qurainy, Collis & Feuer, 2009).

Returning to the complexity of balancing beneficence and autonomy, in this setting, beneficence was felt to be taking precedence over autonomy. Critical voices have

raised that prioritising autonomy is a 'good' in itself (O'Keefe, 2001) however some argue that within a rehabilitation context, many require paternalistic interventions with an aim of working towards autonomy in the longer term (Proot et al, 2000). It is perhaps not as simple as prioritising one or the other, but instead reinforces the need for reflective and deliberate reasoning, rather than reasoning based on default positions because of time and system constraints. Reflecting on the many instances of dubious involvement, choice and informed consent, alongside some of the examples of functional decline during rehabilitative care, perhaps neither autonomy nor beneficence was genuinely realised.

8.5.4. Fairness, justice and rationing

The principle of justice relates to fair distribution of resources and the discussion of this not only received significant emphasis within research findings, but has also permeated earlier discussions of utilitarian and deontological ethics. Rationing, a strategy linked to justice, has been discussed as the withholding of beneficial interventions to patients, mainly for cost-effectiveness reasons (Strech et al, 2009) and finding ways of working in attempt to fairly distribute finite resources (Schubert et al, 2008). The need for healthcare rationing has been described as 'inescapable' (Fleck, 2011; p156) in the context of the already well-documented demographic and social changes and against a complex economic and political background. This inescapable nature of rationing in everyday health care was witnessed during fieldwork observations and permeated decisions about priorities within daily caseloads, management of lists for weekend therapy, listing patients for bed-based rehabilitation, and decisions about referrals to other services.

More critically, rationing has been seen as a way of maintaining the appearance of a public service whilst reducing it in practice (Lipsky, 1980). And although decision-

makers at a macro level – such as politicians and strategists – may indicate that resources can be distributed more fairly through efficiency savings and ridding the system of those who abuse it (Fleck, 2011) – this fieldwork would suggest that practitioners are still left with the inescapable, and often hidden, task of rationing on a patient-by-patient basis.

Examples and extracts from fieldwork not only illuminate instances of rationing, but also interestingly provide insight in to the creation of rules or principles to assist with rationing. Examples of rules included whether people with cognitive impairment could benefit from rehabilitation ('with an MMSE of 10 we can't really put reablement in as she won't learn anything'), or when people were not demonstrating sufficient progress in rehabilitation and therefore interventions should be scaled back ('this is potentially her new baseline, she's not going to get any better. Especially because it's been two consecutive days in a row [that the patient has needed the hoist]').

Interesting, the use of the Mini Mental State Examination (MMSE) as a rationing tool received high profile attention and criticism when it was used in strategic decision-making by NICE on the subject of funding for dementia drugs (Moreira, 2011). Critics challenged the sensitivity of the tool to disease progression and level of functioning and suggested it was open to subjective administration (Ballard, 2006). In recent recommendations about Intermediate Care including Reablement (NICE, 2017), it is specifically highlighted that people should not be excluded from services because they have particular conditions such as dementia. Yet despite criticism of using a tool such as the MMSE to make rationing decisions, and specific policy guidance not to exclude on the basis of cognitive impairment, at a micro-level, practitioners were observed to use this practice to create their own criteria for rationing. Pre-existing cognitive impairment has been recognised as being associated with poorer functional outcomes following a hospital admission and whilst likely to be deeply layered and multi-factorial, the link to limited access to rehabilitation services for this group has been questioned

(Longley et al, 2018). Findings here suggest there could be some truth behind this suggestion.

Principles to assist with rationing were also observed to be based on tacit knowledge and open to subjective interpretation. In the example on page 194, the physiotherapist was required to ration the weekend physiotherapy service, interpreting who would benefit from mobility practice with a physiotherapist or physiotherapy assistant (in comparison to those where nursing staff could manage their mobility) and using knowledge of a specific ward to reason that someone potentially may stay in bed all weekend without physiotherapy-led intervention. In this example there was also a suggestion that the negotiation with the wider physiotherapy team about the value they place on weekend physiotherapy for their own patients would influence the list and rationing of the service as a whole. For all of these reasons, it is easy to see how decisions reached by individual therapists could differ from another professional, influenced by their own tacit knowledge and a different set of contextual factors.

Strategies used to assist with rationing were presented in chapter six, including explicit and established strategies such as waiting lists and triage, alongside less overt strategies to assist with the gatekeeping of a finite rehabilitation resource. Links to the influential work of Lipsky (1980) were hypothesised in the literature review and it is in such examples where health professionals could be seen as 'street-level bureaucrats'. Lipsky describes street-level bureaucrats as having the power to deliver benefits and sanctions and as constantly being torn by the demands of service recipients (in this case, patients and relatives) and by the demands of the organisation (and perhaps the population) to ensure the effectiveness and efficiency of the service. People who find themselves in this position determine how people experience state policies first hand and find themselves acting as mediators between the state and any given individual. Hammell (2007) offers further insight in to this issue for Allied Health Professionals (AHPs), suggesting that there are institutional processes that professionals actively

reinforce which may disempower clients and go against fundamental principles of person-centred practice.

8.5.4.1. Rationing based on appropriate or deserving behaviours

Controversially, Lipsky suggests that citizens who receive public benefits interact with public agents who require certain behaviours of them. In this case, patients interact with health professionals in order to receive interventions to benefit their health and wellbeing, but judgements about appropriate behaviour and conduct may influence the health professional in this important gatekeeping role.

Although small in number and influenced by a wide range of factors, the use of positive descriptions by health professionals observed during fieldwork – such as 'she's lovely', 'she's delightful', 'she's my favourite' - was obvious within handover meetings, weekly multi-disciplinary meetings and within other informal times of discussion and information exchange. In contrast, the creation of accounts around difficult or challenging patients - 'refused assessment', 'not trying', 'wife destructive to care' - was also witnessed with no instances of such accounts being challenged or counter-narratives created At a surface level, such accounts simply emphasise the emotional element of professional practice and that professional decision-making can be influenced by emotion and values.

At a deeper level, insight in to such accounts can be linked to creating narratives around good or bad patients; those who are deserving or undeserving of services. The labelling of good and bad patients is an idea which is supported by research (Sointu, 2017; Dingwall & Murray, 1983) and again is of interest to the social constructionist researcher. Due to the realities of finite resources, it is reasonable to suggest that the narratives about deserving and underserving patients may then form a foundation which informs the allocation of time, care and services. On some occasions, the

narrative of good or bad patients was associated with favourable or less favourable outcomes (presented on page 196).

Through a closer look at the examples in this study, the good patient was understood to be one who was complying with the social norms of the institution, such as adhering to health professional advice and engaging positively in health professional interactions. The bad patient, in comparison, was often one who was interpreted to have refused advice or interventions, and therefore was potentially disrespecting the unspoken hierarchy of the system. These themes are echoed in a study by Sointu (2017) exploring good and bad patients from the perspective of medical students. Sointu suggests that good patients are characterised by features such as being motivated, knowledgeable, compliant and active in their care and these patients were then seen to be afforded more time, care and appreciation.

Finally professionals regularly cited patient motivation as a key influencing factor in their evaluation of rehabilitation potential and it has been hypothesised that motivation may form part of the narrative around a good or deserving patient. It is an area which warrants further exploration, although a narrative created and upheld about a demotivated patient could feasibly equate to rehabilitation potential being quickly ruled out.

8.5.4.2. Manipulating the nature and quality of information to assist with rationing

Lipsky suggests that client demands will be expressed only to the extent to which people are aware that they have a need or condition that could be met by a public agent. He goes on to suggest that withholding information about a service inevitably depresses service demand. Firstly, in relation to this project, neither the term rehabilitation potential, nor the reasoning process to evaluate potential were observed being explicitly explained to patients, which could be said to be putting patients at an immediate disadvantage with limited information.

Lipsky also suggests that patients will only make demands for interventions or services to the extent to which they know they are available or that they could be accessed by them. As outlined in the findings, it was common that information about a service or intervention would not be communicated to patients unless it had already been decided that the person would benefit from this service. In this way, patients deemed not suitable for a rehabilitation resource (such as weekend therapy) would often not be aware of the existence of these services or that they had ever been considered.

In the case of referring for community rehabilitation (such as community physiotherapy), professionals discussed openly that there were occasions when they decided not to refer patients because the waiting list was too long. This decision-making process or outcome was then not observed to be communicated to patients, although and not documented in records. One consequence of this was that patients were never fully aware that they had a need that could be met by this service and therefore did not make any demands for it. However, Lipsky also proposes wider consequences of practices such as this in that, at a macro and strategic level, organisations often ration services at a community and population level based on information from referrals, waiting lists, and contacts. Through practices such as failing to refer, and failing to let people know that services stand ready to assist if patients think they have unmet needs, Lipsky suggests that professionals, and in this case health professionals, are contributing to data which suggests low demand. It is interesting to reflect on this in light of a reduction of bed-based intermediate care provision and perceptions that community rehabilitation was under-resourced.

Where there is a lack of alternatives or patient preferences cannot be acted on because of resource limitations, engagement and involvement should focus on

education about realistic and available options (Kimmel et al, 2017). This was observed in part – for example where referral to the reablement service had been determined to be inappropriate, discussions took place about long term care packages including how to refer and potential costs. However, once again, this was observed to happen in an ad-hoc way and the presentation of alternatives or reasons for a service being unavailable or unsuitable were not always clear.

8.5.4.3. Is justice and fairness realistic for older people?

Despite being the biggest users of acute hospitals, the acute hospital system is increasingly orientated away from the needs of people with complex presentations and co-existing conditions. The high profile that specialisms attract in terms of funding and research, the ongoing separation between physical and mental health services, and the lack of integration of health and social care could also be seen as disadvantaging the largest group of hospital users. The introduction of comprehensive geriatric assessments and development of specialist models of service provision, such as frailty assessment units, or in-reach frailty teams are going some way to recognise and address the issue (Edmans et al, 2013). Evidence suggests such developments may go some way to improve outcomes for this population – such as improving the likelihood that people will be alive and in their own homes at follow-up (Ellis et al, 2017). However, patient and carer perceptions suggest their experience is less positive (Darby et al, 2017) and many of the issues above represent overriding and systemic barriers facing an older person in relation to receiving fair and just services within an acute hospital system.

The issue of resources within the acute hospital system was inescapable during time spent within this field. The closure and re-provision of intermediate care beds and the perceived deficiencies in community rehabilitation services for older people were

frequently discussed by professionals. Professionals shared feelings of pressure that this created and also, through comparisons with the funding allocated to more specialist branches of rehabilitation (such as the stroke rehabilitation ward and the community stroke team), shared the perception that the organisation did not place value on general rehabilitation targeted at the frail, older population.

Wade, in his four-part critical commentary suggests limited funding, low profile and inadequate provision are realities of rehabilitation practice (2015). Returning to neoliberalist politics, it could be argued that the issue of cost effectiveness is the most critical one, and the evidence for cost-effectiveness of rehabilitation is not only limited, but measurements are controversial and the subject of debate. Measurements frequently utilise data in relation to contribution back to society (in terms of return to employment) and reduction in dependency on care services. Not only is the relevance to an older population questionable, but the extent to which this benefit accrues to health services (rather than social care) is not immediately obvious. Coupled with crude estimates that the cost benefits of rehabilitation may take between 1-5 years to pay off (Turner-Stokes, Paul & Williams, 2006), the critical challenge to commissioners is whether spending on rehabilitation for frail older people is a worthwhile endeavour. Although not explicitly acknowledged in this way, this may be suggestive of the 'fair innings' debate discussed in relation to equitable health services, intimating that scarce resources should be allocated to those judged to have the longest life left to live and largest contribution to make back to society (Hicks, 2011; Lane et al, 2017). Reflecting on this, even before entering the system, and before becoming the subject of individual decisions, older people face challenges to fair and just services.

It is important to note that rationing on the basis of age, or a 'fair-innings' attitude was not observed at the level of individual professionals or within clinical teams. Indeed, the desire to give people 'a chance' communicated an intention that health professionals wanted to facilitate positive outcomes for all patients, irrespective of age. However, this

is not to say that issues pertaining to age did not influence professional reasoning in other more implicit ways. There were examples where professionals clearly recognised that, in their attempts to manage the competing demands of their own workloads, they potentially disadvantaged older people. The most obvious examples related to involving older people directly in care and decision-making with professionals recognising that when time was limited, meaningful involvement of older people became more challenging, with short-cuts such as gaining information from relatives, becoming a recognised and default practice.

There were also other critical examples which would benefit from reflection. Professionals within the study suggested that the organisation placed value on fastpaced patient trajectories and encouraged them to prioritise those patients where transfer of care could be achieved most quickly. Occupational therapists clearly stated that they were unable to prioritise patients who already resided in 24-hour care environments. The study also revealed examples of health professionals creating their own justification for the rationing or withholding of services – for example stating cognitive function scores which indicated that progress through rehabilitation was unrealistic; patients who had already received services had already had a 'chance' at rehabilitation; and the time that it would take an older person to participate in a rehabilitation session created more pressure when compared to a patient of a younger age.

Alongside this, professionals discussed conceptualising rehabilitation potential as not just potential to benefit, but potential to get back to a baseline level of function, and yet the contrary outcome of older people being vulnerable to decline during a hospital admission has been discussed in earlier sections (Covinsky et al, 2003). The significance of motivation for, and active engagement within, the rehabilitation process has also been discussed although yet again, literature would suggest that prevalence

of low motivation and mood disorders in hospitalised older people is a significant issue (Royal College of Psychiatrists, 2005).

Considering all of these factors, a significant and challenging landscape emerges of the numerous challenges faced by older people potentially contributing to universal disadvantages when it comes to the allocation of finite rehabilitation resources.

8.5.5. Ethical distress for professionals

The verbatim extracts presented in section 6.2.5 illuminate notable differences between how professionals would like to practice, and the realities of their service provision. The difference between ideal and real-world practice provoked a number of responses, from general feelings of challenge and discomfort, to feelings of guilt, and to actions orientated towards blaming others.

These feelings of tension and challenge were perhaps being experienced in a deeper way by different professionals at different points in time, and could be equated with the concept of 'ethical distress' (Durocher et al, 2016; Bushby, 2015; Mendes, 2017), originating from the seminal work by Jameton (1984). According to Jameton, ethical distress is experienced when a person knows the right thing to do but institutional constraints make it difficult to pursue this course of action. There are differences between situations of ethical uncertainty (where someone may not be sure of the right course of action) and situations of ethical dilemma (where a person faces two or more situations with perceived equally positive or equally negative outcomes). A more recent definition suggests that ethical distress is "the experience of being seriously compromised as a moral agent in practicing in accordance with accepted values and standards. It is a relational experience shaped by multiple contexts, including the sociopolitical and cultural context of the work-place environment" (Varcoe et al, 2012; p59).

One participant directly talked about the issue of compromise in their practice (page 192) and Mendes (2017) suggests that distress emerges because of a compromise in values when the right course of action cannot be taken because of perceived constraints.

Insights shared by participants, such as their desire to talk to patients but talking to families instead (page 184), the desire to focus on wider functional goals rather than just mobility (page 139) and the desire for intensive therapeutic interventions but being limited by time (page 169) all strongly resonate with the professional compromises suggested above and therefore may have been contributors to ethical distress. Examining the examples alongside language used in the descriptions, such as 'uneasy', 'frustration' and 'failing patients' adds further insight in to the personal and professional distress experienced. Alongside this, throughout many aspects of observations and interviews, practitioners retold stories of challenging practice situations, which Carpenter (2010) suggests is indicative of people carrying the remnants of professional compromises or situations they have been unable to satisfactorily resolve.

More globally, the desire to bring about positive outcomes (discussed extensively in section 8.5.3) but feeling compromised and constrained in their attempts to deliver a 'proper' version of rehabilitation could also be linked to ethical distress. The extent to which practitioners were aware of the limited functional improvements suggested in the narratives captured during patient tracking was not fully clear, although the distress and frustrations communicated suggest insight at either a conscious or subconscious level.

Authors who recognise that ethical distress exists for healthcare professionals, also reflect that professionals will use various strategies to manage and minimise the effects. Professionals will use resourceful and creative ways to advocate for patients and potentially 'get round the system' in attempts to find concessions and ways forward

which are acceptable for the patient, to the organisation, and to professionals' own morals and values (Wong, 1992; Carpenter 2010). It is here that Lipsky's detailing of the role of *Street Level Bureaucrats* (1980) once more emerges as influential to this discussion. The presentation above of strategies used by street level bureaucrats was mainly linked to the desire to ration a public resource and attempts to translate overriding strategies and policies in to actions for individuals who require state involvement. However, these strategies were also potentially fulfilling an additional purpose – that is to say, helping to minimise the feelings of powerlessness and distress experienced if service delivery was compromised through systemic pressures.

Although street level bureaucrats are criticised for creating their own rules and subjectively interpreting them in different contexts, the creativity and resourcefulness required to do this suggests a sense of control and empowerment in otherwise disempowering situations. This resourcefulness and creativity could be seen in the imposition of changing the perspective on the concept of baseline (and how professionals circumnavigated the guidance in order to still utilise the word and concept), or how professionals advocated for particular patients in order to facilitate desired outcomes. However, although such strategies may have energising potential and promote feelings of professional accomplishment, there are also inherent dangers and risks potentially resulting in inconsistencies or inequities and a lack of respect for wider roles (Carpenter, 2010).

An additional strategy suggested is the use of standards, guidelines and in particular, professional codes of conduct, to provide guidance in situations of ethical uncertainty (Mendes, 2017). However, on closer examination of codes of conduct, statements about fairness and equality are based on broad recommendations linked to common determinants of inequalities such as race, age and gender (RCOT, 2015; CSP, 2011) with little in the way of strategies or practical support to manage more intricate equality issues. Indeed, Edwards et al (2011) suggest that although codes of conduct attempt to

include broad statements to acknowledge societal obligations and a commitment to collective health and wellbeing, underpinning such codes with the four individualist principles and with a particularly under-developed, un-critical understanding of the concept of justice, mean they offer insufficient support for practitioners to enact true ethical practice. Hammell (2007) also questions whether professional codes of conduct, which suggest commitment to needs led, person-centred services, offer realistic guidance for contemporary practice where professionals are tied by their role of resource gatekeeper, with an allegiance to the institutional system. The participant quote that it is the 'resources that win' highlighted the reality of this institutional allegiance.

To conclude this section, and before progressing to the consideration of professional roles in decision-making, it is important to return to the many influences on the evaluation of rehabilitation potential, a central area of enquiry for this study. These influences, including the influence of ethical dimensions, and the use of strategies to manage some of the pressures, are visually represented in Figure 13. The three inner layers, represent layers of factors considered by professionals as they evaluate whether a person has capacity to benefit from rehabilitative interventions. However, the boxes which surround these layers represent other influences on professional reasoning, with many areas of 'push' and 'pull', and tensions between different influences. The box below the lower line represents areas which were rarely vocalised or observed but were implicit influences on reasoning and decision-making.



	Hidden elements of reasor	ning	
Patient goa	ls Evidence	Criteria	

8.6. The centrality of the physiotherapy role in evaluating rehabilitation potential and making decisions about pathways

The importance of the physiotherapist in evaluating rehabilitation potential clearly emerged from findings and was noted by physiotherapists themselves, by occupational therapy counterparts and by other members of the team. This was linked to the joint focus of physiotherapists on treating acute medical issues and improving mobility, both areas being part of a dominant discourse in this setting. Not only was this value communicated by individual professionals, but institutional developments also emphasised value placed on physiotherapists as the main agents in the delivery of active rehabilitation interventions (for example the 'Green Day/Red Day' initiative described on page 142)

The importance of the physiotherapy role was also linked to the frequency and intensity of involvement in direct patient care in comparison to occupational therapists (Table 6). Information from patient tracking suggested that more patients within the base ward and rehabilitation ward had traditional physiotherapy needs, reflecting earlier and higher intensity physiotherapy input and a more prominent physiotherapy role. Findings such as this are reflected in other studies of patients on acute and rehabilitation wards, where more patients are determined to have physiotherapy needs than occupational therapy needs (Hubbard et al, 2004).

Studies also suggest that older patients tend to have better awareness of the physiotherapy role than the occupational therapy role during an acute hospital admission (Atwal et al, 2007) which could also be linked to more frequent involvement and clearer purpose. In this study, there was evidence of patients more readily associating with the physiotherapy role – for example Patient 5 believed that his physiotherapist had been present on a home visit, even though it was actually the occupational therapist who was present. Whilst not necessarily being significant in and

of itself, it is reflective of the broader importance associated with the physiotherapy role.

8.6.1. Locating occupational therapy and physiotherapy within the acute care hierarchy – issues of power and decision-making authority

Issues of hierarchy and power within health systems could become the focus for a full thesis. However, the hierarchical position of the occupational therapist and physiotherapist role is worthy of reflective discussion here, primarily because of the direct relevance to their authority within rehabilitation decision-making and because a comparative exploration of the difference in occupational therapy and physiotherapy roles was an objective of the study.

Despite many researchers and policy makers bringing attention to issues of power and hierarchy, traditional medical hierarchies persist within healthcare services and systems, perhaps nowhere more obvious than the acute general hospital (Willis, 2006; Nugus et al, 2010). The traditional hierarchy of the doctor being the decision-maker was evident in this setting with occasions noted where the judgement of the doctor was observed to override other professionals; or required to form alliances in order for the opinions of other professionals to hold value (pages 160; 202).

Findings provide support for the notion that physiotherapists had a greater standing within the social setting than occupational therapists. Furthermore, and in relation to the traditional medical hierarchy, it could also be hypothesised that the position of physiotherapists was partly influenced by their proximity to doctors, whose own position and standing was greater still.

Whilst there are studies which acknowledge or allude to the existence of healthcare hierarchies, literature which more specifically explores hierarchies within rehabilitation

settings or issues of power and hierarchy for occupational therapists and physiotherapists are notable by their absence. Yet such observations of position and power within the social space are of clear interest to the social-constructionist researcher. Equally, how professionals mobilise power in their actions – in the case of this research, how they influence and make decisions – is of particular interest here.

With a paucity of contemporary, discipline-specific literature, the seminal work of Pierre Bourdieu (1984; 1989, 1993) provides explanatory theoretical frameworks to support further exploration. Bourdieu discusses his concept of 'field' as a structured space of positions which imposes determinations on all those who enter it and hierarchies within the field are arranged in respect of certain types of power or capital (Bourdieu, 1993). Individuals and groups accumulate 'capital' over time, based on dominant values and discourses, and capital can also accumulate based on the proximity people and groups have in relation to others. Types of capital include economic capital (capital accrued through financial assets and resources), cultural capital (capital accrued through language, possessing cultural objects and acquisition of styles and tastes), social capital (resources mobilised through membership in social networks) and symbolic capital (resources created from social position and prestige). Capital can manifest itself in skills and behaviours which match the standards of dominant institutions and therefore can be utilised to produce meaningful situational advantages.

Applied to the social space of the healthcare team within this context, it is perhaps acceptable to hypothesise that the medical team were, through historical and established professional esteem, and through the dominance of a biomedical model of health and illness, positioned highly in terms of cultural, social and symbolic capital. In this setting, and over time, it could be proposed that physiotherapists had developed proximity to the medical profession by closely affiliating through language and skills and therefore accrued their own multi-dimensional capital in this setting. Physiotherapists and doctors were able to talk with more frequency and regularity

about shared patient cases – linked to physiotherapists knowing the majority of patients at an earlier part of the patient pathway. Physiotherapists also possessed knowledge (and a shared language) of acute medical problems and interventions and collaborated with the medical team to manage some of these concerns.

Perhaps less obvious although worthy of note, physiotherapists shared the same physical environment more frequently with the medical team (that is to say, they were observed to be present on the base ward with more frequency) in comparison to the occupational therapist (who was observed to be more frequently based within a departmental base). All of these elements raised the social, cultural and symbolic capital held by physiotherapists, who could then translate this capital in to value placed on actions and decisions. Bourdieu's principles would suggest that situational advantages can be gained through this and the extracts presented on page 202 suggest that the physiotherapist opinion was held in high esteem. The application of Bourdieu's construct suggests that such positions of authority and power do not emerge by chance or in response to individual personalities or dynamics.

Extending the application of these concepts, it could be further hypothesised that this framework provides insight in to the position of occupational therapists within this social space. Whilst a proximity between medicine and physiotherapy was noted, a distance between medicine and occupational therapy was also observed. This was recognised by one occupational therapist who suggested that their stronger alignment with a social model of health and disability created distance from dominant norms and affected the influence of their profession (page 207). There is also support for this within literature with occupational therapists reporting feeling intimidated by doctors (Lohman, Mu & Scheirton, 2003) and struggling to assert their role in interactions with medical professionals (Robertson, 2012).

In this study, it was acknowledged by occupational therapists themselves, alongside implicitly being acknowledged within the wider team, that occupational therapists were perceived as less influential in the rehabilitation environment and within rehabilitation decision-making. Bourdieu's theories would equate this to occupational therapists possessing lower quantities of capital, particularly symbolic capital. The alliances that occupational therapists formed with others – particularly with physiotherapy (examples discussed on page 209) - represented a way of creating proximity to a group who held more in the way of resources and capital.

Another critical way in which occupational therapists were creating proximity to physiotherapy and medicine, and therefore to a biomedical model of health, was through the alignment with a version of rehabilitation orientated towards improving physical health and particularly mobility. Manifestation of this included words used in their discussions of rehabilitation (pages 138; 154), and that occupational therapists were often observed giving feedback in team meetings in relation to physical elements of functioning, such as transfers and mobility. This is echoed in wider research where occupational therapists are found to express their work in biomedical language because a focus on wider aspects of occupation is not understood or potentially valued (Murray et al, 2015).

Although physiotherapists were perceived as potentially holding more power than occupational therapists, significantly it was not a power that was observed on the surface to be divisive or competitive. However on deeper reflection, although occupational therapists did not appear to be dominated by, or in conflict directly with, physiotherapy or wider colleagues, extracts are suggestive of internal and external struggles (page 206) related to attempts to legitimise knowledge and values which are essentially in contrast with dominant discourses.

It is important to emphasise that alongside such struggles, there was evidence of collaborative relationships. The interview extract on page 210 suggest physiotherapists and occupational therapists emphasise their similarities rather than their differences. There were also features resonant of a concept named as collaborative power (Nugus et al, 2010); a concept to describe the way power is utilised to maximise the potential of both distinctive and overlapping roles. Such features between physiotherapists and occupational therapists included facilitative information-sharing (such as sharing information to help identify when a patient was ready for occupational therapy assessment), and appropriate role distinctiveness and interchangeability (for example, physiotherapists distinguishing their role in mobility assessments, but both professionals contributing to assessments of transfers, mobility in the home environment and moving and handling plans).

These examples provide a platform for professionals to recognise how to harness collaborative power, although this may be reliant on a tacit understanding of roles and influenced by personalities within this context. The emphasis on collaboration rather than conflict should therefore not be taken for granted and could emerge within different teams or at different times.

It is important to note that this analysis of the hierarchies and positions within this particular field, but as situated within the larger thesis with broader research objectives, has, through necessity, presented a brief and concise overview. It does not discuss the position of the many other individuals, groups, organisations or institutions which act within, and influence, the structure of this space. A deeper analysis could, for instance, also illuminate positions adopted (and struggles faced) by physiotherapy, medicine and others in relation to wider dominant values held by organisations and institutions. And perhaps significantly, it could also provide a framework and vocabulary to discuss the critical position of patients and families.

Attention must also be paid to limitations with Bourdieu's version of the organisation of the social world. A frequently cited criticism is the deterministic nature of his theories, with some suggesting that his conceptualisations provide little acknowledgement of how individuals and groups can influence social action or transformation (Fowler, 2000). His concepts are also highly political and therefore attract disciples and extreme critics (Swartz and Zolberg, 2004). However, Bourdieu's concepts have enabled observations about how contrasting sets of knowledge and experience (specifically for occupational therapists and physiotherapists) potentially translated to position and capital within the social space and have presented important opportunities for reflection.

8.6.2. Challenges to both disciplines from systemic constraints

A clear understanding emerged from fieldwork and interviews that both occupational therapists and physiotherapists felt pressure created from systemic constraints. Issues such as a lack of time, the focus on fast-paced turnover and trajectories, and perceptions about wider resources such as access to follow-up services were highlighted by both professions. Many of these issues contributed to examples where both disciplines felt they were unable to provide best (or 'proper') practice for their patients.

Whilst frustrations were evident across both groups, it was perhaps notable that occupational therapists also presented as experiencing challenges with professional identity, compromises to professional practice and issues relating to professional confidence, issues which were not notably obvious from physiotherapists. All of these themes were almost identically echoed in a scoping review of occupational therapy practice in acute hospital settings in Australia (Britton, Rosenwax and McNamara, 2015) and reflect real challenges for the profession in this area of practice. In contrast, for physiotherapists, the extract presented on page 209 suggested that despite frustrations they were managing to practice with a sense of confidence and value.

It is important to note that this may have been influenced by the experience level of the professionals involved in this study and whilst detailed information about time since qualification and grades of staff was not a focus (and deliberately not presented to ensure anonymity in such a localised context), general observations include the fact that the physiotherapy team included a higher grade therapist (Band 7) than the occupational therapy team (Band 6) and that some of the occupational therapists involved were relatively new to their roles. It is also important to note that, as an occupational therapist, I also may have been more focussed on examples pertaining to her own profession

However, as challenges to confidence and professional identify have been recognised as consequences of ethical distress, one explanation could be that occupational therapists were experiencing this distress to a greater extent than physiotherapy counterparts. Literature suggests that occupational therapists in acute care are particularly vulnerable to feelings of distress through occupational alienation. This refers to situations where the purpose of their labour becomes increasingly orientated away from a client-centred approach to practice which focusses on the central tenet of meaningful occupation (Durocher et al, 2016). Furthermore, professional dissatisfaction is reported among occupational therapists who feel they are unable to focus on their key domain of concern (Wilding & Whiteford, 2009). Practice observed in this study, alongside the narrative shared during interviews, suggested that rehabilitation did not focus on wider occupational, social and emotional aspects of health and recovery and this was being keenly felt by the occupational therapists involved.

The orientation of rehabilitation and rehabilitation decision-making towards safety, rather than a wider philosophy to address holistic needs, is increasingly recognised

(Durocher et al, 2016; Durocher & Gibson, 2010). Although this is of concern to all rehabilitation professionals, because safety was also intrinsically linked to mobility, it is likely that this was minimising the impact of this tension for physiotherapists, whilst being felt more acutely by occupational therapy counterparts. At the same time, it could be argued that resilience amongst this group was already at a low due to issues of power and perceived value.

8.7. Chapter summary

Hammersley and Atkinson (2007) discuss that it is not enough for ethnographic writing to be persuasive or rich in descriptive detail, but that it also needs to demonstrate the adequacy of the scholarly claims. It is in this chapter, through links to policy, research and theoretical frameworks, that the presentation of findings, and the many questions raised, are placed within a macro socio-political context, illuminated by wider professional knowledge.

The wider contextual and cultural background presented within this discussion serves to provide explanatory frameworks for values and positions which are evident within rehabilitation practices in this setting and which underpin how the concept of rehabilitation was understood.

Due to the centrality of the concept to the research aim, the meaning and interpretation of rehabilitation potential has been developed further in this chapter, considering the ambiguity and blurring of the concept and raising challenges and questions for professionals and patients. Many of these questions have begun to frame implications for practice outlined in the concluding section.

The language and examples reported in the findings section pertaining to the ethical dimensions of decision-making led to links with a concept of ethical distress which is

reported in wider literature. In part due to the power of this distress communicated through fieldwork experiences, ethical dimensions have received particular emphasis within this interpretation. Critical questions and challenges have again been posed, such as whether justice can be achieved for older people, and the extent to which good outcomes and improvements can be realised. The chapter concludes with a section which compares and reflects on occupational therapy and physiotherapy roles within rehabilitation decision-making and recognises the value and esteem placed on physiotherapy in this context. Once again this paves the way for important implications for practice discussed in the next chapter.

CHAPTER NINE – CONCLUSION

9.1. Chapter introduction

This research set out to explore health professional's decision-making in relation to the assessment of rehabilitation potential of older people in hospital and recommendation of subsequent rehabilitation pathways. It was motivated by my own previous experience of being involved in such decisions and by a lack of evidence which focussed on understanding real-time decision-making in the context in which it occurs. The focus on acute hospital admissions and rehabilitation pathways for a heterogeneous population of older people was developed when a review of the literature highlighted that such issues were largely underexplored for this group, despite being the largest users of acute hospital services.

Because of these themes, a qualitative study was designed, from a social constructivist stand-point, utilising principles of ethnography and case study research. In order to set some parameters for the purposes of data-collection, one ward was identified as the main-unit of analysis and physiotherapists and occupational therapists were the main rehabilitation professionals focussed on within the study.

9.1. Summary of key findings

It is important, at this concluding stage, to revisit the objectives of the study and discuss how the knowledge and understanding generated can be understood in relation to these objectives.

Objective 1 - To explore the meaning of the concepts of 'rehabilitation' and 'rehabilitation potential' to occupational therapists, physiotherapists and patients

Professionals in this setting recognised that the concept of rehabilitation was influenced by wider social and cultural factors such as the public interest in, and relatively higher profile of, rehabilitation of war veterans and rehabilitation of athletes. This could be suggested to lay a foundation for rehabilitation as a process which improves the strength and function of the body. Within the context of this ward and acute hospital, there were further factors which added layers to this conceptualisation. Set against a backdrop of neo-liberal politics, the NHS Trust was driven by targets – for example optimising flow out of the system. Optimising physical function, and more specifically mobility, was seen as being directly linked to such targets.

Physiotherapy was noted as the dominant professional in delivering rehabilitation, linked to their central domain of concern focussing on improving physical abilities and mobility. However, the causality question of 'chicken and egg' requires consideration here – did a dominant physiotherapy role contribute to mobility being the main focus of rehabilitation, or had the emphasis on mobility centralised the physiotherapy role. My own reflection was that elements of both were evident although the central focus on mobility would likely exist and endure because of many other influential factors.

There are many tensions between this conceptualisation of rehabilitation and the philosophy of rehabilitation professionals, who verbally reported a commitment to a multi-disciplinary, holistic, and patient-centred paradigm but recognised that the version they experienced fell someway short of this ideal. More specifically, this reductionist version of rehabilitation as a process to improve mobility posed particular challenges for occupational therapists, although they were observed to align their practice with this dominant discourse.

Alongside this focus on rehabilitation as a process to facilitate improvements in mobility, and physiotherapists as the primary providers of rehabilitation, rehabilitation was often seen as a place, rather than a phase of care. This was problematic in that patients then waited for rehabilitation, were re-prioritised whilst waiting, and sometimes needed a potentially disruptive move within the system if medical needs became a priority.

Unsurprisingly, the meaning of rehabilitation potential was found to be ambiguous, subjective and open to interpretation in relation to contextual factors. However, the localised examples shine a light on how decisions about rehabilitation potential were being enacted and frequently this was associated with a judgement about the potential of a patient to make gains and improvements in mobility. Importantly, this evaluation was found to be intrinsically linked to high-stakes decisions for patients (such as whether they would move to a rehabilitation bed) and those who made decisions about rehabilitation potential therefore could be seen as adopting a high-stakes, sometimes controversial, gatekeeping role.

In part due to limitations in the methodology, the concepts of rehabilitation and rehabilitation potential from the patient perspective were relatively under-explored. However, through my own exposure to the constructs in real-life and real-time, the terms were not regularly explained to, or used with, patients and families. One of the main manifestations of this was a difference in expectations between professionals and patients of what rehabilitation would entail and what potential could realistically be achieved.

Objective 2 - To map the reasoning process of health professionals involved in evaluating rehabilitation potential and when making decisions about rehabilitation pathways for older people

Professionals did not have any objective criteria to assist in their evaluation of rehabilitation potential and the use of assessment tools (such as a frailty assessment) which may have assisted to recognise those older people who could benefit from, or decline during, a rehabilitative phase of care were also not observed to be in use. Instead of criteria, practitioners were observed to use their own rules to assist them to reach and rationalise decisions. In such instances, subjectivity and individual interpretation were particularly noted.

Decision-making involved a tacit reasoning process which was broadly understood in terms of the following stages: gathering baseline information and carrying out initial assessments; awaiting (or providing) supportive and curative interventions to treat medical issues; providing initial rehabilitative interventions and allowing time to monitor response and carryover; and making an evaluation of likely rehabilitation potential to inform the subsequent pathway.

Engagement with a rehabilitative process to inform a decision about rehabilitation potential was an important part of reasoning, in that professionals valued seeing the response to, and carryover from, early rehabilitative efforts in order to evaluate ongoing potential. The critical challenge here was that there was a perceived absence of 'proper' rehabilitative interventions and a facilitative rehabilitation environment, which could suggest that subsequent evaluations were fundamentally problematic.

The process of evaluating rehabilitation potential involved health professionals managing ethical dimensions and, sometimes competing, ethical demands. Health professionals communicated their commitment to the ethical principle of beneficence
and their desire to bring about a good outcome when making decisions about rehabilitation potential and rehabilitation pathways. However, in the absence of an explicit goal-setting process, this could be seen as paternalistic and based on questionable evidence. Whilst maintaining this desire to bring about positive outcomes, professionals were attempting to balance other ethical dimensions including managing the demand for diminishing rehabilitation resources, and working towards fairness and autonomy for older people.

Because of this, the reasoning process therefore developed reasoning and strategies to manage ethical demands and challenges. Some of these strategies were overt, such as the use of waiting lists and triage, although some were more covert, such as creating narratives about deserving patients or manipulating information giving in order to manage demand. The personal ethical distress suggested in the language of health professional interviews suggested that these tensions were not easy to reconcile.

Objective 3 - To compare the occupational therapy and physiotherapy role in evaluations of rehabilitation potential and decisions about rehabilitation pathways

Despite recognising a reductionist focus on mobility, both occupational therapy and physiotherapy professionals presented as aligning their practice with this version of rehabilitation. Physiotherapy tended to take a lead role in the early stages of evaluating rehabilitation potential and attributed this to their model of working of seeing all patients on the base ward and often being involved at an acute stage. Occupational therapists were observed to get involved at later stages, and sometimes this involvement was limited or absent. The multi-disciplinary team all placed value on physiotherapy evaluations of rehabilitation potential and recommendations of rehabilitation pathways. Multiple factors presented as contributing to this – physical improvement was the main (if sometimes implicit) goal of a rehabilitative process and physiotherapists tended to be the main protagonists in delivering these rehabilitative interventions. And because evaluations of rehabilitation potential tended to rely on information from monitoring such interventions, the importance of physiotherapy in evaluating rehabilitation potential and making recommendations about rehabilitation pathways also emerged. Occupational therapists could not always rely on knowledge from engagement with a rehabilitative process – an important part of evaluating rehabilitation potential – because they were not always involved early enough, and sometimes not at all.

Although both groups were subject to the ethical tensions from systemic pressures, this presented as being particularly felt by occupational therapists who also presented as struggling with issues related to the value of, and respect for their role, and their professional identity.

Objective 4 - To describe influences on decisions regarding rehabilitation potential and subsequent pathways

The evaluation of rehabilitation potential was influenced by a number of different factors in this context. Some factors related to the individual patient, such as baseline function, presence of co-morbidities, cognition and motivation, to name a few. Professionals placed particular emphasis on baseline function although critically information about this was sometimes observed to be incomplete or unclear. The reason baseline function was deemed to be important was that it gave an implicit goal about what the patient would have potential to achieve although during the course of fieldwork, professionals were being subjected to pressure to reframe this goal to focus on achieving safety rather than returning to a baseline norm.

Cognition and motivation were also discussed as being particularly influential. It was not observed that some factors were consistently more influential than others, but instead it was recognised that all factors could become important and influential either in different circumstances or for different individual patients. Again, the potential for subjective interpretation was evident.

The patient perspective of their rehabilitation potential and their preferences about pathways were discussed as being desired and deemed to be important, although in reality perhaps not as influential as other factors. Significantly, the concept of rehabilitation potential was not observed as being explicitly discussed with patients and therefore patient influence on, and involvement in, this evaluation would always be limited by this. Furthermore, patients and families were observed as being involved to varying extents in pathway decisions with those who could proactively vocalise this involvement observed as influencing such decisions in the most obvious ways.

Alongside factors individual to patients, wider factors were also influencing evaluations about rehabilitation potential. Perhaps because the concept of rehabilitation was strongly linked to place, and the concept of rehabilitation potential often was synonymous with the potential to benefit from going to a place or accessing a service, the overlap between decisions about potential and decisions about pathways became blurred. A resultant outcome of this was that decisions about rehabilitation potential became heavily influenced by availability of beds and services.

Objective 5 - To explore how patients and families are involved in evaluations of rehabilitation potential and subsequent recommendations regarding treatment and care.

Principles of shared decision-making and meaningful involvement were discussed as being desired but recognised as being challenging within this context. Barriers noted included the time needed for meaningful involvement and the extent to which patients and families wanted to share decision-making responsibility. Indeed, professionals recognised that time spent communicating with patients and families, and on activities such as involving family members in therapy sessions were prioritised out due to time pressures. This was recognised as being particularly pertinent for older people where such activities often involve greater time.

Earlier parts of this concluding section have acknowledged that decisions tended to be professionally-led rather than patient-centred and often linked to professionals giving patients a 'chance' to make gains through bed-based rehabilitation. Whilst acknowledging such decisions were well-intentioned, the small number of patients tracked during their admission showing lack of improvement (and in some cases functional decline, and death), considered in light of wider evidence which suggests that older people may not achieve functional improvements during hospital and bedbased episodes of care, suggests such reasoning requires critical consideration.

The orientation of acute care towards fast-paced trajectories, managing expenditure by utilising less expensive beds where appropriate, and discharging at the point of safety rather than at the point of achieving optimum function, was experienced as being at odds with an environment which fosters collaborative decision-making and patient and family involvement.

9.3. Implications for practice

The construct of rehabilitation in this setting, mainly as a way to optimise physical functioning, primarily as a professionally-led process headed by physiotherapists, and strongly associated with a movement to a place which may (or may not) facilitate intensity and frequency of professional intervention, has many implications for practice. The social constructionist researcher believes that multiple constructions within the social world are possible although dominant constructions emerge in relation to social utility (Gergen, 2009). It is therefore important to pause on the important question of why is it useful that such constructions of rehabilitation and rehabilitation potential have emerged in this field and what purposes are they serving.

Much of the discussion in this thesis has shone a light on the issues that the construct of rehabilitation as a process to improve mobility, and the blurring of an assessment of rehabilitation potential with an internal transfer, are serving organisations and systems who need to optimise flow and minimise cost whilst potentially failing those who are in need of rehabilitation services. Patient examples of functional decline, limited opportunities for meaningful involvement and the messages of distress communicated in professional extracts are powerful pointers towards incompatibility between desired tenets of rehabilitation and the acute hospital environment. Hammell (2006) suggests that contemporary rehabilitation must challenge assumptions that physical gains represent enhancements in quality of life, and that safety and independence are the desired goals of those living with impairments. She continues that rehabilitation should be more focussed on the endeavour of living, in line with definitions of rehabilitation which encourage attention to holistic needs, adjustment, and interaction with environments.

Fieldwork in this context suggested that rehabilitation professionals recognised that the version of rehabilitation was unsatisfactory for many reasons, and indeed critically,

ways of constructing rehabilitation (such as rehabilitation as a different place) were perhaps serving their own purposes to rationalise deficiencies in their own service provision. Durocher et al (2016) suggest that a first step is to notice, document, and have conversations about such dimensions of practice; something where this thesis can contribute to an important and achievable step forwards.

It was noted that, in part due to rehabilitation being conceptualised as a place, and a place outside of the base ward, patients listed for a different place such as the rehabilitation ward, or intermediate care unit, often became less of a priority for rehabilitative interventions in the current place and phase of care. Although implications of this may have been minimised through recent changes in this site to co-locate medical and rehabilitation phases, dissemination of these findings should aim to assist other practitioners and services to recognise when this may be happening. The risks associated with this – including increasing length of stay, increasing likelihood of functional decline, and compromising the patient and family experience of care are significant to note. It is areas such as this where practitioners can aim to influence change without delay.

The interrelated issues of autonomy, patient involvement and collaboration would benefit from attention in practice on many levels. At the level of individual patients, receiving clear information in an accessible way about what rehabilitation may mean and involve and what options for rehabilitation are available, is an obvious and achievable foundation step and would assist in the management of expectations of both patients and families. It is proposed that the simple 4-stage model (presented on page 123) could form the basis of involving patients and families in dialogue about rehabilitation decision-making. It could also provide a framework for documenting reasoning – something which was notably unclear in this setting. Part of this could also include developing a clearer process for obtaining and documenting information about

a patient's baseline, which was assumed to be embedded within practice within this context but would benefit from immediate attention.

Alongside this, honest conversations about what older people would like to work towards to enhance their quality of life, how realistic this may be and which services (if any) may be best placed to facilitate these enhancements would again be important steps in moving towards a different version of rehabilitation with less emphasis on assumed or implicit goals and less controlled by professional decisions. The most commonly reported area to improve goal-setting practices suggested by research is to introduce an education element prior to setting-goals, so patients and families understand the purpose of goal-setting and can participate in the process in a meaningful way (Rose, Rosewilliam and Soundy, 2017).

Although goal-setting conversations could happen on an individual level with an older person following an acute admission, there are many valid reasons why this may not be happening, leading to challenges with implementing this in practice. Indeed professionals know that this is a desirable element of a rehabilitative phase of care and simply reminding or restating is unlikely to change the status quo. Once again, there is a salient reminder here that acute hospital care is perhaps not the most appropriate or facilitative environment for such conversations, and that by the time of an acute admission, the older person may not be in the optimum position to engage.

Therefore, a wider and more creative solution could be to develop the model of advance care planning, adopted already in many places for people with life-limiting diagnoses, for older people in general. This has the potential to provide a framework for honest discussion about ageing, including elements such as when bed-based or hospital care may be required, and how individuals would evaluate their own quality of life. This in itself could contribute to moving towards desired central tenets of

rehabilitation, tailoring interventions to individuals and placing individual goals at the heart of the process.

Many of these implications are fundamentally linked to incongruence between the orientation of an acute hospital setting, and the provision of rehabilitative phases of care for older people. The biomedical model of illness is recognised as dominating acute hospital systems, sustained by neo-liberal politics of targets and efficiencies for episodes of diagnosis and treatment. However, the biomedical model is increasingly discussed as being fundamentally flawed when considering complexity and chronicity of health and wellbeing needs, with arguments for radical change and a move towards a biopsychosocial model growing in strength (Wade and Halligan, 2017).

Whilst recognising that much of this is not easy to change or influence, dialogue that recognises this incongruence and the way that the practice of rehabilitation professionals is being moulded by this is again, an important point of action and reflection. Occupational therapists and physiotherapists – alongside the professionals they work alongside - need support to recognise how their roles have been shaped by contemporary organisational drivers, to recognise the professional artistry that has evolved in response to this and to continue to work towards practice which is aligned with the core values of their professions. Both professions can contribute to leading and championing local change which can potentially inform wider reform. Although pertinent across both groups, this was felt to be particularly salient for occupational therapists who would benefit from developing ways to promote their identity and value in this acute hospital context – an issue noted in this setting but echoed in wider research (Britton, Rosenwax & McNamara, 2015; Wilding and Whiteford, 2009).

The ethical challenges and distress experienced by occupational therapists and physiotherapists as part of their everyday roles – striving for ideals of rehabilitation, acting as gatekeepers for rehabilitation resources, and providing rehabilitative

interventions in the context of highly pressured services – provided important insights in to the realities of practice. Interestingly, although this was implied during earlier stages of fieldwork, this was mainly communicated during interviews with health professionals. This suggests that the creation of reflective spaces helps to give voice to such tensions and realities. The use of clinical narratives, reflection and debriefing have been discussed as ways to manage the effects of ethical tensions, with the suggestion that such processes will increase clarity, help to proactively recognise ethical issues before the demands feel unmanageable, and recognise sources of support (Erler, 2017). Whilst taking time out for reflection on ethical issues may feel counter-intuitive within busy acute environments, it is seen as an essential part of professional practice and one which has direct links to quality of patient care. Managers and educators also play an important role in creating cultures of reflective practice and preparing professionals with strategies to manage ethical challenges in practice.

Space and time for reflective practice could also link to a stronger emphasis on evidence-based practice amongst rehabilitation professionals. Although many elements of practice and decision-making appear to have been linked implicitly to evidence, explicit articulation of such links were notably absent. More explicit use of evidence to underpin dialogue with patients (for example evidence informing positive outcomes or those most likely to benefit), or to encourage new ways of working (for example use of goal-setting approaches or frailty measures) could lead to important service improvements. Equally, collation of evidence to oppose the organisational rhetoric of aiming for safety rather than optimum function could help to challenge unhelpful discourses and assert professional autonomy.

Whilst reflective and evidence-based practice at the level of individuals and teams is an important strategy, practitioners would benefit from wider support from professional bodies. Ethical codes of conduct could be strengthened to provide guidance on the

professional role in making sustainable decisions within the realities of competing and finite resources, to provide constructive support to practitioners in their gatekeeping roles, and to more directly situate professional reasoning in a context of collaboration and partnership with service users.

Recent WHO communications – for example 'Rehabilitation in Health Systems' (2017) and 'Rehabilitation 2030 – a call for action' (2017) - have gone some way to provide an international reminder of the importance of rehabilitation in meeting the contemporary health and social care needs of global populations. However, the emphasis is mainly on the challenges faced by low- and middle-income countries where health investment has been historically poor. Although these challenges are undisputable, the localised reflections above, coupled with the challenging landscape of rehabilitation for older people in the UK, suggest that there are many similar and many different questions facing practice closer to home. The recent NICE guidance on 'Intermediate care Including Reablement' (2017) goes some way to address this although the guidance on rehabilitation in acute care for older people could be a specific area of focus for policy-makers and strategists.

Due to the significance of the concept to the research aim, I return to the term 'rehabilitation potential' and it's place within professional discourse. In basic terms, the concept was not explained to patients, was linked to ethical distress for professionals and was recognised as ambiguous, subjectively interpreted and poorly documented. It is therefore a logical outcome to question the relevance of the term at all. Rather than ongoing rumination on the term rehabilitation potential and its associated evaluation, emphasis could simply be placed on honest communication (including documentation) to collaborate about what people want and need to be able to do, whether all parties think such goals are realistic, and which services are available (if any) to work towards any goals. Whilst influencing some of the wider implications for practice may feel

overwhelming or unmanageable, this simple conclusion could lead to meaningful cultural change.

9.4. Strengths and Limitations of the current study

The immersive ethnographic methodology, placing importance on the social construction of meaning and subsequent decision-making, has enabled examination of the concepts of rehabilitation and rehabilitation potential within the context in which they are formed, and is proposed as a major strength of this study. This is in comparison with wider literature on the topic which has examined the issues through methods which encourage retrospective analysis or discussing the concepts outside of the everyday influences of healthcare practice. An additional strength includes the intentional focus given to occupational therapists and physiotherapists, whose practice and identity is interwoven with rehabilitation ideology and yet where a specific focus and emphasis within research enquiry relating to rehabilitation potential is relatively under-developed. Finally, the importance placed on examining these issues in relation to the non-uniform pathways and heterogeneous population of older people is seen as another important strength of the study and one where extensions to knowledge and understanding are of vital importance.

Limitations of this thesis must begin by acknowledging criticisms of ethnography as a methodology with those key features often presented as strengths, illustrated as weaknesses by others perhaps in equal measures. For example, for some the principle of honing in on focussed areas (in this instance a ward and a small group of professionals and patients) can be reflected as gaining in-depth understanding of a localised context whilst for others it represents small-scale research which cannot be generalised. Equally, the ability to use flexible data gathering methods is highlighted as

an ability to use all methods available to throw light on emerging and salient issues, whilst for others this is seen as unscientific, difficult to replicate and heavily open to researcher bias. And with all forms of qualitative enquiry, it is recognised that data gathering, analysis and subsequent interpretive discussions are my own constructions which are open to challenge. Whilst not contesting that these issues represent real limitations, I have discussed and appraised many elements of decision-making in relation to the choice of method in previous chapters and presented strategies to enhance quality and rigour.

One of the key defining features of ethnography is proposed as prolonged periods of fieldwork and observation (Savage, 2000). Although data collection spanned approximately 13 months, actual points of data gathering involved much shorter time scales with an introductory first phase of 2 weeks, phase 2 fieldwork lasting approximately 8 weeks, and phase 3 involving 5 in-depth interviews within one month and I do feel that longer periods for all phases would have enabled further and deeper understandings in all areas. Also, time gaps between phases perhaps meant that opportunities were missed to gain detailed understanding of some of the organisational changes which took place during this time and potentially changed relationships with participants. The reality was that pragmatic factors of time and workload played a significant part.

Another limitation was the limited representation of the patient perspective in this research, something discussed in earlier sections. Meaningful service user involvement represented a challenge on a number of levels – both through my limited ability to collaborate with patients and carers in a meaningful way as research participants, and through the limited involvement of service users in the wider construction of the research. An important element of my planned activities included informal and formal opportunities to talk to patients and families about aspects of rehabilitation and rehabilitation decision-making. However in reality, I quickly became aware that these

concepts were not obviously being discussed with, or translated for, patients (discussed in more detail in previous chapters) which resulted in discussion about these elements being problematic. Linked to this, my interactions with patients instead became characterised by short and superficial interactions asking what had happened on a particular day or asking if anything had changed. In a recent descriptive study of interactions between patients and health professionals the median interaction was just 36 seconds (Barker et al, 2016) and I was perhaps also being influenced by dominant ways of working in this setting. Also, regarding more formal communication, I had proposed to carry out individual interviews towards the end of an episode of care although these were either declined (in two cases) or were not possible due to patient death (in three cases – summarised in table 2 on page 77). For some of these reasons, the patient and family voice and experience was perhaps under-represented during phases of fieldwork. However, the body of knowledge presented in this thesis would have been richer and more deeply layered with improvements in this area.

Some areas of focus did not explicitly emerge until later stages of analysis, such as the far-reaching influence of the biomedical model of illness and the ethical distress experienced through systemic challenges in practice such as gatekeeping responsibilities. It could be argued that this in itself represents a strength of the research, that I remained open to new perspectives and emerging themes. However, I am also aware that opportunities were missed, particularly during in-depth interviews, to explore these issues more explicitly with those experiencing them.

9.5. Dissemination of findings

The findings of this study are most obviously relevant to those within the field, therefore the first stage of dissemination will be to share the research findings with those who participated in this study. Patients and relatives who indicated an interest in receiving follow-up information will be sent a summary of findings. I will also work with the clinical teams involved to identify the best way to disseminate and discuss findings – including the offer of written summaries, presentations, and reflective workshops.

In my role as an occupational therapy lecturer, I am also keen to identify ways to disseminate findings both within pre-registration and post-qualifying education. Using research findings to prepare students and professionals for the realities of decisionmaking in rehabilitation within acute care, alongside encouraging critical discourse to constructively challenge models of practice and service delivery, could provide important opportunities for professional and service development.

I have also been encouraged by the interest of my immediate supervision team and the interested voices of wider networks in the knowledge which has emerged from this project. Specific topics from the research findings which could potentially become the focus of future publications or conference presentations include the meaning of rehabilitation in acute hospital contexts and working towards justice for older people in acute care. Learning from the process of undertaking this research could also be disseminated, including sharing experiences of being a novice ethnographer in an acute context and challenges to meaningful patient and family involvement in research situated in acute hospitals.

9.6. Recommendations for future research

This research focussed on the meaning of rehabilitation potential as influenced by the context in which it was constructed, and the evaluation of rehabilitation potential as experienced and constructed by occupational therapists and physiotherapists in their daily practice. It did not examine effectiveness of any rehabilitation interventions, or whether judgements about rehabilitation potential were accurate. Research to examine efficacy of interventions and which patients will benefit most are questions of ongoing importance and priority. This is particularly significant when considering that even in contemporary policy examples – such as the NICE Guideline on Intermediate Care and Reablement (2017), many recommendations are underpinned from research in to specialisms such as orthopaedics and stroke.

Work to develop and evaluate criteria or assessments which help to move towards more objective and standardised professional decision-making when considering the allocation of rehabilitation resources for older people is another related area of interest. Another area of note is research to evaluate different models of rehabilitation for older people from the perspective of the patient or service user, which could be particularly relevant within this local context where services are being reconfigured.

Although advance care planning is proposed as a process for any person to have prospective discussions about future health and care needs with family members and health professionals (Sudore et al, 2017), in practice the focus tends to be on discussions about life-sustaining treatments, and much of the existing literature focuses on particular clinical groups, often with established life-limiting conditions, such as Dementia or Cancer. Use with healthy populations in general, and more specifically with well older people who can still communicate their wishes and goals is less understood and has been challenged due to the changing nature of preferences and needs during episodes of illness or when adjusting to new disability. However, an

advance care planning model which focuses on issues such as wishes for hospital admissions, goals for rehabilitation and realistically, moving forwards, which services people would wish to pay for, would add significantly to current understanding and would benefit from research to assist with development and evaluation. Whether the label of advance care planning has now become synonymous with end of life care and terminal illness requires critical consideration when considering how this framework could support discussions and decision-making amongst healthy populations.

Rehabilitation as a place was noted by this research, in the sense that rehabilitative phases of care were often initiated when a patient was moved to a ward or bed outside of the base ward. Practitioners also communicated a belief that facilities were related to the creation of a rehabilitation ethos and the interaction with physical space could be seen to influence elements of role development and decision-making (for example the time spent on the base ward by physiotherapists in comparison to occupational therapists). Hospitals have been identified as 'extraordinary places' due to being outside of everyday geographies which are associated with catastrophic events (Kontos et al, 2015; p120). Research to understand the influence of physical space, built environment and facilities on the ethos of rehabilitation for older people, rehabilitation roles and decision-making could provide ways to develop ethnographic understanding on these topics. This is particularly significant in this local hospital site and in acute hospital systems in general, where functions of wards are frequently reconfigured (supported by the extract on page 155 'when we were a rehab ward'), although physical environments often remains static.

Although professionals shared their feelings of ethical distress, the sharing of consequences and strategies were perhaps less obvious and forthcoming. In simple terms, this could be because I was not drawing focus to this issue or asking direct questions during fieldwork or interviews and this could become a primary focus of future research. However, another reflection could be that whilst practitioners may be

able to recognise and describe ethical dilemmas and distress, they may lack the vocabulary to articulate a deeper analysis of the situation, or may struggle to acknowledge some of their responses to minimising tensions and distress (Carpenter, 2010). Participatory research methods to use and evaluate approaches to assist with this could again be an interesting area of further enquiry.

Alongside this, because of the breadth of the current study, other areas emerged as pertinent although received limited focus within the thematic analysis and subsequent discussion. Understanding the labelling of patients was an interesting, yet small area highlighted within the findings. Equally, hierarchies within rehabilitation teams and the impact on rehabilitation decision-making, and the professional identify of rehabilitation professionals were both smaller themes in a broader presentation of research findings and are again areas of significant interest which could receive more direct and detailed focus in future research.

9.7. Final reflections on the research journey

At different points in the writing of this thesis, I have reflected on the concept of 'self' and how my own experiences were shaping the direction, and indeed integrity, of this research. It is therefore important to conclude with my own final reflections on this personal journey.

One of my main challenges was the journey through ethical approval; balancing the unpredictability and serendipity of ethnography with clear and planned processes for the purposes of research governance was a difficult reality. I can understand why this may discourage ethnographers, and particularly novice ethnographers, from planning this type of research within the social spaces of health and care environments. During the process of ethical approval, I was encouraged to reconsider the involvement of

people who may lack capacity to consent to participate. I continue to reflect that strengthening my understanding of involving people who lack capacity in research in order to gain ethical approval for this element was one of my main achievements and provided some of the richest learning throughout the whole journey. The reality of this study was that this deep planning and learning only applied to the involvement of one patient, although I hope the learning will add long-lasting value to my research career and journey in its entirety.

I have been driven from the outset by a desire for this work to be useful and interesting, a motivation recognised within social scientific research (Michael, 2012). I have frequently questioned whether this could ever be the case: with the context so specific and localised; my interest in a population who, it could be perceived, receive less than favourable attention; and my interest in smaller professional disciplines perhaps at the exclusion of other more dominant groups. Something I have only reflected on at a late stage is that, throughout the research journey, I have managed to communicate to others that the work *is* useful and interesting – both as part of the doctorate approval and progression processes, but perhaps more importantly, to participants, stakeholders and gatekeepers.

My supervisors played an important role in continuing to reassure me that the work held value, interest and utility. But alongside this, they also encouraged me to think critically about my judgements and assumptions about what I thought to be useful and interesting. Supervision, particularly during phases of fieldwork, and then subsequently during data analysis encouraged me to look and relook at things I had perhaps disregarded as of little interest. More realistically, I had perhaps begun to think of things as failures, deviances, or incomplete pictures and may have been tempted to bury them or pass them by. Supervision meetings while I was active within the field, and then sharing anonymised fieldnotes and transcripts were central to this process. I will

certainly remember and apply this learning in future roles within research teams and when contributing to research supervision.

I feel I experienced ongoing tension between theoretical and philosophical research principles and delivering real-world research. Such examples are reflected in earlier sections: for example my stated commitment to immersive research methods although my actual short, and sometimes abrupt, research interactions; and my commitment to public and patient involvement although the frustrations in achieving this in a meaningful way. This second example has been particularly difficult to reconcile, particularly during the process of engagement with critical theory emphasising the need for collaborative approaches to rehabilitation. I feel I was guilty of assuming that there would be recognised ways to facilitate public and patient involvement within the Trust and I would be able to link my own involvement agenda seamlessly to this. However, evidence suggests there are challenges to involvement at all levels – not just at the level of individual researchers - and in future research I will aim to establish much clearer mechanisms for meaningful involvement at the outset rather than making assumptions that established mechanisms will either be effective or indeed exist.

This personal experience of translating theoretical and philosophical principles in to the real world is perhaps mirrored in the experiences of participants within the field. That is to say, practitioners expressed theoretical and philosophical principles and values relating to rehabilitation and rehabilitation decision-making although expressed frustrations and constraints with their ability to enact these principles and live out these values. In this sense, I feel a sense of empathy with how the realities of an acute hospital system can constrain practice – irrespective of whether this practice is research or clinically orientated.

Looking ahead, I feel this can potentially provide me with invaluable learning for some of the challenges of my research journey which are yet to come. When reporting and

disseminating ethnographic research, researchers are often perceived to have been in a position of privilege and power and are then seen to be commenting on, and sometimes judging, a social world they have only experienced for a short time and potentially in a one-dimensional way. However, I feel this essence of shared experience can offer me an important starting point and act as the foundation for a collaborative partnership with practitioners when looking to future practice development and opportunities for research.

9.8. Final Conclusions

In conclusion, this study has prioritised understanding rehabilitation pathways for older people as a core issue for research and professional dialogue, recognising the vital contribution that such pathways make to the health and wellbeing of older people as they recover from episodes of ill-health and as they experience and adjust to increasing frailty. Specifically, the research has examined the meaning of rehabilitation and rehabilitation potential in relation to older people in hospitals, through an immersive methodology to examine the concepts within the social, temporal and organisational context in which they are constructed. Evaluations of rehabilitation potential for older people are primarily understood in relation to the likelihood of physical improvements and the evaluation is often driven by well-intentioned paternalism, both of which are linked to the dominant discourse of acute hospital settings. Despite being a regular and taken-for-granted part of professional discourse, reasoning is poorly articulated both within professional-patient and inter-professional interactions and poorly documented within clinical records. Dissemination of the research findings will aim to highlight such issues as important areas for service and professional development.

REFERENCES

Alaszewski, A. (2006) Using Diaries for Social Research. London: Sage.

Allen, D., Griffiths, L. & Lyne, P. (2004) 'Accommodating health and social care needs: routine resource allocation in stroke rehabilitation'. *Sociology of Health and Illness*. Available online at: <u>https://onlinelibrary.wiley.com/doi/full/10.1111/j.0141-9889.2004.00397.x. Accessed on 26.4.18</u>.

Alvesson, M. and Skoldberg, K. (2010) *Reflexive methodology: New Vistas for Qualitative Research - 2nd edition*. London: Sage Publications.

Al-Qurainy, R., Collis, E. & Feuer, D. (2009) 'Dying in an acute hospital setting: the challenges and solutions'. *The International Journal of Clinical Practice*. Available online at: <u>http://onlinelibrary.wiley.com/doi/10.1111/j.1742-1241.2008.01991.x/abstract</u>. Accessed on 5.12. 17.

Ambrosino, N. & Foglio, K. (1996) 'Selection criteria for pulmonary rehabilitation'. *Respiratory Medicine*. Available online at: <u>https://ac.els-</u> <u>cdn.com/S0954611196901250/1-s2.0-S0954611196901250-main.pdf?_tid=5d7e84da-66a6-435e-8e26-</u> <u>64322829cab4&acdnat=1524818906_136826f1e42e78df51de7345e348a274</u>. Accessed on 27.4.18.

Anderson, G. (2011) The challenge of financing care for individuals with multimorbidities. In: Health Reform: Meeting the challenge of ageing and multiple morbidities. Organisation for Economic Co-operation and Development. Available online at: <u>http://dx.doi.org/10.1787/9789264122314-6-en</u>. Accessed on 26.4.18

Anderson, L., Sharp, G., Norton, R., Dalal, H., Dean, S., Jolly, K., Cowie, A., Zawada, A. & Taylor, R (2017) 'Home-based versus centre-based cardiac rehabilitation'. *Cochrane database of systematic reviews*. Available online at: <u>http://cochranelibrary-wiley.com/doi/10.1002/14651858.CD007130.pub4/full</u>. Accessed on 26.4.18.

Ariss, S. (2015) *National audit of intermediate care: patient reported experiences,* 2015. Available online at: <u>https://www.scie-socialcareonline.org.uk/national-audit-of-intermediate-care-patient-reported-experiences-2015-full-report-of/r/a11G000000CTDEMIA5. Accessed on 5.12.17.</u>

Arling, G., Williams, A. & Kopp, D. (2000) 'Therapy use and discharge outcomes for elderly nursing home residents. *The Gerentologist*: 40 (5); 587-595.

Atwal, A., Tattersall, K., Murphy, S., Davenport, N., Craik, C., Caldwell, K., McIntyre, A. (2007) 'Older adults experiences of rehabilitation in acute health care'. *Scandinavian Journal of Caring Sciences*: 21 (3); 371-378.

Atwal, A., Wiggett, C. & McIntyre, A. (2011) 'Risks with older adults in acute care settings: Occupational Therapists' and Physiotherapists' perceptions'. *British Journal of Occupational Therapy*. Available online at:

http://journals.sagepub.com/doi/pdf/10.4276/030802211X13153015305510. Accessed on 27.4.18.

Bachmann, S., Finger, C., Huss, A., Egger, M., Stuck, A., Clough-Gorr, K. (2010) 'Inpatient rehabilitation specifically designed for geriatric patients: systematic review and meta-analysis of randomised controlled trials'. *British Medical Journal*. Available online at: <u>https://www.bmj.com/content/bmj/340/bmj.c1718.full.pdf</u>. Accessed on 26.4.18.

Bail, K. & Grealish, L. (2016) "Failure to Maintain': A theoretical proposition for a new quality indicator of nursing care rationing for complex older people in hospital'. *International Journal of Nursing Studies*. Available online at: <u>https://ac.els-</u>cdn.com/S002074891630116X/1-s2.0-S002074891630116X-main.pdf?_tid=a43626be-bb0f-11e7-8e24-00000aacb361&acdnat=1509106331_704c2ca8c741ef27eaadde611add89df. Accessed on 27.10.17

Ballard, C. (2006). *Presentation on Alzheimers drugs to in house of commons*. London: Alzheimer's Society.

Barbour, R. (2014) *Introducing qualitative research: a student's guide*. Los Angeles: Sage Publications.

Barnitt, R., Warbey, J. & Rawlins, S. (1998) 'Two case discussions of ethics: Editing the truth and the right to resources'. *British Journal of Occupational Therapy*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/030802269806100201</u>. Accessed on 15.12.17.

Barker, H., Griffiths, P., Mesa-Eguiagaray, I., Pickering, R., Gould, L. & Bridges, J. (2016) 'Quantity and quality of interaction between staff and older patients in UK hospital wards: A descriptive study'. *International Journal of Nursing Studies*. Available online at: <u>https://ac.els-cdn.com/S002074891630102X/1-s2.0-S002074891630102X-main.pdf</u>? tid=effaf5f4-fb7c-11e7-a555-

00000aab0f6b&acdnat=1516190156_19e6e0f416c0b58b8a5d91cfb87f164a. Accessed on 17.1.18.

Bastiaens, H., Van Royen, P., Rotar Pavlic, D., Raposo, V., Baker, R. (2007) 'Older people's preferences for involvement in their own care: A qualitative study in primary health care in 11 European countries'. *Patient Education and Counselling*. Available at: <u>http://ac.els-cdn.com/S0738399107001449/1-s2.0-S0738399107001449-main.pdf?_tid=90b86a80-4889-11e4-9a44-</u>00000aab0f27&acdnat=1412071820_6e4e6c3b39988e48b4cbe3c4ed794f89.

Accessed on 23.12.14.

Baumbusch, J. (2010) 'Conducting critical ethnography in long-term residential care: experiences of a novice researcher in the field'. *Journal of Advanced Nursing*. Available online at: <u>http://onlinelibrary.wiley.com/doi/10.1111/j.1365-2648.2010.05413.x/full</u>. Accessed on 18.1.18.

Bahar-Fuchs, A., Clare, L. & Woods, B. (2013) Cognitive training and cognitive rehabilitation for mild to moderate Alzheimer's disease and vascular dementia. *Cochrane Database of Systematic Reviews*. Available online at: <u>https://www.cochranelibrary.com/cdsr/doi/10.1002/14651858.CD003260.pub2/epdf/full</u>. Accessed on 19.11.18.

Beauchamp, T. & Childress, J. (1989) *Principles of biomedical ethics.* Oxford: Oxford University Press.

Becker, H. S., Geer, B., Hughes, E. C. and Strauss, A. L. (1961) *Boys in white: student culture in medical school.* Chicago: University of Chicago Press.

Blackmer, J. (2000) 'Ethical issues in rehabilitation medicine'. *Scandinavian Journal of Rehabilitation Medicine*. 31: 51-55.

Bloor, M. (2001) 'The Ethnography of Health and Medicine' In: Atkinson, P., Coffey, A., Delamont, S., Lofland, J. & Lofland L. (Eds) (2001) *Handbook of Ethnography*. London: Sage Publications Ltd.

Blumer, H (1969) Symbolic Interactionism: Perspective and Method. *Englewood Cliffs, NJ: Prentice Hall.*

Bourdieu, P. (1984) *Distinction: a social critique of the judgement of taste*. London: Routledge.

Bourdieu, P. (1989) 'Social Space and Symbolic Power'. *Sociological Theory*. 7(1); p14-25.

Bourdieu, P. (1993) *The field of cultural production: essays on art and literature.* Cambridge: Polity Press.

Bowling, A. (2009) *Research methods in health: Investigating health and health services – 3rd edition*. Maidenhead: Open University Press.

Braun, V. & Clarke, V. (2006) 'Using thematic analysis in psychology'. *Qualitative Research in Psychology*: 3(2); 77-101.

Brewer, J. (2000) Ethnography. Buckingham: Open University Press.

Bridges, J., Flatley, M. & Meyer, J. (2010) 'Older People's and relatives' experiences in acute care settings: systematic review and synthesis of qualitative studies'. *International Journal of Nursing Studies*. Available online at: <u>https://ac.els-cdn.com/S0020748909003162/1-s2.0-S0020748909003162-main.pdf?_tid=f6eade39-6dde-46a0-89a9-</u>

<u>5bcc47e87332&acdnat=1524829885_bb8c9a192148e7718f72f340bb589b2a</u>. Accessed on 26.4.18.

Bright, F., Kayes, N., Worrall, L., McPherson, K. (2015) 'A conceptual review of engagement in healthcare and rehabilitation'. *Disability and Rehabilitation*. Available online at:

https://www.tandfonline.com/doi/pdf/10.3109/09638288.2014.933899?needAccess=tru e. Accessed on 26.4.18.

British Society of Rehabilitation Medicine (2017) *About the British Society of Rehabilitation Medicine*. Available online at: <u>https://www.bsrm.org.uk/about-us/about</u>. Accessed on 15.12.7.

Britton, L., Rosenwax, L., McNamara, B. (2015) 'Occupational therapy practice in acute physical hospital settings: evidence from a scoping review'. *Australian Occupational*

Therapy Journal. Available at: <u>http://onlinelibrary.wiley.com/doi/10.1111/1440-1630.12227/epdf</u>. Accessed on 20.10.17.

Brown, L., Forster, A., Young, J., Crocker, T., Benham, A., Langhorne, P. (2015) 'Medical day hospital care for older people versus alternative forms of care'. *Cochrane Database of Systematic Reviews.* Available online at:

http://onlinelibrary.wiley.com/doi/10.1002/14651858.CD001730.pub3/full. Accessed on 20.10.17.

Burr, V. (2003) Social Constructionism. London: Routledge.

Burton, C., Horne, M., Woodward-Nutt, K., Bowen, A., & Tyrrell, P. (2015) 'What is rehabilitation potential? Development of a theoretical model through the accounts of healthcare professionals working in acute rehabilitation services'. *Disability and Rehabilitation.* Available online at:

https://www.tandfonline.com/doi/pdf/10.3109/09638288.2014.991454?needAccess=tru e. Accessed on 26.4.18.

Bushby, K., Chan, J., Druif, S., Ho, K. & Kinsella, E. (2015) 'Ethical tensions in occupational therapy practice: A scoping review'. *British Journal of Occupational Therapy*: 78 (4); 212-221.

Carpenter, C. (2010) 'Moral distress in physiotherapy practice'. *Physiotherapy Theory and Practice:* 26 (2); 69-78.

Chambliss, D. (1996) *Beyond Caring: Hospitals, Nurses and the Social Organization of Ethics*. Chicago: University of Chicago Press.

Chalk, D. & Pitt, M. (2015) 'Fractured neck of femur patients: Rehabilitation and the acute hospita'l. *British Journal of Healthcare Management*. Available online at: <u>https://www.magonlinelibrary.com/doi/full/10.12968/bjhc.2015.21.3.146</u>. Accessed on 30.4.18.

Chan, D., Levi, C., Cordato, D., O'Rourke, F., Chen, J., Redmond, H., Ying-Hua, X., Middleton, S., Pollack, M. & Hankey, G. (2014) 'Health service management study for stroke: a randomized controlled trial to evaluate two models of stroke care'. *International Journal of Stroke*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1111/ijs.12240</u>. Accessed on 27.4.18.

Chang, Y., Peng, L., Lin, M., Lai, H., Chen, L., Hwang, S. & Lan, C. (2011) 'Who determines the rehabilitation needs of care home residents? An observational survey'. *Archives of Gerentology and Geriatrics*: 52 (2); 138-141.

Chapman, G. & Sonnenberg, F. (2000) 'Introduction'. In Chapman, G. & Sonnenberg, F. (Eds) *Decision making in health care: theory, psychology and applications*. Cambridge: Cambridge University Press, 3-19.

Chapparo, C. & Ranka, J. (2008) 'Clinical reasoning in occupational therapy'. In: Higgs, J., Jones, M., Loftus, S. & Christensen, N. (2008) (Eds) *Clinical Reasoning in the Health Professions*. Boston: Elsevier, 265-277.

Charles, C., Gafni, A. & Whelan, T. (1997) 'Shared decision-making in the medical encounter: What does it mean? (or it takes at least two to tango)'. *Social Science and Medicine*: 44 (5); 681-692.

Charmaz, K. (2006) *Constructing grounded theory: A practical guide through qualitative analysis*. London: Sage Publications.

Charmaz, K. & Mitchell, R. (2001) 'Grounded Theory in Ethnography'. In: Atkinson, P., Coffey, A., Delamont, S., Lofland, J. & Lofland L. (Eds) (2001) *Handbook of Ethnography*. London: Sage Publications Ltd, 160-174.

Chartered Society of Physiotherapy (2011) *Code of Members' Professional Values and Behaviour*. Available online at: <u>http://www.csp.org.uk/publications/code-members-professional-values-behaviour</u>. Accessed on 5.12.17.

Clemens, E. & Hayes, H. (1997) 'Assessing and balancing elder risk, safety and autonomy: decision making practices of health care professionals'. *Home Health Care Services Quarterly*: 16 (3); 3-20.

Clouser, K. & Gert, B. (1990) 'A critique of principlism'. *The Journal of Medicine and Philosophy*: 15 (2); 219-236.

Coffey, A. (1999) *The ethnographic self: fieldwork and the representation of identity*. London: Sage Publications.

Colquhoun, H., Lamontagne, M., Duncan, E., Fiander, M., Champagne, C. & Grimshaw, J. (2016) 'A systematic review of interventions to increase the use of standardised outcome measures by rehabilitation professionals'. *Clinical Rehabilitation*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/0269215516644309</u>. Accessed on 22.12.17

Conroy, B., Dejong, G. & Horn, S. (2009) 'Hospital-based stroke rehabilitation in the United States'. *Topics in Stroke Rehabilitation*: 16 (1); 34-43.

Cooke, M., Higgins, J. & Kidd, P. (2003) 'Use of emergency observation and assessment wards: a systematic literature review'. *Emergency Medicine Journal*. Available online at: <u>http://emj.bmj.com/content/emermed/20/2/138.full.pdf</u>. Accessed on 26.4.18.

Covinsky, K., Palmer, R., Fortinksky, R., Counsell, S., Stewart, A., Kresevic, D., Burant, C. & Landefeld, S. (2003) 'Loss of independence in activities of daily living in older adults hospitalised with medical illnesses: Increased vulnerability with age'. *Journal of the American Geriatrics Society*. Available online at: <u>https://s3.amazonaws.com/academia.edu.documents/43792915/Loss_of_Independenc_e_in_Activities_of_Da20160316-15717-</u> <u>1gerech.pdf?AWSAccessKeyId=AKIAIWOWYYGZ2Y53UL3A&Expires=1524135088&</u> <u>Signature=m0K8I2deMMw2VG8FKh%2F3LRmvCfg%3D&response-content-</u> <u>disposition=inline%3B%20filename%3DLoss_of_Independencc_in_Activities_of_Da.pd</u> <u>f. Accessed on 7.7.18</u>

Cresswell, J. (2009) *Research Design: Qualitative, quantitative and mixed methods approaches.* London: Sage Publications Ltd.

Cresswell, J. & Poth, C. (2018) *Qualitative Inquiry and Research Design: Choosing Among Five Approaches – 4th Edition.* London: Sage Publications Ltd.

Crocker, T., Forster, A., Young, J., Brown. L., Ozer, S., Smith, J., Green, J., Hardy, J., Burns, E., Glidewell, E. & Greenwood, D. (2013) 'Physical rehabilitation for older people in long-term care'. *Cochrane Database of Systematic Reviews*. Available online at:

https://www.cochranelibrary.com/cdsr/doi/10.1002/14651858.CD004294.pub3/epdf/full. Accessed on 19.11.18.Crotty, M. (1998) *The Foundations of Social Research: meaning and perspective in the research process.* London: Sage Publications.

Cullen, N., Meyer, M., Aubut, J., Bayley, M., Serrato, J. & Teasell, R. (2014) *Efficacy* and models of care following an acquired brain injury. London: Parkwood Institute.

Cunliffe, A., Gladman, J., Husbands, S., Miller, P., Dewey, M. & Harwood, R. (2004) 'Sooner and healthier: a randomised controlled trial and interview study of an early discharge rehabilitation service for older people'. *Age and ageing*. Available online at: <u>https://academic.oup.com/ageing/article/33/3/246/21165</u>. Accessed on 25.4.18.

Cunningham, C., Horgan, F. & O'Neill (2000) 'Clinical assessment of rehabilitation potential of the older patient: a pilot study'. *Clinical Rehabilitation*: 14; 205-207.

Daniels, N. & Sabin, J. (1998) 'The ethics of accountability in managed care reform'. *Health affairs*. Available online at:

https://search.proquest.com/docview/204622352/fulltextPDF/E46792AD39024A13PQ/1 ?accountid=12860. Accessed on 19.6.18.

Darby, J., Williamson, T., Logan, P. & Gladman, J. (2017) 'Comprehensive geriatric assessment on an acute medical unit: a qualitative study of older people's and informal carer's perspectives of the care and treatment received. *Clinical Rehabilitation*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/0269215515624134</u>. Accessed on 28.12.17.

das Nair, R., Cogger H., Worthington, E. & Lincoln, N. (2016) 'Cognitive rehabilitation for memory deficits after stroke'. Cochrane Database of Systematic Reviews. Available online at:

https://www.cochranelibrary.com/cdsr/doi/10.1002/14651858.CD002293.pub3/epdf/full. Accessed on 19.11.18.

D'Cruz, K., Unsworth, C., Roberts, K., Morarty, J., Turner-Stokes, L., Wellington-Boyd, A., Matchado, J. & Lannin, N. (2016) 'Engaging patients with moderate to severe acquired brain injury in goal setting'. *International Journal of Therapy and Rehabilitation*. Available online at:

https://www.magonlinelibrary.com/doi/pdf/10.12968/ijtr.2016.23.1.20. Accessed on 1.5.18.

DeJong, G., Horn, S., Gassaway, J., Slavin, M. & Dijkers, M. (2004) 'Toward a taxonomy of rehabilitation interventions: using an inductive approach to examine the 'black box' of rehabilitation'. *Archives of Physical Medical Rehabilitation*. Available online at: <u>https://ac.els-cdn.com/S000399930301013X/1-s2.0-S000399930301013X-main.pdf?_tid=f55bae66-e192-11e7-9e81-00000aacb35e&acdnat=1513340876_ccdc905bd9541719a41f2c41d142f91a</u>. Accessed on 15.12.17.

Denson, L., Winefield, H. & Beilby, J. (2012) 'Discharge planning for long-term care needs: the values and priorities of older people, their younger relatives and health professionals'. *Scandinavian Journal of Caring Sciences*. Available online at: <u>http://web.a.ebscohost.com/ehost/pdfviewer/pdfviewer?vid=1&sid=ffccf0cb-23c9-40ed-8a2b-89291381bed4%40sessionmgr4008</u>. Accessed on 26.4.18.

Denzin, N. (1983) 'Interpretive Interactionism'. In Morgan, G. (Ed.) *Beyond Method: Strategies for Social Research.* London: Sage.

Denzin, N. & Lincoln, Y. (1998a) *Strategies of qualitative inquiry*. London: Sage Publications.

Denzin, N. & Lincoln, Y. (1998b) *Collecting and interpreting qualitative methods*. London: Sage Publications.

Department of Health (2005) *The Mental Capacity Act*. Available: <u>http://www.legislation.gov.uk/ukpga/2005/9/contents</u>. Accessed on 26.1.15.

Department of Health (2000) *The NHS Plan: A plan for investment; A plan for reform.* Available at:

Department of Health (2001) National Service Framework for Older People. Available online at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachme nt data/file/198033/National Service Framework for Older People.pdf. Accessed on 25.4.18.

Department of Health (2010) *Equity and Excellence: Liberating the NHS*. Available at: <u>https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/213823/</u><u>dh_117794.pdf</u>. Accessed on 26.9.14.

Department of Health (2012) *The Health and Social Care Act*. Available online at: <u>http://www.legislation.gov.uk/ukpga/2012/7/pdfs/ukpga_20120007_en.pdf</u>. Accessed on 22.4.14.

Department of Health (2013) *The NHS Constitution*. Available online at: <u>https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/170656/</u> <u>NHS_Constitution.pdf</u>. Accessed on 22.4.14.

Department of Health (2014) *The Care Act.* Available online at: <u>http://www.legislation.gov.uk/ukpga/2014/23/pdfs/ukpga_20140023_en.pdf</u>. Accessed on 26.4.18.

Department of Health (2016) *Discharging Older Patients from Hospital: Report by the Controller and Auditor General.* Available online at: <u>https://www.nao.org.uk/wp-content/uploads/2015/12/Discharging-older-patients-from-hospital-Summary.pdf</u>. Accessed on 27.10.17.

Department of Health and Social Care (2018) *The Adult Social Care Outcomes Framework 2018/19: Handbook of Definitions.* Available online at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachme nt_data/file/687208/Final_ASCOF_handbook_of_definitions_2018-19_2.pdf. Accessed on 16.4.18.

Dewey, J. (1929) The quest for certainty. New York: Henry Holt.

Dewing, J. (2002) 'From ritual to relationship: a person-centred approach to consent in qualitative research with older people who have a dementia'. *Dementia*: 1 (2); 157-170.

Dierckx, K., Deveugele, M., Roosen, P. & Devisch, I. (2013) 'Implementation of shared decision making in physical therapy: Observed level of involvement and patient preference'. *Physical Therapy*: 93(10); 1321–1330.

Dingwall, R. & Murray, T. (1983) Categorization in accident departments: 'good' patients, 'bad' patients and 'children'. *Sociology of Health & Illness: 5 (2);* 127–48.

Dodier, N. & Camus, A. (1998) 'Openness and specialisation: dealing with patients in a hospital emergency service'. *Sociology of Health and Illness*: 20 (4); 413-444.

Duncan, P (2010) Values, Ethics and Health Care. London: Sage Publications.

Durocher, E., & Gibson, B, (2010) 'Navigating ethical discharge planning: a case study in older adult rehabilitation'. *Australian Occupational Therapy Journal*: 57 (1); 2-7.

Durocher, E., Gibson, B. & Rappolt, S. (2016) 'Rehabilitation as 'destination triage': a critical examination of discharge planning'. *Disability and Rehabilitation*. Available online at:

https://www.tandfonline.com/doi/pdf/10.1080/09638288.2016.1193232?needAccess=tr ue. Accessed on 27.4.18.

Durocher, E., Kinsella, E., McCorquodale, L., & Phelan, S. (2016) 'Ethical tensions related to systemic constraints: Occupational alienation in Occupational Therapy practice'. *Occupation, Participation and Health*: 64 (4); 216-226.

Edmans, J., Bradshaw, L., Franklin, M., Gladman, J. & Conroy, S. (2013) 'Specialist geriatric medical assessment for patients discharged from hospital acute assessment units: randomised controlled trial'. *British Medical Journal*. Available online at: <u>http://www.bmj.com/content/bmj/347/bmj.f5874.full.pdf</u>. Accessed on 28.12.17.

Edwards, I., Jones, M., Carr, J., Braunack-Mayer, A. & Jensen, G. (2004) 'Clinical Reasoning Strategies in Physical Therapy'. *Physical Therapy*: 84 (4); 312-320.

Edwards, I., Delany, C., Townsend, A. & Swisher, L. (2011) 'New perspectives on the theory of justics: Implications for physical therapy ethics and clinical practice. *Physical Therapy.* Available online at: <u>https://academic.oup.com/ptj/article/91/11/1642/2735107</u>. Accessed on 26.7.18.

Ekdahl, A., Andersson, L. & Friedrichsen, M. (2010) "They do what they think is best for me." Frail elderly patients' preferences for participation in their care during hospitalisation'. *Patient education and counselling*: 80 (2); 233-240.

Ellis, G., Gardner, M., Tsiachristas, A., Langhorne, P., Burke, O., Harwood, R., Conroy, S., Kircher, T., Somme, D., Saltvedt, I., Wald, H., O'Neill, D., Robinson, D. & Shepperd, S. (2017) 'Comprehensive geriatric assessment for older adults admitted to hospital'.

Cochrane Database of Systematic Reviews.

https://www.cochranelibrary.com/cdsr/doi/10.1002/14651858.CD006211.pub3/full. Accessed on 15.8.18.

Elphick, H., Mankad, K., Madan, S., Parker, C. & Liddle, B (2007) 'The determinants of successful in-hospital rehabilitation in people aged 90 years and older'. *Gerentology*. 53 (2); 116-120.

Enderby, P., Pandyan, A., Bowen, A., Hearnden, D., Ashburn, A., Conroy, P., Logan, P., Thompson, C. & Winter, J. (2017) 'Accessing rehabilitation after stroke – a guessing game?'. *Disability and Rehabilitation*. 39 (7); 709-713.

Erler, K. (2017) 'The role of occupational therapy ethics rounds in practice'. *OT Practice*. Available at: <u>https://search.proquest.com/docview/1923231708/fulltextPDF/1865D414649A4041PQ/</u> 1?accountid=12860. Accessed on 2.7.18.

Fleck, L. (2011) 'Just caring: health care rationing, terminal illness, and the medically least well off'. *Journal Of Law, Medicine and Ethics*. Available at: <u>http://web.a.ebscohost.com/ehost/pdfviewer/pdfviewer?vid=1&sid=d6dd8ed3-23b9-4245-9eda-1ca6a03abcf1%40sessionmgr4007</u>. Accessed on 30.10.17

Folstein, M., Folstein, S., & McHugh, P. (1975) 'Mini-Mental State: A practical method for grading the cognitive state of patients for the clinician'. *Journal of psychiatric research.* Available at:

https://www.sciencedirect.com/science/article/pii/0022395675900266?via%3Dihub . Accessed on 19.11.18.

Fonteyn, M. & Fisher, A. (1995) 'Use of think aloud method to study nurses' reasoning and decision making in clinical practice settings'. *The Journal of Neuroscience Nursing*. 27: (2); 124-128.

Foss, C., (2011) 'Elders and patient participation revisited – a discourse analytic approach to older persons' reflections on patient participation'. *Journal of Clinical Nursing*. Available online at: <u>https://onlinelibrary.wiley.com/doi/full/10.1111/j.1365-2702.2010.03505.x</u>. Accessed on 19.6.18.

Fowler, B. (2000) 'Introduction'. In: Fowler, B. (Ed) *Reading Bourdieu on Society and Culture.* London: Blackwell Publishers, 1-21

Gaggiotti, H., Kostera, M. & Kryzyworzeka, P. (2016) 'More than a method? Organisational ethnography as a way of imagining the social'. *Culture and Organisation*. Available online at:

http://www.tandfonline.com/doi/full/10.1080/14759551.2016.1203312?scroll=top&need Access=true. Accessed on 14.6.17.

Garbutt, G. & Davies, P. (2011) 'Should the practice of medicine be a deontological or utilitarian enterprise'. *Journal of Medical Ethics*. Available online at: <u>http://jme.bmj.com/content/early/2011/01/28/jme.2010.036111.full</u>. Accessed on 15.12.17

General Medical Counicl (2014) *First do no harm: teaching medical students about patient safety.* Available online at: <u>https://www.gmc-uk.org/publications/27695.asp.</u> Accessed on 17.1.18

Gergen, K. (2009) *An invitation to social construction - 2nd Edition*. London: Sage Publications.

Gillon, R. (1994) 'Medical ethics: four principles plus attention to scope'. *British Medical Journal*. Available online at:

https://www.ncbi.nlm.nih.gov/pmc/articles/PMC2540719/pdf/bmj00449-0050.pdf. Accessed on 5.12.17.

Gillon, R. (1995) 'Defending the four principles of medical ethics'. *Journal of Medical Ethics:* 21 (6); 323-324.

Gillon, R. (2003) 'Ethics needs principles – four can encompass the rest – and respect for autonomy should be "first among equals". *Journal of Medical Ethics*: 29 (5); 307-312.

Gillon, R. (2015) 'Defending the four principles approach as a good basis for good medical practice and therefore for good medical ethics'. *Journal of Medical Ethics*: 41(10); 111-116.

Gobo, G. & Molle, A. (2017) *Doing Ethnography – 2nd Edition.* London: Sage Publications Ltd.

Goffman, E. (1961) *Asylums: Essays on the Social Situation of Mental Patients and Other Inmates.* New York: Doubleday.

Goncalves-Bradley, D., Lannin, N., Clemson, L., Cameron, I. & Shepperd, S. (2016) 'Discharge planning from hospital'. *Cochrane database of systematic reviews*. Available online at: <u>http://cochranelibrary-</u> wiley.com/doi/10.1002/14651858.CD000313.pub5/epdf</u>. Accessed on 27.4.18.

Gonzalez, M. (2000) 'The four seasons of ethnography: a creation-centred ontology for ethnography'. *International Journal of Intercultural Relations*. Available online at: <u>https://ac.els-cdn.com/S0147176700000201/1-s2.0-S0147176700000201-</u> main.pdf?_tid=6f65a836-fc39-11e7-8672-

<u>00000aacb361&acdnat=1516271107_023d4037702bb37434dd0d5150ae9c91</u>. Accessed on 18.1.18.

Goodwin, D., Pope, C., Mort, M. & Smith, A. (2003) 'Ethics and Ethnography: An experiential account'. *Qualitative Health Research*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/1049732302250723</u>. Accessed on 16.1.18.

Griffiths, P., Edwards, M., Forbes, A., Harris, R., Ritchie, G. (2007) 'Effectiveness of intermediate care in nursing-led in-patient units'. *Cochrane Database of Systematic Reviews*. Available online at: <u>http://cochranelibrary-</u> wiley.com/doi/10.1002/14651858.CD002214.pub3/epdf. Accessed on 26.4.18.

Gubrium, J. & Buckholdt, D. (1982) *Describing Care: Image and Practice in Rehabilitation*. Boston: Oelgeschlager, Gunn and Hain.

Guest, G., MaQueen, K., Namey, E. (2012) *Applied thematic analysis*. London: Sage Publications.

Hagedorn, R. (1996) 'Clinical Decision Making in Familiar Cases: A Model of the Process and Implications for Practice'. *British Journal of Occupational Therapy*. Available online at:

http://journals.sagepub.com/doi/pdf/10.1177/030802269605900508. Accessed on 25.4.18.

Hakkennes, S., Brock, K. & Hill, K. (2011) 'Selection for inpatient rehabilitation after acute stroke: A systematic review of the literature'. *Archives of Physical Medicine and Rehabilitation*: 92; 2057-2070.

Hakkennes, S., Hill, K., Brock, K., Bernhardt, J. & Churilov, L. (2013) 'Selection for rehabilitation after severe stroke: what factors influence rehabilitation assessor decision-making?'. *Journal of Rehabilitation Medicine*. Available online at: <u>https://www.medicaljournals.se/jrm/content/html/10.2340/16501977-1065</u>. Accessed on 27.4.18.

Hammell, K. (2006) *Perspectives on Disability and Rehabilitation*. London: Churchill Livingstone Elsevier.

Hammell, K. (2007) 'Client-centred practice: Ethical obligation of professional obfuscation'. *British Journal of Occupational Therapy*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/030802260707000607</u>. Accessed on 14.3.18.

Hammersley, M. (1990) *Reading ethnographic research – 1st edition*. London: Longman.

Hammersley, M. (1998) *Reading ethnographic research – 2nd edition*. Essex: Addison Wesley Longman Ltd.

Hammersley, M. & Atkinson, P. (1995) *Ethnography: Principles in Practice – 2nd Edition*. London: Routledge.

Hammersley, M. & Atkinson, P. (2007) *Ethnography: Principles in Practice – 3rd Edition*. London: Routledge.

Hammersley, M. & Gomm, R. (2000) 'Introduction'. In Gomm, R., Hammersley. & Foster, P. (Eds) (2000) *Case Study Method*. London: Sage Publications, 1-16.

Health and Care Professions Council (2013) *Occupational Therapists: Standards of Proficiency*. Available at: <u>http://www.hcpc-uk.org.uk/assets/documents/10000512Standards_of_Proficiency_Occupational_Therapists.pdf</u>. Accessed on 26.9.14

Health Research Authority (2013) *HRA Strategy for Public Involvement*. Available online at: <u>https://www.hra.nhs.uk/planning-and-improving-research/best-practice/involving-public/</u>. Accessed on 16.1.18.

Henderson, C., Sheaff, R., Dickinson, A., Beech, R., Wistow, G., Windle, K., Ashby, S. & Knapp, M. (2011) 'Unplanned admissions of older people: exploring the issues'. Available online at: <u>http://www.netscc.ac.uk/hsdr/files/project/SDO_FR_08-1618-136_V01.pdf</u>. Accessed on 19.6.18

Hershkovitz, A. & Brill, S. (2007) 'The association between patients' cognitive status and rehabilitation outcome in a geriatric day hospital'. *Disability and Rehabilitation:* 29 (4); 333-337.

Hershkovitz, A., Jacubovski, O., Alima Bot, M., Oshry, M. & Brill, S. (2010) 'Clock drawing and rehabilitation outcome in hip fracture patients'. *Disability and Rehabilitation*: 32(25); 2113-2117.

Hicks, L., (2011) 'Making Hard Choices: Rationing Health Care Services'. *The Journal of Legal Medicine*. Available online at: <u>http://web.a.ebscohost.com/ehost/pdfviewer/pdfviewer?vid=1&sid=1529fd74-3ac5-488f-b916-4ee72c77f3d3%40sessionmgr4010</u>. Accessed on 1.5.18

Higgs, J. & Jones, M. (1995) 'Clinical reasoning in the health professions'. In: Higgs, J. & Jones, M. (Eds) *Clinical Reasoning in the Health Professions*. Oxford: Butterworth-Heinemann, 3-23.

Higgs, J. & Jones, M. (2000) 'Clinical Reasoning in the Health Professions'. In: Higgs, J. & Jones, M. (Eds) *Clinical Reasoning in the Health Professions – 2nd Edition*. Oxford: Butterworth-Heinemann, 3-14.

Howatson-Jones, L. (2013) *Reflective Practice in Nursing. 2nd ed.* Exeter: Learning Matters Ltd.

Hubbard, R., O'Mahony, S., Cross, E., Morgan, A., Hortop, H., Morse, R. & Topham, L. (2004) 'The ageing of the population: implications for multidisciplinary care in hospital'. *Age and Ageing.* 33(5); 479-482.

Huby, G., Stewart, J., Tierney, A. & Rogers, W. (2004) 'Planning older people's discharge from acute hospital care: linking risk management and patient participation in decision-making'. *Health, Risk and Society*. Available online at: <u>http://web.a.ebscohost.com/ehost/pdfviewer/pdfviewer?vid=1&sid=46ae866a-c567-4455-8618-c81588c9a188%40sessionmgr4009</u>. Accessed on 27.4.18.

Hugman, R. (2005) *New approaches in ethics for the caring professions*. Hampshire: Palgrave McMillan.

Ilett, P., Brock, K., Graven, C. & Cotton, S. (2010) 'Selecting patients for rehabilitation after acute stroke: Are there variations in practice?'. *Archive of Physical Medicine and Rehabilitation*: 91; 788-792.

Jameton, A. (1984) Nursing practice: The ethical issues. New Jersey: Prentice-Hall.

Jasinarachchi, K., Ibrahim, I., Keegan, B., Mathialagan, R., McGoury, J., Phillips, J., & Myint, P. (2009) 'Delayed transfer of care from NHS secondary care to primary care in England: it's determinants, effect on hospital bed days, prevalence of acute medical conditions and deaths during delay, in order adults aged 65 years and over'. *BMC Geriatrics*. Available online at:

https://bmcgeriatr.biomedcentral.com/track/pdf/10.1186/1471-2318-9-4. Accessed on 26.4.18.

Jensen, G., Gwyer, J., Hack, L. & Shepard, K (1999) *Expertise in Physical Therapy Practice*. Boston: Butterworth-Heinemann.

Johns, C. (2013) *Becoming a Reflective Practitioner:* 4th *edition*. Oxford: Blackwell Publishers.

Kennedy, G., Brock, K., Lunt, A. & Black, S. (2012) 'Factors influencing selection for rehabilitation after stroke: A questionnaire using case scenarios to investigate physician perspectives and level of agreement'. *Archive of Physical Medicine and Rehabilitation*: 93; 1457-1459.

Khan, F., Ng, L., Gonzalez, D., Hale, T. & Turner-Stokes, L. (2008) 'Multi-disciplinary rehabilitation programmes following joint replacement at the hip and knee in chronic arthropathy'. *Cochrane database of systematic reviews*. Available online at: <u>http://cochranelibrary-wiley.com/doi/10.1002/14651858.CD004957.pub3/pdf/</u>. Accessed on 26.4.18.

Kimmel, L., Holland, A., Lannin, N., Edwards, E., Page, R., Bucknill, A., Hau, R. & Gabbe, B. (2017) 'Clinician's perceptions of decision making regarding discharge from public hospitals to in-patient rehabilitation following trauma'. *Australian Health Review*. Available online at:

https://search.proquest.com/docview/1923220180/fulltextPDF/459869D23CD442FBPQ /1?accountid=12860. Accessed on 30.4.18.

Kontos, P., Miller, K-L., Cott, C. & Colantonio, A. (2015) 'Therapeutic Landscape: Rethinking "Place" in Client-Centred Brain Injury Rehabilitation'. In: McPherson, K., Gibson, B. & Leplege, A. (Eds) (2015) *Rethinking Rehabilitation: Theory and Practice*. Boca Raton: CRC Press, 117-136.

Kotiadis, G., Carpenter, M. & Mackenzie, M. (2004) 'Examining the effectiveness and suitability of referral and assessment in intermediate care services'. *Journal of Integrated Care*: 12 (4); 42-48.

Kumlien, S., Axelsson, K., Ljunggren, G. & Winblad, B. (1999) 'Stroke patients ready for discharge from acute care – a multi-dimensional assessment of functions and further care'. *Disability and Rehabilitation*: 21 (1); 31-38.

Kus, S., Muller, M. Stobl, R, & Grill, E., (2011) 'Patient goals in post-acute geriatric rehabilitation: goal attainment is an indicator for improved functioning'. Journal of *Rehabilitation Medicine*: 43; 156-161.

Kumar, K., Samuelkamaleshkumar, S., Viswanathan A. & Macaden, A. (2017) 'Cognitive rehabilitation for adults with traumatic brain injury to improve occupational outcomes'. *Cochrane Database of Systematic Reviews*. Available online at; <u>https://www.cochranelibrary.com/cdsr/doi/10.1002/14651858.CD007935.pub2/epdf/full</u>. Accessed on 19.11.18.

Lally, J. & Tullo, E. (2012) 'Engaging older people in decisions about their healthcare: the case for shared decision making'. *Reviews in Clinical Gerontology*: 22; 99-107.

Lane, H., Sarkies, M., Martin, J. & Haines, T. (2017) 'Equity in healthcare resource allocation decision making: a systematic review'. *Social Science and Medicine*: 175; 11-27.

Leach, E., Cornwell, P., Fleming, J. & Haines, T. (2010) Patient centred goal-setting in a subacute rehabilitation setting. *Disability and Rehabilitation*. Available online at: <u>https://www.tandfonline.com/doi/full/10.3109/09638280903036605</u>. Accessed on 1.5.18.

Legare, F., Stacey, D., Turcotte, S., Cossi, M., Kryworuchko, J., Graham, I., Lyddiatt, A., Politi, M., Thompson, R., Elwyn, G. & Donner-Banzhoff, N. (2014) 'Interventions for improving the adoption of shared decision making by healthcare professionals'. *Cochrane Database of Systematic Reviews*. Available online at: <u>http://cochranelibrary-wiley.com/doi/10.1002/14651858.CD006732.pub3/epdf</u>. Accessed on 26.4.18.

Lenze, E., Skidmore, E., Dew, M., Butters, M., Rogers, J., Begley, A., Reynolds, C. & Munin, M. (2007) 'Does depression, apathy or cognitive impairment reduce the benefit of inpatient rehabilitation facilities for elderly hip fracture patients?'. *General Hospital Psychiatry*: 29 (2); 141-146.

Levack, W. (2009) 'Ethics in goal planning for rehabilitation: a utilitarian perspective'. *Clinical Rehabilitation*: 23; 345-351.

Liberati, E., Gorli, M., Moja, L., Galuppo, L., Ripamonti, S. & Scaratti, G. (2015) 'Exploring the practice of patient centred care: the role of ethnography and reflexivity'. *Social Science and Medicine*. Available online at: <u>https://ac.els-</u> <u>cdn.com/S0277953615002038/1-s2.0-S0277953615002038-main.pdf?_tid=df34560a-</u> <u>fa9d-11e7-8c92-</u> <u>00000aacb35e&acdnat=1516094342_6d6af78969228ce974931afa48b3f557</u>. Accessed on 16.1.18.

Lincoln, Y. & Guba, E. (2000) 'The only generalisation is: there is no generalisation'. In Gomm, R., Hammersley, M. & Foster, P. (Eds) *Case Study Method*. London: Sage Publications Ltd: 27-44.

Linker, B. (2016) 'The great war and modern healthcare'. *The New England Journal of Medicine*. Available online at: <u>https://search.proquest.com/docview/1790074535?rfr_id=info%3Axri%2Fsid%3Aprimo</u>. Accessed on 27.4.18.

Lipsky, M. (1980) *Street-Level Bureaucracy: Dilemmas of the individual in public services*. New York: Russell Sage Foundation.

Lofland, J., Snow, D., Anderson, A. & Lofland, L. (2006) *Analysing social settings: a guide to qualitative observation and analysis – 4th Edition*. Belmont, California: Wadsworth Cenage Learning.

Loh, A., Simon, D., Wills, C., Kriston, L., Niebling, W. & Harter, M. (2007) 'The effects of a shared decision-making intervention in primary care of depression: a cluster randomised controlled trial'. *Patient Education and Counselling*. Available online at: <u>https://www.pec-journal.com/article/S0738-3991(07)00140-1/pdf</u>. Accessed on 26.4.18.

Lohman, H., Mu, K. and Scheirton, L. (2003) 'Occupational therapists' perspectives on practice errors in geriatric practice settings'. *Physical & Occupational Therapy in Geriatrics*: 21(4); 21–39.

Longley, V., Peters, D., Swarbick, C. & Bowen, A. (2019) 'What influences decisions about ongoing stroke rehabilitation for patients with pre-existing dementia or cognitive impairment: a qualitative study'. *Clinical Rehabilitation.* Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/0269215518766406</u>. Accessed on 16.8.18.

Lynch, E., Luker, J., Cadilhac, D. & Hillier, S. (2016) 'Inequities in access to rehabilitation: exploring how acute stroke unit clinicians decide who to refer to rehabilitation'. *Disability and Rehabilitation*. Available online at: <u>https://www.tandfonline.com/doi/pdf/10.3109/09638288.2015.1103791?needAccess=tr</u> ue. Accessed on 27.4.18.

Madden, R. (2017) *Being ethnographic: a guide to the theory and practice of ethnography*. Los Angeles: Sage Publications.

Massey, D., Chaboyer, W. & Anderson, V. (2017) 'What factors influence ward nurses' recognition of and response to patient deterioration? An integrative review of the literature. *Nursing Open*. Available online at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC5221430/. Accessed on 27.10.17.

McGlinchey, M. & Davenport, S. (2015) 'Exploring the decision-making process in the delivery of physiotherapy in a stroke unit'. *Journal of Disability and Rehabilitation*. Available online at:

https://www.tandfonline.com/doi/full/10.3109/09638288.2014.962106. Accessed on 25.4.18.

McLean, S. (2014) First do no harm: Law, ethics and healthcare. London: Routledge.

Melis, R., Rikkert, M., Parker, S. & Van Eijken, M. (2004) 'What is intermediate care? An international consensus on what constitutes intermediate care is needed'. *British Medical Journal*. Available online at:

https://www.ncbi.nlm.nih.gov/pmc/articles/PMC509331/pdf/bmj32900360.pdf. Accessed on 25.4.18.

Mendes, A. (2017) 'Managing ethical distress in community nursing practice'. *British Journal of Community Nursing*. Available online at: <u>http://web.b.ebscohost.com/ehost/pdfviewer/pdfviewer?vid=1&sid=1647d3e8-c098-4fbc-acf8-7df99b695bd6%40pdc-v-sessmgr01</u>. Accessed on 20.12.17

Michael, M. (2012) 'When things go wrong'. In Seale, C. (Ed) *Researching Society and Culture: 3rd Edition.* London: Sage Publications Ltd, 544-554.

Milton-Wildey, K. & O'Brien, L. (2010) 'Care of acutely ill older patients in hospital: clinical decision making'. *Journal of Clinical Nursing*: 19; 1252-1260.

Moats, G. (2006) 'Discharge decision-making with older people: the influence of the institutional environment'. *Australian Occupational Therapy Journal*: 53 (2); 107-116.

Moats, G. & Doble, S. (2006) 'Discharge planning with older adults: toward a negotiated model of decision-making'. *Canadian Journal of Occupational Therapy*.

Available online at:

https://search.proquest.com/docview/212910161/fulltextPDF/C492AE5006D54F17PQ/ 1?accountid=12860. Accessed on 26.4.18.

Moreira, T. (2011) 'Health care rationing in an age of uncertainty: A conceptual model'. *Social Science Medicine*. 72; 1333-1341.

Murray, C., Turpin, M., Edwards, I., & Jones, M. (2015). 'A qualitative meta-synthesis about challenges experienced in occupational therapy practice'. *British Journal of Occupational Therapy*: 78 (9); 534-546

Myers, J., Grigsby, J., Teel, C. & Kramer, A. (2009) 'Nurses' assessment of rehabilitation potential and prediction of functional status at discharge from inpatient rehabilitation'. *International Journal of Rehabilitation Research*: 32(3); 264-266.

Nanjayan, S., John, J., Swamy, G., Mitsiou, K., Tambe, A. & Abuzakuk, T. (2014) 'Predictors of change in 'discharge destination' following treatment for fracture neck of femur. *Injury*. Available online at:

https://www.sciencedirect.com/science/article/pii/S002013831400076X. Accessed on 7.7.18.

National Audit Office (2010) *Progress in improving stroke care*. Available online at: <u>https://www.nao.org.uk/wp-content/uploads/2010/02/0910291.pdf</u>. Accessed on 304.18.

National Institute for Health and Care Excellence (2012) *Quality Statement 6: Shared Decision Making*. Available online at:

http://www.nice.org.uk/guidance/qs15/chapter/quality-statement-6-shared-decisionmaking. Accessed on 23rd December 2014.

National Institute for Health and Care Excellence (2013) *What social values should NICE consider when balancing fairness with efficiency*?. Available online at: <u>https://www.nice.org.uk/news/feature/what-social-values-should-nice-consider-when-balancing-fairness-with-efficiency</u>. Accessed on 15.12.17.

National Institute for Health and Care Excellence (2016a) *Multimorbidity: clinical assessment and management* [NG56]. Available online at: <u>https://www.nice.org.uk/guidance/ng56</u>. Accessed on 26.4.18.

National Institute for Health and Care Excellence (2016b) *Major trauma: assessment and initial management* [NG39). Available online at: <u>https://www.nice.org.uk/guidance/ng39</u>. Accessed on 26.7.18.

National Institute for Health and Care Excellence (2016c) *Older people's views on advance care planning*. Available online at: <u>http://www.ncl.ac.uk/media/wwwnclacuk/policyethicsandlifesciences/files/Olderpeoples viewsonadvancecareplanning.pdf</u>. Accessed on 5.12.17.

National Institute for Health and Care Excellence (2017) Intermediate Care Including Reablement [NG74]. Available online at: <u>https://www.nice.org.uk/guidance/ng74</u>. Accessed on 26.4.18

National Institute for Health and Care Excellence (2018) *NICE Guidelines*. Available online at: <u>https://www.nice.org.uk/About/What-we-do/Our-Programmes/NICE-guidalnce/NICE-guidelines</u>. Accessed on 26.4.18.
National Institute for Health Research (2012) *Involve: Briefing Notes for researchers: public involvement in NHS, public health and social care research.* Available online at: <u>http://www.invo.org.uk/wp-</u>

<u>content/uploads/2014/11/9938_INVOLVE_Briefing_Notes_WEB.pdf</u>. Accessed on 16.1.18.

Neyland, D. (2007) Organizational Ethnography. London: Sage Publications.

NHS (2007) Advance care plannig: a guide for health and social care staff. Available online at: <u>http://www.ncpc.org.uk/sites/default/files/AdvanceCarePlanning.pdf</u>. Accessed on 5.12.17.

NHS Digital (2016) *Deaths within 30 days of hospital procedure or emergency admission to hospital – financial year 2014/15.* Available online at: <u>http://digital.nhs.uk/catalogue/PUB20987</u>. Accessed on 5.12.17.

NHS Digital (2017) *Hospital Admitted Patient Care Activity, 2016-17*. Available online at: <u>http://digital.nhs.uk/catalogue/PUB30098</u>. Accessed on 16.4.18.

NHS Digital (2018) *NHS Outcomes Framework Indicators*. Available online at: <u>https://digital.nhs.uk/catalogue/PUB30226</u>. Accessed on 16.4.18.

NHS England (2014a) Safe, compassionate care for frail older people using an integrated care pathway: practical guidance for commissioners, providers and nursing, medical and allied health professional leaders. Available online at: https://www.england.nhs.uk/wp-content/uploads/2014/02/safe-comp-care.pdf. Accessed on 19.4.18.

NHS England (2014b) *Five Year Forward View*. Available online at: <u>https://www.england.nhs.uk/wp-content/uploads/2014/10/5yfv-web.pdf</u>. Accessed on 27.10.17

NHS England (2016a) *Commissioning Guidance for Rehabilitation*. Available at: <u>https://www.england.nhs.uk/wp-content/uploads/2016/04/rehabilitation-comms-guid-16-17.pdf. Accessed on 20.10.17</u>

NHS England (2016b) *Harm Free Care*. Available online at: <u>http://harmfreecare.org/measurement/nhs-safety-thermometer/</u>. Accessed on 15.18.

NHS England (2017) *Good care planning guide for dementia* – *case study: my advance care plan*'. Available online at: <u>https://www.england.nhs.uk/publication/good-care-planning-guide-for-dementia-case-study-my-advance-care-plan/</u>. Accessed on 5.12.17.

NHS England (2018) *Standard General Medical Services Contract 2017/18*. Available online at: <u>https://www.england.nhs.uk/wp-content/uploads/2018/01/17-18-gms-contract.pdf</u>. Accessed on 19.4.18.

NHS Institute for Innovation and Improvement (2010) Safer Care: SBAR – Implementation and training guide. Available online at: <u>https://www.england.nhs.uk/improvement-hub/wp-</u> <u>content/uploads/sites/44/2017/11/SBAR-Implementation-and-Training-Guide.pdf</u>. Accessed on 25.7.18. NHS National Institute for Health Research (2017) *Comprehensive Care: Older People Living with Frailty in Hospitals*. Available online at: <u>https://www.dc.nihr.ac.uk/themed-reviews/Comprehensive-Care-final.pdf</u>. Accessed on 19.4.18

Nugus, P., Greenfield, D., Travaglia, J., Westbrook, J. & Braithwaite, J, (2010). 'How and where clinicians exercise power: interprofessional relations in health care'. *Social Sciences and Medicine*. Available at <u>https://ac.els-cdn.com/S0277953610004284/1-s2.0-S0277953610004284-main.pdf?_tid=f0fcc78c-b597-11e7-9934-00000aab0f02&acdnat=1508505164_bea6bf66c194408565e23eb9d1a15df3</u>. Accessed on 20.10.17.

Ocloo, J. & Matthews, R. (2016) 'From tokenism to empowerment: progressing patient and public involvement in healthcare improvement'. *British Medical Journal Quality and Safety Online*. Available online at:

http://qualitysafety.bmj.com/content/qhc/early/2016/03/18/bmjqs-2015-004839.full.pdf. Accessed on 16.1.18.

O'Connor, A., Stacey, D., Entwhistle, V., Llewellyn-Thomas, H., Rovner, D., Holmes-Rovner, M., Tait, V., Tetroe, J., Fiset, V., Barry, M. & Jones, J. (2003) 'Decision aids for people facing health treatment or screening decisions'. *Cochrane Database of Systematic Reviews*. Available online at:

http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.458.6047&rep=rep1&type=p df. Accessed on 26.4.18.

O'Keefe, S. (2001) 'Autonomy vs welfare? Anatomy of a risky discharge'. *Irish Medical Journal:* 94 (8); 234–236.

O'Reilly, K. (2005) *Ethnographic methods – 2nd edition.* London: Routledge.

O'Reilly, J., Lowson, K., Green, J., Young, J., Forster, A. (2008) 'Post-acute care for older people in community hopsitals – a cost-effectiveness analysis within a multi-centre randomised controlled trial'. *Age and Ageing*. Available online at: <u>https://academic.oup.com/ageing/article/37/5/513/40543</u>. Accessed on 26.4.18.

Parker, G. (2014) Intermediate Care, Reablement or Something Else? A research note about the challenges of defining services. Available online at: https://www.york.ac.uk/inst/spru/pubs/pdf/ICR.pdf. Accessed on 16.4.18

Patton, M. (2015) *Qualitative research and evaluation methods.* London: Sage Publications Ltd.

Polatajko, H., Molke, D., Baptiste, S., Doble, S., Santha, J., Kirsh, B., Beagan, B., Kumas-Tan, Z., Iwama, M., Rudman, D., Thiebeault, R. & Stadnyk, R. (2007) 'Occupational science: imperatives for occupational therapy'. In E. A. Townsend & H. J. Polatajko (Eds.) *Enabling occupation II: Advancing an occupational therapy vision for health, wellbeing, & justice through occupation.* Ottawa: Ontario: Canadian Association of Occupational Therapists, 63-82.

Poole, M., Bond, J., Emmett, C., Greener, H., Louw, S., Robinson, L. & Hughes, J. (2014) 'Going home? An ethnographic study of assessment of capacity and best interests in people with dementia being discharged from hospital'. *BMC Geriatrics*.

Available online at: <u>https://bmcgeriatr.biomedcentral.com/track/pdf/10.1186/1471-2318-14-56</u>. Accessed on 27.4.18.

Pope, C. (2005) 'Conducting ethnography in medical settings'. *Medical Education*: 39; 1180-1187.

Poynter, L., Kwan, J., Sayer, A. & Vassallo, M. (2011) 'Does cognitive impairment affect rehabilitation outcome?'. *Journal of the American Geriatrics Society*: 59 (11); 2108-2111.

Press, Y., Grinshpun, Y., Berzak, A., Friger, M. & Clarfield, M. (2007) 'The effect of comorbidity on the rehabilitation process in elderly patients after hip fracture'. *Archives of Gerontology and Geriatrics*: 45 (3); 281-294.

Prince, M., Knapp, M., Guerchet, M., McCrone, P., Prina, M., Comas-Herrera, A., Wittenberg, R., Adelaja, B., Hu, B., King, D., Rehill, A. & Salimkumar, D. (2014), *Dementia UK: Second Edition - Overview*. Available online at: <u>http://eprints.lse.ac.uk/59437/1/Dementia UK Second edition - Overview.pdf</u>. Accessed 19.11.18.

Proot, I., Abu-Saad, H., de Esch-Janssen, W., Crebolder, H., ter-Meulen, R. (2000) 'Patient autonomy during rehabilitation: the experiences of stroke patients in nursing homes. *International journal of nursing studies*. Available at: <u>https://ac.els-</u> <u>cdn.com/S002074890000080/1-s2.0-S002074890000080-main.pdf?_tid=b8170bdf-</u> <u>3e98-4910-b64e-</u>

<u>ce68fb885207&acdnat=1534756074_30b7394eb4045e2584a00beb82180766</u>. Accessed on 20.8.18.

Putman, K., De Wit, L., Schupp, W., Beyens, H., Dejaeger, E., De Weerdt, W., Feys, H., Walter, J., Louckx, F. & Leys, M. (2007) 'Inpatient stroke rehabilitation: A comparative study of admission criteria to stroke rehabilitation units in four European centres'. *Journal of Rehabilitation Medicine*. Available online at: <u>http://www.ingentaconnect.com/contentone/mjl/sreh/2007/00000039/00000001/art0000</u> <u>5?crawler=true</u>. Accessed on 27.4.18.

Ray, S., & Davidson, S. (2014) *Dementia and cognitive decline: a review of the evidence*. Available online at; <u>https://www.ageuk.org.uk/globalassets/age-uk/documents/reports-and-publications/reports-and-briefings/health--wellbeing/rb_oct14_cognitive_decline_and_dementia_evidence_review_age_uk.pdf</u>. Accessed on 19.11.18.

Rentz, D. (1991) 'The assessment of rehabilitation potential: cognitive factors'. In: Hartke, R. (Ed) *Psychological aspects of geriatric rehabilitation*. Gaithersburg, MD: Aspen.

Reynish, E., Hapca, S., De Souza, N., Cvoro, V., Donnan, P. & Guthrie, B. (2017) 'Epidemiology and outcomes of people with dementia, delirium and unspecified cognitive impairment in the general hospital: prospective cohort study of 10, 014 admissions'. *BMC Medicine*. Available online at: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC5530485/pdf/12916_2017_Article_899.p df. Accessed on 19.11.18.

Robertson, D. (2012) 'Critical thinking and clinical reasoning in new graduate occupational therapists: A phenomenological study'. *PhD Dissertation: Robert Gordon University.* Available online at: https://openair.rgu.ac.uk/bitstream/handle/10059/792/David%20Robertson%20PhD%2 Othesis.pdf?sequence=1. Accessed on 1.5.18.

Rock, P. (2001) 'Symbolic interactionism and ethnography'. In, Atkinson, P., Coffey, A., Delamont, S., Lofland, J. & Lofland, L. (Eds) (2001) *Handbook of Ethnography*. London: Sage Publications Ltd, 26-28.

Rose, A., Rosewilliam, S., & Soundy, A. (2017) 'Shared decision making within goal setting in rehabilitation settings: a systematic review'. *Patient Education and Counselling*. Available online at: <u>https://ac.els-cdn.com/S0738399116303251/1-s2.0-S0738399116303251-main.pdf?_tid=c9e8b6af-2f03-4ae1-82e1-4ae1d73ead1c&acdnat=1523891172_f123455bddb30d6ca7eec4cac69a41fb</u>. Accessed on 16.4.18.

Rosewilliam, S., Roskell, C. & Pandyan, A, (2011) 'A systematic review and synthesis of the quantitative and qualitative evidence behind patient-centred goal setting in stroke rehabilitation'. *Clinical Rehabilitation*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/0269215510394467</u>. Accessed on 5.12.17.

Royal College of Occupational Therapists (2015) *Code of Ethics and Professional Conduct*. Available online at: <u>https://www.rcot.co.uk/sites/default/files/CODE-OF-ETHICS-2015_0.pdf</u>. Accessed on 5.12.17.

Royal College of Psychiatrists (2005) *Who cares wins: improving the outcomes for older people admitted to acute hospital.* Available at: <u>http://www.rcpsych.ac.uk/PDF/WhoCaresWins.pdf</u>. Accessed on: 2.4.15

Savage, J. (2000) 'Ethnography in health care'. *British Medical Journal*. Available online at: <u>https://search.proquest.com/docview/1777609188/fulltext/D6DFD75FBF8E4349PQ/1?</u> accountid=12860. Accessed on 2.7.18.

Savage, J. (2006) 'Ethnographic evidence: The value of applied ethnography in healthcare'. *Journal of Nursing Research*: 11 (5); 383-393.

Scheel-Sailor, A., Post, M., Michel, F., Weidmann-Hugle, T. and Baumann Holze, R. (2017) 'Patients' views on their decision making during inpatient rehabilitation after newly acquired spinal cord injury – a qualitative interview-based study'. *Health Expectations*. Available online at:

https://onlinelibrary.wiley.com/doi/epdf/10.1111/hex.12559. Accessed on 16.4.18

Scheunemann, L. (2011) 'The ethics and reality of rationing medicine'. *Chest.* Available online at <u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3415127/pdf/110622.pdf</u>. Accessed on 19.6.18.

Schön, D. (1991) The Reflective Practitioner. Aldershot: Ashgate Publishing Ltd.

Schubert, M., Glass, T., Clarke, S., Aiken, L., Schaffert-Witvliet, B., Sloane, D. & De Geest, S. (2008) 'Rationning of nursing care and its relationship to patient outcomes: The swiss extension of the international hospital outcomes study'. International Journal for Quality in Health Care. Available online at:

https://academic.oup.com/intqhc/article/20/4/227/1846768. Accessed on 1.5.18

Schulman-Green, D., Naik, A., Bradley, E., McCorkle, R. & Borgardus, S. (2006) 'Goal setting as a shared decision making strategy among clinicians and their older patients'. Patient Education and Counselling. Available online at: https://ac.elscdn.com/S0738399105002752/1-s2.0-S0738399105002752-main.pdf? tid=9478ec91b5ad-4b73-a9a0-26e66b1e429e&acdnat=1523890545 ae0449183634c4a303d0c9be4e13b3da.

Accessed on 16.4.18.

Schwandt, T. (2007) The Sage dictionary of qualitative inquiry – 3^{rd} edition. Los Angeles: Sage Publications.

Seale, C. (2018) 'Introduction'. In: Seale, C. (Ed) Researching Society and Culture. Los Angeles: Sage Publications, 1-4.

Seedhouse, D. (2005) Values-based decision-making for the caring professions. Chichester: Wiley.

Seedhouse, D. (2017) Thoughtful health care: ethical awareness and reflective practice. London: Sage Publications.

Semel, J., Gray, J., Jun Ahn, H., Nasr, H. & Chen, J. (2010) 'Predictors of outcome following hip fracture rehabilitation'. Journal of Injury, Function and Rehabilitation: 2 (9); 799-805.

Shun, P., Bottari, C., Ogourtsova, T. & Swaine, B. (2017) 'Exploring factors influencing occupational therapists' perception of patients' rehabilitation potential after acquired brain injury'. Australian Occupational Therapy Journal: 64; 149-158.

Silber, J., Williams, S., Krakauer, H. & Schwartz, J.(1992) 'Hospital and patient characteristics associated with death after surgery: a study of adverse occurrence and failure to rescue'. Medical Care. 30: (7); 615–629.

Silverman, D. (2000) Doing qualitative research: a practical handbook. London: Sage Publications.

Silverman, D. (2014) Interpreting qualitative data – 5th Edition. London: Sage Publications.

Singh, I., Gallacher, J., Davis, K., Johansen, A., Eeles, E. & Hubbard, R. (2012) 'Predictors of adverse outcomes on an acute geriatric rehabilitation ward'. Age and Ageing. Available online at: https://academic.oup.com/ageing/article/41/2/242/47632/. Accessed on 304.18.

Skeggs, B. (2001) 'Feminist Ethnography'. In Atkinson, P., Coffey, A., Delamont, S., Lofland, J. & Lofland, L. (Eds) Handbook of Ethnography. London: Sage Publications, 426-442.

Smith, T., Hameed, Y., Cross, J., Henderson, C., Sahota, O. & Fox, C. (2015) Enhanced rehabilitation and care models for adults with dementia following hip fracture surgery. Available online at: <u>http://cochranelibrary-</u> wiley.com/doi/10.1002/14651858.CD010569.pub2/epdf</u>. Accessed on 9.4.18.

Snape, D., Kirkham, J., Britten, N., Froggatt, K., Gradinger, F., Lobban, F., Popay, J., Wyatt, K. and Jacoby, A. (2014) 'Exploring perceived barriers, drivers, impacts and the need for evaluation of public involvement in health and social care research: a modified Delphi study'. *British Medical Journal Open Access*. Available online at http://bmjopen.bmj.com/content/bmjopen/4/6/e004943.full.pdf. Accessed on 16.1.18.

Sointu, E. (2017) 'Good patient/bad patient: clinical learning and the entrenching of inequality'. *Sociology of Health and Illness*: 39 (1); 63-77.

Spradley, J. (1980) Participant Observation. London: Holt, Rienhart and Winston.

Stake, R. (2000) 'The case study method in social inquiry'. In Gomm, R., Hammersley. & Foster, P. (Eds) (2000) *Case Study Method*. London: Sage Publications, 19-26.

Strech, D., Persad, G., Marckmann, D. & Danis, M. (2009) 'Are physicians willing to ration health care? Conflicting findings in a systematic review of survey research'. *Health Policy.* Available online at: <u>https://www.sciencedirect.com/science/article/pii/S0168851008002340</u>. Accessed on 1.5.18

Stroke Unit Trialists' Collaboration (2017) *Organised inpatient (stroke unit) care for stroke*. Available online at: <u>http://cochranelibrary-</u>wiley.com/doi/10.1002/14651858.CD000197.pub3/epdf. Accessed on 9.4.18

Stucki, G., Cieza, A. & Melvin J. (2007) 'The international classification of functioning, disability and health (ICF): a unifying model for the conceptual description of the rehabilitation strategy'. *Journal of Rehabilitation Medicine*: 39; 279-85.

Sudore, R., Lum, H., You, J., Hanson, L., Meier, L., Pantilat, S., Matlock, D., Rietjens, J., Korfage, I., Ritchie, C., Kutner, J., Teno, J., Thomas, J., McMahan, R., Heyland, D. (2017) 'Defining advance care planning for adults: a consensus definition from a multidisciplinary delphi panel'. *Journal of Pain and Symptom Management*. Available online at:

https://reader.elsevier.com/reader/sd/3A47E70F38E3913E0FD2A6B21D7175294D20A 7B0A4930201C5DEBF70B7101A3F194B856D1ECF3E62A74DF7ED218EC138. Accessed on 2.7.18.

Sugavanam, T., Mead, G., Bulley, C., Donaghy, M. & van Wijck, F. (2013) The effects and experiences of goal setting in stroke rehabilitation – a systematic review. *Disability and Rehabilitation*. Available online at:

https://www.tandfonline.com/doi/full/10.3109/09638288.2012.690501. Accessed on 15.12.17.

Swartz, D. & Zolberg, V. (2004) 'Introduction: Drawing inspiration from Bourdieu'. In: Swartz, D. & Zolberg, V. (Eds) *After Bourdieu: Influence, critique, elaboration*. New York: Kluwer Academic Publishers, 1-13. The Cambridge Dictionary (2017) *Rehabilitation*. Available online at: <u>https://dictionary.cambridge.org/dictionary/english/rehabilitation</u>. Accessed on 15.12.17.

The Kings Fund (2001) *Rehabilitation and Development Notebook. News Update 7.* London: Kings Fund.

The Kings Fund (2012) Older people and emergency bed use – Exploring variation. Available at: <u>http://www.kingsfund.org.uk/sites/files/kf/field/field_publication_file/older-people-and-emergency-bed-use-aug-2012.pdf</u>. Accessed on 23.5.13

The Kings Fund (2013) Urgent and emergency care – A review for NHS South of England. Available at: <u>http://www.southofengland.nhs.uk/wp-</u> <u>content/uploads/2012/05/Kings-Fund-report-urgent-and-emergency-care.pdf</u>. Accessed on 23.5.13

The Kings Fund (2017) *NHS Hospital Bed Numbers: Past, Present, Future*. Available online at: <u>https://www.kingsfund.org.uk/publications/nhs-hospital-bed-numbers</u>. Accessed on 16.4.18.

The National Audit Office (2016) *Discharging older patients from hospital*. Available online at: <u>https://www.nao.org.uk/wp-content/uploads/2015/12/Discharging-older-patients-from-hospital-Summary.pdf</u>. Accessed on 16.4.18

The Oxford English Dictionary (2017) *Rehabilitation*. Available online at: <u>https://en.oxforddictionaries.com/definition/rehabilitation</u>. Accessed on 15.12.17.

Traballesi, M., Brunelli, S., Pratesi, L., Pulcini, M., Angioni, C. & Paolucci, S. (1998) 'Prognostic factors in rehabilitation of above knee amputees for vascular disease'. *Disability and Rehabilitation*: 20 (10); 380-384.

Turner-Stokes L., Paul, S. and Williams, H. (2006) 'Efficiency of specialist rehabilitation in reducing dependency and costs of continuing care for adults with complex acquire brain injuries'. *Journal Neurology Neurosurgery and Psychiatry*. 77; 634–639.

United Kingdom Homecare Association (2015) *Dementia and homecare: driving quality and innovation.* Available online at: <u>https://www.ukhca.co.uk/downloads.aspx?ID=488</u>. Accessed on 19.11.18.

United Nations (2009) *Convention of the rights of persons with disabilities*. Available online at: <u>https://www.un.org/development/desa/disabilities/convention-on-the-rights-of-persons-with-disabilities/convention-on-the-rights-of-persons-with-disabilities-2.html</u>. Accessed on 19.6.18.

Unsworth, C. (2001) 'Selection for rehabilitation: acute care discharge patterns for stroke and orthopaedic patients'. *International Journal of Rehabilitation Research*: 24 (2); 103-114.

van Manen, M. (1990) Researching Lived Experience: Human Science for an Action Sensitive Pedagogy. New York: SUNY.

Varcoe C, Pauly B, Webster G, & Storch J. (2012). 'Moral distress: Tensions as springboards for action'. *HEC Forum*: 24; 51–52.

Vermunt, N., Harmsen, M., Westert, G., Olde Rikkert, M. & Faber, M. (2017) 'Collaborative goal setting with elderly patients with chronic disease or multi-moribidity: a systematic review'. *BMC Geriatrics*. Available online at: <u>https://bmcgeriatr.biomedcentral.com/track/pdf/10.1186/s12877-017-0534-0</u>. Accessed on 26.4.18.

Vryonides, S., Papastavrou, E., Charalambous, A., Andreou, P. & Merkouris, A. (2015) 'The ethical dimension of nursing care rationing: a thematic synthesis of qualitative studies'. *Nursing Ethics*: 22 (8); 881-900.

Wacquant, L. (1998) 'Pierre Bourdieu'. In: Stones, R. (Ed) (1998) *Key Sociological Thinkers.* London: MacMillan Press Ltd, 215-229.

Wade, D. (2003) 'Selection criteria for rehabilitation services'. *Clinical Rehabilitation*. Available online at:

https://search.proquest.com/docview/200739109?rfr_id=info%3Axri%2Fsid%3Aprimo. Accessed on 27.4.18.

Wade, D. (2005) 'Describing rehabilitation interventions'. *Clinical Rehabilitation*. Available online at: http://journals.sagopub.com/doi/pdf/10.1101/0269215505cr923od_Accessed.on

http://journals.sagepub.com/doi/pdf/10.1191/0269215505cr923ed. Accessed on 15.12.17.

Wade, D. (2009) 'Adverse effects of rehabilitation - an opportunity to increase quality and effectiveness of intervention'. *Clinical rehabilitation*. Available online at: <u>https://search.proquest.com/docview/200739846?OpenUrlRefId=info:xri/sid:primo&accountid=12860</u>. Accessed on 14.3.18

Wade, D. (2015) 'Rehabilitation – a new approach. Overview and Part One: The Problems'. *Clinical Rehabilitation*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/0269215515601174</u>. Accessed on 15.12.17.

Wade, D. (2016) 'Rehabilitation – a new approach. Part Four: a new paradigm, and it's implications'. *Clinical Rehabilitation*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/0269215515601177</u>. Accessed on 15.12.17

Wade, D. & Halligan, P. (2017) 'The biopsychosocial model of illness: a model whose time has come'. *Clinical Rehabilitation*. Available online at: <u>http://journals.sagepub.com/doi/pdf/10.1177/0269215517709890</u>. Accessed on 14.3.18

Walton, M. & Kerridge, I. (2014) 'Do no harm: is it time to rethink the Hippocratic oath?'. *Medical Education*. Available online at:

https://ses.library.usyd.edu.au/bitstream/2123/14742/1/do-no-harm-PP-2014.pdf. Accessedd on 17.1.18

Ward, D., Drahota, A., Gal, D., Severs, M., Dean, T. (2008) 'Care home versus hospital and own home environments for rehabilitation of older people'. *Cochrane Database of Systematic Reviews*. Available online at: <u>http://cochranelibrary-</u> wiley.com/doi/10.1002/14651858.CD003164.pub2/pdf. Accessed on 26.4.18 Waring, J., Marshall, F., Bishop, S., Sahota, O., Walker, M., Currie, G., Fisher, R. & Avery, T. (2014) 'An ethnographic study of knowledge sharing across the boundaries between care processes, services and organisations: the contributions to 'safe' hospital discharge'. *Health Services and Delivery Research*. Available online at: <u>https://www.ncbi.nlm.nih.gov/books/NBK259989/pdf/Bookshelf_NBK259989.pdf</u>. Accessed on 26.4.18.

White, A., Armstrong, P. & Thakore, S. (2010) 'Impact of senior clinical review on patient disposition from the emergency department'. *Emergency Medicine Journal*. Available online at:

https://search.proquest.com/docview/1780099067/fulltextPDF/E6F42E8876F64AA2PQ /1?accountid=12860. Accessed on 26.4.18

Whyte, J. (2009) 'Defining the active ingredients of rehabilitation'. *Health Policy Newsletter*. Available online at:

http://jdc.jefferson.edu/cgi/viewcontent.cgi?article=1626&context=hpn/. Accessed on 15.12.17.

Whyte, J. & Hart, T. (2003) 'Its more than a black box; it's a Russian doll: defining rehabilitation treatments'. *American Journal of Physical Medical Rehabilitation*: 82; 639-52.

Wilding, C. & Whiteford, G. (2009) 'From practice to praxis: Reconnecting moral vision with philosophical underpinnings'. *British Journal of Occupational Therapy*. 72(10): 434–441.

Willis, E. (2006). 'Introduction: taking stock of medical dominance'. *Health Sociology Review*, 15(5), 421-431.

Wong, D. (1992) 'Coping with moral conflict and ambiguity'. Ethics: 102; 763-784

Woods, P (1999) Successful Writing For Qualitative Researchers. London: Routledge.

World Health Organization (2001) *The International Classification of Functioning, Disability and Health* (ICF). Available online at: http://www.who.int/classifications/icf/en/. Accessed on 26.4.18.

World Health Organisation (2017) *Rehabilitation in Health Systems*. Available online at: <u>http://apps.who.int/iris/bitstream/10665/254506/1/9789241549974-eng.pdf?ua=1</u>. Accessed on 20.10.17.

Yar, M., Dix, D. & Bajekal, M. (2006) 'Socio-demographic characteristics of the healthcare workforce in England and Wales – results from the 2001 census'. *Health Statistics Quaterly*. Available online at:

https://www.ncbi.nlm.nih.gov/pubmed/17165469. Accessed on 16.1.18.

Yin, R. (2009) *Case Study Research: Design and Methods – 4th Edition*. California: Sage Publications.

Yin, R., (2014) *Case Study Research: Design and Methods – 5th Edition*. California: Sage Publications.

Yin, R. (2012) *Applications of Case Study Research – 3rd Edition.* California: Sage Publications.

Young, J., Green, J., Forster, A., Small, N., Lowson, K., Bogle, S., George, J., Heseltine, D., Jayasuriya, T. & Rowe, J. (2007) 'Postacute Care for Older People in Community Hospitals: A Multicentre Randomised Controlled Trial'. *Journal of the American Geriatrics Society*. Available online at:

https://onlinelibrary.wiley.com/doi/epdf/10.1111/j.1532-5415.2007.01456.x. Accessed on 26.4.18.

Zhu, M., Chen, W., Hirdes, J. & Stolee, P. (2007) 'The K-nearest algorithm predicted rehabilitation potential better than current Clinical Assessment Protocol'. *Journal of Clinical Epidemiology*. Available online at: <u>https://ac.els-</u>

cdn.com/S0895435607001941/1-s2.0-S0895435607001941-main.pdf? tid=02fba670-5c38-48b8-a768-

bd9f63fa16c9&acdnat=1524756018_68dbbb653c80c2a0782936a6ed11e458. Accessed on 26.4.18.

Appendices

Appendix 1 – Phase 3 Semi-Structured Interview Schedule

Introduction

- Revisit consent
- Revisit purpose of study and recap what has been done so far
- Discuss the format of the interview pose some broad questions to understand your ideas and then share some reflections from my experience within your services and ask for your further thoughts and interpretations in relation to these areas

Recap of Professional Role

 My area of interest is the concept of rehabilitation potential, and in particular, the decision making process and different people's roles within those decisions. But before, we start talking more specifically about decision making and roles, I wonder whether you could start off by **describing your role in general** (probably the role you were doing at the time if this has since changed) – in your own words. This will be helpful context for the later discussion.

Meanings

2. In your words, can you describe what you mean by rehabilitation?

Further discussion points...

From my observations, when talking about rehabilitation, many people were relating this to a process to facilitate improvements in mobility. Do you think this is the case?

Rehabilitation also was often referred to in relation to a place (e.g. someone being listed for or transferred to rehabilitation) – again, do you think this is the case?

Finally, in relation to the meaning of rehabilitation – it also seemed easier for professionals to describe when they didn't think it was happening – does this fit with your experience?

3. In your own words, can you describe what you mean by rehabilitation potential?

Further discussion points...

Again, from my observations, rehabilitation potential seemed to link mainly to the potential to get back to baseline mobility – does this fit with your experience?

However, there were examples of other meanings – for example, the potential to influence discharge outcomes (e.g. to get home, or to go to res care rather than nursing care), the potential to be safer (e.g. with equipment), or potential to reduce dependency on services (e.g. go home with a minimal care package rather than something like 4x visits). Again – does this reflect your experience?

What do you think the impact, if any, might be of these different interpretations?

At one point during my time within your wards, someone compared rehabilitation potential to how we have now come to think of mental capacity – in that we now think of mental capacity as being decision specific, and we could think of rehabilitation potential being goal specific – that is being specific about potential to do what? What do you think about this reflection?

Decision Making

4. What is going through your mind when you think about rehabilitation potential?

Further discussion points...

Some of the things I noticed included:

- Clinical features (co-morbidities, cognition, mood, nutrition)

- Other individual features (baseline function, home environment, social situation)
- resources

- Assessment through engagement during the rehab process (e.g. carryover, consistency, motivation/engagement)

- your professional desire to give people a chance

Does this reflect your experience?

Do you think some of these influences are more important than others (could prompt with a question to identify 3 most important?)

5. Alongside making a judgement about whether someone has rehabilitation potential, you are also thinking about the pathway that might best meet their needs (e.g. the ward, or the service). Is there anything additional going through your mind at this point?

Further discussion points...

Some of the things I noticed include: where can manage any risks, and the availability of resources – does this fit with your experience and are there any examples that would help me to understand these things further?

The opinion of the receiving service also presented as significant – professionals asking themselves what the staff at ***, or the community ** team would think about the appropriateness of the referral. Do you think this is the case?

How would you describe *your* role in decisions about: Rehabilitation potential Pathways

Further discussion points...

One of the things I noticed was that, particularly in the early stages, Physio seems to have an important role in these decisions. Does this reflect your experience? If so, why do you think this may be the case?

7. To what extent do you think the wishes of the patient and family influences these decisions?

Further discussion points...

How would you describe your role in this area of decision making?

One of the things I reflected upon was that perhaps patients and families are not as involved in decisions about their own rehabilitation potential and their pathways as perhaps organisations (or policy drivers) would like to think they could be. Does this reflect your own experience? If so, why do you think this might be the case? Also, the professionals I worked alongside seemed to becoming very accustomed to working with very frail, and often cognitively impaired individuals and I reflected that perhaps talking to families about decisions – rather than directly to patients – may be becoming more of the default position. Do you think this could be the case?

Appendix 2 – Sample Observational Checklist

Record Date/time/place/type of event

Adapted from Spradley (1980):

- Space layout and points to note about the physical setting (diagram if helpful)
- Actors names and relevant details of people present
- Activities description of the various activities of the actors
- Acts specific individual actions or behaviours
- Events -
- Time sequence of events
- Goals what were the actors trying to accomplish note explicit goals (e.g. purpose of meeting) and other observations about motivations or agendas.
- Feelings

Other considerations:

- Interactive patterns
- Language
- Non-verbal behaviours
- Ideational elements anything that reflects beliefs, attitudes or values
- Broader contextual/social information
- Human needs
- Other observations

Appendix 3 – Sample from Fieldnotes (including annotations and memos)

ABARTA Therday 17th May 40 poor in proton (() gam - mot hadare Present - nursing (IX care from another wood) - I x wood du + 1 x cover from another wood OT SW Physice astronat. - Patent have boday (delayed from yesterday), day le while be bace & god arguing realized paterage -while he need an entry call at much fine is many cameral. Decoded not (- Portrant lise to ward . (the lase discussion i of + 0+) - Pabat aining for d/c binck heres one - "wedran trus magbe" - de (mirportations + details smetheres mater) - Rebail whe with single need for the but fairing tore and "no boy Treng that it it hulty she need misty care. could be a threening bool but could the the fast. Nacked by Nr. . No real denses about - Patient & co-ring - b see if harry in late - Robert mendared all night - family feel is I songling mit - Patend ? for the (? for advanced for my objects) read - for planning meeting escape - dearing meeting planning - Rebert proversity with mobility - to renew build the and of ar week Antre patrent for word . Is viry bed noneque - To war for the Afre MOF, durinsia between OF, much a Sur real publication who is requestly see they aberea, why & Leve from shit. CT When Mot, allowing the high able to a sty of the lay for the is a bounder. I was known for a sport reaction of a to fear " That the hay back " " proposed for the stor on the fear " That the hay back " " proposed that she can measure at home back " " proposed that she can measure at home back " " proposed that she can measure at home with Gall carries (wound can be been proved of the store been to feel the store of the store of the store of the store been to feel the store of the store of the store of the store (would be the store of the store of the store of the store (would be the store of the store of the store of the store (would be the because the is seen all all of the store of the store (would be because the is seen all the store of the s

Appendix 4 – Clarifying the analytical position – adapted from Braun and Clarke (2006) and used to discuss data analysis within supervision

Before outlining the stages and steps involved in thematic analysis, Braun and Clarke suggest the researcher must clarify their position with regard to values, assumptions and decision making and answer a series of analytical questions:

- What counts as a theme?
- Is the analysis inductive or theoretical?
- o Will the analysis be identifying semantic or latent themes?
- How does the analysis align with epistemological assumptions?

Presented as the last in their series of questions, I feel the question of how thematic analysis aligns with epistemological assumptions is important to address first. In the example of this data analysis, the research is aligned to a social constructionist epistemology which aims to examine the ways in which events, realities, meanings and experiences are the effects of a range of influences within the particular social context. It is therefore not the goal of the analysis to focus on individual motivations or the link between meanings and individual experiences.

Another question that any researcher utilising thematic analysis should reflect on, and understand their own response to, is 'what counts as a theme?'. The researcher should bring their awareness to themes being prevalent across the entire data set and as entities which capture something important in relation to the research question and aims. In addition, Braun and Clarke suggest is it important to consider the purpose of analysis and ask whether it represents a rich description of the entire data set or a detailed account of one aspect. In the case of this research, it is my intention to provide a full description of the entire data set, again perhaps at the expense of presenting individual cases. This is again, something that an alternative approach such as narrative analysis may have promoted.

A further question from Braun and Clarke is whether the analysis is inductive or theoretical in nature. By this, they suggest that an inductive analysis is a 'bottom-up' approach, not driven by the researcher's theoretical interest and potentially bearing no resemblance to the questions asked of participants or indeed the researcher's objectives. In simple terms, it is a process of coding without trying to fit the data into a pre-existing coding frame. In contrast, an analysis of a more 'theoretical' nature aims to map the data to specific research aims and questions. In relation to this research study, my position here has perhaps been a little more unclear to define. I am open to the emergence of themes outside of the original research aim and objectives, although recognise that it is perhaps unrealistic to free myself from some of the theoretical perspectives which motivated the project and were needed to research existing literature and gain necessary approvals to progress. I have therefore conceptualised a spectrum between inductive and theoretical positions, with

this project proposing to adopt a mid-point position on this continuum aiming to maximise opportunities from both perspectives.

A final question according to Braun and Clarke, which requires reflection and clarification prior to data analysis, is whether the researcher will be identifying semantic or latent themes. At a semantic level, the researcher is primarily identifying themes within the explicit or surface meanings and the researcher is not looking much beyond what has been said or has been written. In comparison, a latent theme instead looks to identify or examine underlying ideas, assumptions and conceptualisations that are theorised as influencing the semantic content of the data. Again, as what has been said and written by participants is only a small part of a wider data set which captures researcher observations of the physical, social and temporal environment and context, the analysis of data set for this project presents as aiming to identify latent themes, which is in alignment with the suggestion that this approach tends to grow from a constructionist paradigm (Burr, 1995).

Appendix 5 – Example from Coding - Code: 'Meaning of Rehabilitation'

Examples from Fieldnotes where the term 'Rehabilitation' being used/described/interpreted

ſ	Physio discussed the 'ideal' of a 24-hour approach to	Page 9
l	rehabilitation – for all staff to be working towards rehab goals.	
l	He would like to implement training with assistant nursing staff	
l	so they could take for walks, practice exercises. Instead, he feel	
l	that different people only have contact with patients for their role	
l	(e.g doing observations). Implication that this 24-hour approach	
ŀ	Bationt who is listed for *** (cog robab unit), by the time there is	Page 10
l	a hed available, may be able to go home from here	Fage 19
l	Rehabilitation perceived as a place?	
ŀ	Going to rehab often gets sold as somewhere they go to for	Page 20
l	intensive therapy but in reality, it might not be that different	1 490 20
ŀ	Go to ward ** for rehab	Page 20
ľ	SW using terms rehabilitation and reablement interchangeably	Page 21
ľ	Other services dealing with rehabilitation issues – e.g. refer to	Page 30
l	POAS for decreased motivation	0
Ī	Rehabilitation linked to improving mobility	
l		
l	PT to nursing staff - Encourage mobility as much as you can	Page 32
l	During discussion with bed managers: "If a patient is moved to	Page 42
l	another elderly bed on a base site, the patient comes off the list	
l	for rehab". Rehabilitation as a place rather than an approach to	
ŀ		D 05
	During an MDT meeting, Consultant mentioned "Query for	Page 65
ŀ	Tenab – as it not already happening there.	Dago 68
l	care unit – some aspects are less than the provision in hospital –	raye oo
l	for example, weekend therapy	
ŀ	Discussion with intermediate care OT – people refer patients	Page 70
l	here for rehab although it is more about discharge planning than	
l	rehab.	
ľ	OT intermediate care unit – we try to look at rehabilitation	Page 70
l	beyond just mobility – e.g. get people to manage own	0
l	medications, catheters etc However, this can be limited by time	
L	constraints	
l	OT intermediate care unit – they are a rehab service although	Page 71
l	feel the dependence of patients has changed i.e. they are more	
l	dependent. Linked to rehab ward who also feel they get patients	
ļ	that other services don't know what to do with	D
l	OI – ward ** "Would like to be able to do 'rehab' but does not	Page 83
ŀ	teel able to	Dama 00
l	OT ward "" - told story of someone sne is working with -	Page 83
l	time spont in bod, bad some difficult conversations. "vou're not	
I	doing this for me, you're doing this for yourself" (but link to OT	
l	thinking she's not doing 'rehab'	
1		•

Physio ward ** – patient 1 – trying to improve consistency with mobility	Page 84
Patient 2 – 4 days after admission, plan in medical entry includes medical actions and "physio" (i.e. physio are the providers of rehab at this early stage)	Page 88
Physio ward ** re: patient 2 – Unfortunately I've not seen a lot of her so far (e.g. this ward not a place for rehab?)	Page 93
Patient who has had 7 ward transfers during the admission and decided to keep on ward ** at the moment. Consultant "There are still issues around his Parkinsons, continence, skin integrity, and to manage his posture/prevent contractures. There is a lot the team can do for quality of life" – wider interpretation of rehabilitation	107
Consultant "best rehab tends to be at home in terms of doing stuff"	107
Training a patient to empty catheter and stoma bags	107
Ward ** Physio about patient 2 – sometimes you think, is it better to get someone home as quickly as possible even if it does mean them being hoisted and perhaps not reaching their potential or is it better to keep them and just keepyou knowto work on what and to try and get them betterif we could keep her a bit longer.	134
"Waiting for rehab as he's still assistance of 1 for mobility"	141
Ward ** staff "people using hospitals like respite"	148
Patient 2 notes – from MDT "More time with OT/PT" – this may be what team think of as rehab but no OT involvement to date	156
Expectations of rehabilitation – ward ** sometimes get anger from relatives if they have unrealistic expectations	160
Junior PT "That will have given her another week of rehab"	170
Ward ** Physio re patient "Feels this patient could make good progress with a period of intensive rehab but don't' feel we have that here". Wish they could do passive movements a few times per day and then give him practice with the standing hoist a couple of times per day but that just cant happen	178
Ward ** Physio – when discussing prioritisation – if we see these patients 2/3 times per day and see these patients lessto direct the resources to those where they can make the best gains instead of just trying to see every patient every day. But that seems at odds with the organisation who want them to see more of the quick patients to turn them around.	178
Discussion with physio about ** closing – thinks this ward will become more like a rehab ward	183
"Can come back in for rehab once fracture has healed"	183
"too sleepy for physio"	185
Asked patient 5 if he was aware why he had gone to ward ** – "to get stronger".	199
Patient 5 – using pedals.	217

Examples from Interviews where the term 'Rehabilitation' being used/described/interpreted

it's a multi-disciplinaryemand goal-ledthat's what I tend to	Interview 1 pg 2
thinkthat if we're looking at proper rehabilitation thenemit	

should be that we're working withthe patients and with their families as much as we can to all be achieving the goals that the patient wants and that we want as well. So it could beeven, you know rehabilitation includes lots of thingsit's notfrom a physio point of view, as I say, we tend to always be thinking are they transferring, can they get in and out of bed independently. But from an Occupational Therapist, then rehabilitation is much more about things like personal care and, you know, managing things at home. Which obviously we dothat's why we work with themEm. But there'syou know, a lot of different areas to rehabilitation. And alsorehabilitationemisn't just the physical side of it as wellthere's the cognitive side of it and theemI think a big part of rehabilitation isemput it this way, rehabilitation works a lot better when people want to do it and they've got motivation to do it and cognitively they're able to understand whyemwe're doing the things that we're doing.	
so, we need to have goalsbut within the team, and that's the whole team, it always does seem to go down to right, transfers, mobility, that seems to be where everyone hones in on	Interview 1 pg 3
But the rehabilitation shouldn't just be about those things, it should beand this is what we're trying to introduceidentifying rehab patients and things like emand this is one of ** thingshe's doing a course andRight, so this is a rehab patient and they don't need to be washed and dressed, you can just give them a bowl and just see what they can do. That's all part of the rehab. Emthey shouldn't have a commode next to their bed, they should be walkingbecause that's all rehab. But I do definitely think that people think ohwalkingemare they transferringthat's the rehab. Emand I think thatemit's a hard thing to get away from.	Interview 1 pg 3
I mean the rehab can take place anywhere but it's thatthey're waiting for (rehab ward), they're waiting for rehab. And I think that things have changed now because we're trying to be much more rehab focussed on the wards because we don't have as many rehab places. So we are still sending people to rehab	Interview 1 pg 3
Emwhere, it would be, oh they are waiting for rehab so we don't really need to see them. Embecauseyeahdefinitely, that has happened. And I think that emit was something that we constantly tried to address on ** because, that's the worst thing to do. Because sometimes the can be waiting for quite a long time	Interview 1 page 4
I think we all want this ideal text book thingyou know, goal-led, get the carers, get the family members, get the patient, do it all together. But it's not always as easy as that. Unfortunately.	Interivew 1 page 7
we've got the community rehab team as well, I don't know whether you've heard of them. That's a newagainit's completely new. And theyemare basicallywhen the ** closed down the teams were made up of support workers who worked at the ** so they've all got a really good background of rehab. The problem that we've got with that team at the moment is that it doesn't have any kind of physio that'semconnected to the team. So at the moment we're not using them that much from the hospital.	Interview 1 page 10

as I keep saying, it is an MDT approach but I think, the focus is often on <i>how are they transferring, how are they mobilising,</i> so then that immediately then comes down to us.	Interview 1 page 12
sometimes more than once a day, several times a day	Interview 1 page 12
And he was absolutely livid because he said 'I was told that my mum was coming to this unit to get 2 hours of physiotherapy a dayand what's happeningshe's not getting anywhere near thisblah blah blah'. Now his mum was in her nineties and was really quite frail and you thinkwell how could you possibly expect that your mum could cope with 2 hours of physiotherapy a day. Because that's just completely unrealistic	Interview 1 page 13
I worked on ward (rehab ward) which was seen as supposedly, a slow-stream rehab ward. EmI think I probably would have described it more as 'we don't really know what to do with this patientthere's a lot of complexitieswe'll send them to ** because they'll have more time to sort them out'	Interview 2 page 1
The physios did more rehab, because there was more physio staff on the ward	Interview 2 page 1
sort of everyday for about 2 weeks	Interview 2 page 1
So then I ended up getting shouted at because people hadn't got the moving and handling plans done and hadn't had their site visits done because I'd been doing <i>proper rehab</i>	Interview 2 page 1
you might have rehab with a younger person that takes you half an hour, forty minutesand the same rehab session with an older person might take you the entire morning. So, you're starting at half 9 and finishing at lunch time, and obviously that givesI thinkif you have twentyI think it was 29 patients I hadyou know, you have to try and balance what's happening with everybody	Interview 2 page 2
Emfor me rehabilitation is somebody getting back something that they've lost. So if they've had an ability beforeemthenmy aim, if I was rehabbing them, would be to try and get them back to that point to where they were before. So if they've dropped 3 points down a scale, I'd be trying to get them back up to their original baseline point	Interview 2 page 2
we're being told we're not allowed to use that terminology now, we're not allowed to say we're getting people back to baseline. It's all about what we can do to get them home safely	Interview 2 page 2
because I think mobility is quite key in terms of all of the other activities of daily living	Interview 2 page 3
for the majority of people, [mobility] is their number one priority of being able to manage something themselves	
I do suspect that there was times when it was probably used	Interview 2 page
Because at the end of the day, they're in hospital because they're poorly so they probably don't feel 100%. Like, how many times would you go out for a run if you had a cold and felt crapsoand that's what we're essentially asking somebody who's in their 80s to do is get up and have a little jog down the	Interview 2 page 12

ward. And they're maybe in pain and they're tired and they haven't had a good nights sleep because the person in the next bed kept trying to get in to bed with them. And then we're going, we just want to walk you over here, and they're just like'go awayI just want to sleep today'	Interview 2 page
they're a green patient. If they're still needing physio they're a green patient, but then if they're waiting for OT then they're red	17
Sorehabilitation to me is looking at an originalI think the word is now banned in the hospitalbaseline as to where the patient was functioning prior to coming in to hospital. And thenworkingobviously after the medical issues have been dealt withworking towards getting them back to that level of function that they were previously at. Emobviously sometimes it may not bewe may not be able to get them back to that baseline so that functional level would drop down. So therefore, you're rehabilitating back to their maximum potential. Emandthat to me in a nutshell is really what rehabilitation is about	Interview 3 page 1
people think that the major thing that people need to be able to do is mobilise. I'm not always sure that that's the case	Interview 3 page 1
A lot of people get told at (emergency hospital), oh you're going down to the base site for intensive rehab and we've got a bed of28 patients with only 2 people on there, 1 assistant, 1 physio. So there's way you could get through and do intensive rehab and do 2 half hour sessions per day for every one of those patients	Interivew 3 page 2
in some respects it is different because they may have some different equipment there. They may have a separate gym thereculture therethe philosophy of how they work there. So for instanceit's hard to reflect on the (new facility) at the moment but the old (intermediate care unit), they used to have to go down to the day room for their meals etc, so it was kind of getting people in to that normal routine and function in everyday. So in that respect it might be different but the actual hands-on may not be that much different	Interview 3 page 2
But you're right, it could be a placeor a philosophy ofwhat do I do in this rehab? And that's what I'm trying to work on at the moment with regard to a service development I'm doing. I think it's changing that culture of rehab on the wards and getting everyone to participate and getting everyone to realise that rehabilitation is their job, including the patients. So what I'm working on is to do a patient information leaflet to say you're now nearing the end of your medical treatment or at the end of your medical treatment and the focus is now on you doing what you need to do	Interview 3 page 2/3
are getting towards that rehab part of their stay and emand then getting the healthcares etc involved. So everything that patient does is then focussed towards function and rehabilitation. So they would no longer give them a washso to say they expect the patient to try and wash themselves. They wouldn't bring a commode by the bed, they would get the patient to walk to the toilet and do it for themselves. That's what I want to create	

on the ward and it takes that whole emphasis aboutoh physio, they're rehab, they're mobility. It's notit's everybody's job to get involved with that. So every little bitlike I say, with the numbers on the ward there's no way that physio even spending 10 minutes with somebody, walk them to the window and back, that's not really rehab. So if we can get everyone engagedthey go to the toilet 5 times a day or whatever, then there's another 5 walks during the day. If we can encourage them to get out of bed themselves then there's transfer practice. The idea is thatto actually start to educate the health care assistants in to best techniques to do that so they've got a basic understanding of 'if I stand back here, or if I held them at that point'then the patient could do it themselves.	
I think if you looked at the wider team and you said 'What's rehab?' to somebody other than the physio's, the OTs etc, those therapists involved in rehabI think they would give a very simplistic idea of what rehab isoh yeah you'll walk with the	Interview 3 page 3
and it'll just have in the plan you know, their medical stuff and it'll say 'for physio' or they'll have 'rehab' and you just think what, what?	Interview 3 page 7
give this guy a chance in a rehab environment where they have got better facilities and maybe can spend a bit more time with him	Interview 3 page 10
I've had the question where, 'what are we sending them down there for, the physio's aren't really seeing that patient?'	Interview 3 page 12
people see Physio, they think 'Physio – rehab', that'sI think that's everybody in the population in general. It usually starts with the guy running on the football pitch with the sponge, oh that's what physio doesphysio rehab people	Interview 3 page 15
some of the ward is meant to be part rehab for people	Interview 4 page 1
I think either building or regaining skills that will give you some sort of function to complete a goal-orientated task. Not necessarily embeing independent, maybe just learning new techniques or different techniques to be able to carry out a certain task	Interview 4 page 1
And in reality, it's not kind of all singing, all dancing you know, getting soldiers to walk again kind of place	Interview 4 page 2
there'll be people on the ward that we're working with and who are improving but because of the bed pressures they'll be sent to rehab	Interview 4 page 2
people judge rehab on mobility because there was the criteria that you must be mobile with 1 person. Sothat's what people were looking for	Interview 4 page 5
if it was going to be quite a complex discharge if there was a lot of problems, it seemed they'll be like they need more time so we'll send them to rehab when there wasn't really a rehab need	Interview 4 page 6
the word rehabthey're going to a rehab ward and there's physio's there and they're going to have therapy and you knowdon't get me wrong, it is obviously a rehab facility, but I	Interview 4 page 8

think it doesincrease family, relative expectations of what's going to happen when they get there	
the way I look at rehabilitation is having time to work with somethingsomebodyon some meaningful activity. Say it'swashing and dressing for example, if someone can't manage that. It's working with them daily, having daily input with that person, practicing tasks, so that they canpossibly become independent or improve in their independence in that area that we're looking at. But that relies on having the time to do that	Interview 5 page 1
all the other areas of rehab have dwindled away and there isn't the funding for it, there isn't the time to look in to other areas of ADLs that would need a bit more time, that we could have assistance to support patients in those roles. And then, mobility is really the last thing that we're left with perhaps	Interview 5 page 1
Particularly as we're OTs it should be around all areas of life but it just doesn't seem to be that way in hospital, it's whether people can get up and walk or not.	Interview 5 page 2
now that I'm on ward (respiratory), we talk about patients needing further rehab and they are referred to a Care of the Elderly ward. And numerous times I say in MDT on those wards they have the same number of physiotherapists and OTs and other disciplines on that ward that we have on ward **, the respiratory ward. But because our ward is specifically medical for respiratory patients, they need the beds for those patients and then patients would go to a Care of the Elderly ward which we also refer to as rehab wards. Andit's no differentthere isn't any increased support there	Interview 5 page 2
because if you say to a patient that you're going to rehab, the family, they really cling on to that and they're expecting some intense rehabilitation	Interview 5 page 3
even if you look at a ward handover it just says 'rehab'which it just sort ofwe all understand it that they need more time with the physio or the OT, that's how it's understood. it could be something that the OT's working with but it would just be termed 'rehab'. And I thinkas I've mentioned on a handover if we see that, we just assume it means they're not back to their baseline mobility	Interview 5 page 4
And the impact that (the new emergency hospital) has on something like rehabis one area it's had such a massive impact on	Interview 5 page 10

Appendix 6 – Faculty Ethical Approval



Professor Kathleen McCourt, CBE FRCN Executive Dean

This matter is being dealt with by:

Dr Nick Neave Director of Ethics Feculty of Heath & Life Sciences Northumborland Building Newcestle upon Tyne NET @ST

17th July 2015

Dear Gemma

Faculty of Health and Life Sciences Research Ethics Committee

Submission Code: HLS-PHW141515

Title: Rehabilitation Potential Of Older People Admitted To An Acute Hospital – Exploring The Decision Making Process Of Health Professionals

Following independent peer review of the above proposal, I am pleased to inform you that Faculty approval has been granted on the basis of this proposal and subject to compliance with the University policies on ethics and consent and any other policies applicable to your individual research. You should also have recent Disclosure & Barring Service (DBS) if your research involves working with children and/or vulnerable adults.

The University's Policies and Procedures are available on the ELP. Organisation name: HLS0002: Research Ethics and Governance

You may now also proceed with your application (if applicable) to:

- NHS R&D organisations for approval. Please check with the NHS Trust whether you require a Research Passport, Letter(s) of Access or Honorary contract(s).
- NHS or Social Care Research Ethics Committee (REC). [They will require a copy of this letter plus the
 ethics panel comments and your response to those comments].

You must not commence your research until you have obtained all necessary external approvals.

The University strongly advises that the supervisor accompany the student when attending an external REC.

All researchers must also notify this office of the following:

- Any significant changes to the study design, by submitting an 'Ethics Amendment Form'
- Any incidents which have an adverse effect on participants, researchers or study outcomes, by submitting an 'Ethical incident Form'
- Any suspension or abandonment of the study;

We wish you well in your research endeavours.

Yours sincerely

Ú, Juni Dr Jcanna Reynolds

Faculty Ethics Coordinator: Department of Public Health and Wellbeing

Pice-Chancellar and Chief Ececutive Professor Andrew Wathey

Northumoria University is the buding name of the University of Northumbria at Newcastle

Appendix 7 – NHS Research Ethics Committee Approval



Jarrow Business Centre Jarrow REC Centre (Room 001) Rolling Mill Road Jarrow NE32 3DT

Telephone: 0191 428 3565

11 November 2015

Miss Gemma Bradley c/o B105, Coach Lane Campus West Northumbria University Benton Newcastle upon Tyne NE7 7XA

Dear Miss Bradley

Study title:

REC reference:

Rehabilitation Potential Of Older People Admitted To An Acute Hospital – Exploring The Decision Making Process Of Health Professionals 15/NE/0322 171751

IRAS project ID: 171751
Thank you for your letter of 3 November 2015, responding to the Committee's request for

Thank you for your letter of 3 November 2015, responding to the Committee's request for further information on the above research and submitting revised documentation.

The further information has been considered on behalf of the Committee by the Chair.

We plan to publish your research summary wording for the above study on the HRA website, together with your contact details. Publication will be no earlier than three months from the date of this opinion letter. Should you wish to provide a substitute contact point, require further information, or wish to make a request to postpone publication, please contact the REC Manager, Ms Gillian Mayer, nrescommittee.northeast-newcastleandnorthtyneside1@nhs.net.

Confirmation of ethical opinion

On behalf of the Committee, I am pleased to confirm a Favourable ethical opinion for the above research on the basis described in the application form, protocol and supporting documentation as revised, subject to the conditions specified below.

Mental Capacity Act 2005

I confirm that the committee has approved this research project for the purposes of the Mental Capacity Act 2005. The committee is satisfied that the requirements of section 31 of the Act will be met in relation to research carried out as part of this project on, or in relation to, a person who lacks capacity to consent to taking part in the project.

Conditions of the favourable opinion

The favourable opinion is subject to the following conditions being met prior to the start of the study.

A Research Ethics Committee established by the Health Research Authority

Appendix 8 - Participant Information for all core staff on base ward

Information for Staff on Ward ***

Study Title: Rehabilitation potential of older people admitted to an acute hospital - exploring the decision making process of health professionals

I would like to inform you of a research study being carried out within your place of work. Before the study commences, it is important that you understand why the research is being done and what it would involve for you. This information leaflet provides background information about the study alongside answering some of the common questions that people have about involvement in research.

Please take time to read the following information. Please contact me if there is anything that is not clear, or you would like more information.

This study is part of study towards a post graduate research degree. It has been reviewed and approved by the research ethics committee of the Faculty of Health and Life Sciences at Northumbria University and the Research and Development department of Northumbria Healthcare Trust.

What is the research study about?

This study will explore the decision making of health professionals who are involved in making decisions about rehabilitation potential and subsequent pathways of care and treatment for older people following an unplanned admission to your hospital. Older people often experience longer hospital stays and have more complex needs than other patient groups, and understanding pathways for this patient group, alongside the decisions that influence those pathways, is increasingly important.

The research will predominantly focus on the decision making of Occupational Therapists and Physiotherapists as these professionals have not been the focus of previous research, although there may be times when the researcher will be involved in activities which also involve other members of staff, patients and carers.

Who is the researcher?

The researcher is a qualified Occupational Therapist who now works as a Senior Lecturer at Northumbria University. She is undertaking this research as part of a Post Graduate Research Degree (PhD).

What does the research involve?

The researcher would like to spend time in your clinical area to understand the decisions made for older people following an unplanned admission. In particular the researcher would like to understand what activities happen to support decisions (assessments, handovers, MDT meetings), the involvement of professionals and patients in such decisions and the influences on these decisions at the time they are being made.

Phase 1 of the research will involve periods of workplace observations to enable the researcher to become familiar with the organisation and the environment and to understand activities and processes which happen within this environment to support decision making.

Phase 2 will involve, the researcher, in collaboration with the ward team, identifying particular patients to 'track' during their admission and will carry out more focussed individual observations of particular assessments and meetings, and interviews with staff and patients. All of these activities will relate to the episode of care for the identified patients.

How does this involve me?

You are a practitioner who is involved in the care and treatment of older people on the ward where the research is taking place. I have been given permission from your trust and from your service managers to carry out this research and would now like to ask for your consent to observe your routine practices.

What am I being asked to do?

You are being asked to be part of Phase 1 of the research. During this stage, the researcher simply wants to understand the organisation that you work within and gain an overview of what activities happen to support decision making for older people following their unplanned admission to hospital. The researcher would like to be present during activities such as ward rounds or multi-disciplinary meetings. At this stage, the researchers role will be mainly non-participatory (i.e. she will mainly observe in an unobtrusive way) although may ask you some questions to further develop her understanding of processes. The researcher will always introduce herself during such activities, explain what she is focussing on and ask all staff present whether they are comfortable and think it is appropriate for her to be present. Please also be reassured that the researcher has permission from your organisation to be present when confidential information is being discussed about patients, staff, or ward activities.

During Phase 2 of the research, the researcher may also ask for your help in identifying appropriate patients to 'track' during their admission. She will fully explain to you the type of patients she would like to include in the research.

What are the benefits of taking part?

On an individual level, it may be that you experience personal benefit in thinking about and discussing the decision making process following an unplanned hospital admission. You could also reflect on the experience of being a participant in a research study and use this as evidence of continuing professional development.

At a professional and organisational level, you will be making an important contribution to the evidence base underpinning the understanding of decision making in relation to rehabilitation and services for older people. The findings of the research could help to inform local service development in this area and contribute to wider developments in relation to these issues.

Are there any disadvantages of taking part?

It may be that you find the process of being observed or discussing your practice uncomfortable. If this is the case, you can ask the researcher to terminate an observation at any time. The researcher will also have contact details of identified people within your organisation who you can talk to if you have found any element of the experience uncomfortable.

Some people worry that the things they say will be identifiable to them. Please be reassured that any information you share will be treated as confidential and will be anonymised within any subsequent reports, projects or publications.

What happens if I don't want to participate?

You are under no obligation to participate in this study and a decision not to take part will have no adverse consequences for you. If you do not want to participate, the researcher will not observe any activities which involve you.

What would happen if I agree and then I change my mind?

Before each contact with the researcher, she will always check that your consent is still ongoing and you are happy for her to be present. You can also ask to withdraw from the study at any time although any anonymised data that was gathered when you were in agreement could still be used within the study.

What about other people present when you are observing me – will they also need to consent?

During phase 1, the researcher will not be directly observing patient or family interactions. However, notices will be on display to inform patients and visitors (including staff visitors) that there is a research project being carried out within the ward.

If the researcher would like to approach patients, family members or wider staff to become more involved in the research during phase 2, she will approach them with further information and ask for full, written consent.

The researcher may be present when information about other patients or staff is being discussed. The researcher has the relevant permissions from the Research and Development department of your Trust to be present when service user or staff information is discussed.

How will information be collected?

During observational stages, data will be collected through notes taken by the researcher.

What will happen to the information that is collected?

All notes (and subsequent electronic files) will be anonymised and will not contain any personal information. They will be stored securely (either electronically on a secure, password protected server through the university, or hard copies in locked storage areas only accessed by the researcher) until the end of the study and then will be destroyed. I would like to reassure you that any information given by you will only be used for the purpose of this research study alone.

How will whatever I say be anonymised?

Please be assured that all the information shared during an observation or an interview will be anonymised. When the study is written up and disseminated, no names will be identified and participants and episodes of observation will be referred to using numbers/codes (e.g. Observation 1/Participant 1). Nobody will be able to identify any individual in the final write-up or subsequent publications.

How will the research be disseminated?

As this research is part of a research degree, the research will be written up as part of the final thesis for the PhD award. A shorter summary will also be shared with all of the teams involved in the study, service managers and other key members of your Trust, alongside members of staff at the university.

I also hope to publish the study in relevant peer-reviewed journals and present the findings both internally to members of academic staff at Northumbria University and externally, at relevant conferences.

Who do I contact if I want to ask more questions about the study?

Please do not hesitate to contact me directly if you require further information or would like to chat about this study.

My contact details are:

Gemma Bradley

Room H215, Coach Lane Campus East, Northumbria University, Coach Lane, Benton, Newcastle Upon Tyne. NE7 7XA

Telephone: 0191 215 6289 Email: gemma.bradley@northumbria.ac.uk

Alternatively, you can contact my principal supervisor:

Dr Katherine Baker

Room ***, Coach Lane Campus West, Northumbria University, Coach Lane, Benton, Newcastle Upon Tye, NE7 7XA

Telephone: 0191 215 6723 Email: Katherine.baker@northumbria.ac.uk

Many thanks for your support

Appendix 9 – Written summary/poster to introduce research to all staff, patients and visitors

Information for all Patients, Carers and Visitors to Ward ***

This summary contains information about a research study which is currently being carried out on this ward. Please take a few moments to read this information

What is the research about?

The research is exploring what happens to older people following an unplanned admission to hospital and is particularly focussing on the decision making of health professionals who are involved in the care and treatment of older people.

Who is the researcher?

The researcher is a registered Occupational Therapist with experience of working with older people in hospitals. She current works as a lecturer at Northumbria University. This means she is independent to the Trust although has full permission to be carrying out the research.

What does this mean for you?

During the course of the research, the researcher is spending time on this unit to understand how decisions are made and will be spending time with different members of staff, and then observing and interviewing staff and patients. At this stage, you are not being asked to do anything although youmay see the researcher during ward activities (for example ward rounds), or spending time with other staff or patients. The researcher will always introduce herself and explain what she is doing. If you do not wish her to be present, please just let her know and please be reassured that this will not affect your care or treatment in any way.

Will I be involved in the research?

At a particular stage in the research, the researcher would like to more directly involve particular staff, patients and carers, because they have particular experiences which are of interest to the project. If this applies to you, the researcher will provide detailed information about what you are being asked to do and will ask for your full informed consent.

How can I ask questions about the research?

If you see the researcher on the ward, please feel free to ask questions directly. If you would like to ask questions but do not know how to contact the researcher, please ask a member of the ward team who can either answer questions directly or pass on questions to the researcher. You can also contact the researcher directly using the details below:

Gemma Bradley

Room H215, Coach Lane Campus East, Northumbria University, Coach Lane, Benton, Newcastle Upon Tyne. NE7 7XA

Telephone: 0191 215 6289 Email: gemma.bradley@northumbria.ac.uk

This study is part of study towards a post graduate research degree. It has been approved by the research ethics committee of the Faculty of Health and Life Sciences at Northumbria University and the Research and Development department of Northumbria Healthcare Trust.

Appendix 10 - Participant Information for Patients and Carers – At a glance summary

Study Title: Rehabilitation potential of older people admitted to an acute hospital exploring the decision making process of health professionals

This information sheet summarises key information about a research study that you are being asked to participate in: The research aims to understand how decisions are made following the admission of an older person to hospital. Many of these decisions affect where that person goes next, or what services they receive and these decisions are of interest. The researcher has full permission from the trust to be on the ward and to talk to you as part of this study. You are being asked to participate because your experience is of particular interest to the researcher. The researcher would like to be present during some of your contacts with health professionals – she will try to simply observe during these contact but is happy to answer any of your questions. She would then like to talk about your experience, ideally before you leave this ward. She will find a private room on the ward to do this and would be happy for a carer or relative to be present with you. If you were happy, the researcher would like to audio-record this interview. The researcher would also like to be able to view your clinical records to review how your care, and decisions about your care, has been documented. She will not take the records off the ward and will not use any identifiable information about you. To give permission for the researcher to do these things, she will ask you to sign a consent form. Even if you give your consent, the researcher will always introduce herself to you again so you understand why she is there and what she is doing. If you have changed your mind or do not want her there that day, you just have to let the researcher know. The researcher will ensure she anonymises any information and stores this securely and confidentially. There will be nothing in any research reports or publications which identifies you. You are under no obligation to take part and if you do not take part, your routine

care will not be affected.
Appendix 11 – Extended Participant Information for Patients, Families and Carers

Study Title: Rehabilitation potential of older people admitted to an acute hospital - exploring the decision making process of health professionals

I would like to tell you about a research study that is currently taking place within your ward.

Please take time to read the following information. Please contact me if there is anything that is not clear, or you would like more information.

This study is part of study towards a post graduate research degree. It has been reviewed and approved by the research ethics committee of the Faculty of Health and Life Sciences at Northumbria University and the Research and Development department of Northumbria Healthcare Trust.

What is the research study about?

This research study will explore decisions made following unplanned admissions to hospital for older people. Older people often experience longer hospital stays and have more complex needs than other patient groups, and understanding experiences for this patient group, alongside the decisions that take place, is increasingly important.

The researcher is working alongside departments and wards within this Trust to identify older people, following an unplanned admission, where decisions regarding future improvements, needs and recommended services will be made. For a small number of these patients, within an identified time period, the researcher would like to be present during assessments and interactions with patients and family members, and during professional activities such as handovers and multi-disciplinary meetings. The researcher would then like to interview both health professionals and patients, asking questions about the decisions made and the decision making process.

The health professionals involved will predominantly include Occupational Therapists and Physiotherapists (as research in the past has focussed on other professionals such as doctors). The study would also like to explore your experience as a patient during the decision making process.

This research is currently being undertaken within the ward that you have been admitted to. You may see the researcher involved in different activities and spending time on the ward.

Who is the researcher?

The researcher is a qualified Occupational Therapist who now works as a Senior Lecturer at Northumbria University. She is undertaking this research as part of a Post Graduate Research Degree (PhD).

How does this affect me?

You have been identified as an older person, currently on the ward for an unplanned hospital admission. The researcher would like to focus on your experience and your interactions with health professionals to gain insight in to the decisions that are made regarding what happens next following your time on this ward.

What am I being asked to do?

The researcher would like to ask you to be involved in three main ways:

Observe the assessments and interactions you have with health professionals the researcher would like to be present when particular members of staff are working with or talking to you. The focus of such observations is to understand how decisions are made within the ward and therefore the researcher will not be gathering any personal information about you, your condition or your situation.

In these situations, the researcher will always identify herself and explain the purpose of what she is doing. Even if you have given your consent at an earlier stage, the researcher will always revisit this consent to make sure you are still happy to be involved.

It is important to reassure you that, irrespective of any involvement, this research will not change the care or treatment that you receive in any way.

Carry out a short interview with you to gain a deeper understanding of your experience – following observations, the researcher would like to carry out a short individual interview with you to gain your perspective on the decisions made about what is likely to happen next after your stay on this ward. Where possible, the researcher will try to do this before you move wards or are discharged home, although as decisions are sometimes made at short notice within hospitals, the researcher may make contact with you after this to carry out the interview.

Depending on the situation, the interview will either take place in a private room on your current ward, in a private room of another ward or unit that you may have moved to, or within your home. You may find it helpful to have a family member or friend present and the researcher will ask you about this before making arrangements.

Give permission for the researcher to access your health records – An

understanding of decision making can be developed by reviewing healthcare records as these documents provide a record of things that have been said, people who have been present and decisions that have been made. Again, please be reassured that the researcher will not be using any personal information about you, your condition, or your situation.

What are the benefits of taking part?

On an individual level, the researcher does not want to influence your experience and therefore it is hoped that you will not notice any difference in the service and care that you receive.

If you do decide to be involved you will be contributing to a more detailed understanding of decision making following unplanned admissions to hospital for older people and this may contribute to future service developments. More specifically, you will be helping to represent the patient perspective in this area; a perspective where current knowledge is limited.

Are there any disadvantages of taking part?

It may be that you find the process of being observed or discussing your experience uncomfortable. If this is the case, you can ask the researcher to terminate an observation or an interview at any time. The researcher will also have contact details of identified people within the Trust who you can talk to if you have found any element of the experience uncomfortable.

Some people worry that the things they say will be identifiable to them, especially when talking about the experience they have had with members of the healthcare team, or on a particular ward. Please be reassured that any information you share will be treated as confidential and will be anonymised within any subsequent reports, projects or publications.

What happens if I don't want the researcher to be present during aspects of my care?

As mentioned above, the researcher will always identify herself and explain the purpose of what she is doing on that day. If you do not feel comfortable with the researcher being present during your assessment or interaction then please just say. The researcher can leave at any point and your normal care, treatment and involvement with services will not be affected in any way.

Can I agree to be involved in some parts of the study and not others?

Ideally, the researcher would like to gather information through all of the methods above and therefore contributing to each stage would be beneficial for the study. However, it may be possible to consent to involvement in some areas and not others. If this is appropriate, the researcher will discuss this with you and the written consent form will clearly identify this. Also, the researcher will check that your consent is still ongoing during any new activity.

What would happen if I agree and then I change my mind?

You can indicate at any point during an observation or interview that you have changed your mind and the researcher will be happy to leave or terminate the interview.

It is important to be aware that the researcher may still use anonymised information in the study up until the point that you withdraw your consent.

Can other people be present?

As with any assessment or intervention with a health professional, it is up to **YOU** who you would like to be present.

What type of information is being collected?

During observations, the researcher will be making notes in relation to what is happening and being discussed. Please be reassured that these notes will not include any personal information relating to you or your health condition but instead will be focussed on the things that you and the health professional are saying and doing to help reach decisions about what happens next in your care.

During the interview, the researcher will ask some structured questions and would like to audio-record your responses. She will then transcribe the recordings to gain a written transcript of your words. A trained transcriber may be asked to transcribe on behalf of the researcher although any files given to this person will not contain identifiable information (e.g. they will be labelled 'participant 1' rather than using your name).

If the researcher accesses your clinical records, she will be particularly looking at how assessments and decisions are documented.

Will whatever I say be confidential between myself and the researcher?

Information will be treated confidentially. However, there are some exceptions to this, where the researcher may have a duty of care to inform other staff of information. For example, if you highlighted a concern about a member of staff or another patient, the researcher would have a duty to inform relevant other staff members.

How will whatever I say be anonymised?

Codes and numbers will be used to anonymise notes and transcripts. Consent forms which contain participant codes will be stored separately to any subsequent data. When the study is written up and disseminated, no names will be identified and nobody will be able to identify the individuals involved in the study.

How will information be stored?

All audio files and notes will be anonymised and will be stored securely until the end of the study and then will be destroyed. Paper-based information will be stored in locked areas and electronic information will be stored on a secure, password protected server. I would like to reassure you that any information given by you will only be used for the purpose of this research study alone.

How will the findings of the research be shared with others?

Only the researcher will have access to any identifiable information although she may share anonymised information with her research team and supervisors during the research process.

As this research is part of a research degree, the research will be written up as part of the final thesis for the award. It is also hoped that findings will be published in relevant journals and presented at conferences. A shorter summary will also be shared with any interested participants, and with the teams involved in the study, including service managers. If you do decide to participate, the researcher will ask if you would like to receive follow-up information about the study after it is finished.

Who do I contact if I want to ask more questions about the study?

Please do not hesitate to contact me directly if you require further information or would like to chat about this study.

If the study raised an issue that I wanted to complain about, who could I contact?

Initially, you could contact the researcher, or the research supervisor using the contact details below. If you wanted to speak to someone not directly involved in the research, you could contact the Patient Advice and Liaison Service (PALS) on 0800 032 0202

My contact details are:

Gemma Bradley

Telephone: 0191 215 6289

Email: gemma.bradley@northumbria.ac.uk

Alternatively, you can contact my principal supervisor (a senior and experienced researcher within my organisation who is supervising my research):

Dr Katherine Baker

Telephone: 0191 215 6723

Email: Katherine.baker@northumbria.ac.uk

Many thanks for your support

Appendix 12 – Patient Consent Form Patient Consent Form

Title of research study: Rehabilitation potential of older people admitted to an acute hospital - exploring the decision making process of health professionals

Name of researcher: Gemma Bradley box

Please initial each

GENERAL
1. I confirm that I have read and understood the information sheet dated (version) for the above study.
 I have had the opportunity to ask questions and have had any questions answered satisfactorily.
 I understand that my participation is voluntary, that I am free to withdraw at any time without giving any reason, but that anonymised information gathered until the point I have withdrawn consent may still be used in the study.
 I understand that my name and details will be kept confidential and will not appear in any printed documents.
OBSERVATION
 I agree for the researcher to observe elements of my care and for notes about these observations to be included in the research
INTERVIEW
6. I agree to be interviewed as part of the above study.
 I am happy for my interview responses to be audio-recorded and am happy for resulting information to be included in the research.
ACCESS TO RECORDS
8. I agree that my clinical records relating to this admission can be accessed by the researcher for the purpose of this study.

Name of Participant Date Signature

Researcher	Date	Signature

(When completed: 1 copy (original) for researcher file; 1 copy for participant)

Appendix 13 – Health Professional Consent Form

Health Professional Consent Form

Title of research study: Rehabilitation potential of older people admitted to an acute hospital - exploring the decision making process of health professionals

Name of researcher: Gemma Bradley

For Involvement in Phase 1:

Please initial each box

1.	I confirm that I have read and understood the information sheet dated (version) for the above study.	
2.	I have had the opportunity to ask questions and have had any questions answered satisfactorily.	
3.	I understand that my participation is voluntary, that I am free to withdraw at any time without giving any reason, but that anonymised information gathered until the point I have withdrawn consent may still be used in the study.	
4.	I agree for my practice to be observed as part of the above study and anonymised notes about this practice included in the research.	

For Involvement in Phase 2:

5.	I agree for my practice, including my work and discussions with patients, to be observed as part of the above study and anonymised notes about this practice included in the research.	
6.	I agree to be interviewed as part of the above study.	
7.	I am happy for my interview responses to be audio-recorded and am happy for resulting information to be included in the research.	
8.	I understand that my name and details will be kept confidential and will not appear in any printed documents.	

Please enter your name and date and sign below:

Name of Participant	Date	Signature	
Researcher	Date	Signature	

(When completed: 1 copy (original) for researcher file; 1 copy for participant)

Appendix 14 - Process for seeking advice from a consultee

Obtain name and contact details of next of kin from healthcare team Telephone or face to face contact with next of kin – introduce researcher role and research project and desire to seek advice about involving their next of kin in the research

Ask the next of kin whether, in their opinion:

- There are other individuals who know the person best and would be better placed to support the individual with this decision
- Whether there is anyone who provides regular care for the person
- Whether the person has granted Lasting Power of Attorney to another person

 \downarrow

- Whether there is a Court Appointed Deputy?

Obtain names and contact details for any identified individuals

If different to next of kin, or any individual named above, also ask for contact details of person named by the potential participant as the person who could best support them to make a decision.

 \checkmark

Provide written and verbal information to the next of kin and any significant other regarding participating in the research project

Ask the next of kin, and any individuals identified as significant from the above information:

- For advice on whether the person should take part in the research project
- What, in their opinion, the person's wishes would be in relation to involvement in the project if they had capacity

All contact with relatives or other individuals to be documented and at least one person to be identified to complete the consultee form

Form to be completed at least 24 hours following initial contact with consultee

Appendix 15 – Consultee Advice Form Advice from consultee regarding service user involvement in research

Title of research study: Rehabilitation potential of older people admitted to an acute hospital - exploring the decision making process of health professionals

Name of researcher: Gemma Bradley box

Please initial each

1.	I confirm that I have been consulted about [name of participant]'s	
	participation in this research study.	
2.	I have had the opportunity to ask questions and have had any	
	questions answered satisfactorily.	
2	I have been called for my advice about whether frame of	
э.	nave been asked for my advice about whether [name of participant] should take part in the project and my response is	
	recorded below:	
4	I have been asked what I think the person's feelings and wishes	
	would be, if they had capacity to decide whether to take part and	
	my response is recorded below:	
5.	I understand that I can request that he/she is withdrawn from the	
	study at any time without giving any reason and without his/her	
	care being affected, however it has been explained that	
	anonymised information could still be used within the study.	
6.	I understand that his/her name and details will be kept	
	confidential and will not appear in any printed documents.	
7.	I understand that the researcher may be present during routine	
	parts of his/her care in an observational role.	
0	Lunderstand that the researcher would like to interview him/her	
0.	and Lagree to be present during this interview or nominate	
	another carer or relative.	
9.	I understand that relevant sections of his/her care record may be	
	iooked at by the researcher	

Please complete the details below:

Name of Participant	
Name of Consultee	

Relationship of Consultee to	
Paticipant	
Date	
Signature	

For the researcher:

*Delete as appropriate			
Taking in to account the assessment of the person and the advice from the			
consultee, I have decided to include/not include* [name of participant] as a			
participant in the research			
Name of Researcher			
Date			
Signature			

(When completed: 1 copy (original) for researcher file; 1 copy for participant; 1 copy for consultee)





Appendix 17 – Example daily handover sheet

NAME	is and the	The Article of the	BACKBROUND	ASSESSMENT	RECOMMENDATION
Bpy 1 61	Admitted 1st-ney PC / Diagnosis:		PMH: COPO, hemis, 4/8P	TFR x1, NDF, 85C (recention)	For de with once day care dackede.
			1	SH74. Cat 2 - sectur. SHOP 10/5	7 Monday, DN for catheter.
	Upper Gi b'ec	a (habbelge		MSU +ve. 12/5 Enema	Perameters in place. It Con at
	UTI		Social: Lives alone, no carer		Positional changes
	Infection	Falls Risk			Referrate Sec 2 & 51' OTh
Not for 2222	18th Me			NEWS: EO: 12th may	
BAY1 B2	Admitted	:2h May	PMH Lowmood, Bypess greit,	NDF, SRC, weather TER ZE 41, SKIN	Physic, analgesia. Family heopy for
	PC / Diagnosis: Organg No pain following fall CT & Xiray NAD.		Prostete Call HT'N, delarkim,	SKIP: NUST:	Extension for Barris Court
					P2 Cod. an in total
			Seciet Si Pater's Ci	-	
	Intection	Falls Risk			Referrals:
	71:181:0	lias		NEWS: BU:14:h May	-
	CINING	yes	Carlot to started at some started at	NET - 1 and each IEC Fight	Landa simond analysis DD Davids
BAY I BO		I CET MAY	PTOP: power projects. N. M. Ar.	SVID- BUIST	Cosator Skoped - Administer, Physic
	PC / Diagno	HSIS.		JULE MUSIC	Brute below dama
	Fail, # puble ra	mus, utmobility			in or rois.
NIGU.			Social: Lines with wife		
[-*-	Infection	Falls Risk			Referrals: Sec 2 [] OT []
	231d May			NEWS: BO:	the second se
Say 1 Bed 4	Admitted	1 illy may	PMH: PD as (warfarm)	src ndš I. SkiNanostina lais on orom	PD meste - Sam. 12 md. 4 pre. For
CARA	PC/ Diagon	\$15		SKIP: MUST. CT L5 & 4 - 9 Result.	onthe review. Up with physics TWOC
and the second	back calo rates	llen			when mobile. Positional changes.
	reacy been letter	ing n	Geniel: Iluca along farath, suspend		mobile i from 1.
	1.4.4	1	acciai: invest alone family support		
	Infection	Fails Risk			Retenals:
	191h may	1 -		NEWS: BO: 13th May	
Bay 1 BS	Admitted	1 - 1 - 1 - 1 - 1	PMH:	SKIN: SKIP: BUSI:	
	PC / Diagno	ale			
			Social:		
	Infection	tion Falls Risk		Referrals:	
				NEWS: BO: 13th Mey	
au 1 BA	A day Manual	14th Ma-	DMH D sintlestone black anaddama	Hanny TER / sth undefinition and filler	DO MOT CAS Obvies Committee
ar roa	Romittee	Linut well	GKLAF HTM wennes insufficiency	continent. SKIN: small carratione	in pad berause of birs. Maltares
the second s	Perblagnost		one w. Hine we obside that to any	SKIP: MITST:	offered
1 bours	isadi pan, ieg weaknesa.				Second County and manufact first
New	-	and the second second	Social:		Hue serve manual to care
	Infection	Falls Risk			Referrals:
	24th May	5		NEWS: BO: 15th May	
ay 3.81	Admitted	10th may	PMH: sycke isw type 2 diabetic # wrist	Trans ZF +2 or 8th NBM Continent	Treated for paracetamol 100 city. Needs
	PCIDIsases	8:	ni partaly sighted	SKIM: bisters back of eg. SKIP:	swe low ax as coughing. Family aware
Statement and	AKI contriest 7	ChealBudgery		MUST: 2	boorty. On 12 T / 12 AS?
	Stosis	Calculately	Contai: Montin Anton et	Campronny of check	Bart Low Canton
		1	OUCHER INCOMENTION		parde cel.
	Intestion	Falls Fisk		NEXAL TP (1) .	Puste (rais: Lawiician []
ol for 2222	19th may	has		NEWS DU:	
ay3 OZ	Admitted	Pity may	PMH, anxiely RA asthma	zišti ndBi storetention i site D2 no	femanyi paoshi oexisa Sistemida may
	PC / Diagno:	sis:		SKIN: SKIP: MUST:	need blood. Rast on table FR by Dr.
Street Stre	uroecosta lali				bet has refused engine.
			Secial if: history		allestances be saint and
	Intection	Falls Right			Rafema's:
	26th may	196		NEWS: BO Web May	
1 01	And the state	1000	The second s	Det D and the local face of the	
ay 3 83	Admitted	10th may	min, pulmonary fibros s alcheimer	Gat G comfort faecing, incont	ore autors
	PC / Diagnos	15	ternisit*raplasty	SKIM-grade 3 eesral prede 2 r bubook	Positions
	abor and inizion	. !			lemity (ag) needs nursing care.
	1			SKIP: 11th way MUST:	Regular mouthcare.
	Infection	Folis Risk	Social recently went into regione	(1	Referrals: satv
				1	

containing Levit

? and service and

beat .